# UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM 8-K

## CURRENT REPORT Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): September 27, 2010

# **Parametric Sound Corporation**

(Exact name of registrant as specified in its charter)

Nevada

(State or Other Jurisdiction of Incorporation) 000-54020 (Commission File Number) 27-2767540 (I.R.S. Employer Identification No.)

1941 Ramrod Avenue, Suite #100 Henderson, Nevada 89014

(Address of Principal Executive Offices)

888-477-2150

(Registrant's telephone number, including area code)

N/A

(Former name or former address, if changed since last report.)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

[\_] Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

[\_] Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14.a-12)

[\_] Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

[\_] Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

#### Agreements Entered into in Connection with Spin-Off from LRAD Corporation

In connection with the previously announced separation and spin-off by LRAD Corporation ("LRAD") of its wholly owned subsidiary, Parametric Sound Corporation (the "Company"), on September 27, 2010, the Company entered into a Separation and Distribution Agreement and other definitive agreements with LRAD, that, among other things, set forth the terms and conditions of the separation of the Company from LRAD and provide a framework for the relationship between the Company and LRAD following the separation. In addition to the Separation and Distribution Agreement, which contains many of the key provisions related to the spin-off of the Company and the distribution of the Company's common stock to LRAD's stockholders, the parties also entered into a Tax Sharing Agreement. Each of these a greements is described below:

#### Separation and Distribution Agreement

On September 27, 2010, the Company entered into the Separation and Distribution Agreement with LRAD that sets forth the Company's agreements with LRAD regarding the principal transactions necessary to separate the Company from LRAD, including: (i) the contribution of substantially all of the assets of LRAD's HSS technology business to the Company; and (ii) the distribution by LRAD, on September 27, 2010 or as soon as practical thereafter, of all outstanding shares of the Company's common stock to LRAD's stockholders in the form of a pro rata dividend of one share of the Company's common stock for every two shares of LRAD's common stock outstanding to stockholders of record as of the close of business on September 10, 2010. This agreement also sets forth the other provisions that govern certain aspects of the Company's relationship with LRAD after the completion of the separation from LRAD and provides for the allocation of assets, liabilities and obligations between the Company and LRAD in connection with the separation. The description of the Separation and Distribution Agreement set forth under this Item 1.01 is qualified in its entirety by reference to the complete terms and conditions of the Separation and Distribution Agreement filed as Exhibit 2.1 hereto.

#### Tax Sharing Agreement

On September 27, 2010, the Company entered into a Tax Sharing Agreement with LRAD that generally governs the parties' respective rights, responsibilities and obligations after the separation with respect to taxes. Under the Tax Sharing Agreement, all tax liabilities resulting or arising from the contribution of LRAD's HSS technology business to the Company and the other separation transactions including the distribution will be borne solely by LRAD and its subsidiaries other than the Company. In addition, under the Tax Sharing Agreement, all tax liabilities (including tax refunds and credits) attributable to LRAD's HSS technology business for any and all periods preceding the separation will be borne solely by LRAD and its subsidiaries other than the Company. All tax liabilities (including tax refunds and credits) otherwise attributes available to LRAD and its subsidiaries other than the Company. All tax liabilities (including tax refunds and credits) attributes available to LRAD and its subsidiaries other than the Company. All tax liabilities (including tax refunds and credits) attributable to LRAD and its subsidiaries other than the Company. All tax liabilities (including tax refunds and credits) attributable to LRAD and its subsidiaries on the Company. All tax liabilities (including tax refunds and credits) attributable to the Company's operation of the HSS technology business for any and all periods following the separation will be borne solely by the Company. Any and all tax attributes, including net operating losses and research and development credits, which exist as of the date of the separation will be retained by LRAD and its subsidiaries other than the Company. The description of the Tax Sharing Agreement set forth under this Item 1.01 is qualified in its entirety by reference to the complete terms and conditions of the Tax Sharing Agreement filed as Exhibit 10.1 hereto.

# Licensing Agreement with Syzygy Licensing LLC

On September 27, 2010, the Company entered into a license agreement (the "License") with Syzygy Licensing LLC ("Syzygy"), a company in which our Chief Executive Officer, Elwood G. Norris, is the majority owner. Mr. Norris has assigned to Syzygy certain patent pending technology and trade secrets related to a new, improved and more cost-effective method of processing media input to create parametric sound output for parametric emitter devices such as those employed by the Company. Mr. Norris has also invented improvements to the emitters complementary with the new processing and control electronics. The Company agreed to reimburse patent, testing and prototype costs incurred to date (an amount estimated at \$90,000) and to pay future patent related costs. These prior costs include up to a maximum of \$25,000 for Mr. Norris' time in producing and testing prototypes and preparing for production. The License provides for future royalities of 5% of revenues from products employing the technology and a term of 20 years or the life of any resulting patent, whichever is greater. In the event no patent covering the licensed technology is issued after four years, then the royalty rate shall reduce to 3% in any territory until or if a patent is issued for any such territory. The Company may not sublicense without the permission of Syzygy, and sublicense royalty rates are subject to future negotiation in good faith. The License may terminate if the Company does not use commercially reasonable efforts to pursue the parametric sound business. The description of the License is qualified in its entirety by reference to the complete terms and conditions of the License field as Exhibit 10.2 hereto.

## Assumption of Facility Lease

On September 27, 2010, the Company assumed pursuant to that certain Assignment and Assumption of Lease Agreement all of the obligations of Syzygy, as lessee under that certain Lease Agreement, dated as of July 1, 2010 by and between Syzygy and Davric Corporation (the "Lease Agreement"). The Lease Agreement covers 4,500 square feet of space that will be used as the Company's corporate headquarters with monthly rent equal to \$4,000. The term of the Lease Agreement began July 1, 2010 and continues for twelve months, unless earlier terminated in accordance with the terms of the Lease Agreement. The description of the Lease Agreement is qualified in its entirety by reference to the complete terms and conditions of the Lease Agreement and the Assignment and Assumption of Lease Agreement filed as Exhibit 10.3 and 103.3.1 hereto

The Company has an oral month to month sublease with an unrelated party for a portion of the space at a rate of \$2,000 per month.

# Debt Financing

The Company has entered into Subscription Agreements, dated as of September 28, 2010 (the "Subscription Agreements"), with ten accredited investors, pursuant to which the Company has issued an aggregate of \$700,000 in 8% Notes, due September 28, 2011 (the "Notes"), and accompanying warrants to purchase an aggregate of 1,400,000 shares of our common stock (the "Warrants"). The Warrants have a five-year term and a cashless exercise provision, subject to certain limitations. The exercise price of the Warrants is fixed as the average of the closing price of the Company's common stock for the first 20 days of trading subject to a minimum exercise price of \$0.10 and a maximum exercise price of \$0.30 per common shar e.

Syzygy converted \$250,000 of amounts the Company owed to Syzygy for spin-off and technology development costs previously paid by Syzygy. Such amount converted does not represent cash available for working capital. Accordingly the net proceeds of the debt financing obtained in cash were \$450,000 which the Company intends to use for working capital. This amount included \$110,000 from entities affiliated with officers and directors purchased for cash on the same terms as other investors. The description of the Subscription Agreements, Notes and Warrants is qualified in its entirety by reference to the complete terms and conditions of the Subscription Agreements, Notes and Warrants filed as Exhibits 10.4, 4.1 and 4.2 hereto, respectively.

#### Indemnification Agreements

Our officers and directors are indemnified as to personal liability as provided by the Nevada Revised Statutes ("NRS"), our articles of incorporation and our bylaws but these are not exclusive and contemplate that agreements be entered into between the Company and our executive officers and directors with respect to indemnification (see "Limitation of Liability and Indemnification of Directors and Officers" beginning on page 58 of the Company's Information Statement, dated September 3, 2010, which is filed as Exhibit 99.1 hereto, and incorporated herein by reference).

Effective on September 27, 2010 the Company entered into indemnification agreements with its directors and executive officers (a total of three persons). The indemnity provided is in addition to that provided by the NRS or any successor statutes and is limited in certain circumstances including adjudication that the indemnitee's breach of fiduciary duty involved intentional misconduct, fraud or a knowing violation of the law. The description of our indemnification agreements set forth under this Item 1.01 is qualified in its entirety by reference to the complete terms and conditions of the Indemnification Agreement filed as Exhibit 10.7 hereto.

#### Item 3.02. Unregistered Sales of Equity Securities.

The information set forth under Item 1.01 above under the heading "Debt Financing" with respect to the issuance of warrants to purchase up to 1,400,000 shares of the Company's common stock is incorporated herein by reference.

# Item 5.01. Changes in Control of Registrant.

(a) Prior to the spin-off described in Item 1.01, the Company was 100% owned by LRAD. After the distribution of 100% of ownership in the Company to stockholders, LRAD has no ownership in and no longer controls the Company. Also see information on resignation of directors and officers described in Item 5.02 below whereupon after the spin-off there are no longer any common directors or officers between LRAD and the Company.

As described in the Company's Information Statement, dated September 3, 2010, which is filed as Exhibit 99.1 hereto, and incorporated herein by reference, the Company's director and President and CEO, Elwood G. Norris, owned immediately after the spin-off, and prior to any purchases of the financing described above, approximately 13% of the registrant. Due to this ownership along with his role as CEO and as one of two directors may allow Mr. Norris the ability to influence control of the Company.

The Company, after rounding up for fractional shares resulting from the spin-off, has 15,306,064 common shares issued and outstanding.

# Item 5.02. Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers.

(b) In connection with the separation of the Company from LRAD, the following directors of Parametric Sound Corporation resigned from the Company's Board, effective as of September 27, 2010: Thomas R. Brown, Helen C. Adams, Laura M. Clague and Raymond C. Smith. Kathy McDermott, the registrant's President, Secretary and Treasurer and its principal financial officer also resigned on September 27, 2010 in connection with the spin-off.

(c) On September 27, 2010 the Board appointed James A. Barnes as Chief Financial Officer, Treasurer and Secretary and Mr. Barnes is the Company's principal financial officer. Mr. Barnes, age 55, has been President of Sunrise Capital, Inc., a private venture capital and consulting firm since 1984. He participated in the recapitalization of LRAD and the founding of e.Digital Corporation, Patriot Scientific Corporation and other technology companies. Since 1999, he has been Manager of Syzygy Licensing LLC, a private technology invention and licensing company owned with Mr. Elwood G. Norris. Since 2000, he has also been a director and Secretary of AirScooter Corporation a non-reporting public company. He previously practiced as a certified public accountant and management consultant with Ernst & Ernst (1976-1977), Touche Ross & Co. (1977-1980) and as a principal in J. McDonald & Co. Ltd., Phoenix, Arizona (1980-1984). He graduated from the University of Nebraska with a B.A. Degree in Business Administration in 1976 and is a certified public accountant. Mr. Barnes is expected to devote only part-time to Parametric Sound.

As Manager of Syzygy, Mr. Barnes participated in the following related party transactions:

- (i) Licensing Agreement described in Item 1.01 above.
- (ii) Assumption of Facility Lease described in Item 1.01 above.
- (iii) Conversion of \$250,000 of amounts previously paid by Syzygy for Company spin-off and technology development costs into the debt and warrant financing described in Item 1.01 above.
   Syzygy is also owed \$25,000 for Mr. Norris' time in producing and testing prototypes and preparing a new product for production.
- (iv) Purchase by Syzygy for cash of an additional \$10,000 of debt and warrant financing (See Item 1.10 above) on the same terms as unaffiliated investors.

As described in the Information Statement, dated September 3, 2010, which is filed as Exhibit 99.1 hereto, and incorporated herein by reference, Sunrise Capital, Inc., a company controlled by Mr. Barnes, is owed \$25,000 for spin-off consulting costs through the distribution date.

(d) On September 27, 2010, Daniel M. Hunter was elected by the Board to serve as a director of the Company. Mr. Hunter, age 59, has been a licensed certified public accountant for the past 34 years. Mr. Hunter served as a director of LRAD from May 2001 until March 2010. He obtained his accounting degree from the University of Utah in 1973. For the past 29 years, Mr. Hunter has operated his own law offices specializing in business and tax law. He obtained his J.D. from the University of Seattle in 1978. The Company expects that Mr. Hunter will serve on the Board's Audit and Compensation Committees.

(e) On September 27, 2010, the Company's Board approved the Parametric Sound 2010 Stock Option Plan (the "2010 Plan") and accompanying Stock Option Grant Notice and Stock Option Agreement. The 2010 Plan provides for the granting of options to purchase up to 3,000,000 shares of the Company's common stock to directors, officers, employees and consultants of the Company. Under the 2010 Plan, the granting of options, exercise prices and terms are determined by the Company's Board, or a committee designated by the Board to administer the 2010 Plan. The term of options granted under the 2010 Plan may not exceed 10 years. No options have been granted under the 2010 Plan. This sum mary description of the 2010 Plan is qualified in its entirety by reference to the complete terms and conditions of the 2010 Plan and the Stock Option Grant Notice and Stock Option Agreement under the 2010 Plan filed as Exhibits 10.5 and 10.6 hereto, respectively. In addition, a description of the 2010 Plan is included under the heading, "Executive Compensation—Description of the 2010 Stock Option Plan" beginning on page 48, of the Company's Information Statement, dated September 3, 2010, which is filed as Exhibit 99.1 hereto, and incorporated herein by reference.

Reference is also made to the payments owed to Mr. Norris and Mr. Barnes for prior services through the distribution date as described in (c) above.

# Item 9.01. Financial Statements and Exhibits.

#### (d) Exhibits.

Exhibit	
Number	<u>Description</u>
2.1	Separation and Distribution Agreement, dated September 27, 2010, by and between LRAD Corporation and Parametric Sound Corporation.
4.1	Form of 8% Notes Due September 28, 2011.
4.2	Form of Stock Purchase Warrant, dated September 28, 2010.
10.1	Tax Sharing Agreement, dated September 27, 2010, by and between LRAD Corporation and Parametric Sound Corporation.
10.2	License and Royalty Agreement, dated September 27, 2010, by and between Syzygy Licensing LLC and Parametric Sound Corporation.
10.3	Lease Agreement, dated July 1, 2010, by and between Syzygy Licensing LLC and Davric Corporation, as assumed by Parametric Sound Corporation as of September 27, 2010.
10.3.1	Assignment and Assumption of Lease dated September 27, 2010 between Syzygy Licensing LLC and the Company.
10.4	Form of Subscription Agreement, dated September 28, 2010.
10.5	Parametric Sound Corporation 2010 Stock Option Plan.
10.6	Form of Stock Option Grant Notice and Stock Option Agreement under the 2010 Stock Option Plan.
10 5	

- 10.7 Form of Indemnification Agreement dated September 27, 2010.
- 99.1 Information Statement, dated September 15, 2010.

# SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this Report to be signed on its behalf by the undersigned thereunto duly authorized.

Date: October 1, 2010

Parametric Sound Corporation

By: /s/ James A. Barnes James A. Barnes Chief Financial Officer, Treasurer and Secretary

#### Exhibit 2.1

# SEPARATION AND DISTRIBUTION AGREEMENT

THIS SEPARATION AND DISTRIBUTION AGREEMENT (the "Agreement"), dated as of September 27, 2010, is entered into by and between LRAD Corporation, a Delaware corporation ("LRAD"), and Parametric Sound Corporation, a Nevada corporation ("Parametric") (each, a "Party," and collectively, the "Parties").

#### RECITALS

WHEREAS, the Board of Directors of LRAD has determined that it is in the best interests of LRAD and its stockholders to separate its HyperSonic Sound ("HSS") business through a tax-free spinoff transaction (the "Separation");

WHEREAS, in order to effect the Separation, the Board of Directors of LRAD has determined that it is in the best interests of LRAD and its stockholders to distribute to its stockholders, on a pro rata basis, all of the Parametric Common Stock (the "Distribution"); and

WHEREAS, the Parties intend in this Agreement to set forth the principal corporate arrangements between the Parties with respect to the Separation and the Distribution;

NOW, THEREFORE, in consideration of the foregoing and the terms, conditions, covenants and provisions of this Agreement, LRAD and Parametric mutually covenant and agree as follows:

#### ARTICLE 1 DEFINITIONS

Section 1.1. General. As used in this Agreement, the following capitalized terms shall have the following meanings:

"Action" means any demand, action, suit, countersuit, arbitration, inquiry, proceeding or investigation by or before any federal, state, local, foreign or international Governmental Authority or any arbitration or mediation tribunal.

"<u>Affiliate</u>" means, when used with respect to a specified Person, a Person that directly or indirectly, through one or more intermediaries, controls, is controlled by or is under common control with such specified Person, including, without limitation, a Subsidiary (as defined below). As used herein, "control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through ownership of voting securities or other interests, by contract or otherwise.

"Agent" shall have the meaning set forth in Section 3.1(a).

"Ancillary Agreements" means all of the agreements, instruments, assignments or other arrangements entered into in connection with the transactions contemplated hereby.

"<u>Assets</u>" means assets, properties, claims and rights (including goodwill), wherever located (including in the possession of vendors or other Third Parties or elsewhere on behalf of the owner), of every kind, character and description, whether real, personal or mixed, tangible, intangible or contingent, in each case whether or not recorded or reflected or required to be recorded or reflected on the records or financial statements of any Person.

"Combined Books and Records" shall have the meaning set forth in Section 5.1(c).

"Commission" means the United States Securities and Exchange Commission or any successor agency thereto.

"Consents" means any consents, waivers or approvals from, or notification requirements to any Third Parties.

"<u>Contract</u>" means any contract, obligation, indenture, agreement, lease, purchase order, commitment, permit, license, note, bond, mortgage, arrangement or undertaking (whether written or oral and whether express or implied) that is legally binding on any Person or any part of its property under applicable Law, but excluding this Agreement and any Ancillary Agreement save as otherwise expressly provided in this Agreement or any Ancillary Agreement.

"Distribution" shall have the meaning set forth in the recitals hereto.

"Distribution Date" means the date upon which the Distribution shall be effective, as determined by the Board of Directors of LRAD.

"Effective Time" means 11:59 p.m. EDT on the Distribution Date at which time the Distribution is effective.

"Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations of the Commission thereunder, all as the same shall be in effect at the time that reference is made thereto.

"Excluded Assets" means any and all Assets of LRAD not expressly transferred to Parametric pursuant to this Agreement, including without limitation the Assets set forth on Schedule 1.1 hereto under the caption "Excluded Assets".

"Form 10" means the registration statement on Form 10 filed by Parametric with the Commission relating to the Parametric Common Stock, as amended from time to time.

"Governmental Approvals" means any notices, reports or other filings to be made, or any consents, registrations, approvals, permits or authorizations to be obtained from, any Governmental Authority.

"Governmental Authority" means any federal, state, local, foreign or international court, government department, commission, board, bureau, agency, official or other regulatory, administrative or governmental authority.

"HSS" shall have the meaning set forth in the Recitals.

"Indemnifying Party" shall have the meaning set forth in Section 4.7(c).

"Indemnitee" shall have the meaning set forth in Section 4.7(c).

"Information" means information, whether or not patentable or copyrightable, in written, oral, electronic or other tangible or intangible forms, stored in any medium, including without limitation, studies, reports, records, books, contracts, instruments, surveys, discoveries, ideas, concepts, know-how, techniques, designs, specifications, drawings, blueprints, diagrams, models, prototypes, samples, flow charts, data, computer data, disks, diskettes, tapes, computer programs or other software, marketing plans, customer names, communications by or to attorneys (including attorney-client privileged communications), memos and other materials prepared by attorneys or under their direction (including attorney work product), communications and materials otherwise related to or made or prepared in connection with or in preparation for any legal proceeding, and other technical, financial, employee or business information or data.

"Intellectual Property" means all intellectual property and industrial property rights of any kind or nature, including all United States and foreign (i) patents, patent applications, patent disclosures, and all related continuations, continuations-in-part, divisionals, reissues, re-examinations, substitutions and extensions thereof, (ii) trademarks and all goodwill associated therewith, (iii) copyrights and copyrightable subject matter, whether statutory or common law, registered or unregistered and published or unpublished, (iv) rights of publicity, (v) moral rights and rights of attribution and integrity, (vi) rights in Software, (viii) trade secrets and all other confidential and proprietary information, know-how, inventions, improvements, processes, formulae, models and methodologies, (viii) rights to personal information, (ix) telephone numbers and internet protocol addresses, (x) applications and registrations for the foregoing, and (xi) rights and remedies against past, present, and future infringement, misappropriation, or other violation of the foregoing.

"Law" means any federal, national, state, provincial, local or similar statute, law, ordinance, regulation, rule, code, order, requirement or rule of law (including common law).

"Liabilities" means any and all debts, liabilities, and obligations, whether accrued or fixed, known or unknown, absolute or contingent, matured or unmatured, reserved or unreserved of any kind or nature whatsoever.

# "LRAD Books and Records" shall have the meaning set forth in Section 5.1(b).

"LRAD Business" means all of the business and operations of LRAD and its Subsidiaries other than the Parametric Business.

"LRAD Common Stock" means the Common Stock, \$0.00001 par value per share, of LRAD.

"LRAD Liabilities" means the Liabilities of LRAD, which, for the avoidance of doubt, shall not include any Parametric Liabilities.

"LRAD Policies" means all insurance policies and insurance Contracts of any kind (other than life and benefits policies or Contracts), together with the rights, benefits and privileges thereunder, current or past, which are owned or maintained by LRAD under which HSS products are covered.

#### "Operations Data" shall have the meaning set forth in Section 5.2.

"Parametric Assets" means (a) any and all Assets reflected in the Parametric Balance Sheet, and any and all Intellectual Property used primarily in the Parametric Business, which have not been disposed of or removed from the Parametric Balance Sheet between the date of the Parametric Balance Sheet and the Distribution Date, (b) any and all Assets acquired by Parametric (or LRAD on behalf of Parametric) after the date of the Parametric Balance Sheet that would be reflected in the balance sheet of Parametric as of the Distribution Date, if such balance sheet was prepared by Parametric in accordance with the same accounting principles under which the Parametric Balance Sheet was prepared, and (c) any other Assets expressly set forth on <u>Schedule 1.1</u> hereto under the caption "Parametric Bassets". For the avoidance of doubt, (a) Parametric Assets shall not include any cash or cash equivalents, accounts receivable (whether related to the Parametric business or otherwise) or any net operating losses, research and development credits or other tax attributes of LRAD and its Subsidiaries whether or not relating to the Parametric Business, (b) Parametric Assets shall not include any Excluded Assets and (c) Parametric Assets shall not include any Excluded Assets and (c) Parametric Assets shall not include any Retained HSS Assets until assigned in accordance with <u>Section 2.2</u>.

"Parametric Balance Sheet" means the [unaudited] balance sheet of Parametric as of March 31, 2010 included in the Form 10 and attached hereto as Schedule 1.2.

"Parametric Books and Records" shall have the meaning set forth in Section 5.1(a).

"Parametric Business" means all of the business and operations of the HSS technology business of LRAD as described in the Form 10.

"Parametric Common Stock," means the Common Stock, \$0.001 par value per share, of Parametric.

"Parametric Liabilities" means (a) any and all Liabilities reflected in the Parametric Balance Sheet, (b) any and all Liabilities resulting from or accrued in the operation of the Parametric Business after the date of the Parametric Balance Sheet that would be reflected in the balance sheet of Parametric as of the Distribution Date if such balance sheet was prepared by Parametric in accordance with the same accounting principles under which the Parametric Balance Sheet was prepared, (c) any and all Liabilities asserted before or after the Distribution Date relating to, arising out of, or resulting from any Parametric relating to, arising out of the Parametric's performance or obligations under any Ancillary Agreement or this Agreement, and (e) any and all Liabilities that are expressly contemplated by this Agreement or any Ancillary Agreement to be transferred to Parametric.

"<u>Party</u>" shall have the meaning set forth in the preamble hereof.

"Person" means any natural person, firm, individual, corporation, business trust, joint venture, association, company, limited liability company, partnership or other organization or entity, whether incorporated or unincorporated, or any governmental entity.

"Record Date" means the close of business on September 10, 2010, the date determined by the LRAD Board of Directors as the record date for the Distribution.

"Retained HSS Assets" shall mean the Assets expressly set forth on Schedule 1.1 hereto under the caption "Retained HSS Assets".

"Retained HSS Liabilities" shall mean all Parametric Liabilities incurred prior to the Distribution Date, including but not limited to any vendor liabilities, warranty liabilities and payroll or other liabilities, and any Liabilities for or resulting from the HSS business retained by LRAD pursuant to Section 2.2.

"Separation" shall have the meaning set forth in the recitals hereto.

"<u>Software</u>" means all computer programs (whether in source code, object code or other form), algorithms, databases and data, and technology supporting the foregoing, and all documentation, including flowcharts and other logic and design diagrams, technical, functional and other specifications, and user manuals and training materials related to any of the foregoing.

"Subsidiary." means any corporation or other organization of which at least a majority of the securities or interests have voting power to elect at least a majority of the board of directors or others performing similar functions with respect to such corporation or other organization is directly or indirectly owned or controlled by a Person.

"Third Party" means any Person other than LRAD, Parametric or Affiliate thereof.

"Third Party Claim" shall have the meaning set forth in Section 4.7(c).

## "Transfer" shall have the meaning set forth in Section 2.1(a).

Section 1.2. <u>References</u>; <u>Interpretation</u>. References in this Agreement to the singular shall include the plural and vice versa and words of one gender shall include the other gender as the context requires. The terms "hereof," "herein," and "herewith" and words of similar import shall, unless otherwise stated, be construed to refer to this Agreement as a whole (including all of the Schedules and Exhibits hereto) and not to any particular provision of this Agreement. Article, Section, Exhibit and Schedule references are to the Articles, Sections, Exhibits and Schedules to this Agreement unless otherwise specified. The word "including" and words of similar import when used in this Agreement (or the applicable Ancillary Agreement) means "including, without limitation," unless the context otherwise requires or unless otherwise specified. The word "or" shall not be exclusive.

# ARTICLE 2 THE SEPARATION

#### Section 2.1. Transfer of Assets

(a) As of the date hereof and with effect immediately prior to the Effective Time, LRAD hereby transfers, contributes, assigns, distributes and conveys to Parametric all of LRAD's right, title and interest in and to the Parametric Assets (the "<u>Transfer</u>") pursuant to, and so as to qualify as a contribution to capital under, section 351 of the Internal Revenue Code of 1986, as amended. Parametric hereby accepts such Transfer from LRAD, effective concurrently therewith.

(b) Except as otherwise specifically set forth in this Agreement or any Ancillary Agreement, Parametric shall not accept, assume or be responsible for any Parametric Liabilities incurred by LRAD prior to or after the Distribution Date.

(c) If at any time (whether prior to or after the Effective Time) either Party hereto shall receive or otherwise possess an Asset that is allocated to any other Person pursuant to this Agreement or any Ancillary Agreement, such Party shall promptly transfer or cause to be transferred, at such Party's expense, for no additional consideration, such Asset, including any and all economic benefits generated from such Asset after the Effective Time, to such Party hereto.

(d) In furtherance of the Transfer, and simultaneously with the execution and delivery of this Agreement LRAD shall execute and deliver, and shall cause its Affiliates to execute and deliver, such bills of sale, stock powers, certificates of title, assignments of contracts and other instruments of transfer, conveyance and assignment as and to the extent necessary to evidence the Transfer.

# (e) LRAD shall be responsible for all Retained HSS Liabilities.

Section 2.2. <u>Retention and Use of Certain HSS Assets</u>. Notwithstanding the foregoing, the Retained HSS Assets will not be assigned by LRAD to Parametric under <u>Section 2.1</u>. Following the Distribution Date, LRAD shall have the right to use the Retained HSS Assets on a royalty-free basis to (a) to fill any orders for such Retained HSS Assets that are received prior to the Distribution Date, (b) accept and fill continuing purchase orders for LRAD's current project with Cardinal Health, Inc. through the completion of such project, (c) accept unsolicited follow-on orders for LRAD's H450 product from current customers through December 31, 2010; provided that LRAD may not actively pursue new H450 or HSS business after the Distribution Date, and (d) provide warranty services for its H450 customers at its cost through any applicable warranty term. All Retained HSS Assets shall be automatically assigned to Parametric as of and when determined by LRAD to be unnecessary to meet the foregoing requirements. LRAD agrees that it will sell such additional HSS products in the ordinary course and without discounts in excess of 25% of its current sales prices.

#### Section 2.3. Governmental Approvals; Consents.

(a) To the extent that the Separation requires any Governmental Approvals or Consents, the Parties shall use commercially reasonable efforts to obtain any such Governmental Approvals and Consents. If and to the extent that the transfer or assignment of any Parametric Assets would violate applicable Law or require any Governmental Approval or Consent, then, unless LRAD shall otherwise determine, the transfer or assignment of such Parametric Assets shall be deemed deferred and any such purported transfer or assignment shall be null and void until such time as all legal impediments are removed and/or each of such Governmental Approval or Consent has been obtained. No Party shall be obligated to pay any consideration to any Third Party from whom any such Consent is requested.

(b) If the transfer of any Parametric Assets intended to be transferred hereunder is not consummated prior to the Effective Time for any reason, then LRAD shall thereafter hold such Parametric Asset for the use and benefit of Parametric entitled thereto if permitted by law. If and when the Governmental Approvals or Consents, or other impediments to transfer, that caused the deferral of transfer of such Asset are obtained or removed, the transfer of the applicable Asset shall be effected in accordance with the terms of this Agreement and/or the applicable Ancillary Agreement.

(c) If the Parties are unable to obtain any required Governmental Approvals or Consents, LRAD shall continue to be bound by such Contract, license or other obligation; provided, however, that LRAD shall not be obligated to extend, renew or otherwise cause such Contract, license or other obligation to remain in effect beyond the term in effect as of the Effective Time. If and when any such Governmental Approval or Consent is obtained or such agreement, lease, license or other rights or obligations shall otherwise become assignable or capable of novation, LRAD shall promptly assign, or cause to be assigned, all rights, obligations and other Liabilities thereunder to Parametric without payment of any further consideration and Parametric, without the payment of any further consideration, shall assume such rights and obligations and other Liabilities.

Section 2.4. <u>Termination of Agreements</u>. Except with respect to this Agreement and the Ancillary Agreements (and agreements expressly contemplated herein or therein to survive by their terms) on behalf of the Parties, the Parties hereby terminate any and all written or oral agreements, arrangements, commitments or understandings, between them, effective as of the Effective Time; and each Party shall, at the reasonable request of the other Party, take, or cause to be taken, such other actions as may be necessary to effect the foregoing.

Section 2.5. <u>Disclaimer of Representations and Warranties</u>. ON BEHALF OF THE PARTIES, THE PARTIES UNDERSTAND AND AGREE THAT, EXCEPT AS EXPRESSLY SET FORTH HEREIN OR IN ANY ANCILLARY AGREEMENT, NO PARTY TO THIS AGREEMENT OR ANY OTHER AGREEMENT OR DOCUMENT CONTEMPLATED BY THIS AGREEMENT, IS REPRESENTING OR WARRANTING IN ANY WAY AS TO THE ASSETS, BUSINESSES OR LIABILITIES CONTRIBUTED, TRANSFERRED, DISTRIBUTED OR ASSUMED HEREBY OR THEREBY, AS TO ANY CONSENTS OR GOVERNMENTAL APPROVALS REQUIRED IN CONNECTION HEREWITH OR THEREWITH, AS TO THE VALUE OR FREEDOM FROM ANY SECURITY INTERESTS OF, OR ANY OTHER MATTER CONCERNING, ANY ASSETS OF SUCH PARTY, OR AS TO THE ABSENCE OF ANY DEFENSES OR RIGHT OF SETOFF OR FREEDOM FROM COUNTERCLAIM WITH RESPECT TO ANY CLAIM OR OTHER ASSET OF ANY PARTY, OR AS TO THE LEGAL SUFFICIENCY OF ANY CONTRIBUTION, DISTRIBUTION, ASSIGNMENT, DOCUMENT, CERTIFICATE OR INSTRUMENT DELIVERED HEREUNDER TO CONVEY TITLE TO ANY ASSET OR THING OF VALUE UPON THE EXECUTION, DELIVERY AND FILING HEREOF OR THEREOF. EXCEPT AS MAY EXPRESSLY BE SET FORTH HEREIN OR IN ANY ANCILLARY AGREEMENT, ALL SUCH ASSETS ARE BEING TRANSFERRED ON AN "AS IS," "WHERE IS" BASIS AND, THE TRANSFEREE SHALL BEAR THE ECONOMIC AND LEGAL RISKS THAT (I) ANY OCOVEYANCE SHALL PROVE TO BE INSUFFICIENT TO VEST IN THE TRANSFEREE GOOD AND MARKETABLE TITLE, FREE AND CLEAR OF ANY SECURITY INTEREST, AND (II) ANY NECESSARY CONSENTS OR GOVERNMENTAL APPROVALS ARE NOT OBTAINED OR THAT THE REQUIREMENTS OF LAWS, CONTRACTS, OR JUDGMENTS ARE NOT COMPLIED WITH.

## Section 3.1. The Distribution.

(a) Subject to Section 3.3. on or prior to the Distribution Date, for the benefit of and distribution to the holders of LRAD Common Stock on the Record Date, LRAD will deliver stock certificates, endorsed by LRAD in blank, to the distribution agent, Interwest Transfer Company (the "<u>Agent</u>"), representing all of the outstanding and issued shares of Param etric Common Stock then owned by LRAD. LRAD shall instruct the Agent to electronically distribute the appropriate number of such shares of Parametric Common Stock to each such holder or designated transferees of such holder.

(b) Subject to Section 3.3, each holder of LRAD Common Stock on the Record Date (or such holder's designated transferee or transferees) will be entitled to receive in the Distribution as of the Effective Time one (1) share of Parametric Common Stock for each two (2) shares of LRAD Common Stock held of record on the Record Date. No fractional shares will be issued. In the event a holder of LRAD Common Stock would otherwise be entitled to a fractional share of Parametric Common Stock, the number of shares of Parametric Common Stock such holder of LRAD Common Stock would otherwise be entitled to a fractional share of Parametric Common Stock. No action by any such stockholder shall be necessary for such stockholder (or such stockholder's designated transferee) to receive the shares of Parametric Common Stock. Parametric and LRAD, as the case may be, will provide to the Agent any and all information required in order to complete the Distribution.

# Section 3.2. Actions in Connection with the Distribution.

(a) Prior to the Distribution Date, LRAD and Parametric shall have prepared and mailed to the holders of LRAD Common Stock such information concerning Parametric, the Parametric Business, operations and management, the Distribution, the Separation and such other matters as LRAD shall reasonably determine and as may be required by law.

(b) Parametric shall have prepared and, in accordance with applicable Law, filed with the Commission the Form 10, including amendments, supplements and any such other documentation which is necessary or desirable to effectuate the Distribution, and Parametric shall have obtained all necessary approvals from the Commission with respect thereto as soon as practicable.

(c) LRAD and Parametric shall take all such action as may be necessary or appropriate under the securities or blue sky laws of the states or other political subdivisions of the United States or of other foreign jurisdictions in connection with the Distribution.

#### (d) DELETED.

(e) Parametric shall have prepared and filed an application for the original listing on the OTC Bulletin Board of the Parametric Common Stock to be distributed in the Distribution, and the OTC Bulletin Board shall have approved such application and provided the appropriate certification to the Commission.

(f) LRAD and Parametric shall take all reasonable steps necessary to cause the conditions set forth in Section 3.3 to be satisfied and to effect the Distribution on the Distribution Date.

Section 3.3. <u>Conditions to Distribution</u>. Subject to <u>Section 3.2</u>, the following are conditions to the consummation of Distribution. The conditions are for the sole benefit of LRAD and shall not give rise to or create any duty on the part of LRAD or the Board of Directors of LRAD to waive or not waive any such condition:

(a) The Form 10 shall have been declared effective by the Commission, with no stop order in effect with respect thereto.

(b) All material Government Approvals and other Consents necessary to consummate the Distribution shall have been obtained and be in full force and effect.

(c) No order, injunction or decree issued by any court or agency of competent jurisdiction or other legal restraint or prohibition preventing the consummation of the Distribution shall be in effect and no other event outside the control of LRAD shall have occurred or failed to occur that prevents the consummation of the Distribution.

(d) The Board of Directors of LRAD shall have authorized and approved the Distribution and not withdrawn such authorization and approval.

(e) No other events or developments shall have occurred that, in the sole discretion of the Board of Directors of LRAD, would result in the Distribution having a material adverse effect on LRAD or on the stockholders of LRAD or not being in the best interest of LRAD and its stockholders.

## ARTICLE 4 COVENANTS

Section 4.1. <u>Solicitation and Hiring</u>. Neither LRAD nor Parametric from the Distribution Date through and including the two (2) year anniversary of the Distribution Date, without prior written consent of the other Party, will solicit, aid, encourage or induce any employee to terminate or breach an employment, contractual or other relationship with the other Party (or its Affiliates), hire or otherwise employ any employee of the other Party; provided, however, that Parametric shall be permitted to hire David Pratt, and provided, further, that nothing in this <u>Section 4.1</u> shall be deemed to prohibit any general solicitation for employment through advertisements and search firms not specifically directed at employees of such other applicable Party.

Section 4.2. <u>Warrant Exercises</u>. If any warrants to purchase shares of capital stock of LRAD that are outstanding on the Distribution Date are exercised following the Distribution Date, then Parametric will issue to the holder of such warrants such number of shares of Parametric Common Stock as is required by the applicable warrant. If the closing price per share of LRAD Common Stock on the date such warrants are exercised is less than the exercise price of such warrants, then LRAD will pay to Parametric an amount equal to the number of shares of LRAD Common Stock issued upon such exercise multiplied by the difference between the exercise price and the closing price per share of LRAD Common Stock on the date the warrants are exercised, up to a maximum of \$0.25 per share of LRAD Common Stock issued. For example if the closing price of LRAD Common Stock on the date of exercise is \$2.50 and warrants to purchase 1,000,000 shares are exercised then LRAD will retain \$2.50 per exercised share (or \$2,500,000) and will pay to Parametric \$0.17 per exercised share of LRAD Common Stock (or \$170,000) for the 500,000 shares Parametric will be required to issue. Section 4.3. <u>Post Closing Cooperation</u>. LRAD agrees for a reasonable time following the Distribution Date to provide to Parametric reasonable training regarding use of certain of the Parametric Assets and manufacturing parts or elements of the HSS products. Parametric agrees to permit LRAD to use the manufacturing equipment that is part of the Parametric Assets until December 31, 2010 to manufacture, using components and parts owned by LRAD, the additional H450 products that LRAD may require to fulfill orders and warranty obligations as permitted hereunder. In addition to the foregoing, each of the Parties agrees to cooperate with the other P arty in good faith to provide for the manufacturing needs of such other Party. Any additional terms and conditions of such cooperation shall be determined by the mutual, good faith agreement of the Parties.

#### Section 4.4. Insurance.

(a) Following the cessation of coverage of HSS products under an LRAD Policy, if (i) an occurrence for which coverage is available under any such LRAD Policy happens prior to the date of the last delivery by LRAD of an HSS product in accordance with this Agreement and (ii) a claim arising therefrom has been or is eventually asserted against Parametric (including any officer, director, employee or agent thereof), so long as such claim is reported by Parametric to the carrier (with a copy to LRAD), in accordance with the reporting provision of the applicable policy, then LRAD will (A) continue to provide Parametric in submitting such claims under the applicable LRAD Policies, provided that Parametric shall be responsible for its portion of any deductibles or self-insured retentions or co-payments legally due and owing relating to such claims. For the avoidance of doubt, if an occurrence for which coverage is available under any such LRAD Policy happens after the Effective Time (and is not attributable and related to an occurrence which occurred prior to the Effective Time), or a claim arising from an occurrence prior to the Effective Time is not reported by Parametric to LRAD on or before the date when such occurrence must be reported to the carrier under the applicable LRAD Policy, then no payment for any damages, costs of defense, or other sums with respect to such claim shall be available to Parametric shall be take covering its operations and products following the Distribution Date.

(b) With respect to all LRAD Policies, Parametric agrees and covenants (on behalf of itself and each Affiliate of Parametric) (i) not to make any claim or assert any rights against LRAD, or the unaffiliated third-party insurers of such LRAD Policies, except as expressly provided under this <u>Section 4.2</u>, and (ii) to otherwise reasonably cooperate with LRAD and take commercially reasonable actions as may be necessary or advisable to assist LRAD in fulfilling its obligations under the applic able LRAD Policies as set forth in this <u>Section 4.2</u>.

## Section 4.5. Confidentiality.

(a) Notwithstanding any termination of this Agreement and subject to Section 4.5(c), for a period of two (2) years from the Distribution Date, each Party agrees to hold, and to cause its respective Affiliates, directors, officers, employees, agents, accountants, counsel and other advisors and representatives to hold, in strict confidence, and undertake all reasonable precautions to safeguard and protect the confidentiality of, all Information concerning the other Party that is in its possession after the Distribution Date or furnished by the other Party or its respective directors, officers, employees, agents, accountants, counsel and other advisors and representatives to this Agreement, any Ancillary Agreement or otherwise, and shall not use any such Information other than for such purposes as shall be expressly permitted hereunder or thereunder, except, in each case, to the extent that such Information has been (i) in the public domain through no fault of such Party or any of their respective directors, officers, employees, agents, accountants, counsel and other advisors and representatives (ii) lawfully acquired from other sources, which are not bound by a confidentiality obligation, by such Party, or (iii) independently generated without reference to any proprietary or conf idential Information of the other Party.

(b) Each Party agrees not to release or disclose, or permit to be released or disclosed, any such Information to any other Person, except its directors, officers, employees, agents, accountants, counsel and other advisors and representatives who need to know such Information and who are informed and advised that the Information is confidential and subject to the obligations hereunder, except in compliance with <u>Section 4.5(c)</u>. Without limiting the foregoing, when any Information is no longer needed for the purposes contemplated by this Agreement or any Ancillary Agreement, each Party will promptly after request of the other Party either (i) destroy all copies of the Information in such Party's possession, custody or control (including any that may be stored in any computer, word processor, or similar device, to the extent not commercially impractical to destroy such copies) including, without limitation, any copies, summaries, analyses, reports, extracts or other reproductions, in whole or in part, of such written, electronic or other tangible material or any other materials in written, electronic or other tangible material prepared by such Party, and/or (ii) return to the requesting Party, at the expense of the requesting Party, all copies of the Information furnished to such Party by or on behalf of the requesting Party.

(c) In the event that either Party (i) determines after consultation with counsel, in the opinion of such counsel that it is required by law to disclose any Information or (ii) receives any demand under lawful process or from any Governmental Authority to disclose or provide Information of the other Party that is subject to the confidentiality provisions hereof, such Party shall notify the other Party prior to disclosing or providing such Information and shall cooperate at the expense of the requesting party (and to the extent legally permissible) in seeking any reasonable protective arrangements requested by such other Party. Subject to the foregoing, the Party that received such request may thereafter (1) furnish only that portion of the confidential Information that is legally re quired, (2) give notice to the other Party of the information to be disclosed as far in advance as is practical, and (3) exercise reasonable best efforts to obtain reliable assurance that the confidential nature of such Information shall be maintained.

#### Section 4.6. Litigation cooperation.

(a) Each of LRAD and Parametric agrees that at all times from and after the Effective Time, if an Action currently exists or is commenced by a third-party with respect to which a Party is a named defendant but such Action is otherwise not a Liability allocated to such named Party under this Agreement or any Ancillary Agreement, then the other Party shall use commercially reasonable efforts to cause such named but not liable defendant to be removed from such Action and such defendant shall not be required to make any payments or contribution in connection therewith.

(b) Each of LRAD and Parametric shall each use commercially reasonable efforts to make available to the other, upon written request, its officers, directors, employees and agents, and the officers, directors, employees and agents thereof, as witnesses to the extent that such individuals may reasonably be required in connection with any legal, administrative or other proceedings in which the requesting Party may be involved. The requesting Party shall bear all out-of-pocket expenses in connection therewith. On and after the Effective Time, in connection with any matter contemplated by this <u>Section 4.6</u>, the Parties will maintain any attorney-client privilege or work product immunity of each Party as required by this Agreement or any Ancillary Agreement.

## Section 4.7. Indemnification

(a) Except as otherwise provided in this Agreement or any Ancillary Agreement, following the Distribution Date, Parametric shall indemnify, defend and hold harmless LRAD and its Affiliates, including each of their respective directors and officers, and each of the heirs, executors, successors and assigns of any of the foregoing (collectively, the "<u>LRAD Indemnitees</u>"), from and against any and all Liabilities and related losses of the LRAD Indemnites relating to, arising out of or resulting from any of (a) the failure of Parametric or its Affiliates or any other Person to pay, perform or otherwise promptly discharge after the Distribution Date any Parametric Liab ilities incurred after the Distribution in accordance with their respective terms, (b) any untrue statement, alleged untrue statement, omission or alleged omission of a material fact in the Form 10, resulting in a misleading statement, with respect to all information contained in the Form 10, and (c) any breach by Parametric of this Agreement or any of the Ancillary Agreements.

(b) Except as otherwise provided in this Agreement or any Ancillary Agreement, following the Distribution Date, LRAD shall indemnify, defend and hold harmless Parametric, and its Affiliates, including each of their respective directors and officers, and each of the heirs, executors, successors and assigns of any of the foregoing (collectively, the "Parametric Indemnitees"), from and against any and all Liabilities and related losses of the Parametric Indemnitees relating to, arising out of or resulting from any of (a) the failure of LRAD or its Affiliates to pay, perform or otherwise promptly discharge after the Distribution Date any LRAD Liabilities, (b) the LRAD Liabilities, (c) the Retained HSS Liabilities, and (d) any breach by LRAD of this Agreement or any of the Ancillary Agreements.

(c) If a party entitled to indemnification hereunder (an "<u>Indemnitee</u>") shall receive notice or otherwise learn of the assertion by a Third Party (including any Governmental Authority) of any claim or of the commencement by any such Person of any Action (collectively, a "<u>Third Party Claim</u>") with respect to which a party required to provide indemnification hereunder (an "Indemnifying Party") may be obligated to provide indemnification to such Indemnitee, such Indemnitee shall give such Indemnifying Party and each Party to this Agreement written notice thereof as soon as reasonably practicable, but no later than thirty (30) days after becoming aware of such Third Party Claim. Any such notice shall describe the Third Party Claim in reasonable detail. If any Party shall receive notice or otherwise learn of the assertion of a Third Party Claim which may reasonably be determined to be a Liability of the Party shall give the other Party to this Agreement written notice thereof within thirty (30) days after becoming aware of such Third Party Claim. Any such notice shall describe the Third Party Claim in reasonable detail. If any Party shall receive notice or otherwise learn of the assertion of a Third Party Claim Written notice thereof within thirty (30) days after becoming aware of such Third Party Claim. Any such notice shall describe the Third Party Claim in reasonable detail. Notwithstanding the foregoing, the failure of any Indemnitee or other Party to give notice as provided in this <u>Section 4.7(c)</u> shall not relieve the related Indemnifying Party of its obligations under this <u>ARTICLE 4</u>, except to the extent that such Indemnifying Party is actually prejudiced by such failure to give notice.

(i) An Indemnifying Party shall be entitled to participate in the defense of any Third Party Claim, at such Indemnifying Party's own expense and by such Indemnifying Party's own counsel; provided that if the defendants in any such claim include both the Indemnifying Party and one or more Indemnitees and in such Indemnitees' reasonable judgment a conflict of interest between such Indemnifying Party exists in respect of such claim, such Indemnifying Party) shall be paid by such Indemnifying Party. Within thirty (30) days after the receipt of notice from an Indemnitee in accordance with Section 4.7 (or sooner, if the nature of such Third Party Claim so requires), the Indemnifying Party shall notify the Indemnifying such Third Party Claim. After notice from an Indemnifying Party to an Indemnitee of its election to assume the defense of a Third Party Claim, such Indemnitee shall have the right to employ separate counsel and to participate in (but not control) the defense, compromise or settlement thereof, but the fees and expenses of such counsel shall be the expense of such Indemnitee in accordance with Section 4.7 (or sooner, if the nature of other from an Indemnifying Party to an Indemnitee of its election to assume the defense of a Third Party Claim, such Indemnitee shall have the right to employ separate counsel and to participate in (but not control) the defense, compromise or settlement thereof, but the fees and expenses of such counsel shall be the expense of such Indemnitees of such counsel shall be the expense of such Counsel shall be

(ii) With respect to any Third Party Claim, the Indemnifying Party and Indemnitees agree, and shall cause their respective counsel (if applicable), to cooperate fully (in a manner that will preserve all attorney-client privilege or other privileges) to mitigate any such claim and minimize the defense costs associated therewith.

(iii) If an Indemnifying Party fails to assume the defense of a Third Party Claim within thirty (30) days after receipt of written notice of such claim, the Indemnitee will, upon delivering notice to such effect to the Indemnifying Party, have the right to undertake the defense, compromise or settlement of such Third Party Claim on behalf of and for the account of the Indemnifying Party, subject to the limitations as set forth in this <u>Section 4.7(c)</u>; provided, however, that such Third Party Claim shall not be compromised or settled without the written consent of the Indemnifying Party, which consent shall not be unreasonably withheld, delayed or conditioned. If the Indemnitee assumes the defense of any Third Party Claim, it shall keep the Indemnifying Party reasonably informed of the progress of any such defense, compromise or settlement. The Indemnifying Party shall reimburse all such costs and expenses of the Indemnitee in the event it is ultimately determined that the Indemnifying Party is obligated to indemnify the Indemnitee to such Third Party Claim. In no event shall an Indemnifying Party be liable for any settlement effected without its consent, which consent will not be unreasonably withheld, delayed or conditioned.

(d) Any claim on account of a Liability or related loss which does not result from a Third Party Claim shall be asserted by written notice given by the Indemnitee to the related Indemnifying Party. Such Indemnifying Party shall have a period of thirty (30) days after the receipt of such notice within which to respond thereto. If such Indemnifying Party does not respond within such sixty (60) day period, such Indemnifying Party shall be deemed to have accepted responsibility to make payment. If such Indemnifying Party rejects such claim in whole or in part, such Indemnitee shall be free to pursue such remedies as may be available to such party as contemplated by this Agreement and the Ancillary Agreements.

(e) In the event of payment by or on behalf of any Indemnifying Party to any Indemnitee in connection with any Third Party Claim, such Indemnifying Party shall be subrogated to and shall stand in the place of such Indemnitee as to any events or circumstances in respect of which such Indemnitee may have any right, defense or claim relating to such Third Party Claim against any claimant or plaintiff asserting such Third Party Claim or against any other person. Such Indemnitee shall cooperate with such Indemnifying Party in a reasonable manner, and at the cost and expense (including allocated costs of in-house personnel) of such Indemnifying Party, in prosecuting any subrogated right, defense or claim.

(f) In the event of an Action in which the Indemnifying Party is not a named defendant, if the Indemnifying Party shall so request, the Parties shall endeavor to substitute the Indemnifying Party for the named defendant, and add the Indemnifying Party as a named defendant if at all practicable. If such substitution or addition cannot be achieved for any reason or is not requested, the named defendant shall allow the Indemnifying Party to manage the Action as set forth in this section and subject to <u>Section 4.7</u> with respect to Liabilities, the Indemnifying Party shall fully indemnify the named defendant against all costs of defending the Action (including court costs, sanctions imposed by a court, attorneys' fees, experts' fees and all other external expenses, and the allocated costs of in-house coursel and other in-house personnel), the costs of any judgment or settlement.

(g) The rights and obligations of each Party and their respective Indemnitees under this <u>Section 4.7</u> shall survive the sale or other transfer by any Party or its Affiliates of any Assets or businesses or the assignment by it of any and all Liabilities.

# ARTICLE 5 ACCESS TO INFORMATION AND SERVICES

## Section 5.1. Provision of Corporate Records.

(a) Except as otherwise provided in any Ancillary Agreement, upon the prior written request by Parametric for specific and identified books and records which relate to (x) Parametric or the conduct of the Parametric Business, as the case may be, up through the Distribution Date, or (y) any Ancillary Agreement to which Parametric and LRAD are parties (the "<u>Parametric Books and Records</u>"), LRAD shall, at the expense of Parametric, provide for the transport of the Parametric Books and Records in its possession or control to a location provided by Parametric as soon as practicable but no later than thirty (30) days following the date of such request, e xcept to the extent such items are already in the possession of Parametric or a Parametric Affiliate.

(b) Except as otherwise provided in any Ancillary Agreement, upon the prior written request by LRAD for specific and identified books and records which relate to (x) LRAD or the conduct of the LRAD Business, as the case may be, up through the Distribution Date, or (y) any Ancillary Agreement to which Parametric and LRAD are parties (the "<u>LRAD Books and Records</u>"). Parametric shall, at the expense of LRAD, provide the transport of the LRAD Books and Records in its possession or control to a location provided by LRAD as soon as practicable but no later than thirty (30) days following the date of such request, except to the extent such items are alrea dy in the possession of LRAD or a LRAD Affiliate.

(c) With respect to books and records that relate to both the Parametric Business and the LRAD Business (the "<u>Combined Books and Records</u>"), (i) the Parties shall use good faith efforts to divide such Combined Books and Records into Parametric Books and Records and LRAD Books and Records, as appropriate, and (ii) to the extent such Combined Books and Records are not so divided, each Party shall each keep and maintain copies of such Combined Books and Records as reasonably appropriate under the circumstances, subject to applicable confidentiality provisions hereof and of any Ancillary Agreement.

Section 5.2. Access to Information. Except as otherwise provided in an Ancillary Agreement, from and after the Distribution Date, LRAD shall provide Parametric and its authorized accountants, counsel and other designated representatives reasonable access and duplicating rights during normal business hours to all records, books, contracts, instruments, computer data and other data and information relating to pre-Distribution operations of the Parametric Business (collectively, "<u>Operations Data</u>.") within LRAD's possession or control (including using reasonable best efforts to give access to persons or firms possessing information) insofar as such access is reasonably required by Parametric for the conduct of the Parametric Business, subject to appropriate restrictions for classified or privileged information. Similarly, except as otherwise provided in an Ancillary Agreement, Parametric shall provide LRAD and its authorized accountants, counsel and other designated representatives reasonable access (including using reasonable best efforts to give access to persons or firms possession, insofar as such access is reasonably required by LRAD for the conduct of the LRAD and its authorized accountants, counsel and other designated representatives reasonable access (including using reasonable best efforts to give access to persons or firms possession, insofar as such access is reasonably required by LRAD for the conduct of the LRAD Business, subject to appropriate restrictions for Operations Data, within Parametric's possession, insofar as such access is reasonably required by LRAD for the conduct of the LRAD Business, subject to appropriate restrictions for classified or privileged information. Operations Data and other designated the second by required by LRAD for the conduct of the LRAD Business, subject to appropriate restrictions for classified or privileged information. Deprations Data and other descuents may be requested under this <u>ARTICLE 5</u> for the legitimate business purposes of either Party, including, with

Section 5.3. <u>Reimbursement</u>. Except to the extent otherwise contemplated in any Ancillary Agreement, a Party providing Operations Data or witness services to the other Party under this <u>ARTICLE</u> 5 shall be entitled to receive from the recipient, upon the presentation of invoices therefor, payments of such amounts, relating to supplies, disbursements and other ou t-of-pocket expenses (at cost) and direct and indirect expenses of employees who are witnesses or otherwise furnish assistance (at cost), as may be reasonably incurred in providing such Operations Data or witness services.

# ARTICLE 6 FURTHER ASSURANCES

Section 6.1. <u>Further Assurances</u>. In addition to and without limiting the actions specifically provided in this Agreement, each of the Parties hereto shall use its reasonable best efforts, prior to, on and after the Distribution Date, to take, or cause to be taken, all actions, and to do, or cause to be done, all things reasonably necessary, proper or advisable under applicable laws, regulations and agreements to consummate and make effective the transactions contemplated by this Agreement and the Ancillary Agreements. Without limiting the foregoing, prior to, on and after the Distribution Date, each Party hereto shall cooperate with the other Party, and without any fur ther consideration, but at the expense of the requesting Party, to execute and deliver, or use its reasonable best efforts to cause to be executed and delivered, all instruments, including instruments of conveyance, assignment and transfer, and to make all filings with, and to obtain all consents, approvals or authorizations of, any Governmental Authority or any other Person under any permit, license, agreement, indenture or other instrument (including any Consents or Governmental Approvals), and to take all such other actions as such Party may reasonably be requested to take by the other Party hereto from time to time, consistent with the terms of this Agreement and the Ancillary Agreements, in order to effectuate the provisions and purposes of this Agreement and the Ancillary Agreements, in order to effectuate the provisions and purposes of this Agreement and the Ancillary Agreements, in order to effectuate the provisions and purposes of this Agreement and the Ancillary Agreements, in order to effectuate the provisions and purposes of this Agreement and the Ancillary Agreements, in order to effectuate the provisions and purposes of this Agreement and the Ancillary Agreements, in order to effectuate the provisions and purposes of this Agreement and the Ancillary Agreements, in order to effectuate the provisions and purposes of this Agreement a

# ARTICLE 7 TERMINATION

Section 7.1. <u>Termination</u>. Notwithstanding anything to the contrary herein, this Agreement may be terminated and the Distribution may be amended, modified or abandoned at any time prior to the Effective Time by and in the sole discretion of LRAD without the approval of Parametric or the stockholders of LRAD. In the event of such termination, no Party shall have any Liability to the other Party or any other Person. After the Effective Time, this Agreement may not be terminated except by an agreement in writing signed by each of the Parties.

# Section 8.1. Counterparts; Entire Agreement.

(a) This Agreement and each Ancillary Agreement may be executed in one or more counterparts, each of which shall be considered one and the same agreement, and shall become effective when each counterpart has been signed by each of the Parties and delivered to the other Party.

(b) This Agreement, the Ancillary Agreements and the Exhibit and Schedules hereto and thereto contain the entire agreement between the Parties with respect to the subject matter hereof, supersede all previous agreements, negotiations, discussions, writings, understandings, commitments and conversations with respect to such subject matter, and there are no agreements or understandings between the Parties other than those set forth or referred to herein or therein. Except with respect to tax matters, in the event of any conflict between the terms and conditions of this Agreement and the terms and conditions of any Ancillary Agreement, the terms and conditions of this Agreement (including amendments hereto) shall control.

Section 8.2. <u>Governing Law</u>. This Agreement, except as expressly provided herein, and, unless expressly provided therein, each Ancillary Agreement, shall be governed by and construed and interpreted in accordance with the laws of the State of Delaware, irrespective of the choice of laws principles of the State of Delaware as to all matters, including matters of validity, construction, effect, enforceability, performance and remedies.

Section 8.3. <u>Tax Matters</u>. Notwithstanding anything to the contrary in this Agreement, the rights and obligations of the Parties with respect to any and all tax matters shall be exclusively governed by the provisions of the Tax Sharing Agreement, except as set forth therein.

Section 8.4. <u>Assignability</u>. The provisions of this Agreement, each Ancillary Agreement and the obligations and rights hereunder shall be binding upon, inure to the benefit of and be enforceable by (and against) the Parties and their respective successors and permitted transferees and assigns. Notwithstanding the foregoing, this Agreement shall not be assignable, in whole or in part, by any Party without the prior written consent of the other Party, and any attempt to assign any rights or obligations arising under this Agreement without such consent shall be null and void; provided, that (i) a Party may assign this Agreement in connection with a merger transaction in which such Party is not the surviving entity or the sale by such Party of all or substantially all of its Assets, and upon the effectiveness of such assignment the assigning Party shall be released from all of its obligations under this Agreement if the surviving entity of such merger or the transferee of such Assets shall agree in writing, in form and substance reasonably satisfactory to the other Party, to be bound by all terms of this Agreement as if named as a "Party" hereto.

Section 8.5. <u>Third Party Beneficiaries</u>. Except for the indemnification rights under this Agreement of any LRAD Indemnitee or Parametric Indemnitee in their respective capacities as such, (a) the provisions of this Agreement and each Ancillary Agreement are solely for the benefit of the Parties and are not intended to confer upon any Person except the Parties any rights or remedies hereunder, and (b) there are no Third Party beneficiaries of this Agreement or any Ancillary Agreement and neither this Agreement nor any Ancillary Agreement shall provide any Third Party with any remedy, claim, liability, reimbursement, claim of action or other right in excess of those existing without reference to this Agreement or any Ancillary Agreement.

Section 8.6. Notices. All notices, requests, claims, demands or other communications under this Agreement or any Ancillary Agreement shall be in writing and shall be deemed to be duly given when (a) delivered in person or (b) upon receipt after being deposited in the United States mail or private express mail, postage prepaid, addressed as follows:

If to LRAD, to: LRAD Corporation 15378 Avenue of Science, Suite 100 San Diego, California 92128 Attn: Thomas R. Brown Facsimile: (858) 676-1112 If to Parametric, to: Parametric Sound Corporation 1941 Ramrod Avenue, Suite 100 Henderson, Nevada 89014 Attn: Elwood G. Norris Facsimile: (888) 639-2150

Either Party may, by notice to the other Party, change the address to which such notices are to be given.

Section 8.7. <u>Severability</u>. If any provision of this Agreement or any Ancillary Agreement or the application thereof to any Person or circumstance is determined by a court of competent jurisdiction to be invalid, void or unenforceable, the remaining provisions hereof or thereof, or the application of such provision to Persons or circumstances or in jurisdictions other than those as to which it has been held invalid or unenforceable, shall remain in full force and effect and shall in no way be affected, impaired or invalidated thereby, so long as the economic or legal substance of the transactions contemplated hereby or thereby, as the case may be, is not affected in an y manner adverse to any Party. Upon such determination, the Parties shall negotiate in good faith in an effort to agree upon such a suitable and equitable provision to affect the original intent of the Parties.

Section 8.8. <u>Publicity</u>. Prior to the Distribution, each of Parametric and LRAD shall consult with each other prior to issuing any press releases or otherwise making public statements with respect to the Separation, the Distribution or any of the other transactions contemplated hereby and prior to making any filings with any Governmental Authority with respect thereto.

Section 8.9. <u>Expenses</u>. Except as expressly set forth in this Agreement or in any Ancillary Agreement, whether or not the Separation or the Distribution is consummated, all Third Party fees, costs and expenses paid or incurred in connection with the Separation and Distribution will be advanced by Elwood G. Norris. All such fees, costs and expenses so advanced shall be repaid by Parametric to Mr. Norris following the Distribution. Notwithstanding the foregoing, in the event LRAD unilaterally determines not to proceed with the Separation and Distribution (except as a result of the Commission not declaring the Form 10 effective or otherwise not provi ding any consent required), then LRAD shall reimburse Mr. Norris the amounts paid by Mr. Norris for expenses incurred by LRAD in connection with the Separation and Distribution.

Section 8.10. <u>Headings</u>. The article, section and paragraph headings contained in this Agreement and in the Ancillary Agreements are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement or any Ancillary Agreement.

Section 8.11. <u>Survival of Covenants</u>. Except as expressly set forth in any Ancillary Agreement, all covenants, representations and warranties contained in this Agreement and each Ancillary Agreement, and liability for the breach of any obligations contained herein, shall survive after the Distribution Date and remain in full force and effect in accordance with their applicable terms.

Section 8.12. Waivers of Default. The failure of either Party to require strict performance by the other Party of any provision in this Agreement or any Ancillary Agreement will not waive or diminish such Party's right to demand strict performance thereafter of that or any other provision hereof.

Section 8.13. <u>Specific Performance</u>. The Parties agree that irreparable damage would occur in the event that the provisions of this Agreement were not performed in accordance with their specific terms. Accordingly, it is hereby agreed that the Parties shall be entitled, in addition to any other remedy or relief to which they may be entitled, to injunctive relief (including provisional or temporary injunctive relief) to enforce specifically the terms and provisions hereof and enforcement of any such award of an arbitral tribunal in any court of the United States, or any other any court or tribunal sitting in any state of the United States or in any foreign country that has ju risdiction.

# Section 8.14. Amendments. This Agreement may not be modified or amended except by an agreement in writing signed by each of the Parties.

Section 8.15. <u>Waiver of Jury Trial</u>. EACH OF THE PARTIES HEREBY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY WITH RESPECT TO ANY COURT PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF AND PERMITTED UNDER OR IN CONNECTION WITH THIS AGREEMENT. EACH OF THE PARTIES HEREBY (A) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF THE OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT, AS APPLICABLE, BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATI ONS IN THIS <u>Section 8.15</u>.

IN WITNESS WHEREOF, the Parties have caused this Separation and Distribution Agreement to be executed by their duly authorized representatives as of the day and year first above written.

LRAD Corporation, a Delaware corporation

/s/ Katherine H. McDermott By: Katherine H. McDermott Its: Chief Financial Officer Parametric Sound Corporation, a Nevada corporation

<u>/s/ James A. Barnes</u> By: James A. Barnes Its: Secretary and Treasurer

# Schedule 1.1 Parametric Assets

additional U	The 28 U.S. patents, three foreign and six pending applications listed below with a book value of approximately \$424,000. The technology related to the six J.S. patents previously abandoned by LRAD:
585991	ed Patents 5, 5885129, 5889870, 6011855, 6016351, 6044160, 6151398, 6304662, 6466674, 6577738, 6584205, 6606389, 6628791, 6850623, 6925187, 7088830, 7109789, 16, 7162042, 7224219, 7224808, 7298853, 7319763, 7343017, 7376236, 7564981, 7596229, 7729498
	ed Foreign Patents 1249615, Japan 4301956 and Mexico 237866
	ed Patent Applications NP, 21980.JP, 22151.PCT.US, 35924.NP, 35933.PCT.US, T8319.CIP2.CON, T8321.PROV.PCT.JP, T9468.PROV, 7029.CIP.PCT.CN.DV and T7029.CNDVHKD
	oned Patents 17, 6108433, 6188772, 6199655, 6229899, 6359990
	The following 4 U.S. trade names with a book value of approximately \$8,746 along with certain additional foreign rights to such names:
	ed Trade Names MP, HSS, HYPERDIRCTIONAL, HYPERSONIC
\$1,257,000,	Parts inventory consisting of emitter film and emitter and electronic parts with a book value of approximately \$21,700 and an original cost of approximately the difference being previous LRAD inventory reserves for excess parts and obsolescence.
	Tooling and molds that have been fully depreciated and have no remaining book value.
•	Manufacturing jigs, testing and other equipment that have been fully depreciated and have no remaining book value.
•	Drawings, technical descriptions, assembly details and component listings.
	Customer, representative and supplier lists.
	Manufacturing methods, knowhow and trade secrets related to HSS and HSS production.
	Excluded Assets
	Cash
	Accounts Receivable
	Retained HSS Assets
•	Parts inventory with a book value of approximately \$73,000 and an original cost of approximately \$164,000 the difference being previous LRAD inventory reserves

for excess parts.

# EXHIBIT 4.1

THIS NOTE HAS BEEN ACQUIRED FOR INVESTMENT AND HAS NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR QUALIFIED UNDER ANY STATE SECURITIES LAWS. THIS NOTE MAY NOT BE SOLD OR TRANSFERRED IN THE ABSENCE OF SUCH REGISTRATION OR QUALIFICATION OR AN EXEMPTION THEREFROM UNDER SAID ACT AND ANY APPLICABLE STATE SECURITIES LAWS.

> SUBORDINATED PROMISSORY NOTE OF PARAMETRIC SOUND CORPORATION

\$[\_\_\_\_]

Henderson, Nevada September 28, 2010

FOR VALUE RECEIVED, Parametric Sound Corporation., a Nevada corporation ("<u>Borrower</u>"), hereby promises to pay to [\_\_\_\_\_\_] ("<u>Holder</u>"), on or before the Maturity Date (as defined in this Note), the principal sum of \$[\_\_\_\_\_\_] or such lesser principal amount thereof as may remain outstanding in lawful money of the United States of America in immediately available funds, and to pay interest from the date of issuance of this Note on the principal amount hereof from time to time outstanding, in like funds, at a rate or rates per annum and payable on such dates as determined pursuant to the terms of this Note. This Note is one of a duly authorized issue of Notes of the Borrower (the "2010 Notes"), limited in aggregate principal amount to \$750,000.

# SECTION 1. MATURITY DATE

The unpaid principal amount of this Note plus all accrued and unpaid interest hereon and all other amounts owed hereunder with respect hereto will be paid in full in cash on or before the one year anniversary date of this Note (the "<u>Maturity Date</u>").

# SECTION 2. TERMS OF THE NOTE

2.1. Interest. This Note will bear simple interest at a rate equal to eight percent (8%) per annum (based on a 365-day year) on the outstanding principal amount hereof, from and including the date of issuance of this Note until the outstanding principal amount hereof together with all accrued and unpaid interest hereon shall be paid in full in cash, in the manner specified in this Note.

2.2. <u>Payment of Principal and Interest on the Note</u>. Borrower shall pay the principal amount of this Note, together with all accrued and unpaid interest thereon on or before the Maturity Date.

2.3. Optional Prepayment of this Note. This Note may be prepaid, at the Borrower's option, at any time prior to the Maturity Date, in whole or in part, without penalty and without advance notice.

2.4. <u>Application of Prepayments</u>. All prepayments permitted to be made hereunder shall include the payment in cash of interest, accrued and unpaid as of the date of repayment, on the principal amount of the portion of this Note so prepaid and shall be applied first to the payment of costs and expenses hereunder, next to accrued and unpaid interest and then to the payment of principal. The Company shall only make principal reductions or prepayments pro rata among the holders of the 2010 Notes. Likewise, any holder of one of the 2010 Notes who receives any payments or proceeds from any distribution in connection with a bankrupt cy, liquidation, reorganization, dissolution, winding-up or similar proceedings, shall be obligated to pro rate such amounts among the other holders of the 2010 Notes as provided in the Subscription Agreement of even date between the Company and the original Holder hereof (the "Subscription Agreement").

# 2.5. <u>Manner and Time of Payment</u>.

2.5.1 All payments by Borrower under this Note of principal, interest and all other amounts hereunder shall be made in same day funds and delivered to Holder not later than 5:00 p.m. (Pacific time) on the date such payment is due, with such payment to be made by wire transfer of immediately available funds to the respective account designated by Holder in writing; provided that funds received by Holder after 5:00 p.m. (Pacific time) shall be deemed to have been paid by Borrower on the next succeeding business day.

2.5.2 Whenever any payment to be made hereunder shall be stated to be due on a day which is not a business day, the payment shall be made on the next succeeding business day and such additional period shall be included in the computation of the payment of interest hereunder.

2.6. <u>Intercreditor Agreement</u>. Relationships between Holder and each of the other holders of the 2010 Notes shall be determined pursuant to the intercreditor agreements set forth in Section 6 and Section 10 of the Subscription Agreement and Section 2.4 herein on a *pari passu* basis. Holder acknowledges that the terms of this Note including the maturity date may be amended by written consent between the Borrower and holders of at least 60% of the outstanding principal amount of the 2010 Notes at the time of the amendment and such amended terms shall be binding on Holder.

## SECTION 3. REPRESENTATIONS AND WARRANTIES OF HOLDER

Holder represents and warrants, by payment for this Note and by accepting delivery of this Note, that:

3.1. <u>Restrictions on Transfer</u>. Holder has been advised that this Note has not been registered under the Securities Act or any state securities laws and, therefore, cannot be resold or otherwise transferred unless it is registered under the Securities Act and applicable state securities laws or unless an exemption from such registration requirements is available. Holder is aware that Borrower is under no obligation to effect any such registration with respect to the Note or to file for or comply with any exemption from registration. Holder is purchasing this Note for its own account and not with a view t o, or for resale in connection with, the distribution thereof in violation of the Securities Act. Holder shall not transfer, pledge, hypothecate or otherwise encumber this Note without the express written consent of Borrower.

3.2. <u>Accredited Investor, etc.</u> Holder has such knowledge and experience in financial and business matters so as to be capable of evaluating the merits and risks of Holder's investment in this Note, is able to incur a complete loss of such investment and is able to bear the economic risk of such investment for an indefinite period of time. Holder is an "accredited investor" as that term is defined in Regulation D under the Securities Act. In making its decision to invest in this Note, Holder has made its own independent credit analysis, and is not relying on the fact that any other pers on or entity is a Holder to Borrower.

# SECTION 4. DEFAULT

4.1. Events of Default. Subject to the provisions of Section 6 of the Subscription Agreement, each of the following shall constitute an event of default (each an "Event of Default"):

4.1.1 Borrower fails to make any payment under this Note no later than ten (10) business days after the same becomes due and payable.

4.1.2 Borrower fails to observe or perform, in any material respect, any covenant contained in this Note, and such failure shall continue unremedied for a period of thirty (30) days after its receipt of written notice by Holder setting forth in reasonable detail the basis for such failure.

4.1.3 Borrower (a) commences any voluntary proceeding under any bankruptcy, reorganization, arrangement, insolvency, readjustment of debt, receivership, dissolution, or liquidation law or statute, of any jurisdiction, whether now or subsequently in effect; (b) is adjudicated insolvent or bankrupt by a court of competent jurisdiction; (c) petitions or applies for, acquiesces in, or consents to, the appointment of any receiver or trustee of Borrower or for all or substantially all of its property or assets; (d) makes an assignment for the benefit of its creditors; or (e) admits in writing its inability to pay its debts as they mature.

4.1.4 There is commenced against Borrower any proceeding under any bankruptcy, reorganization, arrangement, insolvency, readjustment of debt, receivership, dissolution, or liquidation law or statute, of any jurisdiction, whether now or subsequently in effect, and such proceeding remains undismissed for a period of ninety (90) days or Borrower by any act indicates its consent to, approval of, or acquiescence in, such proceeding; or a receiver or trustee is appointed for Borrower or for all or substantially all of its property or assets, and the receivership or trusteeship remains undischarged for a period of ninety (90) days.

4.2. <u>Remedies Upon Event of Default</u>. Subject to the provisions of Section 6 of the Subscription Agreement, upon any Event of Default, then in any such event, and at any time during the continuation thereof, Holder may declare the outstanding principal amount and all accrued and unpaid interest thereon and all costs and expenses hereunder to be, whereupon this Note shall become, immediately due and payable.

# SECTION 5. SUBORDINATION

5.1. Subordination. Any right of Holder to payment of principal or interest from the Borrower shall be subordinated to the claims and rights of the holders of the Senior Debt. "Senior Debt" means all indebtedness of the Borrower outstanding on the date of execution of this Note or thereafter created, incurred or assumed that is designated as Senior Debt by the Board of Directors of the Borrower at the time it is incurred. Any cash payment of principal or interest to Holder shall be collected, enforced or received by Holder as trustee for and paid over to the holders of Senior Debt. Holder agrees that in the event of any payment of principal or interest by the Borrower to Holder by reason of any receivership, insolvency or bankruptcy proceeding, or proceeding for reorganization or readjustment of the Borrower or its properties, or otherwise, then, in any such event, the holders of Senior Debt shall be preferred in the payment of their claims of the holders of Senior Debt shall be first paid and satisfied in full before any payment or distribution of any kind or character, whether in cash or property, shall be made to Holder.

# **SECTION 6. WARRANTS**

6.1. <u>Warrant</u>. Concurrent with the execution of this Note, Holder has been issued a warrant (the "<u>Warrant</u>") which, subject to the terms and conditions thereof, entitles Holder to purchase shares of Borrower's common stock (the "<u>Warrant Shares</u>") at the exercise price set forth in the Warrant for a period of five (5) years.

6.2. Offset. Notwithstanding anything to the contrary set forth herein, upon exercise by Holder of its purchase rights under the Warrant, Borrower, in its sole and absolute discretion, shall be permitted to offset the purchase price for any such Warrant Shares against the outstanding principal amount owing under this Note as of the date of such exercise.

## SECTION 7. MISCELLANEOUS

7.1. Costs. If any legal action is taken to enforce any provision of this Note, then the non-prevailing party shall pay the reasonable costs, including the reasonable fees and disbursements of counsel, incurred by the non-prevailing party in connection with such legal proceeding.

7.2. <u>No Impairment</u>. Borrower will not willfully avoid the observance or performance of any of the terms of this Note.

7.3. <u>Amendments and Waivers</u>. Subject to the terms of Section 6 and Section 10 of the Subscription Agreement, no amendment, modification, termination, waiver or consent to departure of any provision of this Note shall in any event be effective without the prior written consent of Holder and Borrower. Any waiver or consent shall be effective only in the specific instance and for the specific purpose for which it was given. No notice to or demand on Borrower in any case shall entitle any Borrower to any further notice or demand in similar or other circumstances. Any amendment, modification, term ination, waiver or consent effected in accordance with this Section 7.3 shall be binding upon Holder at the time outstanding and each future holder hereof.

7.4. Notices. All notices and other communications required or permitted hereunder shall be in writing and shall be deemed effectively delivered (a) upon personal delivery, (b) upon delivery by an internationally recognized courier, or (c) five (5) days after having been sent by registered or certified mail, postage prepaid. Such notice shall be addressed to the recipient to be notified at the address set forth on the signature page(s) hereto (or at such other address for a recipient as shall be specified in a notice given in accordance with this Section 7.4).

7.5. Transfers. This Note may not be assigned or transferred by Holder to any person or entity and any assignment or transfer in violation of this Section 7.5 shall be void.

7.6. <u>Headings</u>. Section and subsection headings in this Note are included herein for convenience of reference only and shall not constitute a part of this Note for any other purpose or be given any substantive effect.

7.7. <u>Applicable Law</u>. This Note shall be governed by, and shall be construed and enforced in accordance with, the laws of the State of Delaware without regard to the principles of conflicts of laws.

7.8. <u>Successors and Assigns; Subsequent Holders</u>. This Note shall be binding upon the parties hereto and their respective permitted successors and assigns and shall inure to the benefit of the parties hereto and the permitted successors and assigns of Holder. The terms and provisions of this Note shall inure to the benefit of any permitted assignee or transferee of this Note, and in the event of such permitted transfer or assignment, the rights and privileges herein conferred upon Holder shall automatically extend to and be vested in such permitted transferee or assignee, all subject to the terms and conditions her eof.

7.9. Waiver of Jury Trial. Each of the undersigned waives, to the fullest extent permitted by law, trial by jury in any litigation arising out of or related to this Note.

7.10. Entirety. This Note embodies the entire agreement among the parties and supersedes all prior agreements and understandings, if any, relating to the subject matter hereof.

7.11. <u>No Strict Construction</u>. The language used in this Note shall be deemed to be the language chosen by the parties hereto to express their mutual intent, and no rule of strict construction will be applied against any person or entity. The use of the word "including" and "includes" in this Note shall be by way of example rather than by limitation.

7.12. <u>No Third-Party Beneficiaries</u>. Nothing herein expressed or implied is intended or shall be construed to confer upon or give to any person or entity, other than Borrower and Holder and their respective permitted successors and assigns (and to the extent specified in Section 5, holders of any senior debt), any rights or remedies under or by reason of this Note.

7.13. <u>Waiver of Notice, Etc.</u> Borrower hereby waives diligence, presentment, demand, protest and notice of any kind whatsoever, other than as expressly required by this Note. The nonexercise by Holder any of its rights hereunder in any particular instance shall not constitute a waiver thereof in that or any subsequent instance.

[Signature Page Follows]

IN WITNESS WHEREOF, Borrower has executed and delivered this Note as of the date first written above.

# BORROWER:

Parametric Sound Corporation

By: Name:	 	 	 
Title:			

Address:

Accepted by:

HOLDER:

By: \_\_\_\_ Name: \_\_\_\_ Title: \_\_\_\_

Address: \_

# EXHIBIT 4.2

THIS WARRANT AND THE SECURITIES REPRESENTED BY THIS WARRANT HAVE BEEN ACQUIRED FOR INVESTMENT AND HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR QUALIFIED UNDER ANY STATE SECURITIES LAWS. SUCH SECURITIES MAY NOT BE SOLD OR TRANSFERRED IN THE ABSENCE OF SUCH REGISTRATION OR QUALIFICATION OR AN EXEMPTION THEREFROM UNDER SAID ACT AND ANY APPLICABLE STATE SECURITIES LAWS.

# **Parametric Sound Corporation**

# WARRANT CERTIFICATE

No. [\_\_\_]

1.

September 28, 2010

Parametric Sound Corporation, a Nevada corporation (the "*Company*"), for value received, hereby certifies that [\_\_\_\_\_], at [\_\_\_\_\_] (the "*Holder*") or its registered assigns is the registered owner of the Warrant, which entitles the owner thereof to purchase from the Company at any time during the Exercise Period up to [\_\_\_\_\_] duly authorized, validly issued, fully paid and non-assessable shares of Common Stock, par value \$0.001 per share, of the Company ("*Common Stock*"), at the Exercise Price and subject to the terms set forth herein.

Definitions. As used in this Warrant, unless the context clearly requires a different meaning, the following terms have the meanings indicated.

"Exercise Period" means the period commencing on the date of original issuance of this Warrant and continuing through and including the Expiration Date.

"*Exercise Price*" means the per share price at which this Warrant may be exercised, which price shall be equal to the average closing price of the Common Stock for the first twenty (20) days of trading on the OTC Bulletin Board (or such other exchange, market or quotation system on which the Common Stock is then traded) after the spin-off of the Company from LRAD Corporation, provided however, that in no event shall such Exercise Price be less than \$0.10 or greater than \$0.30, as adjusted pursuant to the terms hereof.

"Expiration Date" means the earliest to occur of (a) the fifth anniversary of the date of original issuance of this Warrant or (b) effective date of a Sale or Merger.

"Mandatory Exercise Period" means the period commencing on the six month anniversary of the date of original issuance of this Warrant and continuing through and including the Expiration Date.

"Note" means that certain promissory note of even date herewith issued by the Company to the Holder.

"Sale or Merger" means any of the following: (a) the merger, reorganization or consolidation of the Corporation or any subsidiary of the Company into or with another corporation or other entity, or the issuance and sale by the Company of voting securities, in which or as a result of which the stockholders of the Company immediately preceding such transaction (solely by virtue of their shares or other securities of the Company) shall own fifty percent (50%) or less of the voting securities of the surviving entity, immediately following such transaction; (b) the sale, transfer or lease, whether in a single transaction or pursuant to a series of related transactions or plan, of all or substantially all the assets of the Company; or (c) the sale or exclusive license, whether in a single transaction or pursuant to a series of related transactions or plan of all or substantially all of the intellectual property of the Company.

"Warrant Notice" means a notice to the Company substantially in the form of Annex II hereto.

"Warrant Shares" means the shares of Common Stock that are issuable upon exercise of this Warrant.

# 2. Exercise of Warrants.

2.1. Voluntary Exercise. The Holder may exercise the Warrant in whole or in part, at any time or from time to time during the Exercise Period, by presentation and surrender of the Warrant Certificate to the Company, together with a duly executed Warrant Notice and payment by certified bank check to the order of the Company of the aggregate Exercise Price in exchange for receiving from the Company the requisite number of shares of Common Stock.

2.2. Cashless Exercise. The Holder shall have the option, subject to the Company's option under Section 2.4 (the "Cashless Exercise Option") to exercise this Warrant, in whole but not in part, by the surrender of this Warrant Certificate (and without payment in cash of the Exercise Price) in exchange for a number of whole shares of the Common Stock equal to the product of (a) the number of shares of the Common Stock for which this Warrant is exerciseable as of the business day on which the Warrant Notice is received by the Company (the "Cashless Exercise Date"), multiplied by (b) the Cashless Exercise Ratio (the "Cashless Exercise Price) in exchange for a number of the Cashless Exercise Date"). The "Cashless Exercise Ratio" shall be equal to (i) the Final Price on the Cashless Exercise Date less then Exercise Price divided by (ii) by the Final Price on the Cashless Exercise Date. The "Final Price" means, on any day, the last reported sale price per share of the Common Stock for that day on OTC Bulletin Board (or such other exchange, market or quotation system on which the Common Stock is then traded). The "Cashless Exercise Date" shall be deemed the Exercise Date under the Warrant.

2.3. Mandatory Exercise. If for any period of ten (10) consecutive trading days during the Mandatory Exercise Period (the "Subject Trading Period"), the closing price for the Common Stock exceeds \$0.90, the Company shall have the right, in its sole and absolute discretion, to require the Holder to exercise the Warrant (the "Mandatory Exercise") in accordance with the procedure set forth in Section 2.1 or 2.2, by providing a written request for such Mandatory Exercise within ten (10) days of the end of the Subject Trading Period.

2.4. Offset of Exercise Price against Note. The Company may, in its sole and absolute discretion, elect to offset the aggregate Exercise Price for any Warrant Shares issued upon exercise of this Warrant against the outstanding principal amount owing under the Note as of the Exercise Date.

2.5. Delivery of Certificates. As soon as practicable but not later than ten (10) business days after the Company shall have received such Warrant Notice, the Company shall execute and deliver or cause to be executed and delivered, in accordance with such Warrant Notice, a certificate or certificates representing the number of shares of Common Stock specified in such Warrant Notice, issued in the name of the Holder or in such other name or names of any Person or Persons designated in such Warrant Notice. Except as otherwise set forth in Section 2.2 above, each Warrant Certificates shall be deemed to have been exercised and such share certificate or certificates shall be deemed to have been issued, and such Holder or other Person or Persons designated in such Warrant Notice and payment for the shares purchased thereby shall have been received by the Company (the "Exercise Date").

2.6. Surrender of Warrant. The Holder shall surrender the Warrant Certificate when it delivers the Warrant Notice, and in the event of a partial exercise of any Warrant, the Company shall execute and deliver to the Holder, at the time the Company delivers the share certificate or certificates issued pursuant to such Warrant Notice, a new Warrant Certificate for the unexercised portion.

3. **Fractions, Expenses.** The Company shall not be required to issue fractions of shares upon an exercise of any Warrant. If any fraction of a share would, but for this restriction, be issuable upon an exercise of the Warrant, in lieu of delivering such fractional share the Company shall pay to the Holder, in cash, an amount equal to the fair market value of such fractional share as determined above. The Company shall pay all expenses, taxes and owner charges payable in connection with the preparation, issuance and delivery of certificates for the shares of Common Stock and any new Warrant Certificates, except that if the certificates for the shares of Common Stock or the new Warrant Certificates are to be registered in a name or names other than the name of the Holder, funds sufficient to pay all transfer taxes payable as a result of such transfer shall be paid by the Holder at the time of its delivery of the Warrant Notice or promptly upon receipt of a written request by the Company for such payment.

4. <u>Adjustments</u>. The number of shares of Common Stock for which this Warrant is exercisable, and/or the Exercise Price at which such Common Stock may be purchased upon exercise of this Warrant, shall be subject to adjustment from time to time as set forth in this <u>Section 4</u>. For purposes of clarification, the parties acknowledge and agree that the adjustments described in this <u>Section 4</u> are not intended to, and shall not, trigger any duplicative adjustments to the number of shares of Common Stock and the Exercise Price that may occur pursuant to any other document or instrument of the Company.

4.1. Stock Dividends, Subdivisions and Combinations. If at any time the Company shall (a) take a record of the holders of its Common Stock for the purpose of entitling them to receive a dividend payable in, an instrument convertible into, or other distribution of, additional Common Stock, (b) subdivide its outstanding Common Stock into a larger number of shares of Common Stock, or (c) combine its outstanding Common Stock into a smaller number of shares of Common Stock, then, in each such case, (i) the number of shares of Common Stock for which this Warrant is exercisable immediately after the occurrence of any such event shall be adjusted to equal the number of shares of Common Stock which a record holder of the same number of shares of Common Stock for which this Warrant is exercisable immediately prior to the adjustment) multiplied by the number of shares of Common Stock for which this Warrant is exercisable immediately prior to the adjustment) multiplied by the number of shares of Common Stock for which this Warrant is exercisable immediately after such adjustment.

4.2. Reorganization, Reclassification, Merger, Consolidation or Disposition of Assets. In case the Company shall reorganize its capital, reclassify or recapitalize its capital stock, consolidate or merge with or into another corporation (where the Company is not the surviving corporation or where there is a change in or distribution with respect to the Common Stock), or sell, transfer or otherwise dispose of all or substantially all of its property, assets or business to another corporation and, pursuant to the terms of such reorganization, reclassification, consolidation, merger or disposition or assets, shares of common stock of the successor or acquiring corporation or any cash, shares or stock or other securities or property of any nature whatsoever (including warrants or other subscription or purchase rights) in addition to or in lieu of common stock of the successor or acquiring corporation, reclassification, consolidation, merger or disposition of assets by a holder of the number of shares of such verganization, reclassification, consolidation, merger or disposition of assets by a holder of the number of shares of Common Stock for which such Warrant is exercisable immediately prior to such event. In case of any such reorganization, reclassification, consolidation, merger or disposition of assets, the successor or acquiring corporation (if other than the Company) shall expressly assume in writing the due and punctual observance and performance of each and every covenant and condition of such Warrant to be performed and observed by the Company and all the obligations and liabilities hereunder. The foregoing provisions of this <u>Section 4.2</u> shall similarly apply to successive reorganizations, reclassifications, consolidations, mergers or disposition of assets.

5. Notice of Adjustments. Whenever the number of shares of Common Stock for which a Warrant is exercisable, or whenever the Exercise Price shall be adjusted pursuant to Section 4, the Company shall forthwith prepare a certificate to be executed by the chief financial officer of the Company setting forth, in reasonable detail, the event requiring the adjustment and the method by which such adjustment was calculated, specifying the number of shares of Common Stock for which such Warrant is exercisable and (if such adjustment was made pursuant to Section 4.2) describing the number and kind of any other shares of stock or Other Property for which such Warrant is exercisable and ny change in the Exercise Price, after giving effect to such adjustment or change. The Company shall promptly cause a signed copy of such certificate to be delivered to each Warrant holder in accordance with Section 11.1. Company shall keep at its office copies of all such certificates and cause the same to be available for inspection at said office during normal business hours by any Warrant holder or any prospective purchaser of a Warrant designated by a holder thereof.

6. No Impairment. The Company shall not by any action, including, without limitation, amending its certificate of incorporation or through any reorganization, transfer of assets, consolidation, merger, dissolution, issue or sale of securities or any other voluntary action, avoid or seek to avoid the observance or performance of any of the terms of the Warrants, but will at all times in good faith assist in the carrying out of all such terms and in the taking of all such actions as may be necessary or appropriate to protect the rights of Warrant holders against impairment. Upon the request of a Warrant holder, the Company will at any time during the period a Warrant is outstanding acknowledge in writing, in form reasonably satisfactory to such holder, the continuing validity of the Warrant and the obligations of the Company thereunder.

7. **Transfer of Warrant**. The transferability of this Warrant is subject to the restrictions set forth in any shareholder agreements that apply, or may in the future apply, to the Common Stock issuable upon exercise of this Warrant as of the date of any such proposed transfer; provided that no transfer shall be made that (a) does not comply with all applicable federal and state securities laws, or (b) would require registration or qualification of the Warrant pursuant to the Securities Act of 1933, as amended (the "Securities Act") or any applicable state securities laws. Upon transfer of the Warrant the Holder must deliver to the Company a duly executed Warrant Assignment in the form of Annex I hereto, with funds sufficient to pay any transfer tax imposed in connection with such assignment (if any). The Company shall execute and deliver a new Warrant Certificates in the form of this Warrant Certificate with appropriate changes to reflect such assignment, in the name or names of the assignee or assignees specified in the fully executed Warrant Assignment or other instrument of assignment and, if the Holder's entire interest is not being transferred to assignee, in the name of the Kareant Certificate or Certificates or certificates or exchange of this Warrant Certificate shall be without charge to the Holder (except as provided above with respect to transfer taxes, if any) an d any new Warrant Certificate or certificate be dated the date hereof. The terms "Warrant" as used herein includes all Warrants into which this Warrant (or any successor Warrant) may be exchanged or issued in connection with the transfer or assignment of this Warrant or any successor Warrant.

8. **Reporting.** The Company hereby agrees that it will file any reports required to be filed by it under the Securities Act, or the rules and regulations adopted by the Commission thereunder and that it will use all reasonable efforts to cooperate with each Holder and each holder of Warrant Shares in supplying such information concerning the Company as may be necessary for such Holder or holder to complete and file any information reporting forms currently or hereafter required by the Commission as a condition to the availability of an exemption from the Securities Act for the sale of any Common Stock issu able upon exercise hereof. The Company also agrees that it will take such further action, and supply such information as any Holder may reasonably request to the extent required from time to time to enable the Holder to sell Common Stock without registration under the Securities Act within the limitation of the exemptions provided by Rule 144 under the Securities Act, as such rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the Commission.

9. Lost Certificate. Upon receipt by the Company of evidence satisfactory to it of the loss, theft, destruction or mutilation of any Warrant Certificate, and, in the case of loss, theft or destruction, upon receipt of indemnification satisfactory to the Company, or, in the case of mutilation, upon surrender and cancellation of the mutilated Warrant Certificate, the Company shall execute and deliver a new Warrant Certificate of like tenor and representing the right to purchase the same aggregate number of shares of Common Stock issuable upon exercise hereof. The recipient of any such Warrant Certificate shall reimburse the Company for all reasonable expenses incidental to the replacement of such lost, mutilated or missing Warrant Certificate.

10. Securities Representations of Holder. By accepting the Warrant, Holder represents that it is acquiring the Warrant for its own account for investment purposes and not with the view to any sale or distribution, and that Holder will not offer, sell or otherwise dispose of the Warrant or the Common Stock issuable upon exercise hereof except under circumstances as will not result in a violation of applicable securities laws. Holder further represents and warrants as follows.

10.1. Information Provided to Holder. Holder has been advised that: (a) the offer and sale of this Warrant is intended to be a transaction by an issuer not involving any public offering and thereby exempt from registration under the Securities Act by virtue of Section 4(2) thereof and Regulation D promulgated thereunder or any state securities laws; (b) neither this Warrant nor the Common Stock issuable upon exercise hereof may be transferred without (i) registration under the Securities Act or a valid exemption therefrom and from any applicable state securities laws and (ii) compliance with the restrictions contained in this Warrant, (c) there are substantial risks of loss of investment involved in an investment in this Warrant and the Warrant Shares is presently an illiquid investment and the Holder may be required to bear the economic risk of investment in this Warrant and the Warrant Shares for a substantial period of time; and (d) there is no established market for the Warrant and the Warrant Shares and no assurance has been provided that any public market will develop.

10.2. **Transfer Restrictions.** Holder acknowledges that it has been advised that if Holder proposes to offer, sell or otherwise transfer, pledge or hypothecate all or any part of the Warrant and the Warrant Shares (other than pursuant to Rule 144 under the Securities Act or an effective registration statement under the Securities Act) the Holder must, upon request, deliver to the Company a written opinion of counsel, reasonably satisfactory in form and substance to the Company, that an exemption from the registration requirements of the Securities Act is available.

10.3. Holder Information. Holder (a) has such knowledge and experience in financial and business matters, including investments of the type represented by the Warrant and the Warrant Shares, as to be capable of evaluating the merits of investment in the Company and can bear the economic risk of an investment in the Warrant and the Warrant Shares; (b) has not been furnished with or relied upon any oral representation, warranty or information in connection with the offering of this Warrant and the Warrant Shares; (c) is an "Accredited investor" as such term is defined in Regulation D under the Securities Act; and (d) is acquiring the Warrant and the Warrant Shares for investment purposes only, for its own account and not with a view to, or for resale in connection with, the distribution or other disposition thereof in contravention of the Securities Act or any Blue Sky Laws, without prejudice, however, to the Holder's right at all times, subject as applicable to any shareholders applicable Blue Sky Laws, or under an exemption from such registration available under the Securities Act and other applicable Blue Sky Laws.

#### 11. GENERAL PROVISIONS.

11.1. **Notices**. All notices and other communications required or permitted hereunder shall be in writing and shall be deemed effectively delivered (a) upon personal delivery, (b) upon delivery by an internationally recognized courier, or (c) five (5) days after having been sent by registered or certified mail, postage prepaid. Such notice shall be addressed to the recipient to be notified at the address set forth on the signature page(s) hereto (or at such other address for a recipient as shall be specified in a notice given in accordance with this Section 11.1).

11.2. **Waiver and Amendment.** Any provision of this Warrant may be amended or modified only as set forth in the Note; <u>provided</u>, that without the consent of each holder adversely affected thereby, no such amendment or modification shall (a) change the Expiration Date, change the Exercise Price or the number of shares of Common Stock or other securities for which the Warrant is exercisable, or (b) amend this <u>Section 11.2</u>. No course of dealing between the Company and Holder will operate as a waiver or modification of any party's rights or obligations under this Warrant. No delay or failure on the part of either party in exercising any right or remedy under this Warrant will operate as a waiver of such right or any other right. A waiver given on one occasion will not be construed as a bar to, or as a waiver of, any right or remedy on any future occasion.

11.3. Not a Stockholder. The Warrant shall not entitle the Warrant Holder, prior to the exercise of the Warrant, to any rights as a stockholder of the Company.

11.4. **Governing Law; Jurisdiction.** This Warrant Certificate will be governed by and construed in accordance with the internal laws of the State of Nevada as applied to agreements between residents thereof to be performed entirely within such State, without reference to that body of law relating to conflict of laws or choice of law. With respect to any suit, action or proceeding initiated by the Company or the registered holder of this Warrant Certificate arising out of, under or in connection with this Warrant Certificate, the Company and such holder hereby submits to the non-exclusive jurisdi ction of any state or Federal court sitting in Clark County, Nevada and irrevocably waives, to the fullest extent permitted by law, any objection that it may now have or hereafter obtain to the establishment of venue in any such court in any such suit, action or proceeding.

IN WITNESS WHEREOF, the Company caused this Warrant Certificate to be duly executed and its corporate seal to be hereunto affixed and attested, all as of the day and year first above

written.

## PARAMETRIC SOUND CORPORATION

By:	 	 
Name:		
Title:		

Address: \_\_\_\_\_

# FORM OF ASSIGNMENT

(Relating to Parametric Sound Corporation)

(To be executed by the registered holder if such holder desires to transfer the Warrant Certificate.)

FOR VALUE RECEIVED

RECEIVED \_\_\_\_\_\_ hereby sells, assigns and transfers unto \_

this Warrant Certificate, together with all right, title and interest therein, and does hereby irrevocably constitute and appoint \_\_\_\_\_\_ attorney, to transfer the within Warrant Certificate on the books of the within - named Company, with full power of substitution.

Date: \_\_\_\_\_, 20\_\_\_\_

Signature: \_\_\_\_\_

[Signature Guaranteed:

Signatures must be guaranteed by a member firm of a registered national securities exchange, a member of the National Association of Securities Dealers, Inc., or a commercial bank or trust company having an office or correspondent in the United States.]

# FORM OF WARRANT NOTICE

Dated: \_\_\_\_\_

# To: Parametric Sound Corporation (the "Company")

Reference is made to the Subscription Agreement (the "Agreement") dated [\_\_\_\_\_], 2010. Terms are used herein as therein defined.

\_\_\_\_

The undersigned, pursuant to the provisions set forth in the Agreement and the Warrant Certificate No. \_\_\_, dated \_\_\_\_\_\_, hereby irrevocably elects and agrees to exercise the Warrants pursuant to the terms of Section 2 of the Warrant, being equal to \_\_\_\_\_\_ shares of Common Stock of Parametric Sound Corporation.

[If said number of shares is less than all of the shares purchasable hereunder, the undersigned hereby requests that a new Warrant Certificate representing the remaining balance of the shares be registered in the name of \_\_\_\_\_\_\_, whose address is

The undersigned hereby represents that it is exercising the Warrant for its own account for investment purposes and not with the view to any sale or distribution and that the undersigned will not offer, sell or otherwise dispose of the Warrant or any underlying Warrant Shares in violation of applicable securities laws.

[NAME OF HOLDER]

By:

Name: \_\_\_\_\_ Title: \_\_\_\_\_

[ADDRESS OF HOLDER]

#### TAX SHARING AGREEMENT

This Tax Sharing Agreement (this "<u>Agreement</u>") is entered into as of September 27, 2010 between LRAD Corporation, a Delaware corporation ("<u>LRAD</u>"), and Parametric Sound Corporation, a Nevada corporation and wholly owned subsidiary of LRAD ("<u>Parametric Sound</u>," and together with LRAD, the "<u>Parties</u>"). Capitalized terms used in this Agreement and not otherwise defined herein shall have the meanings ascribed to such terms in the Separation and Distribution Agreement, dated as of the date hereof, by and between LRAD and Para metric Sound (the "<u>Separation Agreement</u>").

## RECITALS

Whereas, LRAD is the common parent corporation of an affiliated group of corporations within the meaning of Section 1504(a) of the Internal Revenue Code of 1986, as amended (the "<u>Code</u>"), that has filed consolidated federal income tax returns;

Whereas, Parametric Sound is a newly-formed, wholly owned subsidiary of LRAD;

Whereas, pursuant to the Separation Agreement, among other things, LRAD will transfer to Parametric Sound all of the Parametric Assets and Parametric Sound will issue to LRAD shares of Parametric Common Stock (the "Contribution");

Whereas, on the Distribution Date at the Effective Time, LRAD will distribute all of the issued and outstanding shares of Parametric Common Stock on a pro rata basis to holders of LRAD Common Stock (the "Distribution");

Whereas, the Parties intend that the Distribution shall qualify as a distribution described in Section 355 of the Code (the "Distribution Tax Treatment");

Whereas, the Parties intend that after the Distribution Parametric Sound will not be a member of LRAD's affiliated group of corporations for federal income tax purposes;

Whereas, the Parties intend that the Contribution, taking into account the Distribution, shall qualify as a series of transfers described in Section 351(a) of the Code or otherwise as a transaction eligible for tax-free treatment under the Code (the "Contribution Tax Treatment"); and

Whereas, the Parties desire to set forth their rights and obligations with respect to Taxes (as defined herein) due for periods before and after the Distribution Date;

NOW, THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereto agree as follows:

#### ARTICLE I. DEFINITIONS

1.01 General. As used in this Agreement, the following terms shall have the following meanings:

"Affiliate" shall have the meaning set forth in the Separation Agreement.

"Agreement" shall have the meaning set forth in the preamble to this Agreement.

"Ancillary Agreements" shall mean the Ancillary Agreements as set forth in the Separation Agreement together with the Separation Agreement.

"Code" shall have the meaning set forth in the recitals.

"Contribution" shall have the meaning set forth in the recitals.

"Contribution Tax Treatment" shall have the meaning set forth in the recitals.

"Dispute" shall have the meaning set forth in Article VIII.

"Distribution" shall have the meaning set forth in the recitals.

"Distribution Date" shall mean the date hereof.

"Distribution Tax Treatment" shall have the meaning set forth in the recitals.

"Effective Time" shall mean 11:59 p.m. EDT on the Distribution Date at which time the Distribution is effective.

"Final Determination" shall mean a determination within the meaning of Section 1313 of the Code or any similar provision of state or local Tax law.

"Governmental Authority" shall have the meaning set forth in the Separation Agreement.

"Liabilities" shall have the meaning set forth in the Separation Agreement.

"LRAD" shall have the meaning set forth in the preamble to this Agreement.

"LRAD Filed Tax Return" shall have the meaning set forth in Section 2.01(a).

"LRAD Indemnitees" shall have the meaning set forth in Section 4.01(b).

"LRAD Taxes" shall have the meaning set forth in Section 2.03(b).

"Parametric Sound" shall have the meaning set forth in the preamble to this Agreement.

"Parametric Assets" shall have the meaning set forth in the Separation Agreement.

"Parametric Business" shall have the meaning set forth in the Separation Agreement.

"Parametric Common Stock" shall have the meaning set forth in the Separation Agreement.

"Parametric Filed Tax Return" shall have the meaning set forth in Section 2.01(b).

"Parametric Indemnitees" shall have the meaning set forth in Section 4.01(a).

"Parametric Taxes" shall have the meaning set forth in Section 2.03(a).

"Parties" shall have the meaning set forth in the preamble to this Agreement.

"Person" shall have the meaning set forth in the Separation Agreement.

"Post-Distribution Period" shall mean any taxable year or other taxable period beginning after the Distribution Date and, in the case of any taxable year or other taxable period that begins before and ends after the Distribution Date, that part of the taxable year or other taxable period that begins at the beginning of the day after the Distribution Date.

"Pre-Distribution Period" shall mean any taxable year or other taxable period that ends on or before the Distribution Date and, in the case of any taxable year or other taxable period that begins before and ends after the Distribution Date, that part of the taxable year or other taxable period through the close of the Distribution Date.

"Separation Agreement" shall have the meaning set forth in the preamble to this Agreement.

"Subsidiary" shall have the meaning set forth in the Separation Agreement.

"Taxes" shall mean (i) all taxes, charges, fees, duties, levies, imposts, rates or other assessments or governmental charges of any kind imposed by any federal, state, local or foreign Governmental Authority, including, without limitation, income, gross receipts, employment, excise, severance, stamp, occupation, premium, windfall profits, environmental, custom duties, property, sales, use, license, capital stock, transfer, franchise, registration, payroll, withholding, social security, unemployment, disability, value added, alternative or add-on minimum or other taxes, whether disputed or not, and including any interest, penalties, charges or additions attributable thereto, (ii) liability for the payment of any amount of the type described in clause (i) above arising as a result of being (or having been) a member of any group or being (or having been) included or required to be included in any Tax Return related thereto, and (iii) liability for the payment of any amount of the type described in clauses (i) or (ii) above as a result of being conterning or otherwise assume or succeed to the liability of any other Person.

"Tax Advisor" shall have the meaning set forth in Article VIII.

"Tax Contest" shall have the meaning set forth in Section 5.01.

"Tax Information Packages" shall mean any information required in order to prepare and file any LRAD Filed Tax Return.

"Tax Return" shall mean any return, report, certificate, form or similar statement or document (including any related supporting information or schedule attached thereto and any information return, amended tax return, claim for refund or declaration of estimated tax) required to be supplied to, or filed with, a Governmental Authority or any bill for or notice related to ad valorem or other similar Taxes received from a Governmental Authority, in each case, in connection with the determination, assessment or collection of any Tax or the administration of any laws, regulations or administrative requirements relating to any Tax.

1.02 **References; Interpretations**. References in this Agreement to the singular shall be held to include the plural and vice versa and words of one gender shall be held to include the other gender as the context requires. The terms "hereof," "herein," and "herewith" and words of similar import shall, unless otherwise stated, be construed to refer to this Agreement (or the applicable Ancillary Agreement) as a whole (including all of the Schedules and Exhibits hereto and thereto) and not to any particular provision of this Agreement (or such Ancillary Agreement). Article, Section, Exhibit, Schedules and Appendices to this Agreement (or the applicable Ancillary Agreement) and words of similar import when used in this Agreement (or the applicable Ancillary Agreement) unless otherwise specified. The word "including," without limitation," unless the context otherwise requires or unless otherwise specified. The word "or" shall not be exclusive. Any definition of or reference to any statute shall be construed as referring also to any rules and regulations promulgated thereunder.

#### ARTICLE II. TAX RETURNS AND TAX PAYMENTS

#### 2.01 Obligations to File Tax Returns.

(a) LRAD shall have the sole and exclusive responsibility for the preparation and filing of each Tax Return that (x) includes either LRAD or Parametric Sound and which is for a Tax year ending on or before the Distribution Date or (y) includes LRAD and that is required to be filed after the Distribution Date (each, a "<u>LRAD Filed Tax Return</u>"); provided, however, that (1) all LRAD Filed Tax Returns shall be prepared on a basis that is consistent with both the Contribution Tax Treatment and the Distribution Tax Treatment, (2) Parametric Sound shall promptly prepare and deliver to LRAD in a manner consistent with past practices pro forma Tax Returns and Tax Information Packages for any taxable period in which Parametric Sound is included in, or any portion of the Parametric Business is reflected on, a LRAD Filed Tax Return, (3) LRAD shall provide to Parametric Sound sufficiently in advance of the due date for the filing thereof, and Parametric Sound shall have a reasonable opportunity to review and comment on, any such LRAD Filed Tax Return (or the relevant portion thereof) to the extent that Parametric Business is or any LRAD Filed Tax Return for the Distribution Date based on an actual or hypothetical closing of the books at the close of the Distribution Date. Parametric Sound hereby irrevocably authorizes and designates LRAD as its agent, coordinator and administrator for the purpose of taking any and all act tions necessary or incidental to the filing of any such LRAD Filed Tax Returns and, except as otherwise provided herein, for the purpose of making payments to, or collecting refunds from, any Governmental Authority in respect of a LRAD Filed Tax Return for which LRAD bears responsibility hereunder and to determine whether any refunds of Taxes to which LRAD may be entitled shall be received by way of refund or credit against the Tax liability of LRAD.

(b) Parametric Sound shall have the sole and exclusive responsibility for the preparation and filing of each Tax Return that is required to be filed after the Distribution Date that includes Parametric Sound or otherwise relates to the Parametric Business that is not a LRAD Filed Tax Return (each, a "<u>Parametric Filed Tax Return</u>"); provided, however, that, except as otherwise required by law, (1) all Parametric Filed Tax Returns shall be prepared on a basis that is consistent with both the Contribution Tax Treatment and the Distribution Tax Treatment, (2) Parametric Sound shall provide to LRAD sufficiently in advance of the due date for the filing thereof, and LRAD shall have a reasonable opportunity to review and comment on, any su ch Parametric Filed Tax Return (or the relevant portion thereof) to the extent that LRAD is responsible for any portion of the Taxes reported on such Parametric Filed Tax Return, and (3) in the case of any Parametric Filed Tax Return that includes Parametric Sound or the Parametric Business only for the portion of the relevant taxable period that begins after the Distribution Date, Taxes shall be allocated to the portion of such taxable period that begins after the Distribution Date.

2.02 <u>Obligation to Remit Taxes</u>. Subject to Section 2.01 and subject always to the ultimate division of responsibility for Taxes set out in Section 2.03, LRAD and Parametric Sound shall each remit or cause to be remitted to the applicable Governmental Authority in a timely manner any Taxes due in respect of any Tax Return that such Party is required to file (or, in the case of a Tax for which no Tax Return is required to be filed, which is otherwise payable by such Party to any Governmental Authority); provided, however, that in the case of any Tax Return, the Party not required to file such Tax Return in immediately available funds the amount of any Taxes ref lected on such Tax Return for which the former Party is responsible hereunder at least two (2) Business Days before payment of the relevant amount is due to a Governmental Authority.

## 2.03 Tax Sharing Obligations and Prior Agreements.

(a) Parametric Sound shall be responsible for the payment of (and shall be entitled to any refund of or credit for) all Taxes (i) that are attributable to Parametric Sound or the Parametric Business for any taxable period, in accordance with the principles set forth in Section 2.01(a)(4), provided, however, that (x) the determination of any such Taxes for any Pre-Distribution Period shall be made treating Parametric Sound or the Parametric Business, as applicable, as a stand-alone corporation, using methods and conventions consistent with past practices, (y) such Taxes shall not include any Taxes incurred by either Party in connection with either the Contribution, and (z) such Taxes shall be net of any Tax attributes attributes attributes to Parametric Sound, the Parametric Business or LRAD that are available (taking into account any Tax liability incurred by LRAD in connection with either the Contribution or the Distribution) to reduce (whether or not they actually reduce) the Tax Liability of either Party for any Pre-Distribution Period or LRAD for any Post-Distribution Period, or (ii) resulting from any breach of or inaccuracy in any representation, covenant or obligation of Parametric Sound under this Agreement (collectively, "Parametric Taxes").

(b) LRAD shall be responsible for the payment of (and shall be entitled to any refund of or credit for) all Taxes (i) that are attributable to LRAD, other than Parametric Taxes, or (ii) resulting from any breach of or inaccuracy in any representation, covenant or obligation of LRAD under this Agreement (collectively, "<u>LRAD Taxes</u>").

(c) If, prior to the Distribution, a deposit (including a payment of estimated Taxes) was made with respect to any Tax for which Parametric Sound is responsible under this Agreement, such deposit shall be assigned to Parametric Sound and Parametric Sound shall be liable only for the amount of such Tax ultimately due in excess of the applicable deposit. To the extent the amount of such deposit exceeds the amount of Tax attributable to such deposit that is ultimately due, then such excess shall be paid to and retained by LRAD.

(d) Refunds received and the amount of credits claimed by one Party with respect to Taxes for which the other Party is responsible under this Agreement, shall be remitted to such other Party within five days after the first Party receives such refund or files the Tax Return claiming such refund or credit, as applicable. In the event that any such credit is subsequently reduced as a result of any adjustment required by any Governmental Authority, such other Party shall pay the amount of such reduction to the first Party within five days of receiving notice of such reduction from the first Party.

(e) At Parametric Sound's request, LRAD shall, at Parametric Sound's expense, use its reasonable best efforts to obtain any refund or credit of a Tax or item included in a LRAD Filed Tax Return to which Parametric Sound is entitled pursuant to this Agreement, including through filing appropriate forms with the applicable Governmental Authority; provided that LRAD shall not be required to comply with such request if LRAD reasonably determines that attempting to obtain such refund or credit will have a material adverse impact on LRAD.

(f) Except as set forth in this Agreement, any and all prior Tax sharing or allocation agreements or practices between LRAD and Parametric Sound shall be terminated with respect to Parametric Sound as of the Distribution Date, and Parametric Sound shall have no any continuing rights or obligations thereunder.

#### 2.04 Amended Returns.

(a) Parametric Sound shall not file any amended Tax Return that includes LRAD.

(b) LRAD shall not file any amended Tax Return that that may increase any Parametric Sound Tax or otherwise give rise to indemnification pursuant to Section 4.01(b).

### ARTICLE III. COVENANTS

3.01 **Parametric Sound Covenants**. Notwithstanding anything else to the contrary contained in this Agreement or any other agreement, Parametric Sound hereby confirms and agrees that Parametric Sound will not take or permit to be taken any action at any time that could jeopardize the Contribution Tax Treatment, the Distribution Tax Treatment or both.

3.02 **LRAD Covenants**. Notwithstanding anything else to the contrary contained in this Agreement or any other agreement, LRAD hereby confirms and agrees that LRAD will not take or permit to be taken any action at any time that could jeopardize the Contribution Tax Treatment, the Distribution Tax Treatment or both. Notwithstanding the foregoing, LRAD may take or permit to be taken any action prohibited by the preceding sentence, subject to, and without limiting or modifying, LRAD's continuing indemnification obligation under Section 4.01(a), if (x) LRAD obtains the written consent of Parametric Sound (which consent shall not be unreasonably withheld) or (y) LRAD obtains a ruling from the Internal Revenue Service or an opinion of a nationally recognized law firm, in form and substance reasonably satisfactory to Parametric Sound, that the taking of such action will not materially adversely affect either the Contribution Tax Treatment or the Distribution Tax Treatment.

#### ARTICLE IV. INDEMNITY OBLIGATIONS AND PAYMENTS

# 4.01 Indemnity Obligations.

(a) Notwithstanding whether any action is permitted or consented to hereunder and notwithstanding anything else to the contrary contained herein, LRAD shall indemnify and hold harmless Parametric Sound, and its directors, officers and employees (collectively, the "Parametric Indemnitees") from and against, and will reimburse the Parametric Indemnitees for (i) all LRAD Taxes and (ii) all Taxes, Liabilities and related losses arising out of, based upon or relating or attributable to any breach of or inaccuracy in any representation, covenant or obligation of LRAD under this Agreement.

(b) Parametric Sound shall indemnify and hold harmless LRAD, and its directors, officers and employees (collectively, the "<u>LRAD Indemnitees</u>") from and against, and will reimburse the LRAD Indemnitees for (i) all Parametric Taxes and (ii) all Taxes, Liabilities and related losses arising out of, based upon or relating or attributable to any breach of or inaccuracy in any representation, covenant or obligation of Parametric Sound under this Agreement.

4.02 Notice. The Parties shall give each other prompt written notice of any payment that may be due to the provider of such notice under this Agreement.

4.03 Treatment of Payments. The Parties agree that any payment made between the Parties pursuant to this Agreement or any other Ancillary Agreement with respect to a Pre-Distribution Period or as a result of an event or action occurring in a Pre-Distribution Period shall be treated, to the extent permitted by law, for all Tax purposes as a nontaxable payment (i.e., a distribution or a capital contribution) made immediately prior to the Distribution.

#### ARTICLE V. TAX CONTESTS

5.01 Notice. LRAD shall promptly notify Parametric Sound in writing upon receipt by LRAD of a written communication from any Governmental Authority with respect to any pending or threatened audit, dispute, suit, action, proposed assessment or other proceeding (a "<u>Tax Contest</u>") concerning any Taxes for which Parametric Sound may be liable under this Agreement. Parametric Sound shall promptly notify LRAD in writing upon receipt by Parametric Sound of a written communication from any Governmental Authority with respect to any Taxes for which LRAD may be liable under this Agreement.

5.02 <u>Control of Contests By LRAD</u>. LRAD shall have the sole responsibility and control over the handling of any Tax Contest, including the exclusive right to communicate with agents of the Governmental Authority and to control, resolve, settle or agree to any deficiency, claim or adjustment proposed, asserted or assessed in connection with or as a result of any such Tax Contest involving (i) any LRAD Filed Tax Return, or (ii) the Contribution or the Distribution or any transaction associated therewith as described in the Separation Agreement. Subject to LRAD's control right, upon request by Parametric Sound shall, at Parametric Sound's expense, be allowed to participate in the handling of any such Tax Contest with respect to any item that may affect the liability of Parametric Sound under this Agreement or that relates to the Contribution Tax Treatment or the Distribution Tax Treatment or delayed. Notwithstanding anything else to the contrary contained herein, in the case of any such Tax Contest relating to the Contribution Tax Treatment, absent a settlement of such Tax Contest pursuant to the preceding sentence, LRAD shall be required to exhaust, at Parametric Sound's expense, all administrative remedies available with respect to such Tax Contest.

5.03 <u>Control of Contests By Parametric Sound</u>. Parametric Sound shall have the full responsibility and control over the handling of any Tax Contest, including the exclusive right to communicate with agents of the Governmental Authority and to control, resolve, settle or agree to any deficiency, claim or adjustment proposed, asserted or assessed in connection with or as a result of any such Tax Contest, involving any Parametric Filed Tax Return. Subject to Parametric Sound's control right, upon request by LRAD, LRAD shall, at LRAD's expense, be allowed to participate in the handling of any such Tax Contest with respect to any item that may affect the liability of LRAD or any subsidiary of LRAD under this Agreement.

#### ARTICLE VI. COOPERATION

6.01 General. Each Party shall fully cooperate with the other Party in connection with the preparation and filing of any Tax Return or the conduct of any Tax Contest (including, where appropriate or necessary, providing a power of attorney) concerning any issues or any other matter contemplated under this Agreement. Each Party shall make its employees and facilities available on a mutually convenient basis to facilitate such cooperation.

6.02 <u>Consistent Treatment</u>. Unless and until there has been a Final Determination to the contrary, each Party agrees not to take any position on any Tax Return, in connection with any Tax Contest or otherwise that is inconsistent with (a) the allocation of Taxes between LRAD and Parametric Sound as set forth in this Agreement or (b) the Contribution Tax Treatment and the Distribution Tax Treatment.

### ARTICLE VII. RETENTION OF RECORDS; ACCESS

7.01 **Retention of Records**; Access. For so long as the contents thereof may become material in the administration of any matter under applicable Tax law, but in any event until the later of (i) the expiration of any applicable statute of limitation and (ii) seven years after the Distribution Date, the Parties shall (a) retain records, documents, accounting data and other information (including computer data) necessary for the preparation and filing of all Tax Returns in respect of Taxes of either LRAD or Parametric Sound for any Pre-Distribution Period or any Post-Distribution Period or for any Tax Contests relating to such Tax Returns, and (b) give to the other Party reasonable access to such records, documents, accounting data and other information (including computer data) and to its personnel (ensuring their cooperation) and premises, for the purpose of the review or audit of such Tax Returns to the extent relevant to an obligation or liability of a Party under this Agreement or for purposes of the preparation or filing of any such Tax Return, the conduct of any Tax Contest or any other matter reasonably and in good faith related to the Tax affairs of the requesting Party. At any time after the Distribution Date that a Party proposes to destroy such material or information, it shall first notify the other Party in writing and the other Party shall be entitled to receive such materials or information proposed to be destroyed.

# ARTICLE VIII.

DISPUTE RESOLUTION

8.01 **Dispute Resolution**. The Parties shall attempt in good faith to resolve any disagreement arising under this Agreement, including any dispute in connection with a claim by a third party (a "Dispute"). Either Party may give the other Party written notice of any Dispute not resolved in the normal course of business. If such a Dispute is not resolved within sixty (60) days following the date on which one Party gives such notice, the Parties shall jointly retain a nationally recognized law or accounting firm, reasonably acceptable to the Parties (the "Tax Advisor"), to act as an ar bitrator in order to resolve the Dispute. The Tax Advisor's determination as to any Dispute shall be made in accordance with the terms of this Agreement and shall be final and binding on the Parties and not subject to collateral attack for any reason (other than manifest error). All fees and expenses of the Tax Advisor shall be shared equally by LRAD, on the one hand, and Parametric Sound, on the other hand.

#### ARTICLE IX.

## MISCELLANEOUS PROVISIONS

9.01 Governing Law. This Agreement, except as expressly provided herein, shall be governed by and construed and interpreted in accordance with the laws of the State of Delaware, irrespective of the choice of laws principles of the State of Delaware as to all matters, including matters of validity, construction, effect, enforceability, performance and remedies.

9.02 <u>Application to Present and Future Subsidiaries</u>. This Agreement is being entered into by LRAD and Parametric Sound. This Agreement shall constitute a direct obligation of each such entity. Articles III and VI of this Agreement shall be deemed to have been readopted and affirmed on behalf of any entity that becomes a Subsidiary of LRAD or Parametric Sound in the future.

9.03 **Further Assurances**. Subject to the provisions hereof, the Parties hereto shall make, execute, acknowledge and deliver such other instruments and documents, and take all such other actions, as may be reasonably required in order to effectuate the purposes of this Agreement and to consummate the transactions contemplated hereby.

9.04 <u>Survival</u>. Notwithstanding any other provision of this Agreement to the contrary, all representations, covenants and obligations contained in this Agreement shall survive until the expiration of the applicable statute of limitations with respect to any such matter (including extensions thereof).

9.05 <u>Addresses and Notices</u>. All notices, consents, requests, instructions, approvals, statements, reports and other communications provided for herein shall be in writing and shall be deemed to be duly given when (a) delivered in person or (b) deposited in the United States mail or private express mail, postage prepaid, addressed as follows:

If to LRAD:

LRAD Corporation 15378 Avenue of Science, Suite 100 San Diego, CA 92128 Attn: Chief Executive Officer

If to Parametric Sound:

Parametric Sound 1941 Ramrod Avenue, Suite #100 Henderson, NV 89014 Attn: Chief Executive Officer

Either Party may, by notice to the other Party, change the address to which such notices are to be given. Notice delivered personally shall be deemed delivered when received by the recipient. Notice given by mail as set out above shall be deemed delivered five calendar days after the date the same is mailed.

9.06 Binding Effect. This Agreement shall be binding upon and inure to the benefit of the Parties and their successors and assigns.

9.07 Waivers of Default. The failure of either Party to require strict performance by the other Party of any provision in this Agreement, or to exercise any right or remedy under this Agreement will not waive or diminish such Party's right to demand strict performance or exercise thereafter of that or any other provision, right or remedy hereof.

9.08 **Invalidity of Provisions**. If any provision of this Agreement or the application thereof to any Person or circumstance is determined by a court of competent jurisdiction to be invalid, void or unenforceable, the remaining provisions hereof, or the application of such provision to Persons or circumstances or in jurisdictions other than those as to which it has been held invalid or unenforceable, shall remain in full force and effect and shall in no way be affected, impaired or invalidated thereby, so long as the economic or legal substance of the transactions contemplated hereby or thereby, as the case may be, is not affected in any manner adverse to any Party. Upon such determination, the Parties shall negotiate in good faith in an effort to agree upon such a suitable and equitable provision to affect the original intent of the Parties.

9.09 <u>Complete Agreement</u>. This Agreement contains the entire agreement between the Parties pertaining to the subject matter of this Agreement and supersedes all prior agreements, negotiations, discussions, writings, understandings, commitments and conversations pertaining thereto and there are no agreements or understandings between the Parties other than those set forth or referred to in this Agreement. In the event of any inconsistency between this Agreement and the Separation Agreement or any other agreements relating to the transactions contemplated by the Separation Agreement, the provisions of this Agreement shall control.

9.10 Construction. The language in all parts of this Agreement shall in all cases be construed according to its fair meaning and shall not be strictly construed for or against any Party.

9.11 No Double Recovery. No provision of this Agreement shall be construed to provide an indemnity or other recovery for any costs, damages or other amounts for which the damaged Party has been fully compensated under any other provision of this Agreement or under any other agreement or action at law or equity. Unless expressly required in this Agreement, a Party shall not be required to exhaust all remedies available under other agreements or at law or equity before recovering under the remedies provided in this Agreement.

9.12 Setoff. All payments to be made by any Party under this Agreement may be netted against payments due to such Party under this Agreement, but otherwise shall be made without setoff, counterclaim or withholding, all of which are hereby expressly waived.

9.13 Counterparts. This Agreement may be executed in one or more counterparts, each of which shall be considered one and the same agreement, and shall become effective when each counterpart has been signed by each of the Parties and delivered to the other Party.

9.14 <u>No Third Party Rights</u>. This Agreement is only intended to allocate the responsibility for certain Taxes between LRAD and Parametric Sound and to address the other Tax matters stated herein. Nothing in this Agreement, express or implied, is intended or shall confer any right, benefit, claim or remedy of any nature whatsoever under or by reason of this Agreement upon any Person other than LRAD and Parametric Sound. LRAD and Parametric Sound acknowledge and agree that the respective rights of the LRAD Indemnitees and the Parametric Indemnitees expressly provided under this Agreement may only be enforced by LRAD and Parametric Sound, respectively.

9.15 Separation Agreement. To the extent not inconsistent with any specific term of this Agreement, the provisions of the Separation Agreement shall apply in relevant part to this Agreement, including Sections 7.1 (Termination), 8.10 (Headings), 8.13 (Specific Performance), 8.14 (Amendments) and 8.15 (Waiver of Jury Trial).

[Signature Page Follows]

IN WITNESS WHEREOF, the Parties have caused this Tax Sharing Agreement to be executed by their duly authorized representatives as of the day and year first above written.

# LRAD CORPORATION

By:	/s/ Katherine H. McDermott
Name:	Katherine H.
	McDermott
Title:	Chief Financial
	Officer

# PARAMETRIC SOUND CORPORATION

By:	/s/ James A. Barnes	
Name:	James A. Barnes	
Title:	Secretary and Treasurer	

## SYZYGY LICENSING LLC – PARAMETRIC SOUND CORPORATION LICENSE AND ROYALTY AGREEMENT

THIS LICENSE AND ROYALTY AGREEMENT ("Agreement") is made on the day and year first written below by and between **Syzygy Licensing, LLC**, a Nevada limited liability company having an address of 8617 Canyon View Drive, Las Vegas, Nevada 89117 ("Licensor") and **Parametric Sound Corporation**, a Nevada corporation having a business address of 1941 Ramrod Avenue, Suite 100, Henderson, Nevada 89014 ("Licensee"). Licensor and Licensee are sometimes collectively referred to herein as "the Parties."

#### RECITALS

WHEREAS, Licensor has developed or owns certain technology relating to improved systems and methods of processing media input to create parametric sound output (the "Technology"), and is the owner of certain intellectual property in connection with this technology;

WHEREAS, Licensee desires to acquire, and Licensor desires to grant, on the terms and conditions set forth in this Agreement, an exclusive, world-wide right for Licensee to use, manufacture, have manufactured, sell, have sold, offer for sale, distribute, import and/or export products incorporating the Technology in and for any and all commercial purposes.

NOW, THEREFORE, in consideration of the mutual covenants and agreements set forth herein, and other valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Parties agree as follows:

## 1. DEFINITIONS

- 1.1 Intellectual Property shall include any pending patent applications or issued patents, inventions, developments, improvements, specifications, trade secrets, practices and procedures, and knowhow owned by Syzygy that relate(s) to or cover(s) the Technology.
- 1.2 Licensed Product(s) shall include any product that is covered by the Intellectual Property.
- 1.3 Licensed Patents means the pending patent application identified in EXHIBIT A, and any patent or patent applications related thereto, including any foreign and non-provisional patents and patent applications relating thereto, and any continuations, continuations-in-part, divisions, reissues, re-examinations, and equivalent patents and patent applications relating to the non-provisional and foreign patent applications and patents.
- 1.4 Improvements means any improvement, enhancement, alteration, or modification of the Technology, whether invented or developed by Licensor or Licensee. Licensor shall own all rights in any improvements developed to the Technology by either party. However, any such Improvements shall be licensed to Licensee in accordance with the terms of this Agreement.
- 1.5 Confidential Information means all confidential information relating to and, where applicable, including the Technology, including, but not limited to, any and all information concerning products, trade secrets, inventions, discoveries, designs, improvements, research, development, specifications, technical data, market data, know-how, including information derived from reports, investigations, research, works-in-progress, and all other information which may be designated as confidential.

2.

3.

- 1.6.1 Calculation of Licensee's Net Sales. Net Sales shall be defined as Licensee's gross receipts or revenue, computed on a cash basis, for each Licensed Product that is sold, distributed, leased, or otherwise transferred for monetary payment to any third party, less the following:
  - **1.6.1.1** The actual cost of freight charges or of freight absorption, if any, separately stated in such invoice;
  - 1.6.1.2 Any trade, quantity or standard trade cash discounts, if any, allowed;
  - 1.6.1.3 Any tax, duties, imposts or other government charge on the sale, transportation, or delivery which is separately stated on the invoice (unless in the nature of a value added tax, which need not be separately stated);
  - 1.6.1.4 Any credit and cash refunds for returned goods; and
  - 1.6.1.5 Any allowances for damaged, obsolete, and defective goods.
- 1.7 Valid Claim means (a) a claim in an issued and unexpired patent included in the Licensed Patents that: (i) has not been held unenforceable, unpatentable or invalid by a decision of a court or other governmental agency of competent jurisdiction, and not subject to appeal, (ii) has not been admitted to be invalid or unenforceable through reissue or disclaimer or otherwise, (iii) has not been lost through an interference, reexamination or reissue proceeding; or (b) a claim of a pending patent application included in the Licensed Patents.

#### GRANT OF LICENSE.

- 2.1 Exclusive License. Subject to all the terms and conditions of this Agreement, Licensor hereby grants to Licensee an exclusive, worldwide license to use, manufacture, have manufactured, sell, have sold, offer for sale, distribute, and import and export the Licensed Products and all Improvements thereto in and for all commercial purposes.
- 2.2 Trademarks, Trade Dress, and Copyrights. Any trademarks or trade dress used by Licensee in connection with the Licensed Products shall inure to the benefit of Licensor. Any copyrights created by Licensee in connection with the Licensed Products shall be owned solely by Licensor.

## TERM AND TERMINATION

- 3.1 Term. The term of this Agreement shall commence upon the formal separation of Parametric Sound Corporation as an independent public entity distinct from and no longer owned by LRAD Corporation (hereinafter referred to as the "Effective Date") and shall continue in force until twenty years from the Effective Date or until the natural expiration of any patent containing a Valid Claim that covers the Technology, whichever period is longer.
- 3.2 Licensor's Option to Terminate or Renegotiate for Failure to Act. Licensor shall have the option to either renegotiate or terminate this Agreement at any time during the Term should Licensee fail to use commercially reasonable efforts to develop, manufacture, market and/or sell the Licensed Products only if such failure continues following reasonably detailed written notice thereof by Licensor to Licensee and a reasonable opportunity to cure. The parties agree that any previous license or assignment of any portion of the Technology from Licensor to Licensee shall revert back to Licensor in the event of termination of this Agreement.

- **3.3 Default.** If either party commits any material default or breach with respect to any of the provisions of this Agreement, or fails to account for or pay to the other party any payment that becomes due hereunder, then the other party shall have the right to terminate this agreement on 60 days' written notice to the other party if the breaching party fails to cure such breach during the sixty (60) day period.
- 3.4 Sale of Remaining Inventory upon Early Termination. Upon termination of this Agreement for any reason, Licensee shall provide Licensor with a written inventory of all Licensed Products in the possession of Licensee, or its sublicensees, or that are in the process of being manufactured. Licensee, and its sublicensees, shall be allowed a period of **one (1) year** to sell off or otherwise dispose of its remaining inventory of Licensed Products, provided that all Royalties under Section 4 are paid for Licensed Products sold or otherwise disposed of to a third party.
- 3.5 Continuing Obligations after Termination. Upon termination of this Agreement, whether by default, cancellation, termination or normal expiration, Licensee shall not be relieved of any duties or obligations to pay all amounts accrued and due hereunder, and said amounts shall be payable at the effective date of the termination. In addition, Sections 1, 3.4, 3.5, 4, 6, 9, 10, 11, 12, 13, 14, 15, 16, 17, 18, and 19 of this Agreement, and any provisions that specifically provide for survival, shall survive any termination of this Agreement whether by default, cancellation, termination or normal expiration.

#### ROYALTY, PAYMENTS, AND REPORTS

- 4.1 In consideration for the rights conveyed by this Agreement, Licensee shall pay the following to Licensor:
  - **4.1.1** Royalty 5.0%. Licensee shall pay Licensor a royalty of five percent (5.0%) of the Net Sales of Licensed Products which are either: i) sold or otherwise transferred for monetary payment prior to four (4) years from the Effective Date of this Agreement; or ii) are covered by a Valid Claim of a Licensed Patent issued in a territory in which the Licensed Products are sold.
  - **4.1.2** Royalty 3.0%. In the event that no patent covering a Licensed Product has issued in a territory after four (4) years from the Effective Date of this Agreement, Licensee shall pay Licensor a royalty of **three percent (3.0%)** of the Net Sales of the Licensed Products in that territory. If a patent issues in a territory after four (4) years from the Effective Date of this Agreement, the royalty shall revert back to **five percent (5.0%)** of the Net Sales for Licensed Products sold in that territory.
  - **4.1.3** Reimbursement of Development Costs. Licensee shall pay to Licensor its costs incurred through the Effective Date in developing, testing and prototyping the Technology. These costs include, but are not limited to, time charged by Mr. Elwood G. Norris at a rate of \$100 per hour, limited to an aggregate of 250 hours prior to the Effective Date. After the Effective Date, Licensee shall bear its own costs to develop, test, prototype, manufacture or otherwise use the Technology as authorized herein and, other than those actions that are reasonably necessary to convey any inventions, developments, improvements, specifications, trade secrets, practices and procedures, and know-how required by this Agreement, Licensor shall have no obligation to perform any such activiti es or make any Improvements to the Technology. The parties acknowledge that Licensee intends to employ Elwood G. Norris on or after the Effective Date.

- **4.1.4 Patent Costs.** Licensee shall reimburse Licensor's past costs, and shall be responsible for reimbursing all of Licensors' future costs, in filing for, prosecuting and maintaining any of the Licensed Patents in the United States. Licensee may request that Licensor file patent applications in additional territories, in which case Licensee shall reimburse Licensor for all costs associated therewith. Should Licensee not elect to request patent filings in additional territories, Licensor may do so at its own expense, and Licensee shall not be obligated to reimburse or otherwise pay for the costs associated therewith.
- 4.1.5 Reasonableness of Royalty Rate. The parties hereby agree that the Royalty payments set forth in this Section represent the value of the licenses granted hereunder for the right to use, manufacture, have manufactured, sell, have sold, offer for sale, distribute, and import and export the Licensed Products and all Improvements. The parties further agree that any Licensed Patents covering the Licensed Products have utility to Licensee both individually and in the aggregate, and that it would be impracticable to identify each specific Licensed Patents being used. The parties further agree that any Intellectual Property other than patents covering the Licensed Products has utility to Licensee, and that it would be impracticable to identify each specific type of Intell ectual Property being used. As such, the parties have elected to (and agree that it is reasonable to) use the aggregate royalty rates provided for in this Section for Licensee's use of the Intellectual Property.
- 4.2 Quarterly Payments. The Royalties set forth in this Section shall accrue on the Net Sales made day-to-day, on a cash receipts basis, in each calendar quarter. On or before the thirtieth day following the end of a calendar quarter (i.e. April 30<sup>th</sup>, July 30<sup>th</sup>, October 30<sup>th</sup>, and January 30<sup>th</sup>), Licensee shall submit payment of all accrued royalties on the Net Sales made in that calendar quarter.
- 4.3 Late Payments. In the event that any payment due hereunder is not received by Licensor within 10 days of the due date for said payment, then Licensor shall be entitled to charge Licensee interest at the rate of five percent (5 %) per annum on said payments accrued from the date such payment was due.
- **4.4 Quarterly Reports.** Licensee shall provide Licensor with a quarterly report on or before the thirtieth day following the end of a calendar quarter (i.e. April 30<sup>th</sup>, July 30<sup>th</sup>, October 30<sup>th</sup>, and January 30<sup>th</sup>) detailing the Net Sales made in that calendar quarter. This quarterly report shall include at least an accounting of the Net Sales made, with any deductions therefore, and of all payments due.
- 4.5 Records. Licensee shall keep records for a reasonable period of time of the Licensed Products manufactured, sold and distributed in sufficient detail to enable the Royalty Payment and any other payments due to Licensor to be determined. In the event that Licensor disagrees with any such accounting, Licensor or its designated representative shall have the right, upon reasonable notice to Licensee, at Licensor's expense and no more often than semi-annually to review and inspect the books and records of Licensee relating to the sale and distribution of the Licensed Products during normal business hours of Licensee at the location where such books and records are maintained, but for no more than the preceding two (2) years. In the event that any inspection by Licensor discloses any deficiencies in payments made to Licensor, then Licensee shall pay such additional amounts to Licensor within twenty (20) days following receipt of written notice thereof, together with all documentation necessary to support such additional payment by Licensee. In the event that there are no such subsequent payments to become due, then Licensee shall be entitled to deduct and offset any such overpayment from the subsequent amounts (20) days following receipt of written notice thereof to pay Licensor.

4.6 Payment in U.S. Dollars. All payments due hereunder shall be made in U.S. Dollars.

5.

4.7 Taxes. Licensee will pay any taxes incurred by it due to the use, manufacture, sale, distribution, or importation or exportation by Licensee of the Licenseed Products.

## FILING, PROSECUTION, MAINTENANCE AND ENFORCEMENT OF LICENSED PATENTS

- 5.1 Patent Protection Applied for by Licensor at Its Discretion. Licensor may apply for patent protection covering the Technology within the United States at its sole discretion. Should Licensor decide not to pursue patent protection in the United States on any portion of the Technology, Licensee may request that Licensor do so at Licensee's expense. Licensee may request that Licensor apply for patent protection outside the United States at Licensee's expense. Should Licensee elect not to apply for patent protection outside the United States, then Licensor may apply for patent protection outside the United States at Licensor's expense.
- 5.2 Licensor to Own All Licensed Patents and Control Prosecution and Maintenance. All Licensed Patents shall be owned, prosecuted, controlled and maintained solely by Licensor.
- 5.3 Licensor may Obtain Patent Protection on Licensee's Improvements. Licensor shall be entitled to file patent protection on Licensee's Improvements at Licensor's sole discretion. Licensee agrees that its agents, principals and employees will provide all reasonable assistance in drafting and prosecuting any such patent, whether or not said agents, principals and employees are named as inventors. Licensee agrees that its agents, principals and employees will execute all documents necessary to transfer ownership of all patents covering the Technology to Licensor. The costs for any such filings shall be governed by the provisions of Section 4.1.4 of this Agreement.
- 5.4 Notification of Infringements of the Licensed Patents. Licensor and Licensee agree to notify each other in writing of any suspected infringement(s) of the Licensed Patents and provide any evidence therefore.
- 5.5 Enforcement and Defense of Licensed Patents. The parties shall confer in good faith regarding the enforcement or defense of any rights under the Licensed Patents, including whether to institute suit for infringement and the strategy for enforcing such rights. Notwithstanding this cooperation, Licensee shall bear the burden of enforcing the Licensed Patents in the United States and any other territories in which it has elected to obtain patent protection to prevent infringements thereof. Licensee shall be solely responsible for all attorneys' fees, costs, and other expenses incurred in the enforcement and defense of the Licensed Patents in the United States and any other territories in which it has elected to obtain patent protection. License value and you other territories in which it has elected to obtain patent protection. License value and you other territories in which it has elected to obtain patent protection. License value and you other territories in which it has elected to obtain patent protection. License value and you other territories in which it has elected to obtain patent protection. License value and you other territories in which it has elected to obtain patent protection. License value and you other territories in which it has elected to obtain patent protection. License value and you other territories in which it has elected to obtain patent protection. License value and you other territories in which it has elected to obtain patent protection. License value and you other territories in which it has elected to obtain patents to the extent that it affects Licenser's rights or causes it damages within a territory.
- 5.6 Marking of Licensed Products. Licensee shall comply with the legal standards and requirements of the applicable country or territory for the marking of the Licensed Products and their packaging with a notice of the Licensed Patents.
- 5.7 **Ownership of Improvements.** Licensor shall own all Improvements made or developed by Licensor or by Licensee.

#### INDEMNIFICATION

- 6.1 Licensee agrees to release, indemnify and hold harmless Licensor, its officers, employees, and agents against any and all losses, expenses, claims, actions, lawsuits, judgments and damages (including attorney's fees through the appellate level) which may be brought against either party, its officers, employees, and agents as a result of or arising out of any claim of infringement with respect to the use, manufacture, sale, distribution, or importation or exportation by Licensee of the Licensed Products as they relate to the Licensed Patents. This section shall continue after the termination of this Agreement.
- 6.2 Licensee agrees to release, indemnify and hold harmless Licensor, its officers, employees, and agents against any and all losses, expenses, claims, actions, lawsuits, judgments and damages (including attorney's fees through the appellate level) which may be brought against either party, its officers, employees, and agents as a result of or arising out of any product liability claim with respect to the use, manufacture, sale, distribution, or importation or exportation by Licensee of the Licensed Products. This section shall continue after the termination of this Agreement.

#### 7. SUBLICENSING

6.

7.1 Licensee may not sublicense its rights under this Agreement unless it obtains Licensor's written permission to do so. Any Licensed Products sold under such a sublicense shall be subject to the provisions of Section 4 of this Agreement; however the royalty rates of any such sublicense may differ from those delineated in Section 4. The parties agree to negotiate royalty rates of any sublicense in good faith with the intent to provide for gross margins to Licensee and Licensor comparable to those delineated in and/or being the consequence of Section 4.

#### 8. ASSIGNMENT

8.1 Licensee may not sell, assign, or transfer its rights under this Agreement unless Licensor provides written permission to do so. Any entity that is the beneficiary of a sale, assignment or transference of this Agreement shall be subject to the provisions of Section 4 of this Agreement.

## 9. CONFIDENTIALITY

9.1 Licensee and Licensor agree that at all times during the term of this Agreement, any extensions thereof, and after termination of the Agreement, they will hold in trust, keep confidential, and not disclose to any third party or make any use or induce or assist others in the use or disclosure of Confidential Information, except as provided for in this Agreement. Notwithstanding the foregoing, Licensee may disclose Licensor's Confidential Information to its employees, agents, contractors, and advisors who have a need to know such information; and Licensor may disclose License's Confidential Information to its employees, agents, contractors, and advisors who have a need to know such information.

# **10. REPRESENTATIONS AND WARRANTIES OF LICENSOR:**

10.1 Licensor makes no representations or warranties to Licensee relating to patentability of the Technology, or potential infringement of any third-party intellectual property that may occur by practicing the Technology.

### 11. GOVERNING LAW

11.1 This Agreement, and all matters relating hereto, including any matter or dispute arising out of the Agreement, shall be interpreted, governed, and enforced according to the laws of the United States of America and the State of Nevada, where applicable, and the parties hereto consent to the jurisdiction of any appropriate federal or state court in and for the State of Nevada to resolve such disputes.

# 12. ATTORNEYS' FEES

12.1 In the event that any party hereto shall be in default or breach of this Agreement, said party shall be liable to pay all reasonable attorneys' fees, court costs, and other related collection costs and expenses incurred by the non-defaulting or non-breaching party in prosecuting its rights hereunder.

# 13. BINDING EFFECT

13.1 This Agreement shall be binding upon, and shall inure to the benefit of, the parties hereto and their respective heirs, legatees, agents, representatives, officers, directors, successors and assigns.

### 14. NO JOINT VENTURE

14.1 The parties hereto expressly disclaim and disavow any partnership, joint venture or fiduciary status or relationship between them and expressly affirm that they have entered into this Agreement as independent contractors and that the same is in all respects an "arms-length" transaction. No party shall be responsible in any way for the debts or obligations of the other party, nor shall either party have the power to obligate or bind the other party in any manner whatsoever.

## 15. CAPTIONS

15.1 The captions and paragraph headings of this Agreement are solely for the convenience of reference and shall not affect its interpretation.

## **16. SEVERABILITY**

**16.1** The parties agree that if any part, term, or provision of this Agreement shall be found illegal or in conflict with any valid controlling law, the validity of the remaining provisions shall not be affected thereby. In the event of the legality of any provision of this Agreement is brought into question because of a decision by a court of competent jurisdiction, the parties agree that either the parties (by written amendment) or the court may narrow the provision or delete it entirely so as to comply with the decision of said court.

## **17. ENTIRE AGREEMENT**

17.1 This Agreement expresses and contains the entire agreement between the parties with respect to the subject matter hereof and supersedes and replaces any prior agreements between the parties with respect thereto. Except as expressly provided in this Agreement, there are no agreements, understanding, inducements or arrangements between the parties relating to the subject matter of this Agreement. No subsequent alteration, amendment, change or addition to this agreement shall be binding upon either party unless reduced in writing and signed by them.

# **18. PREPARATION OF AGREEMENT** 18.1

The parties acknowledge and agree that they have both participated in the preparation of this Agreement and, in the event that any question arises regarding its interpretation, no presumption shall be drawn in favor of or against any party hereto with respect to the drafting hereof.

## 19. NOTICES

19.1 Any and all notices required or provided for hereunder shall be in writing and shall be delivered, either in person or by certified mail, return receipt requested, postage prepaid to the parties at the addresses set forth in the preamble to this Agreement. Each party may change the address at which they receive notices by notifying the other parties in accordance with the provisions of this Section. All notices shall be deemed received upon actual receipt thereof if delivered in person, or within forty-eight (48) hours following the deposit hereof in the United States mail in accordance with the terms of this Section.

IN WITNESS WHEREOF, the parties have executed this Agreement and have made it effective as of the day and year first written below.

Licensor:	Licensee:
By: /s/ JAMES A. BARNES (signature)	By: /s/ ELWOOD G. NORRIS (signature)
Name: JAMES A. BARNES	Name: ELWOOD G. NORRIS
Title: MANAGER	Title: PRESIDENT AND CEO
Date: SEPTEMBER 27, 2010	Date: SEPTEMBER 27, 2010

IMPROVED PARAMETRIC TRANSDUCER AND SIGNAL PROCESSING SYSTEMS AND METHODS

#### Exhibit 10.3

#### LEASE AGREEMENT

THIS AGREEMENT OF LEASE made and entered into this, 1st day of July, 2010 by and between DAVRIC CORPORATION, a Nevada Corporation, hereinafter referred to as "Lessor" and Syzygy Licensing LLC, hereinafter referred to as "Lessee".

WHEREAS, Lessor is the owner of office/warehouse building/s located at 1937 and 1941 Ramrod, Henderson, Nevada.

WHEREAS, Lessee desires to lease space within the said office/warehouse building;

1. <u>PREMISES</u>. Lessor hereby leases to Lessee and Lessee hereby leases from Lessor, for the term, at the rental, and upon the covenants and conditions hereinafter set forth, the premises consisting of an area approximately Four thousand five hundred (4,500) square feet in said office/warehouse building, whose common address is 1941 Ramrod #100, Henderson, Nevada 89014.

Lessee shall, at its own cost and expense, furnish and install, subject to Lessor's approval as to design, style, quality, color, age and conditions, the necessary tenant improvements, the necessary fixtures, equipment and furnishings in the premises for the effective operation of its business.

2. TERM. The term of this lease shall begin as of the 1st day of July, 2010 and shall continue for twelve (12) months unless sooner terminated as hereinafter provided.

#### 3. INTENTIONALLY LEFT BLANK

4. <u>POSSESSION AND USE</u>. Lessee shall occupy and use the demised premises during the term hereof for the purpose of conducting any and all business relating to the manufacturing and assembly of electronic audio equipment.

5. <u>RENT</u>. Lessee shall pay rent as follows to Lessor, at 980 American Pacific Drive, Suite 111, Henderson, Nevada 89014, or such other place as the Lessor may from time to time designate by written notice: \$4,000 per month from July 1, 2010 through June 30, 2011.

The above rental structure shall be payable monthly in advance, on the first (1<sup>st</sup>) day of each calendar month. In the event the monthly rent is not paid within fifteen (15) days, a five (5) percent penalty will be charged.

# 6. SECURITY. Waived by Lessor

7. <u>COVENANT TO OPERATE BUSINESS</u>. Lessee shall continuously occupy and use the premises and be open for business for the purpose specified herein. This requirement shall not apply during times when the premises are untenantable by reason of fire or other casualty, however, Lessee shall continue the operation of its business to the extent reasonable practical from the standpoint of good business during any period of reconstruction or repair.

8. LESSEE'S TAXES. Lessee shall pay all taxes levied on sales, payrolls, inventory, personal property and business or Lessee, and all other taxes pertaining to Lessee's business activities.

9. INSURANCE. Lessee shall, during the entire term at Lessee's sole cost and expense, keep in force by advance payment of premium, property damage insurance in the amount of not less than \$1,000,000.00 for injury to or death of one person as a result of one occurrence, and not less than \$1,000,000.00 per injury to or death of more than one person as a result of one occurrence, insuring Lessee against any liability that may accrue against Lessee on account of any occurrences in or about the premises during the term or in consequence of Lessee's occupancy of the premises and resulting in personal injury or death, said insurance to protect, hold harmless, and indemnify Lessor not only against any and all such liability, but also against all loss, expense, and damage of any and every sort and kind, including costs of investigation and attorney's fees and other cost of defense. Said insurance, or duly executed certificate or certificates for the same, showing compliance to date with the requirements of this section shall at all times be kept on deposit with Lessor. If Lessee failed to comply with such requirement, Lessor may obtain such insurance and keep the same in force and effect, and Lessee shall pay Lessor upon request the premium cost thereof for the term of this lease then unexpired.

10. <u>RELATIONSHIP</u>. Lessee will not at any time pledge the credit of Lessor, or incur any debt, contract or liability in the name of Lessor. Neither party shall, at any time, hold itself out to be the partner, co-venturer, agent, servant or employee of the other in any manner whatsoever, or to have any relationship one with the other, save and excepting that of Lessor and Lessee.

11. UTILITIES. Lessor shall pay for all Common Area Maintenance (CAM) charges, including, electric, gas, water, sewer and trash pick-up services utilized in the premises of Lessee.

12. SIGNS. Lessee shall use on the exterior premises only such signs, advertising placards, names, and insignia, trademarks and descriptive material as shall have first received written approval of Lessor.

13. LESSOR'S REPAIRS. Lessor shall keep and maintain in good repair, the roof of the building, of which the premises are a part, the exterior walls, exterior foundations, and structural parts of the building, landscaping and common exterior ways. Lessor shall not be obligated to make such repairs, however, until after ten (10) days notice from lessee stating the need for such repairs. Upon receipt of such notice, Lessor shall make such repairs with reasonable diligence. Lessor shall not be liable to Lessee, its agents, servants and/or employees for damage or injury to persons or property arising out of the disrepair of the roof, exterior walls, exterior foundations, structural parts of the building, landscaping and common exterior ways. Lessor shall not be obligated to repair any portion of said building by damage caused by any casualty or act of God, except as provided in Article 19 hereof, 'DESTRUCTION'.

14. LESSEE'S REPAIRS. Except as above provided, Lessee shall at its expense, repair and maintain the entire premises in good order, condition and repair. Lessee will make no alterations in, or additions or improvements to the premises, including, but not limited to, sewers, air conditioning and heating, without written consent of Lessor.

15. LESSEE'S IMPROVEMENTS. Any improvements placed upon the demised premises, by Lessee, including, but not limited to carpets, shall, upon installation, become the property of Lessor.

16. <u>ASSIGNING, MORTGAGING, SUBLETTING</u>. Lessee's attempted transfer, assignment, subletting or hypothecation without Lessor's written consent shall be void and confer no rights upon any third person.

17. <u>SUBORDINATION AND ATTORNMENT</u>. Lessee hereby subordinates its rights hereunder to the lien of any mortgage or deed of trust, to any lending institutions, now or hereafter in force against the land and building of which the premises are a part, and upon any buildings hereafter placed upon the land of which the premises are a part, and to all advances made or hereafter to be made upon the security thereof. Lessee shall attorn to the purchaser upon any foreclosure or sale and recognize such purchaser as the Lessor under this lease.

18. <u>BANKRUPTCY - INSOLVENCY</u>. If all or substantially all of the Lessee's assets be placed in the hands of a receiver or trustee, and such receivership or trusteeship continue for a period of thirty (30) days, or should the Lessee make an assignment for the benefit or creditors, or should the Lessee institute any proceedings to be adjudicated a bankrupt, or go to discharge of its debts, or to effect a plan of liquidation, composition or reorganization, or should any involuntary proceedings be filed against the Lessee under any bankruptcy laws, then this lease or any interest in the premises shall not become an asset in any of such proceedings and, Lessor may declare the term hereof ended.

19. DESTRUCTION. Should the premises or the building or which the same are a part be damaged or destroyed in whole or in part by fire, earthquake, or other casualty, so they cannot be repaired within ninety (90) working days to substantially the condition prior to such casualty. Lessor or Lessee may within thirty (30) days thereafter terminate this lease as of the date of such casualty. If by reason of any laws said building cannot be repaired or restored as a building of the same class, then irrespective of the time in which the building can be repaired or restored, Lessor or Lessee at any time within thirty (30) days of such casualty, may terminate this lease as of the date of such casualty. Upon any damage as aforesaid, and if the lease be not terminated as provided, Lessor shall with reasonable diligence restore the premises to substantially the condition prior to the casualty, with such changed as may be required, under any laws, ordinances or regulations. If such damage shall have occurred after the commencement of the term, Lessee shall be entitled to a reasonable diminution of the minimum monthly rental hereunder during the time required for restoration, according to the proportion of the premises rendered untenantable. Lessor shall not be liable for any damages resulting to lessee from the repair and reconstruction of such demised premises or building, or from the termination of this lease as herein provided, not shall Lessee be released thereby or in any such event from any of its obligations hereunder, except to the extent and upon the conditions expressly stated in this Article.

20. <u>EMINENT DOMAIN</u>. If the premises shall be taken under poser of eminent domain, this lease shall terminate as of such taking. If only a part of the premises shall be so taken, Lessee's interest in the premises and its liability for rent hereunder shall abate proportionately after the date of taking. Lessee hereby waives all right to any damages or compensation which may be awarded in any such condemnation proceedings, and hereby assigns to Lessor the right to any such damages or compensation; provided, however, that in any such condemnation authorities for any damages to its fixtures and equipment.

If any condemnation proceedings are take to change the grade of, or to widen the street or sidewalks, or for such other purpose of similar nature as to any of the land upon which the premises are located, Lessee's liability for the payment of rental and its other obligations shall not be diminished or affected.

21. INDEMNITY. Lessee shall indemnify and hold harmless Lessor from any claim or liability of whatsoever character resulting from Lessee's exercise of any rights or privileges hereunder.

22. <u>CONDITION OF PREMISES</u>. Lessor or Lessor's agents have made no representations or promises with respect to the demised premises, the land upon which the premises are located except as expressly set forth herein. The taking possession of the demised premises by Lessee shall be conclusive evidence as against Lessee, that Lessee accepts the said premises and the building of which the same form a part as being in good and satisfactory condition at the time such possession is so taken.

23. <u>DEFAULT BY LESSEE</u>. Time is of the essence. Should Lessee be in default of any rental or other charges for a period of ten (10) days, or be in default in the performance of any other of its promises, covenants or agreements, for ten (10) days after written notice thereof from lessor, or should the Lessee vacate or abandon the premises, in addition to any or all other rights or remedies of the Lessor hereunder and/or by the law provided, Lessor may:

(a) Declare the term hereof ended and re-enter the premises and take possession thereof and remove all persons therefrom, and Lessee shall have no further claim thereon or thereunder; or

(b) Without declaring the lease ended, re-enter the premises and occupy or lease the whole or any part thereof for and on account of the Lessee and upon such terms and conditions, and for such rent as Lessor may deem proper and to collect said rent, or any other rent that may hereafter become payable and apply the same toward the amount due or thereafter to become due from Lessee, and on account of the expenses or such subletting and any other damages sustained by Lessor. Should such rental be less than that herein agreed, Lessee shall pay such deficiency to Lessor in advance on the day of each month herein before specified for payment of minimum monthly rental, and pay Lessor forthwith upon any such releting the costs and expenses the Lessor may incur by reas on thereof, or

(c) Even though it may have relet said premises, thereafter elect to terminate this lease.

Service by Lessor of any notice pursuant to unlawful detainer statutes and the surrender of possession pursuant to such notice shall not (unless the Lessor elects to the contrary at the time of or at any time subsequent to the service of such notice and such election be evidenced by a written notice to Lessee) be deemed to be a termination of this lease. Nothing herein contained shall be construed as obligating the Lessor to relet the whole or any part of the premises. In the event of any entry or taking possession of the premises, as aforesaid, Lessor shall have the right, but not the obligation, to remove therefrom all or any part of the personal property located therein, and may place the same in storage or a public warehouse at the expense and risk of the owner or owners thereof.

If Lessor elects to terminate this lease under (a) or (b) above, Lessor shall be entitled to recover from the Lessee, as damages, the difference, if any, between the then reasonable rental value of the premises for the period of the term reserved in the lease and the amount of rental and other charges payable by the Lessee for the balance of the term of this lease, together with the rent then unpaid, if any.

If the default complained of, other than for the payment of monies, is of such a nature that the same cannot be cured within the ten (10) day period requiring such curing, then such default shall be deemed to be cured if Lessee within such period shall have commenced the curing thereof, and shall continue thereafter with all due diligence to cause such curing and does so complete the same diligently.

Remedies given to Lessor in this article are in addition and supplemental to all other rights or remedies of Lessor under laws then in force.

24. <u>ATTORNEY'S FEES</u>. If any action at law or in equity is instituted by either Lessor or Lessee against the other to enforce any of the terms, covenants, conditions, provisions, and/or rights hereunder, the unsuccessful party to such litigation shall pay to the successful party all costs and expenses, including reasonable attorney's fees incurred therein by such successful party, and if such successful party shall recover judgment in any action or proceeding, such costs, expenses and attorney's fees shall be included in and as a part of such judgment.

In addition to the aforementioned, in the event the business is sold or disposed of in any manner, any and all attorney fees incurred by Lessor as a result of actions by Lessee shall be paid by Lessee.

25. MISCELLANEOUS. The provisions hereof shall bind and inure to the benefit of the parties hereto, and their respective heirs, legal representatives, successors and assigns.

This lease is the entire agreement between the parties hereto. All negotiations and oral agreements acceptable to both parties are included herein.

Captions of Articles are for convenience only, and do not in any way limit or amplify the terms and provisions of this lease.

The words "Lessor" and "Lessee" refer to all such persons or corporations, and the liability of such persons or corporation shall be joint and several.

Any notice on demand given or served by either party to it on the other shall be given or served in writing, addressed as follows: To Lessor, DAVRIC CORPORATION, 980 American Pacific Drive, Suite 111, Henderson, Nevada 89014, and Lessee at the premises. Either party may change its address by giving written notice to the other.

\*\*\*Landlord agrees that this lease will be assumed by Parametric Sound Corporation upon its spin-off as a separate company from LRAD Corporation

IN WITNESS WHEREOF, the parties hereto have affixed their signatures the day and year first above written.

LESSOR:

DAVRIC CORPORATION

<u>/s/ JERRY E. POLIS</u> Jerry E. Polis, President

LESSEE:

SYZYGY LICENSING LLC

<u>/s/ JAMES A. BARNES</u> James A. Barnes, Manager

#### EXHIBIT A RULES AND REGULATIONS

This Rules and Regulations Agreement is entered into as of the date of the Lease Agreement (the "*Lease*"), by and between the undersigned Landlord and Tenant. To the extent of a conflict between the Lease and the rules and regulations, the Lease shall govern and control. Landlord hereby establishes the following rules and regulations for the safety, care and cleanliness of (i) the warehouse areas (hereinafter referred to as the "*Premises*") of any tenant or tenants of the Building (hereinafter referred to as the "tenant"), (ii) the common area, and (iii) the Project in general, or for the preservation of good order:

## A. FOR THE STORE AREAS:

1. All floor areas of the Premises (including, without limitation, vestibules, entrances, air returns, doors, fixtures, windows, and plate glass shall be maintained in a clean, safe, and good condition.

2. All trash, refuse, and other waste materials shall be stored in adequate containers and regularly removed from the Premises. These containers shall not be visible to the general public and shall not constitute a health or fire hazard or nuisance. In the event that any tenant shall fail to remedy such a health or fire hazard or nuisance within five (5) calendar days after written notice by Landlord to tenant, Landlord may remedy and/or correct such health or fire hazard or nuisance at the expense of the tenant involved.

3. No portion of the Premises shall be used for lodging purposes.

4. Neither sidewalks nor walkways shall be used to display, store, or place any merchandise, equipment, or devices, except with Landlord's prior written approval.

5. No public telephone, newsstand, shoeshine stand, or refreshment, vending, or other coin operated machine shall be installed or placed on any sidewalk or walkway area adjacent to the Premises or on the common area without Landlord's prior written approval in each instance.

6. No person or persons shall use the Premises, or any part thereof, for conducting therein a second-hand store, auction, distress, fire, bankruptcy, "going-out-of business," or "lost our lease" sale without Landlord's prior written consent.

7. Landlord shall have the absolute right to enter upon the Premises to perform such cleaning and cleaning of the pipes and drains serving the Premises (including rotor-rooter service) as Landlord shall deem necessary. The tenant shall pay Landlord the estimated costs therefore estimated as part of the Operating Expenses on a monthly basis.

#### B. FOR THE COMMON AREA:

1. Use of the common area shall be in an orderly manner in accordance with directional and/or other signs and/or guides. Roadways shall not be used at a speed in excess of ten (10) miles per hour and shall not be used for parking or stopping, except for immediate loading or unloading of passengers. Walkways shall be used only for pedestrian travel.

2. Customers and invitees of tenants shall not use the parking areas for anything but parking motor vehicles; provided, however, that no tenant or the employees of any tenant shall park in the parking areas. All motor vehicles shall be parked in an orderly manner within the painted lines defining the individual parking places. During peak periods of business activity, Landlord may impose any and all controls Landlord deems necessary to operate the parking lot, including, but not limited to, the length of time for parking use.

3. No person shall use any utility area, truck loading area, or other area reserved for use in conducting business, except for the specific purpose for which permission to use these areas has been given.

4. Without the written consent of the Landlord, no person shall use any of the common area for:

a. Vending, peddling, or soliciting orders; for the sale or distribution of any merchandise, device, service, periodical, book, pamphlet, or other matter;

- b. Exhibiting any sign, placard, banner, notice, or other written material;
- c. Distributing any circular, booklet, handbill, placard, or other material;
- d. Soliciting membership in any organization, group, or association, or soliciting contributions for any purpose;

e. Parading, patrolling, picketing, demonstrating, or engaging in conduct that might interfere with the use of the common area or otherwise be detrimental to any of the business establishments in the Project;

- f. Using the common area for any purpose when none of the business establishments in the Project is open for business;
- g. Discarding any paper, glass, or extraneous matter of any kind, except in designated receptacles; or
- h. Using a sound-making device of any kind or making or permitting any noise that is annoying, unpleasant, or distasteful; or
- i. Damaging any sign, light, fixture, landscaping material, or other improvement or property within the Project.

The above listing of specific prohibitions is not intended to be exclusive, but is intended to indicate that tenant is only to use the common area as a means of access and convenience for shopping at the business establishments in the Project and that such use is subject to the control of Landlord.

#### C. IN GENERAL:

1. Other than seeing eye dogs and the like, no pets shall be allowed in or about the store areas or common area of the Project under Landlord's sole and exclusive control, without Landlord's prior written consent.

2. All tenants and their authorized representatives and invitees shall not loiter in the parking lot or other parts of the common area; nor shall they in any way obstruct the sidewalks, entry passages, pedestrian passageways, driveways, entrances, and/or exits, all of which shall be used only as ingress to and egress from the Premises.

3. Tenants and their authorized representatives and invitees shall not throw cigar or cigarette butts or other substances or litter of any kind in or about the buildings of the Project, except in receptacles placed in it for that purpose.

4. Landlord shall not be responsible to any tenant or to any other person for the non-observance or violation of any of these rules and regulations by any other tenant or other person. All tenants shall be deemed to have read these rules and to have agreed to abide by them as a condition precedent to occupying the Premises.

IN WITNESS WHEREOF this Rules and Regulations Agreement is executed as of the date of the Lease Agreement.

LANDLORD: Davric Corporation

Signature: <u>/s/ JERRY E. POLIS</u> Date: 6/16/10 Print Name: Jerry E. Polis Title: Davric Corporation, Gen'l Partner President TENANT: Syzygy Licensing LLC

Signature: <u>/s/ JAMES A. BARNES</u> Date: 6/16/10 Print Name: James A. Barnes Title: Manager

# ASSIGNMENT AND ASSUMPTION OF LEASE

For good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, Syzygy Licensing LLC, a Nevada limited liability company ("Syzygy"), does hereby assign, grant, bargain, sell, convey and transfer to Parametric Sound Corporation, a Nevada corporation ("Parametric"), all of Syzygy's right, title and interest as lessee in and to that certain lease dated as of July 1, 2010 by and between Davric Corporation, as lessor, and Syzygy, as lessee, together with all amendments thereto (the "Lease") respecting those certain premises commonly known as 1941 Ramrod #100, Henderson, Nevada 89014, and Parametric does hereby agree to assume and make all payments which become due from, and to perform all covenants and conditions which are to be performed by, Syzygy pursuant to the Lease from and after the date hereof.

Executed at San Diego, California, this 27th day of September, 2010.

Syzygy Licensing LLC

By: <u>/s/ JAMES A. BARNES</u> Its: <u>Manager</u>

Parametric Sound Corporation

By: <u>/s/ ELWOOD G. NORRIS</u> Its: <u>CEO</u>

#### PARAMETRIC SOUND CORPORATION A Nevada Corporation

THE OFFERING OF SECURITIES DESCRIBED HEREIN HAS NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED. THIS OFFERING IS MADE PURSUANT TO RULE 506 OF REGULATION D UNDER SECTION 4(2) OF SAID ACT, WHICH EXEMPTS FROM SUCH REGISTRATION TRANSACTIONS NOT INVOLVING A PUBLIC OFFERING. FOR THIS REASON, THESE SECURITIES WILL BE SOLD ONLY TO INVESTORS WHO MEET CERTAIN MINIMUM SUITABILITY QUALIFICATIONS DESCRIBED HEREIN. THE SECURITIES OFFERED HEREBY HAVE NOT BEEN APPROVED OR DISAPPROVED BY THE SECURITIES AND EXCHANGE COMMISSION, ANY STATE SECURITIES AUTHORITY OR ANY OTHER REGULATORY AUTHORITY, NOR HAS ANY OF THE FOREGOING PASSED UPON OR ENDORSED THE MERITS OF THIS DOCUMENT. ANY REPRESENTATION TO THE CONTRARY IS UNLAWFUL.

A SUBSCRIBER SHOULD BE PREPARED TO BEAR THE ECONOMIC RISK OF AN INVESTMENT IN THE COMPANY FOR AN INDEFINITE PERIOD OF TIME BECAUSE THE SECURITIES HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR THE LAWS OF ANY OTHER JURISDICTION, AND, THEREFORE, CANNOT BE SOLD UNLESS THEY ARE SUBSEQUENTLY REGISTERED OR AN EXEMPTION FROM REGISTRATION IS AVAILABLE. THERE IS NO OBLIGATION OF THE ISSUER TO REGISTER THE SECURITIES UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR THE LAWS OF ANY OTHER JURISDICTION. TRANSFER OF THE SECURITIES IS ALSO RESTRICTED BY THE TERMS OF OPERATING AGREEMENT OF THE COMPANY.

#### PARAMETRIC SOUND CORPORATION A Nevada Corporation

#### SUBSCRIPTION AGREEMENT

Parametric Sound Corporation 1941 Ramrod Avenue, Suite #100 Henderson, Nevada 89014 Attention: James Barnes, Secretary and Treasurer

#### Subscription to acquire Securities of Parametric Sound Corporation

# Gentlemen:

то∙

Re:

The undersigned subscribing investor (the "*Investor*" or the "*undersigned*") understands that Parametric Sound Corporation, a Nevada corporation (the "*Company*"), is offering (the "*Offering*") to issue up to \$750,000 in aggregate principal amount of subordinated promissory notes (the "*Notes*") and warrants to purchase up to 1,500,000 shares of the Company's common stock (the "< font style="DISPLAY: inline; FONT-WEIGHT: bold; FONT-STYLE: italic">Warrants", together with the Notes, the "*Securities*"). The undersigned understands that the Company is newly formed pursuant to Chapter 78 of the Nevada Revised Statutes. The undersigned further understands that (i) the Offering is being made without registration thereof under the Securities Act of 1933, as amended (the "*Securities Act*"), and is being made only to "accredited investors" (as defined in Rule 501(a) of Regulation D under the Securities Act), and (ii) the undersigned will not be deemed to have purchased any of the Securities unless and until such time as all of the following conditions to closing have occurred: (A) this Subscription Agreement and such other documentation as is requested by the Company and accepted by the Company; (B) the purchase price for the Securities has been delivered pursuant to instructions provided by the Company (C) the proposed spin-off of the Company's common stock to the stockholders of LRAD Corporation shall have been completed and the related Form 10 registration statement of the Company shall have been declared effective by the Securities and Exchange Commission, with no stop order in effect with respect thereto and (D) closing documents in form and substance satisfactory to the Company and its counsel have been executed and delivered.

1. <u>Subscription</u>. Subject to the terms and conditions hereof, the undersigned hereby tenders a subscription for the Securities in the amount set forth on the signature page hereto or such lesser amount as the Company shall choose to accept pursuant to <u>Section 2</u> below (the "*Subscription*"). Subject to the Company's acceptance pursuant to <u>Section 2</u> and the satisfaction of the conditions to Closing set forth in <u>Section 3</u>, the entire Subscription will be due and payable at the Closing (as defined below).

2. Acceptance of Agreement. It is understood and agreed that the Company shall have the right to accept or reject this Subscription Agreement and shall have the right to accept or reject all or part of the Subscription in the Company's sole and absolute discretion (the amount accepted shall thereafter be the undersigned's Subscription for all purposes hereof). Subscriptions need not be accepted in the order received, and the Securities may be allocated among subscribers in the Company's sole and absolute discretion. Notwithstanding anything in this Subscription Agreement to the contrary, the Company shall have no obligation to issue any of the Securities to any person who is a resident of a jurisdiction in which the issuance of Securities to him, her or it would constitute a violation of the securities, "blue sky" or other similar laws of such jurisdiction (collectively referred to as the "Blue Sky Laws"). This Subscription Agreement and the Subscription Agreement and the undersigned's Subscription as provided in the form attached hereto as the "Acceptance". Upon acceptance of this Subscription Agreement and the undersigned's Subscription by the Company, the Acceptance ". Upon acceptance of this Subscription Agreement to and adoption of all of the terms of this Subscription Agreement.

#### 3. Closing; Conditions to Closing.

(1)

(a) <u>Time and Place of Closing</u>. The closing of the sale and purchase of the Securities (the "*Closing*") shall take place at the offices of the Company, or at such other place and on such date as shall be selected by the Company. Upon satisfaction by the undersigned of its obligations hereunder, at Closing, the Company shall deliver to the undersigned a Note in substantially the form attached hereto as <u>Exhibit A</u> in the principal amount subscribed for hereunder and a Warrant i n substantially the form attached hereto as <u>Exhibit B</u> entitling the undersigned to purchase two (2) shares of the Company's common stock for every one dollar (\$1.00) of principal amount of the Note subscribed for hereunder at the exercise price designated therein.

(b) Undersigned's Conditions to Closing. The undersigned's obligations hereunder are subject to the fulfillment, prior to or at the Closing, of each of the following conditions:

Representations and Warranties. The representations and warranties of the Company contained in this Subscription Agreement shall be true and correct at the

Closing.

(2) **Performance of the Company.** The Company shall have performed and complied with all agreements and conditions required by this Subscription Agreement to be

performed or complied with by it prior to or at the Closing.
(c) <u>Company's Conditions to Closing</u>. The Company's obligations hereunder are subject to acceptance by the Company of the undersigned's Subscription, and to the

fulfillment, prior to or at the Closing, of each of the following conditions:

(1) <u>Representations and Warranties</u>. The representations and warranties of the undersigned contained in this Agreement shall be true and correct at the Closing.

(2) **Purchase Price.** The purchase price for the Securities shall have been delivered pursuant to instructions provided by the Company.

(3) <u>Spin-off Effective</u>. The proposed spin-off of the Company's common stock to the stockholders of LRAD Corporation shall have been completed and the related Form 10 registration statement of the Company shall have been declared effective by the Securities and Exchange Commission, with no stop order in effect with respect thereto.

(4) **Proceedings and Documents.** All proceedings in connection with the transactions contemplated hereby and all documents and instruments incident to such transactions shall be satisfactory in substance and form to the Company and its counsel, and the Company and its counsel shall have received all such counterpart originals or certified or other copies of such documents as the Company may reasonably request.

(5) **Performance of Covenants.** Each of the covenants and obligations that the undersigned is required to comply with or perform at or prior to Closing shall have been complied with and performed in all material respects.

# 4. **<u>Representations and Warranties by the Company.</u>** The Company represents, warrants and agrees as follows:

(a) Organization and Standing of the Company. The Company is duly and validly organized and validly existing as a corporation under the laws of Nevada, and has all requisite power and authority under such laws to conduct its business as currently conducted.

(b) <u>Compliance with Other Instruments</u>. The Company is not in violation of any term of this Subscription Agreement nor is it in material violation of any term of any other mortgage, indenture, contract, agreement, instrument, judgment, decree, order, statute, rule or regulation which is applicable or to which it is bound.

(c) Litigation. There is no action, suit, proceeding or investigation pending or, to the Company's knowledge, currently threatened against the Company that questions the validity of this Agreement or the right of the Company to enter into this Subscription Agreement, or to consummate the transactions contemplated hereby, or that might result, either individually or in the aggregate, in any material adverse changes in the assets, condition, affairs or prospects of the Company, financially or otherwise.

# 5. Representations and Warranties of the Undersigned. The undersigned hereby represents and warrants to the Company as follows:

(a) The undersigned either (i) is an "accredited investor" within the meaning of Securities and Exchange Commission ("SEC") Rule 501 of Regulation D, as presently in effect, or (ii) (A) certifies that the undersigned is not a "U.S. person" within the meaning of SEC Rule 902 of Regulation S, as presently in effect, and that the undersigned is not acquiring the Securities for the account or benefit of any U.S. person, (B) agrees to resell the Securities only in accordance with the provisions of Regulation S, pursuant to registration under the Securities Act of 1933, as amended (the "Securities Act")), or pursuant to an available exemption from registration and agrees not to engage in hedging transactions with regard to such Securities unless in compliance with the Securities Act, (C) agrees that the Securities (or any certificate representing such Securities) issued to the undersigned shall contain a legend to the effect that transfer is prohibited except in accordance with the provisions of Regulation S, pursuant to registration under the Securities Act or pursuant to an available exemption from registration and that hedging transactions involving such Securities may not be conducted unless in compliance with the Securities Act, or pursuant to registration under the Securities Act, or pursuant to registration under the Securities Act, or pursuant to an available exemption from registration.

(b) The Securities are being acquired for the undersigned's own account for investment, with no intention of distributing or selling any portion thereof and not with a view to any distribution thereof within the meaning of the Securities Act, and will not be transferred by the undersigned in violation of the Securities Act or the then applicable rules or regulations thereunder. No one other than the undersigned has any interest in or any right to acquire the Securities.

(c) By reason of the undersigned's business or financial experience, or that of the undersigned's professional advisor, the undersigned is capable of evaluating the merits and risks of an investment in the Company and of protecting its own interests in connection with the transaction.

(d) The undersigned has received and carefully read and understands this Subscription Agreement and the exhibits hereto.

(e) No representations or warranties have been made to the undersigned by the Company any officer of the Company, or any agent or promoter of said persons or entities, other than as set forth herein.

(f) The undersigned has been afforded an opportunity to ask questions of and receive answers satisfactory to the undersigned from the Company, and the officers of the Company concerning the terms and conditions of the Offering, and the Company has made available all additional information which the undersigned has requested.

(g) The address set forth on the signature page hereto is the undersigned's true and correct residence, if an individual, or principal place of business, if an entity other than an individual.

(h) The undersigned has investigated the acquisition of the Securities to the extent the undersigned deemed necessary or desirable and the Company has provided the undersigned with any assistance it has requested in connection therewith.

(i) The undersigned has full power and authority to make the representations referred to in this Subscription Agreement, to purchase the Securities and to deliver and comply with the terms of this Subscription Agreement.

(j) The undersigned acknowledges and is aware of the following:

(1) Investment in the Company is speculative and involves a high degree of risk of loss.

(2) The Securities have not been registered under the Securities Act or any Blue Sky Laws, and the transfer thereof is restricted by the Securities Act and applicable Blue Sky Laws. The Securities will not be, and the undersigned will have no rights to require that the Securities be, registered under the Securities Act. There will be no public market for the Securities, and the undersigned may not be able to avail itself of the provisions of Rule 144 of the Securities Act with respect to the Securities.

(3) No state or federal agency has made any finding or determination as to the fairness of the terms of the offering and sale of the Securities.

(k) The execution and delivery of this Subscription Agreement and any other document contemplated hereby, the consummation of the transactions contemplated hereby, and the performance of the undersigned's obligations hereunder, will not conflict with, or result in any violation of or default under, any provision of any charter, by-laws, trust agreement, operating agreement or other governing instrument applicable to the undersigned, or any agreement or other instrument to or by which the undersigned or the properties of the undersigned is a party or may be bound.

() If executing this Subscription Agreement in a representative or fiduciary capacity, the undersigned represents and warrants on behalf of the person, partnership, trust, estate, corporation, or other entity for whom the undersigned is acting in this matter, that such person, partnership, trust, estate, corporation or other entity has full right and power to execute, deliver, and perform this Subscription Agreement, and that each of the representations contained in this Subscription Agreement are true and accurate with respect to such person, partnership, trust, estate, corporation or other entity.

(m) The representations and warranties above, and in the attached Confidential Investor Questionnaire are true and accurate as of the date hereof and shall be true and accurate as of the date of Closing and shall survive such date. If in any respect such representation and warranties shall not be true and accurate prior to Closing, the undersigned shall give immediate notice of such fact to the Company by facsimile or telegram, specifying which representations and warranties are not true and accurate and the reasons therefor.

(n) The undersigned is in compliance with the requirements of Executive Order No. 133224, 66 Fed. Reg. 49079 (Sept. 25, 2001) (the "*Order*") and other similar requirements contained in the rules and regulations of the Office of Foreign Assets Control, Department of the Treasury ("*OFAC*") and in any enabling legislation or other Executive Orders or regulations in respect thereof (the Order and such other rules, regulations, legislation, or orders are collectively called the "*Orders*"). Neither the undersigned nor any beneficial owner of the undersigned:

(1) is listed on the Specially Designated Nationals and Blocked Persons OFAC List maintained by OFAC pursuant to the Order and/or on any other list of terrorists or terrorist organizations maintained pursuant to any of the rules and regulations of OFAC or pursuant to any other applicable Orders (such lists are collectively referred to as the "OFAC Lists");

(2) is a Person who has been determined by competent authority to be a Person with whom a U.S. Person is prohibited from transacting business, whether such prohibition arises under U.S. law, regulation, executive orders or any lists published by the United States Department of Commerce, the United States Department of Treasury or the United States Department of State including any agency or office thereof;

(3) is owned or controlled by, or acts for or on behalf of, any Person on the OFAC Lists or any other Person who has been determined by competent authority to be a Person with whom a U.S. Person is prohibited from transacting business, whether such prohibition arises under U.S. law, regulation, executive orders or any lists published by the United States Department of Commerce, the United States Department of Treasury or the United States Department of State including any agency or office thereof; or

(4) is under investigation by any governmental authority for, or has been charged with, or convicted of, money laundering, drug trafficking, terrorist-related activities, any crimes which in the United States would be predicate crimes to money laundering, or any violation of any Anti-Money Laundering Laws.

For purposes of this Subscription Agreement, the term "Anti-Money Laundering Laws" means those laws, rules, regulations, orders and sanctions, state and federal, criminal and civil, that (i) limit the use of and/or seek the forfeiture of proceeds from illegal transactions; (ii) limit commercial transactions with designated countries or individuals believed to be terrorists, narcotic dealers or otherwise engaged in activities contrary to the interests of the United States; or (iii) are designed to disrupt the flow of funds to terrorist organizations. Such laws, regulations and sanctions are deemed to include the Executive Order Number 13224 on Terrorism Financing (September 23, 2001), the Patriot Act; the Currency and Fo reign Transactions Reporting Act (also known as the Bank Sectory Act, 31), the Trading with the Enemy Act, 50 U.S.C. Appx. Section 1 et seq., the International Emergency Economics Powers Act, 50 U.S.C. Section 1701 et seq., and the sanction regulations promulgated pursuant thereto by OFAC, as well as laws relating to prevention and detection of money laundering in 18 U.S.C. Sections 1956 and 1957, as amended. As used in this <u>Section 5(n)</u>, the term (y) "Person" means any individual, corporation, partnership, limited liability company, unincorporated organization, government or any agency or political subdivision thereof or any other form of entity; and (z) "U.S. Person" means any Person that is a United States or its constituent states or territories, or an entity, regardless of where organized, with a principal place of business within the United States or any of its territories.

(0) THE UNDERSIGNED UNDERSTANDS THAT THE LEGAL AND ECONOMIC MATTERS RELATING TO ITS INVESTMENT IN THE COMPANY ARE COMPLEX AND THAT THE UNDERSIGNED IS FREE TO SEEK INDEPENDENT PROFESSIONAL GUIDANCE OR COUNSEL WITH RESPECT THERETO. THE UNDERSIGNED HAS EITHER SOUGHT SUCH ADVICE OR COUNSEL OR DETERMINED, AFTER CAREFULLY READING THE MEMORANDUM, THE OPERATING AGREEMENT, THIS SUBSCRIPTION AGREEMENT, AND THE OTHER DOCUMENTS AND AGREEMENTS CONTEMPLATED HEREBY AND THEREBY, TO FOREGO SUCH ADVICE. THE UNDERSIGNED ACKNOWLEDGES THAT COUNSEL TO THE COMPANY IS NOT ACTING AS COUNSEL TO THE UNDERSIGNED.

(p) THE UNDERSIGNED IS NOT RELYING UPON ANY WRITTEN OR ORAL REPRESENTATION OR ADVICE FROM THE COMPANY, THE COMPANY OR ANY OF THEIR RESPECTIVE AGENTS OR REPRESENTATIVES, WITH RESPECT TO (A) ANY LEGAL, TAX, ECONOMIC OR OTHER CONSIDERATIONS INVOLVED IN THE UNDERSIGNED'S INVESTMENT IN THE COMPANY OR (B) THE MEANING OR EFFECT OF ANY TERMS OR CONDITIONS OF THIS SUBSCRIPTION AGREEMENT OR ANY OTHER DOCUMENT OR AGREEMENT CONTEMPLATED HEREBY.

(q) THE UNDERSIGNED HAS AND WILL RELY ONLY ON THE UNDERSIGNED'S OWN TAX ADVISORS AND FINANCIAL PLANNERS WITH RESPECT TO THE U.S. FEDERAL, STATE, LOCAL AND FOREIGN TAX CONSEQUENCES ARISING FROM THE ACQUISITION AND/OR HOLDING OF THE SECURITIES.

The undersigned acknowledges that the undersigned understands the meaning and legal consequences of the representations and warranties made by the undersigned herein and in the attached Confidential Investor Questionnaire (the terms of which are incorporated herein by this reference) and that the Company is relying on such representations and warranties in making its determination to accept or reject this Subscription. The undersigned hereby agrees to indemnify and hold harmless the Company from and against any and all loss, damage or liability due to or arising out of a breach of any such representation or warranty of the undersigned.

6. Intercreditor Agreement. Each Note issued to Investor is one of a duly authorized issue of Notes of the Company limited in aggregate principal amount to \$750,000. Each Investor understands and agrees that a default is not an Event of Default (as defined in the Note) until the holders of at least 25% in aggregate principal amount of the Notes then outstanding notify the Company of such default and the Company does not cure it within thirty (30) days after receipt of such notice, which must specify the default, demand that it be remedied and state that it is a "Notice of Default." If an Event of Default occurs and is continuing, holders of the Note(s) by notice to the Company, may declare the principal of and accrued interest on the Notes to be due and payable immediately; provided, however, that the holders of at least 60% in aggregate principal amount of the Notes then outstanding, by written notice to the Company, may rescind and annul such declaration and its consequences. The Company shall only make principal reductions pro rata among the holders of the Notes based on principal outstanding. Likewise any Note holder who receives any payments or proceeds from any distribution in connection with a bankruptcy, liquidation, reorganization, dissolution, winding-up or similar proceedings, shall be obligated to pro rate such amounts among the other holders of the Notes.

7. <u>Time; No Revocation</u>. Time shall be of the essence in this Subscription Agreement. The undersigned agrees that this Subscription Agreement and any agreement of the undersigned made hereunder is irrevocable, and that this Subscription Agreement shall survive the death or legal incapacity of the undersigned.

8. Notices. All notices or other communications given or made hereunder shall be in writing and shall be deemed effectively given: (i) upon personal delivery to the party to be notified, (ii) when sent by confirmed telex or facsimile if sent during normal business hours of the recipient, if not, then on the next business day, (iii) five (5) days after having been sent by registered or certified mail, return receipt requested, postage prepaid; or (iv) one (1) day after deposit with a nationally recognized overnight courier, specifying next day delivery, with written verification of receipt, if addressed to the Comp any, at 1941 Ramrod Avenue, Suite #100, Henderson, Nevada 89014, Attn: James Barnes, or to the undersigned at the address set forth on the signature page hereto, or at such other place as a party may designate by written notice to the other party in accordance with this <u>Section 8</u>.

9. <u>Survival of Agreements, Representations and Warranties</u>. All agreements, representations and warranties contained herein or made in writing by or on behalf of a party thereto in connection with the transactions contemplated by this Subscription Agreement shall survive the execution and delivery of this Subscription Agreement, any investigation at any time made by any party, and the sale and purchase of the Securities and payment therefor.

10. <u>Amendment and Waiver</u>. This Agreement and the terms of the Notes may be amended or modified upon the written consent of the Company and holders of at least sixty percent (60%) in the aggregate principal amount of the Notes then outstanding and any such amendments shall be binding upon all Note holders as if such holder had consented and agreed to such amendment. The obligations of the Company and the rights of the holders of the Notes and this Agreement may be waived only with the written consent of the holders of at least sixty percent (60%) in the aggregate principal amount of the Notes then outstanding.

11. <u>Successors and Assigns</u>. This Subscription Agreement shall be binding upon and inure to the benefit of the parties and their successors and permitted assigns. If the undersigned is more than one person, their obligations shall be joint and several, and the agreements, representations, warranties and acknowledgments herein contained shall be deemed to be made by and be binding upon each such person and its successors and assigns. Notwithstanding the foregoing, this Subscription Agreement is not transferable or assignable by the undersigned and any such attempted transfer or assignment shall be null and void.

12. <u>Complete Agreement of the Parties</u>. This Subscription Agreement and the other agreements or documents referred to herein or therein contain the entire agreement of the parties, and there are no representations, covenants or other agreements except as stated or referred to herein and in such other agreements or documents.

13. <u>Severability</u>. Any term or provision of this Subscription Agreement that is invalid or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such invalidity or unenforceability without rendering invalid or unenforceable the remaining terms or provisions of this Subscription Agreement or affecting the validity or enforceability of any of the terms or provisions of this Subscription Agreement in any other jurisdiction.

14. <u>Counterparts</u>. This Subscription Agreement may be executed in any number of counterparts, each of which shall be an original but all of which taken together shall constitute one agreement.

15. Governing Law. This Agreement shall be governed by and construed under the laws of the State of Nevada as applied to agreements among Nevada residents entered into and to be performed entirely within Nevada.

16. General. Signatures transmitted by facsimile shall be effective and binding as if an original. By executing the signature page to this Subscription Agreement, the undersigned agrees to be bound by the foregoing.

[SIGNATURE PAGE FOLLOWS]

## PARAMETRIC SOUND CORPORATION A Nevada Corporation

# SIGNATURE PAGE TO SUBSCRIPTION AGREEMENT

Very truly yours,	Subscription Amount: \$
Print Name of Investor By	If Applicable:
<u>0;</u> Signature Print Title (if applicable)	Print Name of Joint Investor or Other Person Whose Signature Is Required
Social Security Number or Federal Employer Identification Number	By ; Signature
Address:	Print Title (if applicable)
Telephone No	Social Security Number or Federal Employer Identification Number
Facsimile No	Address:
	Telephone No
	Facsimile No
Date:, 2010	Email Address:

## ACCEPTANCE

\_ Dollars (\$\_\_\_\_\_

\_\_\_) upon the terms and

# ACCEPTED BY:

Parametric Sound Corporation	
a Nevada corporation	
By:	_
Its:	_
Dated:	

## 2010 STOCK OPTION PLAN OF PARAMETRIC SOUND CORPORATION

## 1. Purposes of the Plan

The purposes of the 2010 Stock Option Plan (the "Plan") of Parametric Sound Corporation, a Nevada corporation (the "Company"), are to:

(a) Encourage selected employees, directors and consultants to improve operations and increase profits of the Company;

(b) Encourage selected employees, directors and consultants to accept or continue employment or association with the Company or its Subsidiaries; and

(c) Increase the interest of selected employees, directors and consultants in the Company's welfare through participation in the growth in value of the common stock of the Company (the "Shares").

Options granted under the Plan ("Options") may be "incentive stock options" ("ISOs") intended to satisfy the requirements of Section 422 of the Internal Revenue Code of 1986, as amended, and the regulations thereunder (the "Code"), or "non-qualified stock options" ("NQSOs").

## 2. Eligible Persons

Every person who at the date of grant of an Option is an employee of the Company or of any Subsidiary (as defined below) of the Company is eligible to receive NQSOs or ISOs under the Plan. Every person who at the date of grant is a consultant to, or non-employee director of, the Company or any Subsidiary (as defined below) of the Company is eligible to receive NQSOs under the Plan. The term "Subsidiary" as used in the Plan means a subsidiary corporation as defined in the applicable provisions (currently Sections 424(f), respectively) of the Code. The term "employee" (within the meaning of Section 3401(c) of the Code) includes an officer or director who is an employee of the Company. The term "consultant" includes persons employed by , or otherwise affiliated with, a consultant to the Company.

## 3. Stock Subject to the Plan; Maximum Number of Grants

Subject to the provisions of Section 6(a)(i) of the Plan, the total number of Shares which may be issued under Options granted pursuant to the Plan shall not exceed three million (3,000,000) Shares. The Shares covered by the portion of any grant under the Plan which expires unexercised shall become available again for grants under the Plan.

## 4. Administration

(a) The Plan shall be administered by either the Board of Directors of the Company (the "Board") or by a committee (the "<u>Committee</u>") to which administration of the Plan, or of part of the Plan, may be delegated by the Board (in either case, the "<u>Administrator</u>"). The Board shall appoint and remove members of such Committee, if any, in its discretion in accordance with applicable laws. If necessary in order to comply with Rule 16b-3 under the Exchange Act and Section 162(m) of the Code, the Committee shall, in the Board's discretion, be comprised solely of "non-em ployee directors" within the meaning of said Rule 16b-3 and "outside directors" within the meaning of Section 162(m) of the Code. The foregoing notwithstanding, the Administrator may delegate nondiscretionary administrative duties to such employees of the Company as it deems proper, and the Board, in its absolute discretion, may at any time and from time to time exercise any and all rights and duties of the Administrator under the Plan.

(b) Subject to the other provisions of the Plan, the Administrator shall have the authority to, in its discretion: (i) grant Options; (ii) determine the fair market value of the Shares subject to Options; (iii) determine the exercise price of Options granted; (iv) determine the persons to whom, and the time or times at which, Options shall be granted, and the number of shares subject to each Option; (v) interpret the Plan; (vi) prescribe, amend and rescind rules and regulations relating to the Plan; (vii) determine the terms and provisions of each Option granted (which need not be identical), including but not limited to, the time or times at which Options; (ix) adhere (with the consent of the optionee) the exercise date of any Option; (x) authorize any person to execute on behalf of the Company any instrument evidencing the grant of an Option; and (xi) make all other determinations deemed necessary or advisable for the administration of the Plan. The Administrator may delegate nondiscretionary administrative duties to such employees of the Company as it deems proper.

(c) All questions of interpretation, implementation, and application of the Plan shall be determined by the Administrator. Such determinations shall be final and binding on all persons.

## 5. Granting of Options; Option Agreement

(a) No Options shall be granted under the Plan after 10 years from the date of adoption of the Plan by the Board.

(b) Each Option shall be evidenced by a written stock option agreement, in form satisfactory to the Administrator, executed by the Company and the person to whom such Option is granted.

(c) The stock option agreement shall specify whether each Option it evidences is an NQSO or an ISO.

## 6. Terms and Conditions of Options

Each Option granted under the Plan shall be subject to the terms and conditions set forth in Section 6(a). NQSOs shall also be subject to the terms and conditions set forth in Section 6(b), but not those set forth in Section 6(c). ISOs shall also be subject to the terms and conditions set forth in Section (c), but not those set forth in Section 6(b).

(a) Terms and Conditions to Which All Options Are Subject. All Options granted under the Plan shall be subject to the following terms and conditions:

(i) Changes in Capital Structure. Subject to Section 6(a)(ii), if the stock of the Company is changed by reason of a stock split, reverse stock split, stock dividend, or recapitalization, combination or reclassification, appropriate adjustments shall be made by the Board in (1) the number and class of shares of stock subject to the Plan and each Option outstanding under the Plan, and (2) the exercise price of each outstanding Option; provided, however, that the Company shall not be required to issue fractional shares as a result of any such adjustments. Each such adjustment shall be subject to approval by the Board in its sole discretion.

(ii) Corporate Transactions. In the event of the proposed dissolution or liquidation of the Company, the Administrator shall notify each optione at least 30 days prior to such proposed action. To the extent not previously exercised, all Options will terminate immediately prior to the consummation of such proposed action; provided, however, that the Administrator, in the exercise of its sole discretion, may permit exercise of any Options prior to their termination, even if such Options were not otherwise exercisable. In the event of a merger or consolidation of the Company with or into another corporation or entity in which the Company does not survive, or in the event of a all or substantially all of the assets of the Company in which the stockholders of the Company receive securities of the acquiring entity or an affiliate thereof, all Options shall be assumed or equivalent options shall be substituted by the successor corporation (or other entity); provided, however, that if such successor does not agree to assume the Options or to substitute equivalent options therefor, the Administrator, in the exercise of its sole discretion, may permit the exercise of any of the Options prior to consummation of such event, even if such Options were not otherwise exercisable.

(iii) *Time of Option Exercise*. Subject to Section 5 and Section 6(c)(iii), Options granted under the Plan shall be exercisable in accordance with a schedule as specified in the written stock option agreement relating to such Option. In any case, no Option shall be exercisable until a written stock option agreement in form satisfactory to the Company is executed by the Company and the optionee.

(iv) Option Grant Date. The date of grant of an Option under the Plan shall be the date as of which the Administrator approves the grant.

(v) *Nontransferability of Option Rights.* Except with the express written approval of the Administrator which approval the Administrator is authorized to give only with respect to NQSOs, no Option granted under the Plan shall be assignable or otherwise transferable by the optionee except by will, by the laws of descent and distribution or pursuant to a qualified domestic relations order. During the life of the optionee, an Option shall be exercisable only by the optionee.

(vi) *Payment*. Except as provided below, payment in full, in cash, shall be made for all stock purchased at the time written notice of exercise of an Option is given to the Company, and proceeds of any payment shall constitute general funds of the Company. The Administrator, in the exercise of its absolute discretion, may authorize any one or more of the following additional methods of payment:

(1) delivery by the optionee of Shares already owned by the optionee for all or part of the Option price, provided the fair market value (determined as set forth in Section 6(a)(x)) of such Shares being delivered is equal on the date of exercise to the Option price, or such portion thereof as the optionee is authorized to pay by delivery of such stock;

(2) the surrender of Shares then issuable upon exercise of the Option, provided the fair market value (determined as set forth in Section 6(a)(x)) of such Shares is equal on the date of exercise to the Option price, or such portion thereof as the optionee is authorized to pay by surrender of such stock;

(3) cancellation of indebtedness of the Company to the optionee or waiver of compensation due or accrued to optionee for services rendered; and

(4) if and so long as the Shares are registered under Section 12(b) or 12(g) of the Exchange Act, and to the extent permitted by law, delivery of a properly executed exercise agreement or notice, together with irrevocable instructions to a brokerage firm designated or approved by the Administrator to deliver promptly to the Company the aggregate amount of proceeds to pay the option exercise price and any withholding tax obligations that may arise in connection with the exercise, all in accordance with the regulations of the Federal Reserve Board.

(vii) Termination of Employment. If for any reason other than death or permanent and total disability, an optionee ceases to be employed by the Company or any of its Subsidiaries (such event being called a "Termination"), Options held at the date of Termination (to the extent then exercisable) may be exercised in whole or in part at any time within three months of the date of such Termination, or such other period of not less than 30 days after the date of such Termination as is specified in the Option Agreement or by amendment thereof (but in no event after the Expiration Date); provided, however, that if such exercises of the Option would result in liability under Section 16(b) (but in no event after the Expiration Date). If an optionee dies or becomes permanently and totally disabled (within the meaning of Section 22(e)(3) of the Code) while employed by the Company or a Subsidiary or within the period that the Option remains exercisable after Termination, Options then held (to the extent then exercisable) may be exercised, in whole or in part, by the optionee, by the optionee's personal representative or by the person to whom the Option is transferred by devise or the laws of descent and distribution, at any time within twelve months after the death or twelve months after the permanent and total disability of the optionee or any longer period specified in the Option Agreement or by amendment thereof (but in no event after the Expiration Date). For purposes of this Section 6(a)(vii), "employment" includes service as a director or as a consultant. For purposes of this Section 6(a)(vii), an optionee's employment shall not be deemed to terminate by reason of sick leave, military leave or other leave of absence approved by the Administrator, if the period of any such leave does not exceed 90 days or, if longer, if the optionee's right to reemployment or any Subsidiary is guaranteed either contractually or by statute.

(viii) Withholding and Employment Taxes. At the time of exercise of an Option and as a condition thereto, or at such other time as the amount of such obligations becomes determinable (the "<u>Tax Date</u>"), the optionee shall remit to the Company in cash all applicable federal and state withholding and employment taxes. Such obligation to remit may be satisfied, if authorized by the Administrator in its sole discretion, after considering any tax, accounting and financial consequences, by the optionee's (1) delivery of a promissory note in the required amount on such terms as the Administrator d eems appropriate, (2) tendering to the Company previously owned Shares or other securities of the Company with a fair market value equal to the required amount, or (3) agreeing to have Shares (with a fair market value equal to the required amount) which are acquired upon exercise of the Option withheld by the Company.

(ix) Other Provisions. Each Option granted under the Plan may contain such other terms, provisions and conditions not inconsistent with the Plan as may be determined by the Administrator, and each ISO granted under the Plan shall include such provisions and conditions as are necessary to qualify the Option as an "incentive stock option" within the meaning of Section 422 of the Code.

(x) Determination of Value. For purposes of the Plan, the fair market value of Shares or other securities of the Company shall be determined as follows:

(1) Fair market value shall be the closing price of such stock on the date before the date the value is to be determined on the principal recognized securities exchange or recognized securities market on which such stock is reported, but if selling prices are not reported, its fair market value shall be the mean between the high bid and low asked prices for such stock on the date before the date the value is to be determined (or if there are no quoted prices for such date, then for the last preceding business day on which there were quoted prices).

(2) In the absence of an established market for the stock, the fair market value thereof shall be determined in good faith by the Administrator, with reference to the Company's net worth, prospective earning power, dividend-paying capacity and other relevant factors, including the goodwill of the Company, the economic outlook in the Company's industry, the Company's position in the industry, the Company's management and the values of stock of other corporations in the same or similar line of business.

(xi) Option Term. Subject to Section 6(c)(iii), no Option shall be exercisable more than 10 years after the date of grant, or such lesser period of time as is set forth in the stock option agreement (the end of the maximum exercise period stated in the stock option agreement is referred to in the Plan as the "Expiration Date").

(b) Terms and Conditions to Which Only NQSOs Are Subject. Options granted under the Plan which are designated as NQSOs shall be subject to the following terms and conditions:

(i) Exercise Price.

(1) Except as set forth in Section 6(b)(i)(2), the exercise price of an NQSO shall be not less than 85% of the fair market value (determined in accordance with Section 6(a)(x)) of the stock subject to the Option on the date of grant.

(2) To the extent required by applicable laws, rules and regulations, the exercise price of a NQSO granted to any person who owns, directly or by attribution under the Code (currently Section 424(d)), stock possessing more than ten percent of the total combined voting power of all classes of stock of the Company or of any Subsidiary (a "<u>Ten Percent Stockholder</u>") shall in no event be less than 110% of the fair market value (determined in accordance with Section 6(a)(x)) of the stock covered by the Option at the time the Option is granted.

(c) Terms and Conditions to Which Only ISOs Are Subject. Options granted under the Plan which are designated as ISOs shall be subject to the following terms and conditions:

(i) Exercise Price.

(1) Except as set forth in Section 6(c)(i)(2), the exercise price of an ISO shall be determined in accordance with the applicable provisions of the Code and shall in no event be less than the fair market value (determined in accordance with Section 6(a)(x)) of the stock covered by the Option at the time the Option is granted.

(2) The exercise price of an ISO granted to any Ten Percent Stockholder shall in no event be less than 110% of the fair market value (determined in accordance with Section 6(a) (x)) of the stock covered by the Option at the time the Option is granted.

(ii) *Disqualifying Dispositions*. If stock acquired by exercise of an ISO granted pursuant to the Plan is disposed of in a "disqualifying disposition" within the meaning of Section 422 of the Code (a disposition within two years from the date of grant of the Option or within one year after the transfer such stock on exercise of the Option), the holder of the stock immediately before the disposition shall promptly notify the Company in writing of the date and terms of the disposition and shall provide such other information regarding the Option as the Company may reasonably require.

(iii) Term. Notwithstanding Section 6(a)(xi), no ISO granted to any Ten Percent Stockholder shall be exercisable more than five years after the date of grant.

## 7. Manner of Exercise

(a) An optionee wishing to exercise an Option shall give written notice to the Company at its principal executive office, to the attention of the officer of the Company designated by the Administrator, accompanied by payment of the exercise price and withholding taxes as provided in Sections 6(a)(vi) and 6(a)(viii). The date the Company receives written notice of an exercise hereunder accompanied by payment of the exercise price will be considered as the date such Option was exercised.

(b) Promptly after receipt of written notice of exercise of an Option and the payments called for by Section 7(a), the Company shall, without stock issue or transfer taxes to the optionee or other person entitled to exercise the Option, deliver to the optionee or such other person a certificate or certificates for the requisite number of shares of stock. An optionee or permitted transferee of the Option shall not have any privileges as a stockholder with respect to any shares of stock covered by the Option until the date of issuance (as evidenced by the appropriate entry on the books of the Company or a duly authorized transfer agent) of such shares.

## 8. Employment or Consulting Relationship

Nothing in the Plan or any Option granted hereunder shall interfere with or limit in any way the right of the Company or of any of its Subsidiaries to terminate any optionee's employment or consulting at any time, nor confer upon any optionee any right to continue in the employ of, or consult with, the Company or any of its Subsidiaries.

## 9. Conditions Upon Issuance of Shares

Shares shall not be issued pursuant to the exercise of an Option unless the exercise of such Option and the issuance and delivery of such shares pursuant thereto shall comply with all relevant provisions of law, including, without limitation, the Securities Act of 1933, as amended (the "Securities Act").

## 10. Non-Exclusivity of the Plan

The adoption of the Plan shall not be construed as creating any limitations on the power of the Company to adopt such other incentive arrangements as it may deem desirable, including, without limitation, the granting of stock options other than under the Plan.

### 11. Amendments to the Plan

The Board may at any time amend, alter, suspend or discontinue the Plan. Without the consent of an optionee, no amendment, alteration, suspension or discontinuance may adversely affect outstanding Options except to conform the Plan and ISOs granted under the Plan to the requirements of federal or other tax laws relating to incentive stock options. No amendment, alteration, suspension or discontinuance shall require stockholder approval unless (a) stockholder approval is required to preserve incentive stock option treatment for federal income tax purposes or (b) the Board otherwise concludes that stockholder approval is advisable.

## 12. Effective Date of Plan; Termination

The Plan shall become effective upon adoption by the Board; provided, however, that no Option shall be exercisable unless and until written consent of the stockholders of the Company, or approval of stockholders of the Company voting at a validly called stockholders' meeting, is obtained within twelve months after adoption by the Board. If such stockholder approval is not obtained within such time, Options granted hereunder shall be of the same force and effect as if such approval was obtained except that all ISOs granted hereunder shall be treated as NQSOs. Options may be granted and exercised under the Plan only after there has been compliance with all applicable federal and state securities laws. The Plan shall terminate within ten years from the date of its adoption by the Board.

## PARAMETRIC SOUND CORPORATION 2010 STOCK OPTION PLAN

## STOCK OPTION GRANT NOTICE AND STOCK OPTION AGREEMENT

Parametric Sound Corporation, a Nevada corporation (the "<u>Company</u>"), pursuant to its 2010 Stock Option Plan (the "<u>Plan</u>"), hereby grants to the holder listed below ( "<u>Holder</u>"), an option to purchase the number of shares of the Company's Common Stock set forth below (the "<u>Option</u>"). This Option is subject to all of the terms and conditions as set forth herein and in the Stock Option Agreement attached hereto as Exhibit A (the "<u>Stock Option Agreement</u>") and the Plan, each of which are incorporated herein by reference. Unless otherwise defined herein, the terms defined in the Plan shall have the same defined meanings in this Grant Notice and the Stock Option Agreement.

Holder	
Grant No.	
Date of Grant	
Shares Subject to Option	
Exercise Price Per Share	\$ per share
Expiration Date	
Type of Option	o Non-Qualified Stock Option o Incentive Stock Option (within the meaning of Section 422 of the Code)
Vesting Schedule:	
Vesting Date	Number of Shares Vested

By his or her signature and the Company's signature below, Holder agrees to be bound by the terms and conditions of the Plan, the Stock Option Agreement and this Grant Notice. Holder has reviewed the Stock Option Agreement, the Plan and this Grant Notice in their entirety, has had an opportunity to obtain the advice of counsel prior to executing this Grant Notice and fully understands all provisions of this Grant Notice, the Stock Option Agreement and the Plan. Holder hereby agrees to accept as binding, conclusive and final all decisions or interpretations of the Administrator of the Plan upon any questions arising under the Plan or the Option.

Parametric Sound Corporation:

By: \_\_\_\_ Name: \_\_ Title: \_\_\_ Address: Holder:

Name:	
Indille:	_
Title:	_
Address:	_

[] 60;

## EXHIBIT A TO STOCK OPTION GRANT NOTICE

## PARAMETRIC SOUND CORPORATION STOCK OPTION AGREEMENT

Pursuant to the Stock Option Grant Notice ("<u>Grant Notice</u>") to which this Stock Option Agreement (this "<u>Agreement</u>") is attached, Parametric Sound Corporation, a Nevada corporation (the "<u>Company</u>"), has granted you an Option under the Company's 2010 Stock Option Plan (the "<u>Plan</u>") to purchase the number of shares of the Company's common stock ("<u>Stock</u>") indicated in the Grant Notice at the exercise price indicated in the Grant Notice. Capitalized terms not explicitly defined in this Agreement shall have the meanings specified in the Plan and the Grant Notice.

1. Vesting. Except as otherwise provided in the Plan, this Option will become vested and exercisable in such amounts and at such times as provided in the Grant Notice.

## 2. Exercise of the Option.

2.1 <u>Method of Exercise</u>. You may exercise the vested portion of this Option at any time prior to the expiration of the Option by delivering a notice of exercise in such form as may be designated by the Company from time to time together with the exercise price to the Secretary of the Company, or to such other person as the Company may designate, during regular business hours and prior to the expiration of the Option, together with such additional documents as the Company may then require pursuant to the terms of the Plan.

2.2 <u>Method of Payment</u>. Payment of the exercise price may be by: (a) cash (or check); [(b) delivery of Stock already owned by you for all or part of the Option price, provided the fair market value (determined as set forth in the Plan) of such Stock being delivered is equal on the date of exercise to the Option price, or such portion thereof as you are authorized to pay by delivery of such Stock;] [(c) the surrender of Shares then issuable upon exercise of the Option, provided the fair market value (determined as set forth in the Plan) of such Shares is equal on the date of exercise to the Option price, or such portion thereof as you are authorized to pay by surrender of such Stock;] [(d) cancellation of indebtedness of the Company to you or waive r of compensation due or accrued to you for services rendered;] [(e) if and so long as the Shares are registered under Section 12(b) or 12(g) of the Exchange Act, and to the extent permitted by law, delivery of a properly executed exercise agreement or notice, together with irrevocable instructions to a brokerage firm designated or approved by the Administrator to deliver promptly to the Company the aggregate amount of proceeds to pay the option exercise price and any withholding tax obligations that may arise in connection with the exercise, all in accordance with the regulations of the Federal Reserve Board;] or (f) a combination of the above methods, as the Company may designate from time to time.

2.3 <u>Tax Withholding</u>. By exercising this Option you agree that as a condition to any exercise of this Option, the Company may withhold from your pay and any other amounts payable to you, or require you to enter an arrangement providing for the payment by you to the Company of any tax withholding obligation of the Company arising by reason of (1) the exercise of this Option; (2) the lapse of any substantial risk of forfeiture to which the Stock is subject at the time of exercise; or (3) the disposition of Stock acquired upon such exercise.

2.4 <u>Responsibility for Exercise</u>. You are responsible for taking any and all actions as may be required to exercise this Option in a timely manner and for properly executing any such documents as may be required for exercise in accordance with such rules and procedures as may be established from time to time. By signing this Agreement you acknowledge that information regarding the procedures and requirements for this exercise of the Option is available to you on request. The Company shall have no duty or obligation to notify you of the expiration date of this Option.

2.5 <u>Special Tax Consequences</u>. If you have been granted an Incentive Stock Option, you acknowledge that, to the extent that the aggregate fair market value of stock with respect to which "incentive stock options" (within the meaning of Section 422 of the Code, but without regard to Section 422(d) of the Code), including the Option, are exercisable for the first time by you during any calendar year (under the Plan and all other incentive stock option plans of the Company and any Subsidiary) exceeds \$100,000, such options shall be treated as non-qualified options. You further acknowledge that the rule set forth in the preceding sentence shall be applied by taking options into account in the order in which they were granted. For purposes of these rules, the fair market value of stock shall be determined as of the time the option with respect to such stock is granted.

3. Securities Law Compliance. Notwithstanding anything to the contrary contained herein, this Option may not be exercised unless the Stock issuable upon exercise of this Option is then registered under the Securities Act or, if such Stock is not then so registered, the Company has determined that such exercise and issuance would be exempt from the registration requirements of the Securities Act. In addition, you acknowledge that the Plan is intended to conform to the extent necessary with all provisions of the Securities Act and the Exchange Act and any and all regulations and rules promulgated by the Securities and Exchange Commission thereunder, including without limitation Rule 16b-3. Notwithstanding anything herein to the contrary, the Plan shall be administered, and the Option is granted and may be exercised, only in such a manner as to conform to such laws, rules and regulations. To the extent permitted by applicable law, the Plan and this Agreement shall be deemed to the extent necessary to conform to such laws, rules and regulations.

4. **Termination of the Option**. The term of this Option commences on the Date of Grant (as specified in the Grant Notice) and expires and shall no longer be exercisable upon the earliest of: (a) the Expiration Date indicated in the Grant Notice; (b) the tenth (10th) anniversary of the Date of Grant; (c) if you owned (within the meaning of Section 424(d) of the Code), at the time the Option was granted, more than ten percent (10%) of the total combined voting power of all classes of stock of the Company or any Subsidiary, the fifth (5<sup>th</sup>) anniversary of the Date of Grant, (d) the last day for exercising the Option following termination of your service as described in the Plan; or (e) a change of control, to the extent provided in the Plan.

5. **Option Not a Service Contract**. This Option is not an employment or service contract and nothing in this Agreement, the Grant Notice or the Plan shall be deemed to create in any way whatsoever any obligation on your part to continue in the service of the Company, or of the Company to continue your service with the Company. In addition, nothing in your Option shall obligate the Company, its stockholders, board of directors, officers or employees to continue any relationship which you might have as a director or consultant for the Company.

6. **Representations**. By executing the Grant Notice to which this Agreement is attached, you hereby warrant and represent that you are acquiring this Option for your own account and that you have no intention of distributing, transferring or selling all or any part of this Option except in accordance with the terms of this Agreement. You also hereby warrant and represent that you have either (i) preexisting personal or business relationships with the Company or any of its officers, directors or controlling persons, or (ii) the capacity to protect your own interests in connection with the grant of this Option by virtue of the business or financial expertise of any of your professional advisors who are unaffiliated with and who are not compensated by the Compa ny or any of its affiliates, directly or indirectly.

7. Notices. Any notices provided for in this Agreement, the Grant Notice or the Plan shall be given in writing and shall be deemed effectively given upon receipt or, in the case of notices delivered by the Company to you, five (5) days after deposit in the United States mail, postage prepaid, addressed to you at the last address you provided to the Company.

8. Governing Law. This Agreement shall be administered, interpreted and enforced under the laws of the State of Nevada without regard to conflicts of laws thereof. Should any provisions of this Agreement be determined by a court of law to be illegal or unenforceable, the other provisions shall nevertheless remain effective and shall remain enforceable.

9. Amendment. The Board may amend the Option at any time, provided no such amendment may adversely affect the Option or any unexercised portion of the Option without your consent unless such amendment is necessary to comply with any applicable law or government regulation. No amendment or addition to this Agreement shall be effective unless in writing or in such electronic form as may be designated by the Company.

10. Governing Plan Document. Your Option is subject to this Agreement, the Grant Notice and all the provisions of the Plan, which are hereby made a part of this Agreement, and is further subject to all interpretations, amendments, rules and regulations which may from time to time be promulgated and adopted pursuant to the Plan. In the event of any conflict between the provisions of this Agreement, the Grant Notice and those of the Plan, the provisions of the Plan shall control.

11. Rights as Stockholder. You shall not be, nor have any of the rights or privileges of, a stockholder of the Company in respect of any shares purchasable upon exercise of any part of the Option unless and until such shares shall have been issued by the Company to you.

12. Shares to Be Reserved. The Company shall at all times during the term of the Option reserve and keep available such number of shares of Common stock as will be sufficient to satisfy the requirements of the Option.

## [FORM OF] INDEMNIFICATION AGREEMENT

THIS INDEMNIFICATION AGREEMENT (this "<u>Agreement</u>") is made this 27th day of September, 2010, between Parametric Sound Corporation, a Nevada corporation (the "<u>Company</u>"), and \_\_\_\_\_\_\_, an individual ("<u>Indemnitee</u>").

## RECITALS

WHEREAS, Indemnitee is a member of the board of directors ("Board" or "Board of Directors") and/or an officer of the Company;

WHEREAS, the Corporation has adopted bylaws ("Bylaws") providing for the indemnification of the directors and executive officers of the Company ("Covered Persons");

WHEREAS, the Bylaws and Nevada Revised Statute Sections 78.751 and 78.7502 (the "State Statutes") specifically provide that they are not exclusive, and thereby contemplate that agreements may be entered into between the Company and a Covered Person with respect to indemnification of such Covered Person;

WHEREAS, Indemnitee is willing to serve, to continue to serve, and to take on additional service for and on behalf of the Company on the condition that Indemnitee is indemnified as set forth in this Agreement;

WHEREAS, it is intended that Indemnitee shall be paid promptly by the Company all amounts necessary to effectuate in full the indemnity provided in this Agreement; and

WHEREAS, to induce Indemnitee to continue to serve as a director and/or officer of the Company, the Company has determined and agreed to enter into this Agreement with Indemnitee.

NOW, THEREFORE, in consideration of Indemnitee's continued service as a director and/or officer of the Company after the date hereof, and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Company and Indemnitee hereby agree as follows:

## AGREEMENT

1. <u>Indemnification of Indemnitee</u>. The Company hereby agrees to hold harmless and indemnify Indemnitee to the fullest extent authorized or permitted by the provisions of the State Statutes, or any successor statute or amendment thereof, or any other statutory provisions authorizing or permitting such indemnification that is adopted after the date of this Agreement.

2. Additional Indemnity. Subject only to the exclusions set forth in Section 3 of this Agreement, the Company hereby further agrees to hold harmless, indemnify and defend Indemnitee:

(a) against any and all expenses (including fees for attorneys, accountants, private investigators, court and transcript costs, fees and expenses of witnesses, travel expenses and all other like disbursements or expenses reasonably incurred by or for Indemnitee), judgment damages, fines, penalties and amounts paid in settlement (including all interest assessments and other charges paid or payable in connection with or in respect of such judgment, fines, penalties or amounts paid in settlement) actually and reasonably incurred by or for Indemnitee in connection with any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (including an action by or in the right of the fact that at the time of the act or omission which is the subject matter of such Covered Action Indemnitee was a director, officer or employee of the Company; and

(b) otherwise to the fullest extent as may be provided to Indemnitee by the Company under the non-exclusivity provisions of the Bylaws of the Company, the State Statutes or any employment agreement between the Company and Indemnitee. The provisions of this Agreement are in addition to, and not in limitation of, the provisions of such Bylaws, the State Statutes or any employment agreement between the Company and Indemnitee.

3. Limitations on Additional Indemnity. No indemnity pursuant to Sections 1 and 2 of this Agreement shall be paid by the Company to the extent that:

(a) payment therefor is actually made to Indemnitee under a valid and collectible insurance policy or policies, except with respect to any excess amount due to Indemnitee beyond the amount of payment to Indemnitee under such insurance policy or policies. Notwithstanding the availability of such insurance policy or policies, Indemnitee also may claim indemnification from the Company pursuant to this Agreement by assigning to the Company in writing any claims of Indemnitee under such insurance policy or policies to the extent of the amount Indemnitee is paid by the Company;

(b) Indemnitee is indemnified by the Company otherwise than pursuant to this Agreement;

(c) final judgment is rendered against Indemnitee for the payment of dividends or other distributions to stockholders of the Company in violation of the provisions of Subsection 2 of Nevada Revised Statutes Section 78.300, as amended;

(d) final judgment is rendered against Indemnitee for an accounting of profits made from the purchase or sale by Indemnitee of securities of the Company pursuant to the provisions of Section 16(b) of the Securities Exchange Act of 1934, as amended (the "Act"), or other similar provisions of any federal, state or local statutory law;

(e) Indemnitee's conduct giving rise to the claim for indemnification is finally adjudged by a court of competent jurisdiction to have been a breach of fiduciary duty which involved intentional misconduct, fraud or a knowing violation of the law; and/or

(f) except as otherwise provided in this Agreement, in connection with all or any part of a suit or other proceeding which is initiated or maintained by or on behalf of Indemnitee, or any suit or other proceeding by Indemnitee against the Company or its directors, officers, employees or other agents, unless (i) such indemnification is expressly required by Nevada law; (ii) the suit or other proceeding was expressly authorized by an official act of the Board of Directors of the Company or (iii) such indemnification is provided by the Company, in its sole discretion, pursuant to the powers vested in the Company under Nevada law.

4. <u>Continuation of Indemnity</u>. All agreements and obligations of the Company contained in this Agreement shall continue during the period Indemnitee is a Covered Person, and shall continue thereafter for so long as Indemnitee shall be subject to any possible claim or threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative, by reason of the fact that Indemnitee was a Covered Person.

5. Advancement of Expenses. In the event Indemnitee incurs costs or expenses in connection with the defense of any such civil, criminal, administrative or investigative action, suit or proceeding (including any costs or expenses incurred for any appeal therefor), the Company agrees to pay such costs or expenses within 30 calendar days of submission of bills or vouchers for such costs or expenses, provided that Indemnitee delivers to Company prior to such payment a written undertaking by or on behalf of Indemnitee to repay the amount paid by the Company, including amounts paid in settlement, if it is ultimately determined by a court of competent jurisdiction that Indemnitee is not entitled to be indemnified by the Company for such expenses under the provisions of the State Statutes, the Bylaws, this Agreement or otherwise. However, in the case of an action brought against Indemnitee by the Company pursuant to the provisions of Section 16(b) of the Act, or other similar provisions of any federal, state is approved by the Board of Directors by a majority vote of a quorum consisting of directors who are not parties to the action, suit or proceeding, or, if such a quorum cannot be obtained, by independent legal counsel in a written opinion that such indemnification is proper in the circumstances.

6. <u>Presumptions and Effect on Certain Proceedings</u>. Upon making a request for indemnification, Indemnitee shall be presumed to be entitled to indemnification under this Agreement. The termination of any action, suit or proceeding by judgment, order, settlement, arbitration award, conviction or by a plea of nolo contendere or its equivalent shall not affect this presumption except as may be provided in Section 3 of this Agreement.

7. Notification and Defense of Claim. Promptly after receipt by Indemnitee of notice of the commencement of any action, suit or proceeding, if a request with respect thereto is to be made against the Company under this Agreement, Indemnitee shall notify the Company of the commencement thereof; but the failure by Indemnitee to notify the Company will not relieve the Company of any liability which it may have to Indemnitee under this Agreement or otherwise. With respect to any such action, suit or proceeding as to which Indemnitee notifies the Company as required herein:

## (a) The Company shall be entitled to participate therein at its own expense;

(b) Except as otherwise provided below, to the extent that it may wish, the Company, jointly with any other indemnifying party similarly notified, shall be entitled to assume the defense of Indemnitee with counsel reasonably satisfactory to Indemnitee. After notice from the Company to Indemnitee of its election to assume the defense of Indemnitee in the action, suit or proceeding, the Company will not be liable to Indemnitee under this Agreement for any legal or other expenses subsequently incurred by Indemnitee in connection with the defense thereof, other than reasonable costs of investigation or as otherwise provided below. Indemnitee shall have the right to employ its own counsel in such action, suit or proceeding, but the fees and expenses of such counsel incurred after notice from the Company of its assumption of the defense shall be at the sole expense of Indemnitee unless (i) the employment of counsel by Indemnitee at the Company's expense has been authorized in writing by the Company; (ii) Indemnitee shall have reasonably concluded, upon advice of counsel experienced in such matters, that there may be a conflict of interest between the Company and Indemnitee in the conduct of the defense of such action; or (iii) the Company shall not in fact have employed counsel to assume the defense of such action, suit or proceeding. In each such instance set forth in (i) through (iii) of this paragraph (b), the reasonable cost of Indemnitee's counsel shall be borne by the Company. Notwithstanding the foregoing, the Company shall not be entitled to assume the defense of any action, suit or proceeding brought against Indemnitee by or on behalf of the Company or as to which Indemnitee shall have reasonably made the conclusion provided in (ii) above; and

(c) The Company shall not be liable to indemnify Indemnitee under this Agreement for any amounts paid in settlement of any action or claim effected without the Company's prior written consent. The Company shall not settle any action or claim in any manner that would impose any penalty or limitation on Indemnitee without Indemnitee's prior written consent. Neither the Company nor Indemnitee will unreasonably withhold consent to any proposed settlement.

## 8. Enforcement.

(a) The Company expressly confirms and agrees that it has entered into this Agreement and assumed the obligations imposed on the Company hereby in order to induce Indemnitee to continue as a Covered Person, and acknowledges that Indemnitee is relying on this Agreement in continuing in such capacity.

(b) In the event Indemnitee is required to bring any action to enforce his or her rights or to collect moneys due under this Agreement, the Company shall advance Indemnitee all of Indemnitee's reasonable fees and expenses in bringing and pursuing such action. Indemnitee shall be responsible for reimbursement to the Company of such advance unless it is ultimately determined by a court of competent jurisdiction that Indemnity is entitled to be indemnified by the Company for such fees and expenses under the provisions of the Statutes, the Bylaws, this Agreement or otherwise.

9. No Employment Rights. Nothing in this Agreement is intended to confer on Indemnitee any right to continue in the employ of the Company for any period of time or to interfere with or otherwise restrict in any way the rights of the Company or of Indemnitee, which rights are hereby expressly reserved by each, to terminate Indemnitee's service at any time and for any reason, with or without cause, except as may be provided otherwise in an agreement, if any, between the Company and Indemnitee.

10. <u>Severability</u>. Each of the provisions of this Agreement are separate and distinct and independent of one another, so that if any provision of this Agreement shall be held by a court of competent jurisdiction to be invalid or unenforceable for any reason, such invalidity or unenforceability shall not effect the validity or enforceability of the other provisions of this Agreement. If any provision of this Agreement is so held to be invalid or unenforceable, the parties agree that the court making such determination shall have the power to amend such provision or to delete specific words or phrases so that such provision shall then be enforceable to the fullest extent permitted by law unless such change is contrary to the intent of the parties her eto.

11. <u>Subrogation</u>. In the event of payment to Indemnitee under this Agreement, the Company shall be subrogated to the extent of the amount of such payment to all rights of recovery of Indemnitee, who shall execute all papers required and shall do everything that may be necessary or reasonable to secure such rights, including, without limitation, the execution of such documents necessary or reasonable to enable the Company to effectively bring suit to enforce such rights.

12. Governing Law. This Agreement shall be governed by, and construed in accordance with, the laws of the State of Nevada without resort to conflict of laws principles.

13. <u>Notices</u>. All notices, requests, demands and other communications hereunder shall be in writing and shall be deemed to have been duly given (i) when delivered by hand and receipted for by the party to whom said communication shall have been directed or (ii) if mailed by certified or registered mail with postage prepaid, on the third business day after the date on which said communication is so mailed and addressed to the appropriate party at the following address (as such address may be changed by a party by delivering notice of such change in the manner set forth in this Section 13):

If to Indemnitee:

If to the Company:

Parametric Sound Corporation 1941 Ramrod Avenue Suite #100 Henderson, NV 89014

14. <u>Binding Effect; Amendment</u>. This Agreement shall be binding on the parties, their heirs, personal representatives, successors and assigns, and shall inure to the benefit of Indemnitee, his or her heirs, personal representatives and assigns, and to the benefit of the Company, its successors and assigns. No amendment, modification, termination or cancellation of this Agreement shall be effective unless in a writing signed by both parties.

IN WITNESS WHEREOF, the parties hereto have executed this Indemnification Agreement as of the date first above written.

INDEMNITEE:

By:

COMPANY: Parametric Sound Corporation, a Nevada corporation

By:

/s/ Elwood G. Norris Chief Executive Officer



## Dear LRAD Corporation Stockholder:

As previously announced, the Board of Directors of LRAD Corporation has approved a plan to spin-off its newly formed subsidiary, Parametric Sound Corporation, as a new independent, standalone public company holding specified assets that comprise most of our HyperSonic Sound (HSS) technology business. The spin-off was approved by LRAD Corporation stockholders at a special meeting held on June 2, 2010.

The spin-off of Parametric Sound is scheduled to occur effective on September 27, 2010. Holders of record of shares of LRAD Corporation common stock as of the close of business on September 10, 2010 will receive one share of Parametric Sound common stock for every two shares of LRAD Corporation common stock held. Following the spin-off, LRAD Corporation stockholders will own (i) shares of Parametric Sound and (ii) shares of LRAD Corporation. LRAD Corporation will continue to operate its existing sound business, including LRAD and other technologies, but excluding the HSS technology sound business to be held by Parametric Sound. No action is required on your part to receive shares of Parametric Sound common stock.

You will not be required to pay anything or provide any consideration for the shares of Parametric Sound or to surrender any LRAD Corporation shares. We expect the distribution of shares will be electronically credited to your account by your broker or other nominee and physical stock certificates will be issued only to those holding physical certificates in registered form on the record date. No fractional shares will be issued. Any fractional shares of Parametric Sound that would otherwise be issued will be rounded up to the next whole share.

We anticipate that, following the spin-off, quotations for shares of Parametric Sound common stock will be available on the OTC Bulletin Board (OTCBB) quotation and trading system, and that a ticker symbol will be assigned for the Parametric Sound common stock shortly before quotations for the Parametric Sound common stock first become available. It may be difficult to trade Parametric Sound common stock due to the potential for low and sporadic trading activity on the OTCBB quotation and trading system. LRAD Corporation common stock will continue to be listed on the Nasdaq Capital Market under the symbol "LRAD."

I encourage you to read carefully the enclosed information statement, which is being provided to all LRAD Corporation stockholders. It describes the spin-off in detail and contains important business and financial information about Parametric Sound.

I want to thank you for your continued support of LRAD Corporation and we look forward to your support of Parametric Sound in the future.

Sincerely,

/s/ THOMAS R. BROWN Thomas R. Brown President and Chief Executive Officer LRAD Corporation

September 15, 2010



## Dear Parametric Sound Stockholder:

We are very pleased to welcome you as a stockholder of our company, Parametric Sound Corporation. Our goal is to realize the potential of parametric directed sound, primarily targeting digital signage, point-of-purchase, in-store networks and related markets and applications. We also see the opportunity to apply directed sound to new markets.

Since retiring last year from an active role at LRAD Corporation, I have wanted to see the HSS technology achieve greater commercial success. In January 2010, I began evaluating the HSS product offering. The evaluation resulted in the invention of a new electronic processing method to improve the sound from HSS emitters. I also recognized that our directed sound system needs to be more cost competitive with other audio systems. Accordingly, we plan to introduce a new, more competitively priced, product line.

Promptly following the spin-off, Syzygy Licensing LLC, a company in which I am the majority owner, will license to Parametric Sound new patent pending technology I have invented. We plan to use this new technology to complement HSS emitters to produce at lower cost a directional sound solution capable of higher sound volume with improved clarity, wider frequency response and less distortion than the current HSS product offering.

We are excited about our prospects and believe we will be in a better position to realize growth opportunities as an independent, stand-alone company. Looking forward, we intend to rapidly develop and commercialize our new product line, and market it to existing and new customers. We believe the improved sound and reduced system cost will create new demand and open new markets for directional sound applications.

We anticipate that, following the spin-off, quotations for shares of Parametric Sound common stock will be available on the OTCBB quotation and trading system, and that a ticker symbol for our common stock will be assigned shortly before quotations for our common stock first become available. It may be difficult to trade Parametric Sound common stock due to the potential for low and sporadic trading activity on the OTCBB quotation and trading system.

You are invited to learn more about our company by reading the enclosed information statement. On behalf of our Board of Directors, I thank you in advance for your support as we focus on delivering long-term stockholder value as an independent, stand-alone company.

Sincerely,

/s/ ELWOOD G. NORRIS Elwood G. Norris Chief Executive Officer Parametric Sound Corporation

September 15, 2010

## INFORMATION STATEMENT



## **Parametric Sound Corporation**

Common Stock (par value \$0.001 per share)

This information statement is being furnished in connection with the distribution to holders of common stock, par value \$0.00001 per share, of LRAD Corporation ("LRAD Corporation") of all the outstanding shares of common stock, par value \$0.001 per share, of Parametric Sound Corporation ("Parametric Sound").

We are currently a subsidiary of LRAD Corporation. Following the spin-off, our business will consist of specified assets that currently comprise most of the HSS technology business of LRAD Corporation.

Shares of our common stock will be distributed to holders of LRAD Corporation common stock of record as of the close of business on September 10, 2010, which will be the record date. These stockholders will receive one share of our common stock for every two shares of LRAD Corporation common stock held on the record date. We expect the spin-off of our shares will be electronically credited to you or to your brokerage firm on your behalf and physical stock certificates will be issued only to registered holders of physical certificates on the record date. The spin-off will be tax-free for stockholders; however, we do not have any ruling from the U.S. Internal Revenue Service nor do we have a favorable opinion by our accounting firm o r any other expert confirming the spin-off 's tax-free status. You should consult your own tax advisor as to the particular consequences of the spin-off to you.

We are not asking you for a proxy and you are requested not to send us a proxy. LRAD Corporation stockholders will not be required to pay or provide any consideration for the shares of our common stock to be received by them in the spin-off or to surrender or exchange shares of LRAD Corporation common stock in order to receive our common stock or to take any other action in connection with the spin-off.

No public market currently exists for Parametric Sound common stock. Following the spin-off, we currently anticipate that quotations for our common stock will be available on the OTCBB quotation and trading system, and that a ticker symbol for our common stock will be assigned for our common stock shortly before quotations for our common stock first become available. It is possible that a limited trading market, known as a "when-issued" trading market, may develop on or shortly before the record date for the spin-off and that regular trading, to the extent available on the OTCBB quotation and trading system, will begin on the first trading day after the effective date of the spin-off. LRAD Corporation will continue to list its common stock on the Nasdaq Capital Market under the symbol "LRAD" foll owing the spin-off.

In reviewing this information statement, you should carefully consider the matters described under "Risk Factors" beginning on page 11 for a discussion of certain factors that should be considered by holders of our common stock.

Neither the Securities and Exchange Commission nor any state securities commission has approved or disapproved of these securities or determined if this information statement is truthful or complete. Any representation to the contrary is a criminal offense.

This information statement does not constitute an offer to sell or the solicitation of an offer to buy any securities.

The date of this information statement is September 2, 2010.

This information statement was first mailed to LRAD Corporation stockholders on or about September 15, 2010.

## <u>CONTENTS</u>

Executive Summary	1
Questions and Answers About the Spin-Off	7
Risk Factors	11
Special Note About Forward-Looking Statements	
The Spin-Off	19
	20
Dividend Policy	27
Capitalization and Financing	27
Unaudited Pro Forma Financial Statements	
Managements, Discussion and Analysis of Discussion Condition and Develop of Converting	28
Management's Discussion and Analysis of Financial Condition and Results of Operations	34
Business	39
Management	
	46
Director Compensation	48
Executive Compensation	10
	48
Security Ownership by Certain Beneficial Owners and Management	51
Our Relationship with LRAD Corporation After the Spin-Off	
	52
Certain Relationships and Related Party Transactions	55
Legal Proceedings	56
Description of Our Capital Stock	
	56
Limitation of Liability and Indemnification of Directors and Officers	58
Description of Indebtedness	
-	59
Where You Can Find More Information	59
Index to Historical Financial Statements	33
	60

## EXECUTIVE SUMMARY

The following is a summary of the material terms of the transaction described in this information statement. This summary is qualified in its entirety by the more detailed information set forth elsewhere in this information statement, including our financial statements and notes thereto. We urge you to read carefully the entire information statement, especially the risks discussed under "Risk Factors" and our financial statements.

Unless the context otherwise requires, references in this information statement to "Parametric Sound," "we," "us" and "our" refer to Parametric Sound Corporation; references to "LRAD Corporation" refer to LRAD Corporation and its subsidiaries, excluding Parametric Sound after giving effect to the spin-off. The transaction in which we will be spun-off from LRAD Corporation and become an independent, stand-alone company is sometimes referred to in this information statement as the "separation," the "distribution" or the "spin-off."

## **Our Company**

We are a sound technology company focused on delivering directed parametric sound solutions to customers, primarily targeting digital signage, point-of-purchase, in-store networks and related markets and applications that benefit from sound that can be focused and controlled in specified locations.

Parametric sound technology creates sound "in the air." Audible sound is generated along an ultrasonic column using frequencies above the normal range of hearing. This parametric sound beam is highly directional and maintains sonic intelligibility over longer distances than traditional loudspeakers. Our technology is compatible with any media input but beams focused sound where you want it and nowhere else. We own important intellectual property establishing us as the leader in the field of parametric non-linear acoustics for sound reproduction.

A directed sound solution includes (a) the ability to accept various user media sources as input, typically a computer, a DVD player, radio, television or microphone, and (b) custom processing and control electronics allowing an amplifier to drive speakers (called emitters in parametric sound applications). We intend to continue the pioneering work performed in this field by LRAD Corporation. Since 2004, LRAD Corporation has sold over 11,000 HSS systems, mostly the HSS H450 product model. We plan to replace the H450 model with an improved solution employing improved emitters complemented by new patent-pending processing and control electronics developed by our Chief Executive Officer, Elwood G. Norris, and that will be licensed by us following the distribution. Our prototypes evidence improved solution frequ ency response and reduced distortion with higher sound volume from less power input. Due to reduced component costs and manufacturing complexity, we expect to produce new produces at a reduced cost compared to the H450.

We believe improved sound quality produced at reduced cost differentiates us from competitors and opens new market opportunities for directed sound solutions. The existing emitter design has been proven repeatable, and we believe our new processing and control electronics package can be readily assembled by any number of electronic component manufacturers reducing the overall risk of introducing our new generation of products.

## Our Strengths and Challenges

We believe the following competitive strengths position us for future operating success, growth and profitability:

- We pioneered parametric sound, and there is limited competition in this market niche.
- · We have a strong intellectual property position in this market.
- We have an improved electronics solution that enhances our product offering in this market.
- · Through earlier marketing efforts, we have gained an understanding of the needs of customers and have designed our new product with additional features to meet those needs.
- · Our focus only on parametric sound affords us greater marketing flexibility to meet customer requirements.
- · We have limited overhead that allows for efficient results for our stockholders.
- · Our management team has a strong technical base in this technology and experience in bringing products to market.

We expect to face significant challenges and uncertainties, such as the following, to achieve our goals:

- We need to rapidly, profitably and successfully develop, produce and market new products now in the early stage of production development.
- · Our products must meet the needs of existing and new customers, and we need to grow revenues sufficient to sustain profitable operations.
- We have limited personnel and financial resources to develop required business functions, including research and development, production, marketing, sales, distribution, service and administration.
- We will be required to obtain sufficient financing until we are able to produce revenues and profitability to sustain future operations.
- We face the uncertainties and risks facing any new business startup including but not limited to the risk factors described in the section entitled "Risk Factors" starting on page 11.

#### **Our Strategy**

Our goal is to realize the potential of parametric directed sound primarily targeting digital signage, point-of-purchase, in-store networks and related markets and applications. We aim to produce a product line to meet the requirements of these customers. The key elements of our strategy include:

- Produce a product line meeting customer requirements for easy and flexible installation using content from a range of media sources, complementing equipment such as video devices, kiosks and vending machines.
- · Develop a turnkey manufacturing relationship to produce our products, thereby reducing a need for manufacturing space and production personnel.
- Develop relationships with customers requiring large numbers of products, including value added resellers (VARs), original equipment manufacturers (OEMs) and distributors that focus on specific end user solutions.

We also plan to explore uses for our technology in the emerging stereoscopic three-dimensional (3D) video market for computers, homes and theaters. The growth of 3D viewing creates opportunities for new 3D sound systems. We intend to explore the use of parametric speakers to reproduce binaural recordings. Binaural recordings use special microphones to capture directional sound information during recording and typically must be replayed through headphones. Parametric speakers reduce loudspeaker cross-talk that interferes with binaural reproduction from conventional stereo speakers. We believe our speakers complement or enhance digital signal processing (DSP) techniques such as ambiophonics that are designed to emulate headphones for binaural sound reproduction from two speakers.

#### Our Industry

With the growth of directed advertising, such as digital signage, point-of-purchase and in-store TV networks, we believe sound clutter and noise pollution are increasing where conventional sound products and speakers are used. In-store display advertising tends to irritate customers if too intrusive or loud and annoy workers due to repetition. Our products will be designed to achieve focused, controlled sound that targets only those customers situated in specified locations such that nearby customers and store clerks do not hear the message. We believe the ability to focus sound is a driving feature of our sound solution.

Digital signage enables advertisers to deliver more focused and effective advertisements. Digital signage solutions generally consist of the replacement of static or passive signs or displays with video screen devices, such as cathode ray tube (CRT), plasma, liquid crystal display (LCD) or light-emitting diode (LED), containing still or motion video images. The image content, usually advertising or information can be changed via network control of each individual sign. Directed audio is used to contain or focus the audio within a defined space, generally to the person or persons in that defined space, to eliminate objections to the audio content from nearby workers and to lower the overall audio noise level in a confined space. Our product helps digital signage integrators integrate and install digital signage advertisi ng solutions for in-store networks for national and international retailers.

We believe that clean, lightweight, energy-efficient and lower cost LED LCD screens are driving a new wave in digital signage. Sales of commercial flat panel screens are experiencing rapid growth. We believe the industry is maturing with more major companies now engaged in consolidating hardware, networking and content into end-user solutions that are economical and effective.

We do not believe parametric directed sound has yet met the needs of this market that requires higher sound volume, better clarity and a more economical product than offered in the past. We believe we have lost business in the past due to insufficient sound volume, insufficient clarity and unacceptable selling price. We believe our new product offering will be a competitive and attractive alternative to traditional sound systems used in our target markets.

Summary of the Spin-Off The following is a summary of the terms of the spin-off. Please see "*The Spin-Off*" beginning on page 20 for a more detailed description of the matters described below.

Distributing company	LRAD Corporation, a Delaware corporation.
Distributed company	Parametric Sound Corporation, a Nevada corporation ("Parametric Sound"), owns most of the assets of the HSS technology business of LRAD Corporation and following the distribution will license technology developed by our Chief Executive Officer. Parametric Sound's principal executive offices are located at 1941 Ramrod Avenue, Suite #100, Henderson, Nevada 89014.
Distribution ratio	Each holder of LRAD Corporation common stock will receive a dividend of one share of Parametric Sound common stock for every two shares of LRAD Corporation common stock held on the record date. Any fractional shares will be rounded up to the next whole share.
Securities to be distributed	Approximately 15,305,728 shares of Parametric Sound common stock, which will constitute all of the outstanding shares of Parametric Sound common stock immediately after the spin-off (based on the 30,611,456 shares of LRAD Corporation common stock that we expect to be outstanding on the record date). If holders of outstanding stock options of LRAD Corporation exercise their stock options on or before the record date, we will be required to issue up to 1,847,743 additional shares of Parametric Sound common stock (based on 3,695,486 options that are exercisable as of June 30, 2010), and we will not receive any proceeds from the option exercise.
Record date	The record date is the close of business, September 10, 2010. To receive shares of Parametric Sound common stock in the spin-off, holders of LRAD Corporation common stock must be stockholders as of the close of business on the record date.
Distribution date	The distribution date will be on or about September 27, 2010.
Relationship between Parametric Sound and LRAD Corporation after the spin-off	Following the spin-off, each of LRAD Corporation and Parametric Sound will be an independent, publicly-traded company. We have entered into an agreement with LRAD Corporation primarily relating to the transfer of most of its HSS technology business assets to us, including the transfer of emitter manufacturing equipment, parts inventory, test equipment, fixtures, tooling, patents and trade secrets, see " <i>The Spin-Off – Contributed</i> <i>Assets</i> " beginning on page 22. LRAD Corporation may complete certain HSS business with current customers after the distribution date. We have agreed to reimburse Syzygy Licensing LLC (& #8220;Syzygy"), an entity in which our Chief Executive Officer, Elwood G. Norris is majority owner, an estimated \$200,000 for the legal, accounting, consulting, mailing, filing and other costs associated with the spin-off, including the special meeting of stockholders of LRAD Corporation approving the spin-off and the preparation and distribution of this information statement. After the distribution, other than some initial and limited manufacturing and technical support, we do not expect any material transition services between Parametric Sound and LRAD Corporation. We do not expect to share any employees or contract for any services from LRAD Corporation. For additional information regarding our relationship with LRAD Corporation after the spin-off, see " <i>Our Relationship with LRAD Corporation</i> <i>After the Spin-Off</i> " beginning on page 52.

License of Improvements to Parametric Technology	We intend to enter into a license agreement with Syzygy promptly following the distribution. Mr. Norris has assigned to Syzygy certain patent pending technology and trade secrets related to a new and more cost-effective method of controlling and processing media input to create parametric sound output for parametric emitter devices such as those employed by us. He has also invented improvements to the current HSS emitter. Pursuant to the license agreement, we will reimburse patent, prototype and testing costs incurred to date (an amount estimated at \$115,000) and to pay future patent related costs. These prior costs include up to a maximum of \$25,000 for Mr. Norris' time in producing and testing prototypes and preparing for production. The license will provide for royalties of not more than 5% of net sales from pr oducts employing the technology. For additional information regarding the license arrangement, see "Business-License" beginning on page 41.
Funding of Parametric Sound after the spin-off	Parametric Sound is the recipient of most of the business assets of LRAD Corporation's HSS technology business. These assets consist of emitter manufacturing equipment, parts inventory, test equipment, fixtures, tooling, patents and trade secrets. LRAD Corporation will be responsible for any warranty liability for any HSS product sales that it has made or makes completing certain HSS business with current customers.
	We have limited prospects for near-term revenues in part due to our plans to introduce a new product line. We will receive no cash or liquidity at the distribution date.
	We expect to fund a minimum of \$350,000 and a maximum of \$750,000 of subordinated note financing with accompanying stock purchase warrants at the distribution date. The proceeds of this financing are intended to fund the spin-off and related costs described above and provide initial working capital to start the process of bringing our new product line to production. We have received nonbinding, verbal indications of interest from seven individual accredited investors to provide an aggregate of \$475,000 in cash financing and from Syzygy to convert \$200,000 of the amount owed to Syzygy. We have received a written commitment from Mr. Norris that at least \$350,000 will be funded at the distribution date through conversion of amounts owed to Syzygy and additional cash on the same terms as other investors. Conversio n or repayment of amounts owed to Syzygy will not represent cash available for working capital.See " <i>Capitalization and Financing</i> " beginning on page 27.
	We expect that we will need to refinance this debt and/or arrange additional debt or equity or equity-based financing in the first 12 months after the distribution date. Management has not yet determined the amounts required in future financings, which will depend in part on the time required to bring the new product line to production and customer acceptance, if any, of the new product. There is no assurance we will be successful in arranging financing for our company or the terms of any such financing. Any future financing could be dilutive to existing Parametric Sound stockholders. See " <i>Risk Factors</i> " beginning on page 11 and " <i>Description of Indebtedness</i> " beginning on page 59.
Dividend policy	The declaration and amount of future dividends, if any, will be determined by our Board of Directors and will depend on our financial condition, earnings, capital requirements, financial covenants, regulatory constraints, industry practice and other factors our Board of Directors deems relevant.

Pursuant to the spin-off, we will be separated from LRAD Corporation and become a separate, publicly-traded company. The spin-off involves the following steps:

Approval of the spin-off by the stockholders of LRAD Corporation at a special stockholders meeting held on June 2, 2010.
 Entry into a Separation and Distribution Agreement (the "Separation Agreement") to effect the separation. For more information on this agreement, see "Our Relationship with LRAD Corporation After the Spin-Off" beginning on page 52.
 Prior to or at the distribution date, the transfer of certain assets related to our business (LRAD Corporation's HSS technology business) from LRAD Corporation to us, see "The Spin-Off—Contributed Assets" beginning on page 22.
 Declaration of the effectiveness of our registration statement on Form 10, of which this information statement to its stockholders.
 A minimum of \$350,000 in debt financing proceeds will be contributed to Parametric Sound at the distribution date.
 Following the distribution, adoption by our Board of Directors of the Parametric Sound Corporation 2010 Stock Option Plan and related stock option agreement.
 Following the distribution, reimbursement by Parametric Sound to Syzygy of an estimated \$200,000 for the legal, accounting, consulting, mailing, filing and other costs associated with the spin-off and the preparation and distribution of this information statement.
 Following the distribution, upon execution of a licensing agreement with Syzygy, Parametric Sound will reimburse Syzygy an estimated \$115,000 for patent, testing and prototyping costs incurred through the distribution date.
 Following the distribution, the application for listing and quotation of Parametric Sound common stock on the OTCBB.

For a further explanation of the spin-off, see "The Spin-Off" beginning on page 20.

## **Corporate Information and Structure**

Parametric Sound is a Nevada corporation that was formed on June 2, 2010. Our Chief Executive Officer, Elwood G. Norris, and entities affiliated with him beneficially owned approximately 13% of LRAD Corporation's outstanding common stock as of May 31, 2010 and will beneficially own the same percentage of our common stock immediately following the spin-off. Our principal executive offices are located at 1941 Ramrod Avenue, Suite #100, Henderson, Nevada 89014 and our telephone number is (888) 477-2150. Our Internet address following the spin-off will be www.parametricsound.com. Information contained on, or that will be accessible through, our website does not and will not constitute a part of this information statement or the registration statement on Form 10 of which this information statement is a part.

This information statement contains product names, trademarks and trade names of our company, LRAD Corporation and other business entities and organizations.

## QUESTIONS AND ANSWERS ABOUT THE SPIN-OFF

## Q: Why am I receiving this document?

A: LRAD Corporation is delivering this document to you because you were a holder of LRAD Corporation common stock on the record date for the distribution of our shares of common stock. Accordingly, you are entitled to receive one share of our common stock for every two shares of LRAD Corporation common stock that you held on the record date. No action is required for you to participate in the distribution.

## Q: What is the spin-off?

A: The spin-off is the overall transaction of separating our company from LRAD Corporation, which will be accomplished through a series of transactions resulting in us owning certain assets comprising most of the HSS technology business of LRAD Corporation. The final step of the transaction will be the pro rata distribution of our common stock to holders of LRAD Corporation's common stock. We refer to this last step as the "distribution." For additional information regarding these transactions, see "*The Spin-Off—Manner of Effecting the Spin-Off*" beginning on page 22.

## Q: Who is Parametric Sound Corporation?

A: Up to the time of the distribution, we will be a wholly owned subsidiary of LRAD Corporation that will receive LRAD Corporation's HSS technology business as part of the spin-off transaction. Following the spin-off, we will be a separate, publicly-traded company. We are a sound technology company focused on delivering directed parametric sound solutions.

## Q: Why is LRAD Corporation separating our businesses and distributing our stock?

A: LRAD Corporation's Board of Directors believes the separation will provide the benefits set forth below under the caption "*The Spin-Off—Reasons for the Spin-Off*" beginning on page 20, and that achieving those benefits could result in greater aggregate value to LRAD Corporation stockholders who retain both their LRAD Corporation and Parametric Sound shares than would be obtained under the current structure.

## Q: Why is the separation of the two companies structured as a spin-off?

A: LRAD Corporation's Board of Directors believes that a spin-off of our shares is a cost-effective and tax-efficient way to separate the companies. We believe the spin-off will be tax-free for stockholders. However, we do not have any ruling from the U.S. Internal Revenue Service or a favorable opinion by our accounting firm confirming the spin-off's tax-free status. You should consult your own tax advisor as to the particular consequences of the spin-off to you.

#### Q: What is the record date for the distribution?

A: The record date is September 10, 2010, and ownership will be determined as of the close of business, on that date. When we refer to the "record date," we are referring to that time and date.

## Q: What will be our relationship with LRAD Corporation after the spin-off?

A: Following the spin-off, each of LRAD Corporation and Parametric Sound will be an independent, publicly-traded company. We have entered into an agreement with LRAD Corporation primarily relating to the transfer of most of its HSS technology business assets to us, including the transfer of emitter manufacturing equipment, parts inventory, test equipment, fixtures, tooling, patents and trade secrets. After the distribution, other than some initial and limited manufacturing and technical support, we do not expect any material transition services between Parametric Sound and LRAD Corporation. We do not expect to share any employees or contract for any services from LRAD Corporation. For additional information regarding our relationship with LRAD Corporation after the spin-off, see "*Our Relationship with LRAD Corporation After the Spin-Off*" beginning on page 52.

## Q: When will the spin-off be completed?

A: Shares of our common stock will be distributed on or about September 27, 2010. We refer to this date as the "distribution date."

## Q: Can LRAD Corporation decide to cancel the distribution of our common stock?

A: Yes. The distribution is conditioned upon satisfaction or waiver of certain conditions. LRAD Corporation has the right to terminate the distribution, even if all conditions have been met, if at any time LRAD Corporation's Board of Directors determines, in its sole discretion, that LRAD Corporation and its stockholders are better served by retaining Parametric Sound as a wholly owned subsidiary or that business conditions are such that it is not advisable to complete the spin-off. Business conditions that could cause LRAD Corporation's Board of Directors to terminate the spin-off include, among other things, deterioration in business value or a decline in the equity market valuation for similar type businesses. If the distribution is canceled, then the license agreement between Parametric Sound and Syzygy for the new HSS technology will not be executed. LRAD Corporation may be unwilling or unable to continue to develop its parametric sound HSS business or license the new technology from Syzygy, and Syzygy may be unwilling to license the new technology to LRAD Corporation.

## Q: What will happen to the listing of LRAD Corporation's common stock?

A: Nothing. LRAD Corporation common stock will continue to be listed on the Nasdaq Capital Market under the symbol "LRAD."

## Q: Will the spin-off affect the market price of my LRAD Corporation shares?

A: This is unknown. As a result of the spin-off, it is possible that the trading price of LRAD Corporation shares immediately following the distribution could be lower than immediately prior to the distribution because the trading price will no longer reflect the value of our business. However, given the historical losses incurred by the HSS product line while it was part of LRAD Corporation, this could result in an increase in the trading price following the spin-off. Until the market has fully analyzed the operations of LRAD Corporation without this business segment, the price of LRAD Corporation shares may fluctuate significantly. Furthermore, the combined trading prices of LRAD Corporation more stock and our common stock after the distribution may be higher or lower than the trading price of LRAD Corporation common stock after the distribution may be higher or lower than the trading price of shares of our common stock after the spin-off, and sales of a substantial number of shares of our common stock may adversely affect the market price of our common stock" on page 17.

## Q: What will LRAD Corporation stockholders receive in the spin-off?

A: In the spin-off, LRAD Corporation stockholders will receive one share of our common stock for every two shares of LRAD Corporation common stock they own as of the record date. No fractional shares of our common stock will be issued in the spin-off. Any fractional shares will be rounded up to the next whole share. Immediately after the spin-off, LRAD Corporation stockholders will still own all of LRAD Corporation's current business segments, but they will own them as two separate investments rather than as a single investment. After the spin-off, the certificates and electronic brokerage book entry interests representing the "old" LRAD Corporation common stock will represent such stockholders' interests in LRAD Corporation's businesses (other than our business) following the spin-off. The certificates and electronic brokerage book-entry interests representing our common stock that stockholders receive in the spin-off will represent their interest in our business only.

### Q: How will fractional shares be treated?

Any fractional shares of our common stock that would otherwise be issuable in the spin-off will be rounded up to the next whole share. You may have to pay taxes on any shares that you receive as a result of the rounding up of fractional shares. For a discussion of the material U.S. federal income tax consequences of the spin-off, including the rounding up of fractional shares, please see *"The Spin-Off—Material U.S. Federal Income Tax Consequences"* beginning on page 24.

## Q: What does an LRAD Corporation stockholder need to do now?

A: LRAD Corporation stockholders do not need to take any action. We urge you to read this information statement carefully. LRAD Corporation stockholders have no appraisal or dissenters' rights in connection with the spin-off. LRAD Corporation is not seeking a proxy from any stockholders, and you are requested not to send us a proxy. LRAD Corporation stockholders will not be required to pay anything or provide any consideration for our shares distributed in the spin-off or to surrender any shares of LRAD Corporation common stock. LRAD Corporation stockholders should not send in their LRAD Corporation share certificates. LRAD Corporation stockholders will automatically receive their shares of our common stock when the spin-off is effected.

## Q: How will Parametric Sound fund its operations following the spin-off?

A: Parametric Sound has no assurance of continued revenues and has no plans to produce the HSS H450 product line after the distribution date. Accordingly, there is no assurance of future revenues. Future revenues depend on successfully introducing a new product line.

We expect to fund a minimum of \$350,000 and a maximum of \$750,000 of subordinated note financing with accompanying stock purchase warrants at the distribution date. We have nonbinding, verbal indications of interest from seven individual accredited investors to provide an aggregate of \$475,000 in cash financing and from Syzygy to convert \$200,000 of the amount owed to Syzygy. We have received a written commitment from Mr. Norris that at least \$350,000 will be funded at the distribution date through conversion of amounts owed to Syzygy and additional cash on the same terms as other investors. The proceeds of this financing are intended to fund the spin-off and related costs described above and provide initial working capital to start the process of bringing our new product line to production. Conversion or repayment of amounts owed to Syzygy will not represent cash available for working capital. See "*Capitalization and Financing*" beginning on page 27.

We expect that we will need to refinance this debt and/or arrange additional debt or equity or equity-based financing in the first 12 months after the distribution date. Management has not yet determined the amounts required in future financings, which will depend in part on the time required to bring the new product line to production and customer acceptance, if any, of the new product. There is no assurance we will be successful in arranging financing for our company or the terms of any such financing. Any future financing could be dilutive to existing Parametric Sound stockholders. See *"Risk Factors"* beginning on page 11 and *"Description of Indebtedness"* beginning on page 59.

## Q: What will happen to outstanding LRAD Corporation stock options?

A: Holders of LRAD Corporation stock options that continue to be employed by LRAD Corporation after the spin-off will continue to hold those options. LRAD Corporation's Board of Directors and the Compensation Committee for LRAD Corporation's stock plan, pursuant to their authority under the plan, have determined not to make any adjustment to outstanding stock options. They determined the spin-off should have minimal impact on the long-term value of the options and, accordingly, no adjustment is required to preserve the intrinsic value of the equity awards. LRAD Corporation stock options will not be converted into stock options or granted any right to receive shares of Parametric Sound. Accordingly, unless an LRAD Corporation stock option holder exercises his or her outstanding option prior to the record date of September 10, 2010, such option holder will not participate in the distribution with respect to shares subject to the option.

## Q: Are there risks associated with owning our common stock?

A: Yes. Our business is subject to both general and specific risks relating to our operations. In addition, our spin-off from LRAD Corporation presents risks relating to our becoming an independent, public company as well as risks relating to the nature of the spin-off itself. See "*Risk Factors*" beginning on page 11.



# Q: What are the U.S. federal income tax consequences of the spin-off to LRAD Corporation stockholders?

A: We believe the spin-off will be tax-free for stockholders; however, we do not have any ruling from the U.S. Internal Revenue Service nor do we have a favorable opinion by our accounting firm or any other expert confirming the spin-off is tax-free status. We believe stockholders will apportion their tax basis in LRAD Corporation common stock between such LRAD Corporation common stock and our common stock received in the spin-off in proportion to the relative fair market values of such stock at the time of the spin-off. Further, we believe that a stockholder's holding period for our common stock received in the spin-off will be rounded up to the next whole share. You may have to pay taxes on any shares that you receive as a result of the rounding up of fractional shares. See "The Spin-Off—Material U.S. Federal Income Tax Consequences of the Spin-Off" beginning on page 24. You should consult your own tax advisor as to the particular consequences of the spin-off to you.

# Q: What if I want to sell my LRAD Corporation common stock or my Parametric Sound common stock?

A: You should consult with your own financial advisors, such as your stockbroker, bank or tax advisor. We do not make any recommendations on the purchase, retention or sale of shares of LRAD Corporation common stock or our common stock to be distributed. If you do decide to sell any shares, you should make sure your stockbroker, bank or other nominee understands whether you want to sell your LRAD Corporation common stock or your Parametric Sound common stock after it is distributed, or both.

# Q: When will I be able to trade shares of Parametric Sound common stock?

A: There is not currently a public market for our common stock. Trading in shares of our common stock could begin on a "when-issued" basis on or shortly before the record date and before the distribution date, and "regular way" trading may then begin on the first trading day following the distribution date or at a later date subject to requisite regulatory approvals. If trading does begin on a "when-issued" basis, you may purchase or sell our common stock after that time, but your transaction will not settle until after the distribution date. On the first trading day following the distribution date, when-issued trading with respect to our common stock will end and regular way trading will begin. We cannot predict the trading prices for our common stock before or after the distribution date.

### Q: Where will trading prices of shares of Parametric Sound common stock be listed?

A: We anticipate that, following the spin-off, quotations for shares of Parametric Sound common stock will be available on the OTCBB quotation and trading system, and that a ticker symbol will be assigned for the Parametric Sound common stock first become available. It may be difficult to trade Parametric Sound common stock due to the potential for low and sporadic trading activity on the OTCBB quotation and trading system.

### Q: Do you intend to pay dividends on your common stock?

A: Not at this time.

### **Q:** Where can LRAD Corporation stockholders get more information? A: Before the distribution, if you have any questions relating to the distribution, you should contact:

LRAD Corporation Investor Relations 15378 Avenue of Science, Suite 100 San Diego, California 92128 (858) 676-1112

After the distribution, if you have any questions relating to our common stock, you should contact:

Parametric Sound Corporation Investor Relations 1941 Ramrod Avenue, Suite #100 Henderson, Nevada 89014 (888) 477-2150

# Q: Who will be the distribution agent for the spin-off?

A: Interwest Transfer Company, Inc. will be the distribution agent for the spin-off. The distribution agent can be contacted at: Interwest Transfer Company, Inc.

1981 Murray Holladay Road, Suite 100 Salt Lake City, UT 84117 Phone: (801)272-9294

### RISK FACTORS

In addition to the information contained elsewhere in this information statement, you should consider carefully the following risk factors related to Parametric Sound and the spin-off. If any of the risks described below actually occur, our business, financial condition, results of operations, cash flows and stock price could be materially adversely affected. This information statement also contains forward-looking statements that involve risks and uncertainties. Our actual results could differ materially from those anticipated in these forward-looking statements as a result of certain factors, including the risks faced by us described below and elsewhere in this information statement. See "Special Note About Forward-Looking Statements" on page 19.

## **Risk Factors Relating to Our Business**

# We have a history of operating losses, expect additional losses and may not achieve or sustain profitability.

We have a history of operating losses and expect additional losses as we introduce our new product line and until we achieve revenues and resulting margins to offset our operating costs. Our net losses for the fiscal years ended September 30, 2009 and 2008 were \$986,426 and \$2,149,732, respectively, and our net loss for the nine months ended June 30, 2010 was \$226,880. Our ability to achieve future profitability is dependent on a variety of factors, many outside our control. Failure to achieve profitability or sustain profitability, if achieved, may require us to continue to raise additional funding which could have a material negative impact on the market value of our common stock.

# Our independent auditors have expressed substantial doubt about our ability to continue as a going concern.

In their audit opinion issued in connection with our balance sheets as of September 30, 2009 and 2008 and our related statements of operations, changes in LRAD Corporation net investment and cash flows for the years then ended, our independent registered public accounting firm stated that our significant recurring net losses and our requirement to secure new financing raised substantial doubt about our ability to continue as a going concern. We have prepared our financial statements on a going concern basis which contemplates the realization of assets and the satisfaction of liabilities in the normal course of business for the foreseeable future. Our financial statements do not include any adjustments that would be necessary should we be unable to continue as a going concern and, therefore, be required to liquidate our assets and discharge our liabilities in other than the normal course of business and at amounts different from those reflected in our financial statements. If we are unable to continue as a going concern, our stockholders may lose a substantial portion or all of their investment.

# Disruption and fluctuations in financial and currency markets could have a negative effect on our business.

Financial markets in the United States, Europe and Asia have experienced extreme disruption in recent years, including, among other things, extreme volatility in security prices, severely diminished liquidity and credit availability, rating downgrades of certain investments and declining valuations of others. There can be no assurance that there will not be a further deterioration in financial markets and confidence in major economies, which can then lead to challenges in the operation of our business. These economic developments affect businesses such as ours in a number of ways. Concern about the stability of the financial markets and the strength of counterparties has led many lenders and institutional investors to reduce or cease to provide credit to businesses and consumers, and illiquid credit markets have advers ely affected the cost and availability of credit. The current tightening of credit in financial markets adversely affects the ability of commercial customers to finance purchases and operations and could result in an absence of any orders or spending for our planned products as well as create supplier disruptions. We are unable to predict the likely duration and severity of the current disruption in financial markets and adverse economic conditions and the effects they will have on our business and financial condition.

We expect to purchase a number of key components and subassemblies from foreign suppliers. Consequently, we are subject to the impact economic conditions can have on such suppliers and subject to fluctuations in foreign currency exchange rates. Increases in our cost of purchasing these items could negatively impact our financial results if we are not able to pass these increased costs on to our customers.



### We must develop a customer base in order to grow our business.

We do not expect many recurring customers from LRAD Corporation's prior HSS product business. To develop and grow our business, we must develop relationships with new customers and obtain and fulfill orders from new customers. We cannot guarantee that we will be able to develop a customer base. Further, even if we do obtain new customers, we cannot guarantee that those customers will purchase sufficient quantities of our product at prices that will enable us to recover our costs in acquiring those customers and fulfilling those orders. Our ability to sell our products depends on a number of factors, including:

	our ability to manufacture reliable products that have the features required by our customers;
	our ability to develop relationships with new customers that will lead to sales of our products;
•	our ability to develop and expand new markets for directed sound products; and
	our ability to develop international product distribution directly or through strategic partners.

# We plan to introduce new products based on new technology. If commercially successful products do not result or are not produced in a timely manner, we may be unprofitable or forced to cease operations.

Our proposed product line is in production planning and has not yet been introduced to market. We may not successfully and timely produce commercially viable sound systems due to the inherent risks of technology development, new product introduction, limitations on financing, manufacturing problems, competition, obsolescence, inability to attract key technical personnel and other factors. We have not produced any revenues from our new product line, and we cannot guarantee revenues in the future. The development and introduction of our product line may take longer than anticipated, and the introduction of our products, if any, could be subject to delays. Customers may not accept our products and may elect to purchase products from competitors. We may experience quality control problems with our products, and we may not be able to resolve any problems in a timely and cost-effective manner. Products employing our technology may not achieve market acceptance. Unanticipated technical obstacles may arise at any time and result in lengthy and costly delays or result in a dtermination that further exploitation of our technology is not feasible. If we do not successfully exploit our technology, our financial condition, results of operations and business prospects would be adversely affected.

# Our products are still under design and have not yet been produced in any production quantity, and we may incur significant and unpredictable warranty costs as these products are produced.

Our planned product line has not yet been produced. Our technology is substantially different from proven, mass-produced sound transducer designs. We may incur substantial and unpredictable warranty costs from post-production product or component failures. We generally expect to warrant our products to be free from defects in materials and workmanship for a period of up to one year from the date of purchase. Future warranty costs could further adversely affect our financial position, results of operations and business prospects.

# Our ability to develop, manufacture and market new products will be dependant upon our ability to use the technology that is subject to the license from Syzygy. While we expect to enter into the license promptly following the spin-off, the license may not be executed and we may be unable to develop our planned new products.

Promptly following the spin-off, we intend to license from Syzygy new patent pending technology owned by Syzygy. We plan to use this new technology to complement HSS emitters to produce at lower cost our new product line. The license has not been executed as of the date of this information statement and may not be executed following the spin-off and if it is executed it may terminate if we do not use commercially reasonable efforts to pursue the parametric sound business. Although we expect to enter into the license promptly following the distribution and to avoid a termination for non-use, there is no assurance that the license will be executed or that we will have the benefit of the technology that is the subject of the license. If we are unable to obtain the benefits of the technology owned by Syzygy, we will be unable to develop new products and grow our business, which will have a material adverse effect on our business, liquidity, results of operation and financial position.

### Our initial planned revenues will be derived from a single product category and two initial products.

If we are successful in producing initial commercial products, we will be dependent on one core product category and two initial products to generate revenues. We cannot assure you that these or other products will achieve consumer acceptance, nor can we provide assurance of the levels of sales they can achieve, if any. The loss of this product category would have a material adverse effect on our business, results of operations, financial condition and liquidity.

# We cannot predict our future operating results. Our quarterly and annual results will likely be subject to fluctuations caused by many factors, any of which could result in our failure to achieve our expectations.

We expect our proprietary sound reproduction products and technologies will be the source of substantially all of our future revenues. Revenues, if any, from our proprietary sound reproduction products and technologies are expected to vary significantly due to a number of factors. Many of these factors are beyond our control. Any one or more of these factors, including those listed below, could cause us to fail to achieve our revenue expectations. These factors include:

	our ability to develop and supply sound reproduction components to customers, distributors, VARs or OEMs or in the future to license our technologies;
	market acceptance of, and changes in demand for, our products or our customers' products;
	gains or losses of significant customers, distributors or strategic relationships;
	unpredictable volume and timing of customer orders;
	the availability, pricing and timeliness of delivery of components for our products;
	fluctuations in the availability of manufacturing capacity or manufacturing yields and related manufacturing costs;
•	timing of new technological advances, product announcements or introductions by us, by OEMs or licensees and by our competitors;
	product obsolescence and the management of product transitions and inventory;

- unpredictable warranty costs associated with our products;
- · installation or order delays by customers, distributors, OEMs or production delays by us or our suppliers;
- seasonal fluctuations in sales;
- general consumer electronics industry conditions, including changes in demand and associated effects on inventory and inventory practices;
- general economic conditions that could affect the timing of customer orders and capital spending and result in order cancellations or rescheduling; and

· general political conditions in this country and in various other parts of the world that could affect spending for the products that we intend to offer.

Some or all of these factors could adversely affect demand for our products or technologies and, therefore, adversely affect our future operating results.

# Our expenses may vary from period to period, which could affect quarterly results and our stock price.

If we incur additional expenses in a quarter in which we do not experience increased revenue, our results of operations will be adversely affected and we may incur larger losses than anticipated for that quarter. Factors that could cause our expenses to fluctuate from period to period include:

the timing and extent of our research and development efforts;
 investments and costs of maintaining or protecting our intellectual property;
 the extent of marketing and sales efforts to promote our products and technologies; and the timing of personnel and consultant hiring.

### Many potential competitors who have greater resources and experience than we do may develop products and technologies that make ours obsolete.

Technological competition from other and more established electronic and sound system manufacturers is significant and expected to increase. Most of the companies with which we expect to compete have substantially greater capital resources, research and development staffs, marketing and distribution programs and larger facilities than us, and many of them have substantially greater experience in the production and marketing of products. In addition, one or more of our competitors may have developed or may succeed in developing technologies and products that are more effective than any of ours, rendering our technology and products obsolete or noncompetitive.

# Sound reproduction markets are subject to rapid technological change. Our success will depend on our ability to continually develop and introduce new technologies.

Technology and standards in the sound reproduction market evolve rapidly, making timely and cost-effective product innovation essential to success in the marketplace. The introduction of products with improved technologies or features may render our technologies obsolete and unmarketable. If we cannot develop products in a timely manner in response to industry changes, or if our technologies do not perform well, our business and financial condition will be adversely affected. The life cycles of our technologies are difficult to estimate. As a result, our technology, even if successful, may become obsolete before we recoup our investment.

# Our competitive position will be seriously damaged if we cannot maintain and obtain patent protection for important differentiating aspects of our products or otherwise protect intellectual property rights in our technology.

We rely on a combination of contracts and trademark, patent and trade secret laws to establish and protect our proprietary rights in our technology. However, we may not be able to prevent misappropriation of our intellectual property, our competitors may be able to independently develop similar technology and the agreements we enter into to protect our proprietary rights may not be enforceable.

Our success, in part, depends on our ability to obtain and enforce intellectual property protection of our technology, particularly our patents. There is no guarantee any patent will issue on any patent application that we have filed or may file. Claims allowed from existing or pending patents may not be of sufficient scope or strength to protect the economic value of our technologies. Further, any patent that we may obtain will expire, and it is possible that it may be challenged, invalidated or circumvented. If we do not secure and maintain patent protection for our technology and products, our competitive position could be significantly harmed. A competitor may independently develop or patent technologies that are substantially equivalent or superior to our technology.

As we expand our product line or develop new uses for our products, these products or uses may be outside the protection provided by our current patent applications and other intellectual property rights. In addition, if we develop new products or enhancements to existing products we cannot assure you that we will be able to obtain patents to protect them. Even if we do receive patents for our existing or new products, these patents may not provide meaningful protection, or may be too costly to enforce protection. In some countries outside of the United States where our products may be sold or licensed, patent protection is not available. Moreover, some countries that do allow registration of patents do not provide meaningful redress for violations of patents. As a result, protecting intellectual property in these countries is difficult and our competitors may successfully sell products in these countries that infringe on our intellectual property.

We may initiate claims or litigation against third parties in the future for infringement of our proprietary rights or to determine the scope and validity of our proprietary rights or the proprietary rights of our competitors. These claims could result in costly litigation and divert the efforts of our technical and management personnel. As a result, our operating results could suffer and our financial condition could be harmed.

# Our competitive position will be seriously damaged if our products are found to infringe on the intellectual property rights of others.

Other companies and our competitors may currently own or obtain patents or other proprietary rights that might prevent, limit or interfere with our ability to make, use or sell our products. As a result, we may be found to infringe the intellectual property rights of others. The electronics industry is characterized by vigorous protection and pursuit of intellectual property rights or positions, resulting in significant and often protracted and expensive litigation. In the event of a successful claim of infringement against us and our failure or inability to license the infringed technology, our business and operating results could be adversely affected. Any litigation or claims, whether or not valid, could result in substantial costs and diversion of our resources. An adverse result from intellectual property litigation are or more of the following:

- cease selling, incorporating or using products or services that incorporate the challenged intellectual property; • obtain a license from the holder of the infringed intellectual property right, which license may not be available on reasonable terms, if at all; and
  - redesign products or services that incorporate the disputed technology.

If we are forced to take any of the foregoing actions, we could face substantial costs and shipment delays and our business could be seriously harmed. Although we carry general liability insurance, our insurance may not cover potential claims of this type or be adequate to indemnify us for all liability that may be imposed.

In addition, it is possible that our customers or end users may seek indemnity from us in the event that our products are found or alleged to infringe the intellectual property rights of others. Any such claim for indemnity could result in substantial expenses to us that could harm our operating results.

# Our technology is subject to government regulation, which could lead to unanticipated expense or litigation.

Our sound technology emits ultrasonic vibrations and is regulated by the Food and Drug Administration. In the event of certain unanticipated defects in our products, a customer or we may be required to comply with FDA requirements to remedy the defect and/or notify consumers of the problem. This could lead to unanticipated expense, and possible product liability litigation against a customer or us. Any regulatory impediment to full commercialization of our parametric sound technology, or any of our other technologies, could adversely affect our results of operations.

### We may face personal injury and other liability claims that harm our reputation and adversely affect our sales and financial condition.

Since our planned products include ultrasonic emissions along with the acoustic output, customers or others could claim damage to human hearing or human health irrespective if the product is used properly or improperly, such as when the product is used at extreme close ranges for long periods of exposure. A person claiming injury in connection with the use of our products may bring legal action against us to recover damages on the basis of theories including personal injury, negligent design, dangerous product or inadequate warning. We may also be subject to lawsuits involving allegations of misuse of our products. While we plan to obtain product liability insurance, it may be too costly for us or may be unavailable. If we are able to obtain product liability insurance coverage, it may be insufficient to pay all such claims. We may not have sufficient resources to satisfy any product liability claims not covered by insurance which would materially and adversely affect our financial position. Significant litigation could also result in a diversion of management's attention and resources and negative publicity.

# We expect to rely on outside manufacturers and suppliers to provide a large number of components and sub-assemblies incorporated in our products and may rely on third-party turnkey production in the future.

Our products have a number of components and subassemblies we expect to be produced by outside suppliers. In addition, for certain of these items, we qualify only a single source, which can magnify the risk of shortages and decrease our ability to negotiate with our suppliers on the basis of price. In particular, we depend on one piezo-film supplier to provide expertise and materials used in our proprietary emitters. If shortages occur, or if we experience quality problems with suppliers, then our production schedules could be significantly delayed or costs significantly increased, which would have a material adverse effect on our business, liquidity, results of operation and financial position.

We plan to have sub-assemblies produced by third-party manufacturers and may contract for complete production on a turnkey basis. We have no manufacturer relationship contracted at this time. We may not be able to obtain acceptable manufacturing sources on a timely basis. In addition, from time to time we may change manufacturers. Any new manufacturer engaged by us may not perform as expected. An extended interruption in the supply of our products could result in a substantial loss of sales. In addition, any actual or perceived degradation of product quality as a result of our reliance on third-party manufacturers may have an adverse effect on sales or result in increased warranty costs, product returns and buybacks. Failure to maintain quality manufacturing could reduce future revenues, adversely affecting financial c ondition and results of operations.

### Current environmental laws, or laws enacted in the future, may harm our business.

Our operations are subject to environmental regulation in areas in which we conduct business. Our product design and procurement operations must comply with new and future requirements relating to the materials composition of our products, including restrictions on lead, cadmium and other substances. We do not expect that the impact of these environmental laws and other similar legislation adopted in the U.S. and other countries will have a substantial unfavorable impact on our business. However, the costs and timing of costs under environmental laws are difficult to predict.

# Errors or defects contained in our products, failure to comply with applicable safety standards or a product recall could result in delayed shipments or rejection of our products or damage to our reputation and expose us to regulatory or other legal action.

Any defects or errors in the operation of our products may result in delays in their introduction. In addition, errors or defects may be uncovered after commercial shipments have begun, which could result in the rejection of our products by our customers, damage to our reputation, lost sales, diverted development resources, increased customer service and support costs and warranty claims, any of which could harm our business. Third parties could sustain injuries from our products, and we may be subject to claims or lawsuits resulting from such injuries. There is a risk that these claims or liabilities may exceed, or fall outside the scope of, our insurance coverage. We may also be unable to obtain adequate liability insurance in the future. Because we are a small company, a product recall would be particularly harmful to us. We have limited financial and administrative resources to effectively manage a product recall, and it would detract management's attention from implementing our core business strategies. A significant product defect or product declar customer adversely affect our brand image, causing a decline in our sales, and could reduce or deplete our financial resources.



# Inadequate internal controls and accounting practices could lead to errors, which could negatively impact our business, financial condition, results of operations and cash flows.

We will need to establish internal controls and management oversight systems. Our small size and limited personnel and consulting resources will make doing so more challenging than for more established entities. We may not be able to prevent or detect misstatements in our reported financial statements due to system errors, the potential for human error, unauthorized actions of employees or contractors, inadequacy of controls, temporary lapses in controls due to shortfalls in transition planning and oversight resource contracts and other factors. In addition, due to their inherent limitations, such controls may not prevent or detect misstatements in our reported financial results as required under SEC rules, which could increase our operating costs or impair our ability to operate our business. Controls may also become inadequate due to changes in circumstances. It will be necessary to replace, upgrade or modify our internal information systems from time to time. If we are unable to implement these changes in a timely and cost-effective manner, our ability to capture and process financial transactions and support our customers as required may be materially adversely impacted, which could harm our business.

### Loss of our key management and other personnel could impact our business.

Our business is substantially dependent on our chief executive officer, Elwood G. Norris, and other key personnel. The loss of Mr. Norris or any of these key personnel could materially adversely affect our business, financial condition, results of operations and cash flows. In addition, competition for skilled and non-skilled employees among companies like ours is intense, and the loss of skilled or non-skilled employees or an inability to attract, retain and motivate additional skilled and non-skilled employees required for the operation and expansion of our business could hinder our ability to conduct research activities successfully, develop new products, attract customers and meet customer shipments.

# **Risk Factors Relating to the Spin-Off**

### We may be unable to achieve some or all of the benefits that we expect to achieve from our separation from LRAD Corporation.

As a stand-alone, independent public company, we believe that our business will benefit from, among other things, allowing our management to design and implement corporate policies and strategies that are based primarily on the characteristics of our business, to focus our financial resources wholly on our own operations and to implement and maintain a capital structure designed to meet our own specific needs. However, we may not be able to achieve some or all of the benefits expected as a result of the spin-off.

Additionally, by separating from LRAD Corporation, there is a risk that our company may be more susceptible to stock market fluctuations and other adverse events than we would have been before the spin-off due to a reduction in market diversification. As a separate, stand-alone entity, we may be unable to access financial and other resources on terms as favorable as those available to us before the separation. Furthermore, as a stand-alone company, we will not be able to enjoy certain benefits from LRAD Corporation's operating diversity, borrowing leverage and available capital for investments.

# If the distribution, together with certain related transactions, fails to qualify for tax-free treatment for U.S. federal income tax purposes, then our stockholders, we and/or LRAD Corporation might be subject to significant tax liability.

If the spin-off fails to qualify for tax-free treatment, LRAD Corporation would be treated as if it had sold the common stock of our company for its fair market value, resulting in a taxable gain to the extent of the excess of such fair market value over its tax basis in our stock. In general, our initial public stockholders would be treated as if they had received a taxable distribution equal to the fair market value of our common stock that was distributed to them. For additional information, see "Material U.S. Federal Income Tax Consequences of the Spin-Off" beginning on page 24.

# Our operations depend on the availability of additional financing. After the spin-off, we will not be able to obtain financing from LRAD Corporation.

Following the spin-off, we may not have sufficient liquidity to support the development of our business. Parametric Sound may require additional financing for liquidity, capital requirements and growth initiatives. After the spin-off, LRAD Corporation will not provide funds to us. Accordingly, we will depend on our ability to generate cash flows from operations, borrow funds and issue securities in the capital markets to maintain and expand our business. We may need to incur debt on terms and at interest rates that may not be as favorable as those historically available to LRAD Corporation. In addition, future events may prevent us from borrowing funds or raising any equity or equity-based capital. An inability to obtain financing on favorable terms could have a negative effect on our results of operations, cash f lows and financial condition.

### Our ability to operate our business may suffer if we do not, quickly and cost-effectively, establish our own financial, administrative and other support functions to successfully operate as a standalone entity.

Historically, our business has relied on financial, administrative and other resources of LRAD Corporation. After the spin-off, we will need to immediately create our own financial, administrative and other support systems or contract with a third party(s) to provide these resources. Any services we establish or contract for may not be sufficient to meet our needs, and we may not be able to obtain these services at cost-effective prices or on favorable terms. Any failure or significant downtime in our financial or administrative systems could prevent us from paying our employees, billing our customers or performing other administrative services on a timely basis and could materially harm our business or operations.

# Our historical financial information or pro forma financial statements may not be indicative of our future results as an independent company.

The historical information we have included in this information statement may not reflect what our results of operations, financial position and cash flows would have been if we had been an independent company during the periods presented or be indicative of what our results of operations, financial position and cash flows may be in the future. Our results could have been dramatically different as a stand-alone company, and these statements may not be a reliable indicator of what our results of operations, cash flows and financial condition will be in the future.

For additional information about the past financial performance of our business and the basis of the presentation of the historical combined financial statements, see "Management's Discussion and Analysis of Financial Condition and Results of Operations" beginning on page 34 and the historical financial statements and the accompanying notes included elsewhere in this information statement.

# **Risk Factors Relating to Our Common Stock**

### There may not be an active trading market for shares of our common stock.

There is no public trading market for shares of our common stock. We cannot predict the extent to which investor interest in our company will lead to the development of an active trading market in our common stock or how liquid such a market might become. It is possible that, after the spin-off, an active trading market will not develop or continue, and there can be no assurance as to the price at which our common stock will trade. The initial share price of our common stock may not be indicative of prices that will prevail in the future.

In addition, because of the significant changes that will take place as a result of the spin-off, the trading market for both our common stock and LRAD Corporation's common stock after the spin-off may be significantly different from that for LRAD Corporation's common stock before the spin-off. The market may view us as a "new" company after the spin-off, and we will not be the subject of research analyst coverage. The absence of research analyst coverage can adversely affect the market value and liquidity of an equity security.

# We cannot predict the price range or volatility of our common stock after the spin-off, and sales of a substantial number of shares of our common stock may adversely affect the market price of our common stock.

From time to time, the market price and volume of shares traded of companies in the industries in which we operate experience periods of significant volatility. Company-specific issues and developments generally affecting our industries or the economy may cause this volatility. The market price of our common stock may fluctuate in response to a number of events and factors, including:

•	general economic, market and political conditions;
•	quarterly variations in results of operations or results of operations that are below public market analyst and investor expectations;
	changes in financial estimates and recommendations by securities analysts;
	operating and market price performance of other companies that investors may deem comparable;
	press releases or publicity relating to us or our competitors or relating to trends in our markets; and
	sales of common stock or other securities by insiders.

In addition, broad market and industry fluctuations, investor perception and the depth and liquidity of the market for our common stock may adversely affect the trading price of our common stock, regardless of actual operating performance.

Sales or distributions of a substantial number of shares of our common stock in the public market or otherwise following the spin-off, or the perception that such sales could occur, could adversely affect the market price of our common stock. After the spin-off, all of the shares of our common stock, other than the shares held by executive officers and directors, will be eligible for immediate resale in the public market. Investment criteria of certain investment funds and other holders of our common stock may result in the immediate sale of our common stock after the spin-off to the extent such stock no longer meets these criteria. Substantial selling of our common stock, whether as a result of the spin-off or otherwise, could adversely affect the market price of our common stock.

We cannot assure you as to the price at which our common stock will trade after the distribution date. Until our common stock is fully distributed and an orderly market develops in our common stock, the price at which our common stock trades may fluctuate significantly and may be lower or higher than the price that would be expected for a fully distributed issue.

# We may issue additional common stock in the future. The issuance of additional common stock may reduce the value of your common stock.

We may issue additional shares of common stock without further action by our stockholders. Moreover, the economic and voting interests of each stockholder will be diluted as a result of such issuances. Although the number of shares of common stock that stockholders presently own will not decrease, such shares will represent a smaller percentage of the total shares that will be outstanding after the issuance of additional shares. The issuance of additional shares of common stock may cause the market price of our common stock to decline.

# Sales of common stock issuable on the exercise of any future options or warrants may lower the price of our common stock.

We intend to adopt a stock option plan promptly following the distribution, which will authorize the grant of options to purchase up to 3,000,000 shares of our common stock to our employees, directors and consultants. We also intend to issue warrants on up to 1,500,000 common shares at the distribution date as part of our debt financing and may issue common stock purchase warrants or other securities convertible into common stock in the future. The issuance of shares of common stock issuable upon the exercise or conversion of convertible securities, options or warrants could cause substantial dilution to existing holders of common stock, and the sale of those shares in the market could cause the market price of our common stock to decline. The potential dilution from the issuance of these shares could negatively a ffect the terms on which we are able to obtain equity financing.

# We may issue preferred stock in the future, and the terms of the preferred stock may reduce the value of your common stock.

We are authorized to issue up to 1,000,000 shares of preferred stock in one or more series. Our Board of Directors may determine the terms of future preferred stock offerings without further action by our stockholders. If we issue preferred stock, it could affect your rights or reduce the value of your common stock. In particular, specific rights granted to future holders of preferred stock could be used to restrict our ability to merge with or sell our assets to a third party. Preferred stock terms may include voting rights, preferences as to dividends and liquidation, conversion and redemption rights and sinking fund provisions.

### The payment of dividends will be at the discretion of our Board of Directors.

The declaration and amount of future dividends, if any, will be determined by our Board of Directors and will depend on our financial condition, earnings, capital requirements, financial covenants, regulatory constraints, industry practice and other factors our Board deems relevant. See "Dividend Policy" on page 27 for additional information on our dividend policy following the spin-off.

# Our inability to comply with internal control requirements included in Section 404 of the Sarbanes-Oxley Act of 2002 could have a negative impact on our stock price.

As directed by Section 404 of the Sarbanes-Oxley Act of 2002, the SEC adopted rules requiring public companies to include a report of management on internal controls over financial reporting in their annual reports on Form 10-K. If we are unable to conclude that we have effective internal controls over financial reporting as required by Section 404 of the Sarbanes-Oxley Act of 2002, investors may lose confidence in the reliability of our financial statements which could result in a decrease in the value of our securities. We have not yet begun a formal process to establish or evaluate our internal controls over financial resources, substantial uncertainty exists regarding our ability to comply with these requirements within the prescribed time frame.

# SPECIAL NOTE ABOUT FORWARD-LOOKING STATEMENTS

This information statement and other materials filed or to be filed by us and LRAD Corporation, as well as information in oral statements or other written statements made or to be made by us and LRAD Corporation, contain statements, including in this document under the captions "*Executive Summary*," "*Questions and Answers About the Spin-Off,*" "*Risk Factors,*" "*The Spin-Off,*" "*Capitalization and Financing,*" "*Management's Discussion and Analysis of Financial Condition and Results of Operations,*" "*Business*" and "*Our Relationship with LRAD Corporation After the Spin-Off,*" that are, or may be considered to be, for ward-looking statements. All statements that are not historical facts, including statements about our beliefs or expectations, are forward-looking statements. You can identify these forward-looking statements by the use of forward-looking words such as "outlook," "believes," "expects," "potential," "continues," "may," "will," "should," "seeks," "approximately," "predicts," "intends," "plans," "estimates," "anticipates," "foresees" or the negative version of those words or other comparable words and phrases. Any forward-looking statements contained in this information statement are based on our historical performance and on current plans, estimates and expectations. The inclusion of this forward-looking information should not be regarded as a representation by us or any other person that the future plans, estimates or ex pectations contemplated by us will be achieved. There may be events in the future that we are not able to predict accurately or control. The factors listed under "Risk Factors," as well as any cautionary language in this information statement, provide examples of risks, uncertainties and events that may cause our results to differ materially from the expectations we describe in our forward-looking statements could have a material adverse effect on our business, results of operations and financial position.

Forward-looking statements, whether express or implied, are not guarantees of future performance and are subject to risks and uncertainties that can cause actual results to differ materially from those currently anticipated due to a number of factors, which include, but are not limited to:

- · risks that we may not have sufficient capital in the amounts and at the times needed to finance our business;
- · risks inherent in spin-offs, including those related to the capital resources required for business risks, legal risks and risks associated with the tax and accounting treatment of spin-off transactions;
- · risks associated with our future revenue source dependent on a new product line not yet in production;
- · risks that any future potential revenue opportunities from customers may not materialize to any meaningful degree or at all;
- · risks of delays or unforeseen technical obstacles in arranging production and bringing our new product line to market;
- · risks of operating as an independent, stand-alone company and loss of certain benefits associated with being owned as part of a larger company;
- · risks that the expected benefits of the spin-off may not be fully realized or may take longer to realize than anticipated;
- · absence of a public market for our common stock;

- · our ability to attract and retain qualified personnel and key employees;
- our ability to establish our own financial, administrative and other support functions;
- · difficulty in predicting the timing or outcome of new product development efforts;
- · the amount and timing of costs associated with research and development of our product line;
- · our ability to generate operating revenue;
- · market adoption of any new products;
- the competitive nature of the sound reproduction industry;
- our ability and that of our suppliers to comply with laws, regulations and standards, and the application and interpretation of those laws, regulations and standards, that govern or affect the
  electronics and sound reproduction industries, the non-compliance with which may delay or prevent the development of products;
- · the availability and price of acceptable raw materials and components from third-party suppliers;
- · volatility in the financial markets;
- any adverse outcome in litigation to which we may become a party;
- · general economic, political and business conditions that adversely affect our company or our suppliers or any company to which we sell our products;
- · changes in costs, including changes in labor costs and raw material prices;
- the impact on our product development of patents and other proprietary rights licensed or owned by us; and
- · the ability to successfully have our products manufactured in an efficient, time-sensitive and cost-effective manner.

You should read this information statement completely and with the understanding that actual future results may be materially different than expectations. All forward-looking statements made in this information statement are qualified by these cautionary statements. These forward-looking statements are made only as of the date of this information statement, and neither LRAD Corporation nor we undertake any obligation (and we and LRAD Corporation expressly disclaim any such obligation), other than as may be required by law, to update or revise any forward-looking statements to reflect changes in assumptions, the occurrence of unanticipated events or changes in future operating results over time or otherwise.

### THE SPIN-OFF

The Board of Directors of LRAD Corporation has determined that a spin-off of our business from LRAD Corporation and the distribution of our common stock to LRAD Corporation stockholders is in the best interests of LRAD Corporation and its stockholders. The Board of Directors of LRAD Corporation determined that separating the HSS technology business from its other operations would create the opportunity to complement improved emitter technology with new electronic processing technology and better position the new entity to thrive under its own management focus and long-term growth plans. The Board of Directors believes the separate entities have the opportunity to create more long-term stockholder value individually than through the current combined entity.

We believe the spin-off will be tax-free for stockholders; however, we do not have any ruling from the U.S. Internal Revenue Service nor do we have a favorable opinion by our accounting firm confirming the spin-off's tax-free status. You should consult your own tax advisor as to the particular consequences of the spin-off to you. LRAD Corporation's Board of Directors will establish record and payment dates for the spin-off shortly before the completion of the distribution.

### **Reasons for the Spin-Off**

LRAD Corporation's Board of Directors believes the spin-off will separate businesses with different product characteristics that require management to pursue different operating and business strategies. The separation is intended to benefit stockholders by allowing each separate company to maximize the performance of its business assets through undivided senior management focus on and capital allocation to its respective businesses.

The Board of Directors of LRAD Corporation considered the following potential benefits in making the determination to effect the spin-off and determined that the spin-off will:

- Allow us to capitalize on the opportunity to complement existing HSS emitter technology with new processing and control electronics developed independently by Mr. Norris. Parametric Sound intends to introduce a new product line based on the combined technology.
- Separate the risk of new parametric sound product development and introduction, including the likely operating losses generally associated with such product development, from LRAD Corporation's LRAD and other business and operating results.
- Reduce internal competition for capital and enhance the possibility of raising capital for the parametric sound business. Recently, the availability of financial and personnel resources for the HSS business has been limited because LRAD Corporation's business strategy has been focused on building its LRAD business. As an independent company, we will have direct access to the public and private capital markets to allow us to seek to finance our operations and growth without having to compete with LRAD Corporation's LRAD and other businesses with respect to financing.
- Allow management of each separate company to (1) design and implement corporate strategies and policies that are based primarily on the business characteristics of that company, (2) maintain a
  sharper focus on core business and growth opportunities, (3) concentrate their financial resources wholly on their own operations. LRAD Corporation's management believes that a separate focus
  on these items could allow each company to generate stockholder value not currently being realized.
- Allow each separate company to recruit and retain employees pursuant to compensation policies that are appropriate for their respective lines of business. As a separate, publicly-traded company
  with our own executive management team, we may be able to attract greater media attention and press coverage, which could strengthen our ability to promote our product brand.
- Provide both companies heightened strategic flexibility to form strategic business alliances in their target markets, unencumbered by considerations of the potential impact on the other business.
- Create common equity shares for Parametric Sound including options after the distribution, providing the appropriate incentive mechanisms to motivate and reward our management, employees and
  consultants. The common stock of the independent, publicly-traded Parametric Sound will have a value that reflects the efforts and performance of our management and employees. As a result,
  following the distribution we will be able to develop better incentive programs to attract and retain key employees through the use of stock-based and performance-based incentive plans that more
  directly link their compensation with our financial performance. These programs will be designed to more directly reward employees based on our performance.
- Increase transparency and clarity into the different businesses of LRAD Corporation and Parametric Sound. The investment community, including the respective analysts, stockholders and investors
  of LRAD Corporation and Parametric Sound will be better able to evaluate the merits and future prospects of each company. This will enhance the likelihood that each company will receive
  appropriate market recognition of its individual performance and potential.

Neither we nor LRAD Corporation can assure you that, following the spin-off, any of these benefits will be realized to the extent anticipated or at all. For a description of the factors that might impact our ability to achieve these benefits, see "*Risk Factors*" beginning on page 11. LRAD Corporation's Board of Directors also considered a number of other factors in evaluating the spin-off, including:

- our capital structure;
- the one-time and on-going costs of the spin-off;
- the diversion of management attention resulting from the time and effort necessary to complete the spin-off;
- the effects of a possible overlap during the transition from the existing product sold by LRAD Corporation and the introduction of new product;
- · the risk that the businesses of the two companies may overlap or compete in the future;
- the risk that Parametric Sound may fail as an independent entity due to a lack of financing, lack of new product success or for any variety of other reasons including but not limited to those described in "Risk Factors" and the impact, if any, that such a failure would have on LRAD Corporation; and
- the risk that the combined trading prices of our common stock and LRAD Corporation common stock after the distribution may be lower than the trading price of LRAD Corporation common stock before the distribution.



LRAD Corporation's Board of Directors concluded that the potential long-term benefits of the spin-off outweigh these factors, and that separating us from LRAD Corporation in the form of what management believes is a tax-free distribution is appropriate and advisable.

## Manner of Effecting the Spin-Off

The general terms and conditions relating to the spin-off will be set forth in the Separation Agreement between us and LRAD Corporation. The spin-off is scheduled to be effective at the close of business on the distribution date, which is September 27, 2010. The record date is September 10, 2010. To receive shares of Parametric Sound common stock in the spin-off, holders of LRAD Corporation common stock must be stockholders as of the close of business, on the record date. Each LRAD Corporation common stockholder will receive one share of our common stock for every two shares of LRAD Corporation common stock they owned on the record date. Each share of our common stock that is distributed will be validly issued, fully paid and nonass essable and free of preemptive rights. See "Description of Our Capital Stock" on page 56.

We expect the spin-off of our shares will be made in electronic brokerage book-entry form and physical stock certificates will be issued only to registered holders on the record date. For such LRAD Corporation stockholders who own LRAD Corporation common stock in registered form, the transfer agent will issue and mail physical certificate representing the shares of Parametric Sound common st ock. For stockholders who own LRAD Corporation common stock through a broker or other nominee, we expect their shares of Parametric Sound common stock will be credited to their accounts by the broker or nominee.

LRAD Corporation stockholders will not be required to pay or provide any consideration for shares of our common stock received in the spin-off or to surrender or exchange shares of LRAD Corporation common stock in order to receive our common stock or to take any other action in connection with the spin-off. LRAD Corporation stockholders have no appraisal or dissenters' rights in connection with the spin-off.

Fractional shares of our common stock will not be issued to LRAD Corporation stockholders as part of the spin-off nor credited to brokerage accounts. Any fractional shares of our common stock that would otherwise be issuable in the spin-off will be rounded up to the next whole share. You may have to pay taxes on any shares that you receive as a result of the rounding up of fractional shares. For a discussion of the material U.S. federal income tax consequences of the distribution, including the receipt of a whole share in lieu of fractional shares, please see "*—Material U.S. Federal Income Tax Consequences*" beginning on page 24.

After the spin-off, many of our stockholders may hold odd lots or blocks of less than 100 shares of our common stock. An investor selling an odd lot may be required to pay a higher commission rate than an investor selling round lots or blocks of 100 shares.

# TO RECEIVE SHARES OF OUR COMMON STOCK IN THE SPIN-OFF, YOU MUST BE A HOLDER OF LRAD CORPORATION COMMON STOCK AT THE CLOSE OF BUSINESS ON THE RECORD DATE

### **Contributed Assets**

In connection with the distribution, LRAD Corporation has contributed or will contribute to us certain business assets as described below. It will effect this contribution by transferring or assigning assets to us. LRAD Corporation will have no interest in our assets and business and will have no obligation with respect to our liabilities after the distribution. Similarly, we will have no interest in the assets of LRAD Corporation's other businesses and no obligation with respect to the liabilities of LRAD Corporation's retained businesses after the distribution.



The assets contributed include the following specific assets along with a general assignment of the business assets of the HSS business:

- 28 U.S. Patents, 3 foreign patents and 6 pending patents related to the HSS technology with a book value of approximately \$473,000 at June 30, 2010. The technology related to six additional U.S. patents previously abandoned by LRAD Corporation will also be part of the assignment;
- 4 U.S. trade names (including HSS® and HyperSonic®) along with certain additional foreign rights to such names;
- and electronic parts with a book value of \$-0- and an original cost of approximately \$1.2 million, the difference being previous LRAD Corporation inventory allowances for excess parts, with an
  estimated \$146,000 of such parts usable in future production based on our new product design;
- tooling and molds that have been fully depreciated and have no remaining book value;
- manufacturing jigs, testing and other equipment that has been fully depreciated and has no remaining book value;
- · drawings, technical descriptions, assembly details and component listings;
- · customer, representative and supplier lists; and
- · manufacturing methods, know-how and trade secrets related to HSS and HSS production.

Mr. Norris has filed for patent coverage on his new inventions that we intend to use in our new product line. We intend to evaluate during early production patents that we deem important to protecting continuing technology and rights. We may elect to abandon and write-down the value of additional patents upon completion of this review. We expect this review to be completed by September 30, 2010. Generally, patents are legally abandoned by not paying the periodic renewal fees.

LRAD Corporation is not transferring certain parts and electronic component inventory intended for remaining H450 production and warranty service. See "Our Relationship with LRAD Corporation After the Spin-Off" beginning on page 52.

### **Results of the Spin-Off**

After the spin-off, we will be a separately traded, public company. Immediately following the spin-off, we expect to have approximately 1,029 record holders of shares of our common stock based on the number of beneficial and record holders, respectively, of shares of LRAD Corporation common stock on August 17, 2010. We expect that approximately 15,305,728 shares of our common stock will be issued and outstanding following the completion of the spin-off based on the number of outstanding shares of LRAD Corporation common stock as of the close of business on August 17, 2010 and the distribution ratio of one share of Parametric Sound common stock for every two shares of LRAD Corporation common stock. No fractional shares of our common stock will be issued in the distribution. Any fractional shares of our common stock that would otherwise be issuable will be rounded up to the next whole share. The actual number of shares to be distributed will be determined on the record date and will reflect any exercise of LRAD Corporation options or warrants or other share issuances prior to the record date.

LRAD Corporation and Parametric Sound will be parties to an agreement that governs the spin-off and the future relationship between the two companies. For a more detailed description of this agreement, please see "Our Relationship with LRAD Corporation After the Spin-Off" beginning on page 52.

The spin-off will not affect the number of outstanding shares of LRAD Corporation common stock or any rights of LRAD Corporation stockholders, although it may affect the market value of each outstanding share of LRAD Corporation common stock.

### Material U.S. Federal Income Tax Consequences of the Spin-Off

The following is a summary of certain U.S. federal income tax consequences to LRAD Corporation, the holders of LRAD Corporation common stock, us and the holders of our common stock after the spin-off as of the date hereof. This summary does not discuss all tax considerations that may be relevant to stockholders in light of their particular circumstances, nor does it address the consequences to stockholders subject to special rules under the U.S. federal income tax laws, such as stockholders subject to the alternative minimum tax, tax-exempt entities, non-resident alien individuals, foreign entities, foreign trusts and estates and beneficiaries thereof, stockholders who acquire shares as compensation for services (including holders of LRAD Corporation restricted stock who did not make a Section 83(b) election), banks, in surance companies, other financial institutions, traders in securities that use mark-to-market accounting, and dealers in securities or commodities. In addition, this summary does not address any state, local or foreign tax consequences. This summary is based upon provisions of the Internal Revenue Code (the "Code"), and regulations, rulings and judicial decisions, as of the date hereof. Those authorities may be changed, perhaps retroactively, so as to result in U.S. federal income tax consequences different from those summarized below.

If a partnership holds LRAD Corporation or our common stock, the tax rule of a partner will generally depend upon the status of the partner and the activities of the partnership. If you are a partner of a partnership holding LRAD Corporation or our common stock, you should consult your tax advisors.

# All stockholders should consult their own tax advisors concerning the specific tax consequences of the spin-off of our common stock to holders of LRAD Corporation common stock in light of their particular circumstances. This summary is not intended to be, nor should it be construed to be, legal or tax advice to any particular investor.

We have no ruling of any kind whatsoever from the IRS to the effect that the spin-off will qualify as a tax-free transaction under Section 355 of the Code and/or a tax-free reorganization under Section 368(a)(1)(D) of the Code. In part due to the non-binding nature of private letter rulings, and in part due to the time and expense required to obtain such a ruling, we have determined not to request a private letter ruling from the IRS with respect to the spin-off. We have also not requested an opinion from our accounting firm or any other expert confirming the tax-free status of the transaction.

Therefore, on the basis of the opinion of our management only, and assuming that LRAD Corporation common stock is a capital asset in the hands of an LRAD Corporation stockholder on the distribution date:

- · holders of LRAD Corporation common stock will not recognize any income, gain or loss as a result of the receipt of shares of our common stock in the spin-off;
- holders of LRAD Corporation common stock will apportion the tax basis of their LRAD Corporation common stock between such LRAD Corporation common stock and our common stock received in the spin-off in proportion to the relative fair market values of such stock at the time of the spin-off;
- the holding period for our common stock received in the spin-off by holders of LRAD Corporation common stock will include the period during which such holders held the LRAD Corporation common stock with respect to which the spin-off was made; and
- · neither we nor LRAD Corporation will recognize gain or loss as a result of the spin-off.

Current federal tax regulations also generally provide that if an LRAD Corporation stockholder holds different blocks of LRAD Corporation common stock (generally shares of LRAD Corporation common stock purchased on different dates or at different prices), the aggregate basis for each block of LRAD Corporation common stock purchased or acquired on the same date and at the same price will be allocated, to the greatest extent possible, between the shares of our common stock received in the spin-off in respect of such block of LRAD Corporation common stock, and such block of LRAD Corporation common stock, in proportion to their respective fair market values, and the holding period of the shares of our common stock received in the spin-off in respect of such block of LRAD Corporation common stock will include the holding period of such block of LRAD Corporation common stock are received in the spin-off with respect to a particular block of LRAD Corporation common stock are received in the spin-off with respect to a particular block of LRAD Corporation common stock, are received in the spin-off with respect of a particular block of LRAD Corporation common stock, are received in the spin-off in respect of a particular block of LRAD Corporation common stock are received in the spin-off with respect of a particular block of LRAD Corporation common stock are received in the spin-off with respect of a particular block of LRAD Corporation common stock are received in the spin-off in the spin-off in respect of a particular block of LRAD Corporation common stock are received in the spin-off in the spin-off in respect of a particular block of LRAD Corporation common stock are received in the spin-off to the total number of shares of LRAD Corporation common stock are received that distributed to the LRAD Corporation stock holder in the spin-off to the total number of shares of LRAD Corporation common

If you receive a whole share of Parametric Sound in lieu of a fractional share, you will be treated as though you first received a distribution of the fractional share in the spin-off taxable at the market price on the date of distribution. You will generally recognize capital gain or loss, provided that the fractional share is considered to be held as a capital asset, measured by the difference between the market value for such fractional share and your tax basis in that fractional share, as determined above. Such capital gain or loss will be long-term capital gain or loss if your holding period (as determined above) for such fractional share is more than one year on the distribution date.

If the distribution were not to qualify as a tax-free spin-off, each LRAD Corporation stockholder receiving shares of our common stock in the spin-off would be treated as if such stockholder had received a distribution in an amount equal to the fair market value of our common stock received, which would result in (1) a taxable dividend to the extent of such stockholder's pro rata share of LRAD Corporation's current and accumulated earnings and profits, (2) a reduction in such stockholder's basis in LRAD Corporation common stock to the extent the amount received exceeds such stockholder's share of earnings and profits and (3) a taxable gain to the extent the amount received exceeds the sum of the amount treated as a dividend and the stockholder's basis in LRAD Corporation common stock. Any such gain would generally be a capital gain if the LRAD Corporation common stock is held as a capital asset on the distribution date. In addition, LRAD Corporation would recognize a taxable gain to the extent the fair market value of our common stock.

Even if the spin-off otherwise qualifies for tax-free status under Section 355 of the Code, LRAD Corporation could recognize taxable gain if the spin-off is determined to be part of a plan or series of related transactions pursuant to which one or more persons acquire, directly or indirectly, stock representing a 50% or greater interest in either LRAD Corporation or Parametric Sound following the spin-off. Under the Code, any acquisitions of LRAD Corporation or Parametric Sound within the four-year period beginning two years before the date of the spin-off are presumed to be part of such a plan. Regulations issued by the IRS, however, provide mitigating rules in many circumstances. Nonetheless, a merger, recapitalization or acquisition, or issuance or redemption of our common stock after the spin-off could, in some cir cumstances, be counted toward the 50% change of ownership threshold.

There are other restrictions imposed on us under current U.S. federal tax law for spin-offs with which we will need to comply in order to preserve the favorable tax rule of the distribution, such as continuing to own and manage our parametric sound business and limitations on sale or redemptions of our common stock or other property following the distribution. If you are a "significant distributee" with respect to the spin-off, you are required to attach a statement to your federal income tax return for the year in which the spin-off occurs setting forth our name and IRS employer identification number, LRAD Corporation's name and IRS employer identification number, the date of the spin-off, and the fair market value of the shares of our common stock that you receive in the spin-off. Upon request, LRAD Corporation will provide the information necessary to comply with this reporting requirement to each stockholder of record as of the close of business on the record date. You are a "significant distributee" with respect to the spin-off if you own at least 5% of the outstanding shares of LRAD Corporation common stock immediately before the spin-off. You should consult your own tax advisor concerning the application of this reporting requirement in light of your particular circumstances.

# Listing and Trading of Our Common Stock

There is currently no public market for our common stock. Following the spin-off, we currently anticipate that quotations for our common stock will be available on the OTCBB quotation and trading system, and that a ticker symbol for our common stock will be assigned for our common stock shortly before quotations for our common stock first become available. It is possible that a limited trading market, known as a "when-issued" trading market, may develop on or shortly before the record date for the spin-off and that regular trading, to the extent available on the OTCBB quotation and trading system, will begin on the first trading day after the effective date of the spin-off. When-issued trading refers to a sale or purchase made conditionally because the security has been authorized but not yet issued. On the first trading day following the distribution date, when-issued trading with respect to our common stock will end and regular way trading will begin. Regular way trading refers to trading after a security has been issued and typically involves a transaction that settles on the third full business day following the date of the transaction.

We cannot predict what the trading prices for our common stock will be before or after the distribution date. We also cannot predict any change that may occur in the trading price of LRAD Corporation common stock as a result of the spin-off. Until our common stock is fully distributed and an orderly market develops in our common stock, the price at which it trades may fluctuate significantly and may be lower or higher than the price that would be expected for a fully distributed issue. See *"Risk Factors Relating to Our Common Stock"* beginning on page 17.

The shares of our common stock distributed to LRAD Corporation stockholders will be freely transferable except for shares received by persons who may be deemed to be our "affiliates" under the Securities Act of 1933, as amended ("Securities Act"). Persons that may be considered affiliates of us after the spin-off generally include individuals or entities that control, are controlled by or are under common control with us. This may include some or all of our officers and directors as well as our principal stockholders. Persons that are our affiliates will be permitted to sell their shares only pursuant to an effective registration statement under the Securities Act, or an exemption from the registration requirements of the Securities Act, such as the exemptions afforded by Section 4(1) of the Securities Act or Rule 144 thereunder.

# **Spin-off Conditions and Termination**

We expect that the spin-off will be effective on the distribution date, September 27, 2010, provided that, among other things:

- the SEC has declared effective our registration statement on Form 10, of which this information statement is a part, under the Securities Exchange Act of 1934, as amended, and no stop order relating to the registration statement is in effect; and
- no action, proceeding or investigation shall have been instituted or threatened before any court or administrative body to restrain, enjoin or otherwise prevent the consummation of the spin-off, and no restraining order or injunction issued by any court of competent jurisdiction shall be in effect restraining the consummation of the spin-off.

The fulfillment of the foregoing conditions will not create any obligation on LRAD Corporation's part to effect the spin-off, and the Board of Directors of LRAD Corporation has reserved the right to amend, modify or abandon the spin-off and the related transactions at any time before the distribution date. The Board of Directors of LRAD Corporation may also waive any of these conditions.

LRAD Corporation has the right to terminate the distribution, even if all conditions have been met, if at any time LRAD Corporation's Board of Directors determines, in its sole discretion, that LRAD Corporation and its stockholders are better served by retaining Parametric Sound as a wholly owned subsidiary or that business conditions are such that it is not advisable to complete the spin-off. Business conditions that could cause LRAD Corporation's Board of Directors to terminate the spin-off include, among other things, deterioration in business value or a decline in the equity market valuation for such businesses. If the distribution is canceled, then the license agreement between Parametric Sound and Syzygy for the new HSS technology will not be executed. LRAD Corporation may be unwilling or unable to continue to develop its parametric sound HSS business or license the new technology from Syzygy, and Syzygy may be unwilling to license the new technology to LRAD Corporation.

# **Reason for Furnishing this Information Statement**

This information statement is being furnished solely to provide information to LRAD Corporation stockholders who will receive shares of our common stock in the spin-off. It is not and is not to be construed as an inducement or encouragement to buy or sell any securities. We believe that the information contained in this information statement is accurate as of the date set forth on the cover. Changes may occur after that date and neither LRAD Corporation nor Parametric Sound undertakes any obligation to update the information except in the normal course of our respective public disclosure obligations.

# DIVIDEND POLICY

The declaration and amount of future dividends, if any, will be determined by our Board of Directors and will depend on our financial condition, earnings, capital requirements, financial covenants, regulatory constraints, industry practice and other factors our Board deems relevant.

# CAPITALIZATION AND FINANCING

### **Capitalization**

The following table sets forth our unaudited capitalization as of June 30, 2010 on an historical basis and on a pro forma basis to give effect to the spin-off. This table should be read in conjunction with "Management's Discussion and Analysis of Financial Condition and Results of Operations" and our historical and unaudited pro forma financial statements and notes thereto included elsewhere in this information statement. For an explanation of the pro forma adjustments made to our historical financial statements for the spin-off, please see "Unaudited Pro Forma Financial Statements."

		June 30, 2010		
	Historical		Pro Forma	
	(Unaudited)		(Unaudited)	
Cash	\$ -		\$	35,000
Notes payable, net	\$ -		\$	304,400
Stockholders' equity:				
LRAD Corporation, net investment	595,883			-
Preferred stock, \$0.001 par value,				
1,000,000 shares				
authorized, none issued and				
outstanding	-			-
Common stock, \$0.001 par value,				
50,000,000 shares				
authorized, 15,305,728 shares				
issued and outstanding	-			15,306
Additional paid-in capital	-			503,783
Accumulated deficit				(307,500)
Total capitalization	\$ 595,883		\$	515,989

Upon the closing of the spin-off, LRAD Corporation's net investment in Parametric Sound will be redesignated as Parametric Sound stockholders' equity and will be allocated between common stock and additional paid-in capital based on the number of shares of Parametric Sound common stock outstanding at the closing of the spin-off. We have assumed for purposes of the pro forma financial statements a distribution ratio of one share of our common stock for each two outstanding share of LRAD Corporation common stock. And we have not assumed any issuance related to LRAD Corporation's outstanding stock purchase warrants and the minimal effect of the rounding up of fractional shares only determinable at the distribution.

### **Financing**

We expect to fund a minimum of \$350,000 and a maximum of \$750,000 of subordinated note financing with accompanying stock purchase warrants at the distribution date. We have nonbinding, verbal indications of interest from seven individual accredited investors to provide an aggregate of \$475,000 in cash financing and from Syzygy to convert \$200,000 of the amount owed to Syzygy. We have received a written commitment from Mr. Norris that at least the minimum amount of the financing of \$350,000 is funded at the distribution date through conversion of amounts owed to Syzygy and additional cash on the same terms as other investors. These funds are intended to fund the spin-off and related costs and provide initial operating working capital to start the process of bringing our new product line to production. Conversion or repayment of amounts owed to Syzygy will not represent cash available for working capital.

The terms of the debt financing consist of subordinate promissory notes due in one year from the date of distribution with interest at 8%. Investors will be issued a warrant exercisable for two shares of our common stock for each \$1.00 of debt financing, a total of 700,000 warrants if the minimum note financing is funded (1,500,000 warrants if the maximum financing is funded). The warrants will have a five-year term and contain a cashless exercise provision, subject to certain limitations. The exercise price of the warrants shall be fixed as the average of the closing price of our common stock for the first 20 days of trading subject to a minimum exercise price of \$0.10 and a maximum exercise price of \$0.30 per common share.

We may call the warrants for exercise after six months if thereafter the closing price of our common stock for any 10 consecutive trading days exceeds \$0.90 per share as long as notice is given within ten days of the end of any such period. We may offset any note principal against the warrant exercise price, thereafter any remaining warrants may be exercised on a cash or cashless basis at the option of the holder.

## UNAUDITED PRO FORMA FINANCIAL STATEMENTS

The unaudited pro forma financial statements reported below consist of an unaudited pro forma balance sheet as of June 30, 2010 and unaudited pro forma statements of operations for the nine months ended June 30, 2010 and for the year ended September 30, 2009. The unaudited pro forma financial statements reported below should be read in conjunction with our "Management's Discussion and Analysis of Financial Condition and Results of Operations," our audited annual historical financial statements and the notes thereto, and the unaudited interim financial statements and the notes thereto, all of which are included elsewhere in this information statement.

The Unaudited Pro Forma Balance Sheet and the Unaudited Pro Forma Statements of Operations included in this information statement have been derived from the audited and unaudited historical financial statements included elsewhere in this information statement and do not purport to represent what our financial position and results of operations actually would have been had the distribution and related transactions occurred on the dates indicated or to project our financial performance for any future period. LRAD Corporation did not account for us as, and we were not operated as, a separate, stand-alone entity, subsidiary, division or segment for the periods presented.

The pro forma balance sheet adjustments assume that our spin-off from LRAD Corporation occurred as of June 30, 2010 and reflect (a) the new license agreement, (b) recording of spin-off costs and technology costs to be reimbursed to Syzygy (c) contribution of assets by LRAD Corporation less assets retained by LRAD Corporation and (d) the minimum \$350,000 of new financing (see "Capitalization and Financing") and estimated valuation of related warrants. The amount of inventory transferred to us will be reduced by any usage between June 30, 2010 and the distribution and by the amount retained by LRAD Corporation to complete allowed orders. The actual amount we receive may be different than that estimated and will be determined at the distribution date with any excess not used by LRAD being transferred to us at a future date.



The pro forma adjustments to the statements of operations for the nine months ended June 30, 2010 and for the year ended September 30, 2009 assume that the spin-off occurred as of October 1, 2008 and reflect adjustments noted above and described in the Notes to Unaudited Pro Forma Financial Statements below. The pro forma statements of operations do not reflect:

any future patent or inventory impairment that may result from transition to our new product line.

the value and expected use of approximately \$146,000 of inventory with no book value as a result of previous excess part and obsolescence reserves.

the effect of any possible manufacturing cost savings or improved gross margins from our new product design if successfully produced.

the effect on revenues if the new product line is priced as planned at a lower unit selling price than in reported periods or the effect on revenues of customer acceptance, if any, of the new product.

any potential decrease in manufacturing overhead or operating costs management believes may be achievable or any increased incremental expenses, over the amount of general allocation of corporate overhead, associated with being an independent, public company as more fully discussed in the notes below.

the effect of non-cash expense of any new stock option grants that may be made from the Parametric Sound 2010 Stock Option Plan that we intend to adopt following the distribution compared to the amounts included in the historical financial statements.

any financing cost of capital such as interest or other expenses associated with funding operations beyond the costs of the minimum \$350,000 of new financing described below. Actual financing costs could be significantly more than the amounts assumed and reflected below. The historical financial statements included in this information statement do not reflect any additional cost of capital and additional capital is expected to be required beyond the minimum of \$350,000 described below and no costs of such capital is included in the pro forma financial statements.

Accordingly the historical financial information we have included in this information statement and the pro forma information included below may not reflect what our results of operations, financial position and cash flows would have been had we been an independent company during the periods presented or be indicative of what our results of operations, financial position and cash flows may be in the future when we are an independent company. Results could have been dramatically different and these financial statements may not be a reliable indicator of what our results of operations, cash flows and financial condition actually may be in the future.

# Parametric Sound Corporation Pro Forma Balance Sheet (Unaudited) As of June 30, 2010

		Pro Forma					
	1	Historical		Adjustments		Pro Forma	
ASSETS							
Current assets:							
Cash	\$	-	\$	350,000 (A)	\$	35,000	
Clon	Ŷ		\$	(315,000) (C)	Ψ	55,000	
Accounts receivable, net		79,713	\$	(79,713) (B)		-	
Inventories, net		110,032		(110,032) (B)		-	
Total current assets		189,745				35,000	
Patents, net		473,489		7,500 (C)		480,989	
Total assets	\$	663,234	\$	(147,245)	\$	515,989	
LIABILITIES AND STOCKHOLDERS' EQUITY							
Current liabilities:							
Accounts payable	\$	5,907	\$	(5,907) (B)	\$	_	
Accrued liabilities	Ŷ	61,444	Ŷ	(61,444) (B)	Ŷ	-	
Notes payable, net of discount		-	\$	304,400 (A)		304,400	
Total current liabilities		67,351	•			304,400	
Stockholders' equity:							
LRAD Corporation, net investment		595,883		(122,394) (B)		-	
+**F******************************		,		(473,489) (D)			
Preferred stock, \$0.001 par value; 1,000,000 shares authorized;				( -// ( /			
none issued and outstanding		-		-		-	
Common stock, \$0.001 par value; 50,000,000 shares authorized;							
15,305,728 shares issued and outstanding		-		15,306 (D)		15,306	
Additional paid-in capital		-		503,783 (A) (D)		503,783	
Accumulated deficit		-		(307,500) (C)		(307,500)	
Total stockholders' equity		595,883				211,589	
Total liabilities and stockholders' equity	\$	663,234	\$	(147,245)	\$	515,989	

See accompanying notes to unaudited pro forma financial statements

# **Parametric Sound Corporation** Pro Forma Statement of Operations (Unaudited) For the Nine Months Ended June 30, 2010

	TT 1			Pro Forma		
	Historical		Adjustments		Pro Forma	
Revenues:						
Product sales	\$	512,369	\$	(512,369) (1) \$	-	
Other revenue		7,709		(7,709) (1)	-	
Total revenues		520,078			-	
Cost of revenues		441,403	\$	(441,403) (2)	-	
Gross profit		78,675			-	
Operating expenses:						
Selling, general and administrative		215,348			215,348	
Research and development		90,207		375 (3)	90,582	
Total operating expenses		305,555			305,930	
Loss from operations		(226,880)			(305,930)	
Interest expense		-		55,200 (4)	55,200	
Loss before income taxes		(226,880)			(361,130)	
Provision for income taxes		-		(5)	-	
Net loss	\$	(226,880)		\$	(361,130)	
Loss per common share - basic and diluted				(6) \$	(0.02)	
Weighted average common shares outstanding - basic and diluted					15,290,853	

See accompanying notes to unaudited pro forma financial statements

# Parametric Sound Corporation Pro Forma Statement of Operations (Unaudited) For the Year Ended September 30, 2009

	I	Historical		Pro Forma djustments	Pro Forma	
Revenues:	-			·		
Product sales	\$	586,693	\$	(586,693) (1) \$	-	
Other revenue		20,413		(20,413) (1)	-	
Total revenues		607,106			-	
Cost of revenues		638,763		(638,763) (2)	-	
Gross loss		(31,657)		<u> </u>	-	
Operating expenses:						
Selling, general and administrative		839,971			839,971	
Research and development		114,798		500 (3)	115,298	
Total operating expenses		954,769			955,269	
Loss from operations		(986,426)			(955,269)	
Interest expense		-		73,600 (4)	73,600	
Loss before income taxes		(986,426)			(1,028,869)	
Provision for income taxes		-		(5)	-	
Net loss	\$	(986,426)		\$	(1,028,869)	
Loss per common share - basic and diluted				(6) \$	(0.07)	
- Weighted average common shares outstanding - basic and diluted					15,268,712	

See accompanying notes to unaudited pro forma financial statements

### Notes to Unaudited Pro Forma Financial Statements

Please note that, due to guidelines governing the preparation of pro forma financial statements, the pro forma financial statements do not reflect certain estimated incremental expenses associated with being an independent, public company. These additional expenses are estimated to be approximately \$60,000 for the nine months ended June 30, 2010. The estimated incremental expenses, over the amount of general allocation of corporate overhead in the statements, associated with being an independent, public company include directors' and officers insurance, transfer agent and regulatory fees, investor and public relations, and legal and audit costs. Pro forma guidelines recommend that we record the minimum debt financing in our best-efforts offering, accordingly \$350,000 of debt has been assumed along with related wa rrants assuming the floor exercise price of \$0.10 per share is also the market price. The actual warrant exercise price could vary significantly as could the ultimate value recorded for such warrants. Any increase in value of the warrants would not change the cash interest rate on the debt but would increase non-cash amortization from the value of the warrants recorded as a debt discount amortized over the debt term.

These unaudited pro forma financial statements reflect all adjustments that, in the opinion of management, are necessary to present fairly the pro forma results of operations and financial position. The pro forma adjustments to the accompanying historical financial information at June 30, 2010 and for the nine months ended June 30, 2010 and for the fiscal year ended September 30, 2009 are described below:

### Balance Sheet at June 30, 2010

- (A) Reflects the minimum debt financing of \$350,000 as if funded on June 30, 2010. The \$45,600 value assigned to warrants issued with the debt is included in paid-in capital and as a debt discount. For illustrative purposes, the warrants are valued assuming the minimum floor price is the market price and using volatility of 80% (historical volatility of a selected peer group), a term of 5 years and a risk-free rate of 2.5%. Assuming the maximum exercise price and a market price of \$0.30 per share would value the warrants, using the same assumptions, at \$136,800, which would reduce pro forma net debt (\$350,000 less the unamortized note discount) at June 30, 2010 from a net of \$304,400 to \$213,200 and correspondingly increase stockholders' equity from \$211,589 to \$302,789. On a pro forma basis, noncash interest expense would increase by \$68,400 for the nine months ended June 30, 2010 and \$91,200 for the year ended September 30, 2009, compared to the amounts reported in the illustrated pro forma statements of operations. If the market price exceeds the warrant value and the note discount could further increase and the discount amortization would increase noncash interest expense over the term of the debt. Changes to the valuation of the warrants has no impact on cash debt service requirements.
- (B) The LRAD Corporation net investment account represents the cumulative investments in, distribution from, and losses of our company. LRAD Corporation is retaining accounts receivable and is responsible for accounts payable and accrued liabilities at the spin-off. It is expected that there will be no book value of HSS inventory remaining at the spin-off. Accordingly, the pro forma balance sheet reflects adjustments to reduce accounts receivable, accounts payable, accrued liabilities and inventory to zero on the pro forma balance sheet.
- (C) Reflects reimbursement to Syzygy of \$200,000 in estimated spin-off costs (effectively nonrecurring organization costs) and \$115,000 reimbursement to Syzygy of estimated patent and pre spin-off research costs. The patent costs estimated at \$7,500 are capitalized and other research costs are considered as an expense. These costs are not included in the pro forma statements of operations for the respective periods as they are considered nonrecurring charges resulting directly from the transaction and would be duplicative of research and development costs otherwise incurred. Actual costs incurred and reimbursed could differ from our estimates depending on the timing of the actual distribution and actual costs incurred for legal, accounting, pr inting, mailing, research and other costs.
- (D) Common stock issued and outstanding was calculated assuming a distribution ratio of one share of our common stock for every two shares of LRAD Corporation common stock. The pro forma number of shares is based on the number of shares of LRAD Corporation outstanding at June 30, 2010. The actual number of our shares outstanding will not be known until the actual distribution date.

### Statements of Operations

- (1) For pro forma presentation, historical H450 revenues have been eliminated. Substantially all historical revenues have resulted from the H450 product line and Parametric Sound has no plans to continue to produce or sell H450 products. Accordingly, historical H450 revenues are not indicative of revenues that may be obtained from the future sale of the proposed new product line, when and if distributed. At separation, LRAD Corporation will finish certain H450 business in process, including the fulfillment of orders received prior to the separation and unsolicited orders received from current customers through December 31, 2010, and will continue to fulfill continuing purchase orders for a current project with Cardinal Health through the completion of such project.
- (2) To correspond with the elimination of H450 product line revenues, the corresponding cost of sales has been eliminated for pro forma presentation. Parametric Sound has no existing manufacturing costs or overhead other than planned royalties of 5% based on the new license agreement
- (3) Reflects amortization of additional patent costs assuming a 15 year estimated useful life.
- (4) Reflects interest of 8% on the minimum debt financing assuming it had been funded at the beginning of each period and amortization of the value of warrants as a note discount amortized over the term of the debt. Assumes the minimum balance remained outstanding for the period presented.
- (5) No tax benefit is recognized as any deferred tax assets resulting from the net losses are offset by a full valuation allowance as those tax benefits are not likely to be realized.(6) Pro forma basic net loss per share is computed as if the shares of our common stock were issued and outstanding for the periods presented, assuming a distribution ratio of one share
- of our common stock for every two shares of LRAD Corporation common stock. The pro forma number of shares is based on the number of shares of LRAD Corporation outstanding for the respective periods presented. The actual number of our basic shares outstanding will not be known until the distribution date. The dilutive effect of stock options for employees was excluded from the calculation of diluted loss per share as the effect would have been antidilutive.

## MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

You should read the following discussion in conjunction with the financial statements and other financial information included elsewhere in this information statement. The following discussion may contain forward-looking statements that reflect our plans, estimates and beliefs. Our actual results could differ materially from those discussed in these forward-looking statements. Factors that could cause or contribute to these differences include, but are not limited to, those discussed below and elsewhere in this information statement, particularly in "Risk Factors" and "Special Note About Forward-Looking Statements."

We are a sound technology company focused on delivering directed parametric sound solutions to customers primarily in digital signage, point-of-purchase, in-store networks and related markets that benefit from sound that can be focused and controlled in specified locations. We are currently preparing a new product line and do not expect to report revenues until production quantities are available for sale to customers. There can be no assurance regarding the timing or amount of future revenues from this product line, if any.

### **Basis of Presentation**

### Separation from LRAD Corporation

We were incorporated under the laws of the State of Nevada on June 2, 2010 as a wholly-owned subsidiary of LRAD Corporation. We will have no material assets or activities as an independent, stand-alone company until LRAD Corporation contributes to us the business described in this information statement, which is expected to occur immediately prior to the distribution of shares. Following the spin-off, we will be an independent, stand-alone company holding most of LRAD Corporation's HSS business. Prior to the spin-off, we will enter into a Separation Agreement and other ancillary agreements with LRAD Corporation to complete the separation of our business from LRAD Corporation. These agreements will also govern the relationship between us and LRAD Corporation after the distribution of taxes and other liabilities and obligations attributable to periods prior to the distribution. After the spin-off, we expect to operate as an independent, public company. We will not contract with LRAD Corporation for any specific services. However, we and LRAD Corporation will cooperate with respect to certain administrative, manufacturing know-how transfer and other services reasonably necessary to complete the separation following the distribution date. We will enter into a license agreement under which we will exclusively license certain patent pending technology and trade secrets from Syzygy, an entity in which our Chief Executive Officer, Elwood G. Norris is majority owner. We currently anticipate that, initially following the spin-off, beginning on page 52 and "Unaudited Pro Forma Financial Statements" beginning on page 28.

Our financial statements were prepared on a carve-out basis from LRAD Corporation. The carve-out financial statements include allocated operating expenses of \$954,769 and \$2,081,231 for the years ended September 30, 2009 and 2008, respectively, and \$305,555 and \$761,286 for the nine months ended June 30, 2010 and 2009, respectively. The allocations were based on management's estimates of efforts expended on the respective operations of Parametric Sound and LRAD Corporation. For example, the carve-out combined financial statements reflect allocations of employee related costs (including salary, benefits and related overhead) for research and general and administrative services. These allocations were based upon various assumptions and estimates, and actual results may differ from these allocations, assumptions and estimates. However, the carve-out combined financial statements do not reflect additional expenses we expect to incur as an independent, stand-alone company.

### Going Concern

Our financial statements have been prepared on a going concern basis which contemplates the realization of assets and the satisfaction of liabilities in the normal course of business for the foreseeable future. Although normal asset impairment adjustments have been made as required by generally accepted accounting principles, the accompanying financial statements do not include any adjustments that would be necessary should we be unable to continue as a going concern and, therefore, be required to liquidate our assets and discharge our liabilities in other than the normal course of business and at amounts different from those reflected in the accompanying financial statements.

As a result of significant recurring net losses that we have incurred through September 30, 2009 and 2008 and our need for significant new financing, the report of our independent auditors dated June 23, 2010 includes a going concern explanatory paragraph that expresses substantial doubt about our ability to continue as a going concern. If funding from internal sources or from public or private financing is insufficient to fund the business at necessary levels, or if we are unable to develop, market and commercialize our new product line and otherwise develop our business, we will have to substantially cut back our development efforts, reduce or discontinue our operations or cease to exist. Any of these results would be detrimental to the value of our common stock. We can make no assurances that we will be a ble to obtain development and working capital financing on favorable terms or at all, or that our technologies, products or business operations will develop and provide us with funds to continue operations, our going concern uncertainty may affect our ability to raise additional capital, and may also affect our relationships with suppliers and customers. Stockholders should carefully examine our financial statements.

# Challenges, Opportunities, and Uncertainties

To date, our business has been operated as a part of LRAD Corporation's sound products business. Upon completion of the spin-off, we will incur the burden and incremental expenses associated with being an independent, stand-alone public company without any ongoing support or services from LRAD Corporation. Accordingly, we will be required to establish and grow business functions including research and development, production, marketing, sales, distribution, service and administration. Until we generate revenues and margins or obtain additional financing, we expect to have limited personnel to accomplish these functions and will primarily rely on Mr. Norris and the part-time services of Mr. Barnes along with outside consultants and suppliers we intend to engage for production and certain other services. Should we not reach an employment or agency relationship with David Pratt, the current Director of Business Development responsible for HSS business at LRAD Corporation (see "Key Personnel" on page 46), then we intend to hire a replacement to coordinate and direct our sales and marketing activities with existing and new HSS customers. Given our limited personnel, there is risk and uncertainty whether we can timely accomplish required functional activities and achieve important milestones, including introducing new products and obtaining orders from existing and new customers.

Although over 11,000 HSS systems, mostly the H450 product model, have been sold since 2004, we expect future revenues following the spin-off to be from new products not yet in production. If we do not timely produce successful products, future revenues could be delayed. We are unable to predict the market acceptance of our new products or the level of future sales, if any. However, due to our new technology and reduced component costs, we expect to offer improved HSS systems at more competitive prices than the H450, and we believe this creates new opportunities for directed sound. We believe there are opportunities to grow the HSS business for use in growing markets for digital signage, point-of-purchase, in-store networks and related markets and applications. However, we have not yet commenced marketing our new products.

Since we are receiving no cash, receivables or other working capital from LRAD Corporation in connection with the spin-off, we will be reliant on new debt financing to provide initial working capital. We will need additional capital to finish development and marketing of our new product line and working capital to produce product for sale to customers. Obtaining required additional financing in the future could be a significant management challenge and failure to secure necessary financing would have a material adverse affect on our company. Our ability to continue as a going concern is dependent upon achieving a profitable level of operations and until then obtaining additional financing.

Given our limited personnel and financial resources we face significant challenges in establishing, operating and growing our new business. The continued global economic downturn could increase the challenges in operating our business. We expect we will need to continue to innovate new applications for our sound technology, develop new products to meet diverse customer requirements and identify and develop new markets for our products.

## **Critical Accounting Policies and Estimates**

The preparation of financial statements in accordance with accounting principles generally accepted in the United States, which we refer to as U.S. GAAP, requires us to make estimates and judgments that affect the reported amounts of assets, liabilities, revenue and expenses, and related disclosure of contingent assets and liabilities. On an on-going basis, we evaluate our estimates, including those related to bad debts, impairments, warranty liabilities and contingencies. We base our estimates on historical experience and on various other assumptions we believe to be reasonable under the circumstances. Actual results may differ from these estimates under different assumptions or conditions.

Some of our accounting policies require higher degrees of judgment than others in their application. These include revenue recognition, reserves for uncollectible accounts receivable, warranty liabilities, impairments, contingencies and stock-based compensation. We believe the following critical accounting policies affect our more significant judgments and estimates used in the preparation of our financial statements:

# Balances and Expenses Assignment and Allocations

The accompanying financial statements include all revenues and cost of revenues directly attributable to the HSS product business. However, in circumstances where costs are shared, management used certain estimates to allocate expenses incurred by LRAD Corporation on behalf of the business included in the financial statements. The most significant expenses are as follows:

- Direct costs of HSS product development, sales and administrative personnel are included along with associated stock-based compensation and allocated employee benefits on a specific identification basis.
- HSS patent amortization costs, including impairments, are included in research and development costs.
- Certain shared administrative salaries and costs including accounting, payroll, human resources, information technology and other services along with legal, auditing and other administrative costs were allocated based on estimates of the proportionate cost incurred by LRAD Corporation related to the Parametric Sound business. Other costs including facility and occupancy costs were allocated proportionately based on revenues.

Management believes that the assumptions and methods of allocation used underlying the financial statements are reasonable. However, the costs as allocated to us are not necessarily indicative of the costs we would have incurred if we had operated as a stand-alone entity.

#### Revenue Recognition and Product Costs

Product revenues reflect amounts received from customers from the sale of HSS products. Product sales to customers, including resellers, are recognized in the periods that products are shipped to customers (FOB shipping point) or received by customers (FOB destination), when the fee is fixed or determinable, when collection of resulting receivables is probable and there are no remaining obligations on our part. Our customers do not have the right to return product unless the product is found to be defective. Product costs include direct manufacturing costs and allocated overhead which require estimates to allocate various costs to product results.

#### Impairments

Our inventory is comprised of raw materials, assemblies and finished products. We must periodically make judgments and estimates regarding the future utility and carrying value of our inventory. The carrying value of our inventory is periodically reviewed and impairments, if any, are recognized when the expected future benefit from our inventory is less than its carrying value.



Intangible assets consist of patents that are amortized over their estimated useful lives. We must make judgments and estimates regarding the future utility and carrying value of intangible assets. The carrying values of such assets are periodically reviewed and impairments, if any, are recognized when the expected future benefit to be derived from an individual intangible asset is less than its carrying value.

# Stock-based Compensation

We account for share-based compensation in accordance with the provisions of Accounting Standards Codification (ASC) 718, "Compensation—Stock Compensation" requiring the measurement and recognition of compensation expense for all share-based payment awards made to employees based on estimated fair values. ASC 718 requires the use of subjective assumptions, including expected stock price volatility, forfeitures and the estimated term of each award. If actual results differ significantly from our estimates, stock-based compensation expense and our results of operations could be materially impacted.

### Warranty Liabilities

We establish a warranty reserve based on anticipated warranty claims at the time product revenue is recognized. This reserve requires us to make estimates regarding the amount and costs of warranty repairs we expect to make over a period of time. Factors affecting warranty reserve levels include the number of units sold, anticipated cost of warranty repairs, and anticipated rates of warranty claims. If actual results differ significantly from our estimates, cost of sales and our results of operations could be materially impacted.

# **Revenues, Product Costs and Operating Expenses**

During the periods presented, most of our revenues and related product costs were for our HSS H450 product. At the separation, LRAD Corporation will finish certain H450 business in process, including the fulfillment of orders received prior to the separation, unsolicited orders received from current customers through December 31, 2010, and any future orders received from Cardinal Health, Inc. through the completion of their current project, and we will not recognize any revenues or costs or liabilities for such sales. Although we own the rights and could produce and sell H450 product in the future, we have no current plans to do so. Our future revenues and costs are expected to be from our new product line. Although we expect former customers of LRAD Corporation's HSS business to be fruture customers, there can be no assurance thereof. Due to the anticipated changes in our product, past sales and historical margins are not indicative of future results. Based on our costing of prototype components and estimated assembly costs, we are planning product pricing targeting gross margins of at least 40% but actual results could differ materially from our plans. We have not yet begun to market our products and, accordingly, cannot predict the degree of product acceptance by existing or new customers or the level of future product revenues, if any.

Our operating expenses have historically included (a) selling, general and administrative expenses and (b) research and development expenses. Research and development expenses comprise the costs incurred in performing research and development activities on our behalf, including salaries and benefits, facilities expenses, depreciation, overhead expenses, patent amortization, contract services and other outside expenses. As described above, our past expenses required allocations of costs because we were not accounted for as a separate unit of LRAD Corporation. The cost structure of our company after the spin-off will be significantly different than our historical costs. We expense sulling, general and administrative expenses to be lower than in the past due to fewer personnel and our smaller size as a start-up company. The actual level of future selling, general and administrative expenses will be dependent on staffing levels, elections regarding the use of outside resources, public company and regulatory costs, the impact of noncash stock-based compensation costs and other factors, some outside our control.

The scope and magnitude of our future research and development expenses are difficult to predict at this time given the changes being made to our planned product line and other management and organizational changes resulting from the spin-off. Again, historical expenditures are not indicative of future expenditures.

While we expect our initial operating costs to be approximately \$20,000 per month, these costs could increase rapidly as we introduce our products and expand our research and development, production, distribution, service and administrative functions in future months. We may also incur future financing costs and substantial noncash stock-based compensation costs depending on future option grants which are impacted by stock prices and other valuation factors.

### **Results of Operations**

The following is a discussion of the results of our operations for the nine months ended June 30, 2010 and 2009 and the years ended September 30, 2009 and 2008.

### Comparison of Results of Operations for the Nine Months Ended June 30, 2010 and 2009

### Revenues

Revenues of HSS products increased \$127,083, or 32%, to \$520,078 in the nine months ended June 30, 2010 from \$392,995 in the nine months ended June 30, 2009. The sales growth is primarily due to a program with Cardinal Health, Inc. for a kiosk application for in-store pharmacies, which contributed revenue of \$207,670 and \$0 in the nine months ended June 30, 2010 and 2009, respectively. Future revenues for this Cardinal Health, Inc. program will continue to be fulfilled by LRAD Corporation.



# **Gross Profit**

Gross profit for the nine months ended June 30, 2010 was \$78,675, or 15% of revenues, compared to a gross loss of \$21,346 of revenues for the nine months ended June 30, 2009. The increase in gross margin is primarily due to the increased revenues during the period and an increase in the inventory reserve in the prior year.

### Selling, General and Administrative Expenses

Selling, general and administrative expenses for the nine months ended June 30, 2010 decreased by \$450,769 to \$215,348, from \$666,117 during the nine months ended June 30, 2009. The decrease in expenses is primarily due to \$57,770 of lower salaries and benefits, \$414,085 of lower non-cash share based compensation expense due to stock options becoming fully vested and lower staffing, offset by an increase in allocation for support services due to the increase in sales.

We incurred non-cash share-based compensation expenses allocated to selling, general and administrative expenses in the nine months ended June 30, 2010 and 2009 of \$3,901 and \$417,986, respectively.

### **Research and Development Expenses**

Research and development expenses for the nine months ended June 30, 2010 decreased by \$4,962 from \$95,169 in the nine months ended June 30, 2009 to \$90,207 during the nine months ended June 30, 2010. We did not invest any development staff in this product line during these periods. The expenses for research and development pertained to patent amortization expenses and the impairment of patents that were associated with patents that were no longer consistent with our business strategy. We incurred \$3,436 lower patent impairment expense during the nine months ended June 30, 2010 compared to the same period for 2009.

### Net Loss

Our net loss for the nine months ended June 30, 2010 decreased by \$555,752 to \$226,880, compared to a net loss of \$782,632 for the nine months ended June 30, 2009. The decreased loss was the result of higher revenues and lower operating expenses.

### Comparison of Results of Operations for the Years Ended September 30, 2009 and 2008

### Revenues

Revenues decreased \$136,238, or 18%, to \$607,106 in the year ended September 30, 2009, compared to \$743,344 in the year ended September 30, 2008. Sales decreased to several resellers in Europe, Asia and the U.S., partially offset by an increase in sales to Cardinal Health, Inc. in the year ended September 30, 2009 for a program to install kiosks using HSS speakers in their in-store pharmacies. Future orders for this Cardinal Health, Inc. program will continue to be fulfilled by LRAD Corporation.

#### Gross Loss

Gross loss for the year ended September 30, 2009 was \$31,657, compared to a gross loss of \$68,501 for the year ended September 30, 2008. The decrease in gross loss is primarily due to a lower reserve for inventory obsolescence and a lower allocation of expenses due to lower revenues in the year ended September 30, 2009 than in the year ended September 30, 2008.

### Selling, General and Administrative Expenses

Selling, general and administrative expenses for the year ended September 30, 2009 decreased by \$500,678 to \$839,971, from \$1,340,649 during the year ended September 30, 2008. The decrease in expenses is primarily due to \$107,784 of lower salaries and benefits, \$219,657 of lower non-cash share based compensation expense from stock options becoming fully vested and lower staffing, and \$169,996 lower allocated expenses due to an overall reduction in LRAD Corporation spending.

We incurred non-cash share-based compensation expenses allocated to selling, general and administrative expenses in the year ended September 30, 2009 and 2008 of \$444,664 and \$664,321, respectively.

### **Research and Development Expenses**

Research and development expenses for the year ended September 30, 2009 decreased by \$625,784 to \$114,798 from \$740,582 in the year ended September 30, 2008. During the second quarter of fiscal 2008, we significantly reduced our research and development staff supporting the HSS product line. As a result, our year over year expenses decreased by \$302,052 for salaries and benefits, in addition to a decrease of \$131,971 for non-cash share-based compensation expense, \$9,178 for lower patent amortization and \$155,858 for lower patent impairment expense associated with patents that were no longer consistent with our business strategy.

We incurred non-cash share-based compensation expenses in research and development of \$131,971 in the year ended September 30, 2008. There was no similar expense recorded in the year ended September 30, 2009.

### Net Loss

Our net loss for the years ended September 30, 2009 and 2008 was \$986,426 and \$2,149,732, respectively. The decrease in the loss was primarily due to lower operating expenses.

### Liquidity and Capital Resources

Overviev

Historically, our sole source of liquidity has been funding from LRAD Corporation. Following the spin-off, we expect our primary source of liquidity will be the proceeds from our debt financing and any revenue generated from the sale of products after introduction of our new product line.

### Capital Requirements

At the separation date, LRAD Corporation will contribute substantially all of its HSS business assets to us, which consists primarily of patents; inventory, which was previously nearly fully reserved for in LRAD Corporation's financial statements; and fully depreciated fixed assets. We will receive no accounts receivable or any cash from LRAD Corporation. Accordingly, our liquidity will be dependent on our debt financing. In the future, we expect to use certain inventory in future production thus reducing our cash production costs on initial production.

Assuming only the \$350,000 minimum debt financing committed by Mr. Norris is funded at the spin-off, we will have limited cash for working capital and will require additional financing. We expect we will require a minimum of \$325,000 in additional capital to maintain planned operations for the first 12 months following the distribution date. We believe these resources will be funded from our debt financing and that some of these resources may be obtained from selling our new product line to existing and new HSS business customers. There is no assurance we can obtain any required additional debt financing on the terms of the current financing or other terms or that any future debt or equity financing will be available to us. Failure to secure additional f inancing or to quickly produce and sell our new products could have a material adverse effect on our operations and force us to curtail our planned activities.

In addition to reimbursing Syzygy an estimated \$200,000 of spin-off costs and \$115,000 of technology related development costs, we expect that our short-term liquidity requirements will include approximately \$60,000 of incremental public company expenses, \$50,000 for additional design and tooling for our new product line and \$10,000 for production working capital. We anticipate that our operating costs will initially approximate \$20,000 per month but demands for operating and working capital funds could grow rapidly based on decisions regarding staffing, development, production, marketing and other functions and based on factors outside our control.

We currently have no other commitments requiring liquidity. However within twelve months, should we not generate sufficient funds from operations or arrange other financing to retire our debt financing then we will be required to renegotiate or refinance such debt and there is no assurance we can do so on acceptable terms. Failure to timely obtain any required additional financing in the future will have a material adverse affect on our company. Our ability to continue as a going concern is in substantial doubt and is dependent upon achieving a profitable level of operations and until then obtaining additional financing.

Our future capital requirements, cash flows and results of operations could be affected by and will depend on many factors that are currently unknown to us, including:

•	-	the timing of the availability of our new product line for sale to customers;
·		the timing and extent of any market acceptance of our products;
•		the costs, timing and outcome of planned production and required customer and regulatory compliance of our new products;
•		the costs of preparing, filing and prosecuting our patent applications, maintaining and enforcing our issued patents and defending intellectual property-related
	claims;	
•		the costs and timing of additional product development;
•		the costs, timing and outcome of any future warranty claims or litigation against us associated with any of our products; and
•		the timing and costs associated with any new financing.



## **Operating Activities**

During the nine months ended June 30, 2010 and 2009, we reduced our investment from LRAD Corporation from operating activities by \$115,463 and \$7,903. Our investment from LRAD Corporation for the nine months ended June 30, 2010 was reduced by \$338,584 from reduced inventory and \$1,245 from increased accounts payable. The investment from LRAD Corporation was increased as a result of the net loss of \$226,880, reduced by expenses not requiring the use of cash of \$85,481, \$49,803 increase in accounts receivable, \$9,709 increase in warranty settlements and \$23,455 decrease in accrued payroll. Our investment from LRAD Corporation for the nine months ended June 30, 2009 decreased by \$7,903 as a result of \$343,068 lower inventory, and increased as a result of the net loss of \$782,632, reduced by expenses not requiring the use of cash of \$517,549, \$6,191 for reduced accounts payable, \$41,607 from higher accounts receivable, \$363 increase in warranty settlements and \$21,921 for lower accrued payroll.

# Investing Activities

We used cash of \$10,051 and \$6,481 during the nine months ended June 30, 2010 and 2009 for investments in patents.

# Financing Activities

Financing activities used cash of \$105,412 during the nine months ended June 30, 2010, and \$126,456 for the fiscal year ended September 30, 2009 and generated cash of \$577,576 for the fiscal year ended September 30, 2008. All cash and investments are held and managed by LRAD Corporation. Accordingly, cash used to pay our expenses or cash collected from customers by LRAD Corporation on our behalf are recorded as an increase or decrease in the LRAD Corporation net investment (capital deficiency). Other than cash received from customers, LRAD Corporation has funded our research and development expenses. Thus, our financing activities primarily represent the operating expenses funded by LRAD Corporation.

### **Contractual Obligations**

We have no material contractual obligations. We will be obligated to pay to Syzygy up to a 5% royalty on future product sales for use of their patent pending technology and trade secrets included in the new product line.

### Effects of Inflation

We do not believe that inflation has had a material impact on our business, revenues or operating results during the periods presented.

### **Recent Accounting Pronouncements**

There have been no recent accounting pronouncements or changes in accounting pronouncements during the period ended June 30, 2010, or subsequently thereto, that we believe are of potential significance to our financial statements.

### BUSINESS

### Overview

We are a sound technology company focused on delivering directed parametric sound solutions to customers primarily in digital signage, point-of-purchase, in-store networks and related markets that benefit from sound that can be focused and controlled in specified locations.

A directed sound solution includes (a) the ability to accept various user media sources as input, typically a computer, a DVD player, radio, television or microphone, and (b) custom processing and control electronics allowing an amplifier to drive speakers (called emitters in parametric sound applications). We intend to continue the pioneering work performed in this field by LRAD Corporation. Since 2004, LRAD Corporation has sold over 11,000 HSS systems, mostly the HSS H450 product model. We plan to replace the H450 model with an improved solution employing improved emitters complemented by new patent-pending processing and control electronics developed by our Chief Executive Officer, Elwood G. Norris, and that will be licensed by us following the distribution. Our prototypes evidence improved sound clarity, wider freq uency response and reduced distortion with higher sound volume from less power input. Based on our costing of prototype components and estimates of assembly costs, we expect to produce new products at a reduced cost compared to the H450.

We believe improved sound quality produced at reduced cost differentiates us from competitors and opens new market opportunities for directed sound solutions. The existing emitter design has been proven repeatable, and we believe our new processing and control electronics package can be readily assembled by any number of electronic component manufacturers reducing the overall risk of introducing our new generation of products.

### History

LRAD Corporation pioneered a new paradigm in sound production based on well-known principles of physics. Parametric or nonlinear acoustics use changing pressures in air to produce sound indirectly by carrying content into the air along ultrasonic frequencies.

Parametric sound technology has gone through various iterations of both emitter designs and amplifier and processing electronics over the last 15 years. Although we consider that LRAD Corporation was a leader in this niche market, we believe the potential for applications of parametric directed sound have not been realized primarily due to limitations of the current product and its relatively high cost compared to sound systems for targeted applications. Further, due to rapid growth in its LRAD business, LRAD Corporation has not focused new resources on its HSS business in part due to its lower product margins and limited market acceptance.

Mr. Norris was the original inventor of LRAD Corporation's HSS technology. In April 2005, he was awarded the \$500,000 Lemelson-MIT Prize for the HSS acoustics innovation. In April 2009, he stepped down as Chairman of LRAD Corporation, in September 2009, he retired from an active role at LRAD Corporation and in June 2010, he resigned as a director to focus on Parametric's business. LRAD Corporation in the last several years has focused most of its financial, technical and marketing resources on its LRAD business that is now driving profitable operations. In January 2010, Mr. Norris began discussions with LRAD Corporation about an HSS license and in the course of due diligence independently invented new processing and control electronics that he believes complements the current emitter design resulting in a new, im proved and more economical product. After discussions with LRAD Corporation's Board of Directors, approved a plan to spin-off would be the best means of improving HSS technology. Mr. Norris then agreed to take an active role in our company. In April 2010, LRAD Corporation's Board of Directors approved a plan to spin-off the HSS business assets by contributing them to a newly formed subsidiary, Parametric Sound Corporation. The spin-off was approved by LRAD Corporation stockholders on June 2, 2010.

Since January 2010, Mr. Norris has spent significant time evaluating the parametric sound technology, inventing new concepts and testing new processing and control electronics and prototypes along with changes to the emitters to improve sound reproduction. His goal is a directional sound solution more competitive in cost to existing audio systems for targeted applications, but with the benefit of directionality. Mr. Norris has filed for patent protection on these innovations, and the new technology and trade secrets will be licensed through Syzygy, to Parametric Sound (see "*License*" on page 41).

### Technology

The common speaker types in use today such as dynamic, electrostatic, ribbon and other transducer-based designs, are direct radiating, and are fundamentally a piston action, directly pumping air molecules into motion to create audible sound waves we hear. Parametric, or nonlinear acoustics, on the other hand, create sound "in the air." Audible sound is generated along an ultrasonic column using frequencies above the normal range of hearing. This parametric sound beam is highly directional and maintains sonic clarity and intelligibility over longer distances than traditional loudspeakers. Our technology is compatible with any media input but beams focused sound where you want it and nowhere else.

Parametric sound employs ultrasonic frequencies to carry content, such as music and voice, into the air. Proprietary ultrasonic emitters, or transducers, which convert electrical energy to high frequency acoustical energy, produce these ultrasonic frequencies beyond the range of hearing. These ultrasonic emitters are used in lieu of loudspeakers to emit a custom-generated ultrasonic wave with the proper difference frequency characteristics to produce audible sound within and throughout a tightly formed beam. Audible sound is not created on the surface of the ultrasonic emitter—a significant departure from a direct radiating loudspeaker. Instead, the audible sound is generated in the air itself and is focused and directed. For example, if the acoustic beam is directed towards a wall, the sound first emanates from the surface of the wall, not from the emitter, as it would be with a conventional loudspeaker. Or, if the acoustic beam is directed to a person, the sound is created at the person. This directionality allows sound to be focused or "beamed" in space or diffused from a surface in a variety of ways to produce desired effects. Also, the sound does not spread at the same rate over distance as it does with raditional speakers. This unique feature provides improved intelligibility at selected distant points with less energy than traditional speakers, creating the ability to communicate directed sound at longer distances.

A typical sound solution for our targeted applications includes (a) inputs to accept a user media source, typically a computer, a DVD player, radio or microphone, and (b) an amplifier and related control electronics to drive speakers. A parametric sound solution includes (a) the ability to accept various user media sources as input, typically a computer, a DVD player, radio, television or microphone, (b) custom processing and control electronics allowing an amplifier to drive emitters and (c) ultrasonic emitters to produce sound in the air. Parametric solutions to date, including those of LRAD Corporation and our competitors, have demonstrated lower sound quality and volume, higher distortion and higher costs than standard sound systems in comparable applications.

LRAD Corporation developed proprietary emitter technology employing custom thin piezo film manufactured pursuant to our formulation. LRAD Corporation's intellectual property includes the film formulation and methods to reliably produce emitters using this custom film. While we believe the current emitter design and production techniques can be improved over time, the current emitters have proven reliable in the current H450 product and our new processing and control electronics package has been specifically designed to improve the output and performance of the existing emitter technology. We depend on a single piezo-film supplier, Measurement Specialties Inc., to provide expertise and materials used in our proprietary emitter s. We believe that the other components incorporated from our products can be obtained from multiple supply sources. As such, we do not believe that our other current suppliers are material to the operation of our business at this time.

LRAD Corporation's HSS electronics employed sophisticated digital processing algorithms requiring expensive components. We believe the processing and control electronics invented, designed and developed by Mr. Norris address some of the competitive issues facing this technology. We further believe the new electronics system complements the current emitter and uses less power to generate higher output providing less stress and potential for emitter failure. Our new products are intended to offer customers improved sound quality and higher volume with wider frequency response and reduced distortion while being simpler and more economical to produce.

### License

We intend to enter into a license agreement with Syzygy promptly following the distribution Syzygy is an entity in which Mr. Norris is majority owner. Mr. Norris has assigned to Syzygy certain patent pending technology and trade secrets related to a new, improved and more cost-effective method of processing media input to create parametric sound output for parametric emitter devices such as those employed by us. He has also invented improvements to the emitters complementary with the new processing and control electronics. We have agreed to reimburse patent, testing and prototype costs include up to a maximum of \$25,000 for Mr. Norris' time in producing and testing prototypes and preparing for production.

The license, which has not been executed as of the date of this information statement, but is expected to be executed promptly following the distribution, will provide for future royalties of 5% of revenues from products employing the technology and a term of 20 years or the life of any resulting patent, whichever is greater. In the event no patent covering the licensed technology is issued after four years, then the royalty rate shall reduce to 3% in any territory until or if a patent is issued for any such territory. We may not sublicense without the permission of Syzygy and sublicense royalty rates are subject to future negotiation in good faith. The license will be entered into only if the distribution is successfully completed and after effectiveness may terminate if we do not use commercially reasonable efforts t o pursue the parametric sound business.

Although we expect to enter into the license promptly following the distribution, there is no assurance that the license will be executed or that we will have the benefit of the technology that is the subject of the license. If we are unable to obtain the benefits of the technology owned by Syzygy, we will be unable to develop new products and grow our business, which will have a material adverse effect on our business, liquidity, results of operation and financial position.

### Products

The HSS H450 product has a 5 by 10 inch emitter surface and has been produced by LRAD Corporation since 2005. While we will own all rights and certain inventory to continue to build the H450 product, we have no current plans to do so. We have designed a new generation of HSS products and are in early stages of production development for the following two initial products:

<u>HSS-S2150</u> – a stereophonic system employing two 5 by 10 inch emitters. Unlike the H450, the electronics are separate from the emitters connected by essentially standard speaker wire allowing for a great variety of installations and flexibility to connect to multiple media sources.

HSS-M1150 – is a monaural system from one 5 by 10 inch emitter. Similar to the HSS-S2150, this product features electronics separate from the emitter providing greater installation flexibility.

We expect to develop and introduce additional product models in the future. We also expect to have the flexibility to customize products for individual customers including features such as a wireless interface, performance monitoring and other features.

### Strategy

Our goal is to realize the potential of directed sound primarily targeting digital signage, point-of-purchase, in-store networks and related markets and applications. We aim to produce a product line to meet the requirements of these customers. The key elements of our strategy include:

- Produce a product line meeting customer requirements for easy and flexible installation using content from a range of media sources, complementing equipment such as video devices, kiosks and vending machines.
  - Develop a turnkey manufacturing relationship to produce our products, thereby reducing a need for manufacturing space and production personnel.
- Develop relationships with customers requiring large numbers of products, including value added resellers (VARs), original equipment manufacturers (OEMs) and distributors that focus on specific end user solutions.

We also plan to explore uses for our technology in the emerging stereoscopic three-dimensional (3D) video market for computers, homes and theaters. The growth of 3D viewing creates opportunities for new 3D sound systems. We intend to explore the use of parametric speakers to reproduce binaural recordings. Binaural recordings use special microphones to capture directional sound information during recording and typically must be replayed through headphones. Parametric speakers reduce loudspeaker cross-talk that interferes with binaural reproduction from conventional stereo speakers. We believe our speakers complement or enhance digital signal processing (DSP) techniques such as ambiophonics that are designed to emulate headphones for binaural sound reproduction from two speakers.

Our research and development strategy is to continue to develop innovative directed acoustic solutions and design new product solutions for our target markets.

# Manufacturing and Suppliers

We have substantially completed our electronic design and component selection. We are testing our new electronics system while preparing for initial production. We have employed a professional and experienced industrial design and turnkey product development firm for mechanical and manufacturing design of our product, to produce pilot production models and support manufacturing startup. We expect to employ certain existing emitter parts, molds and manufacturing processes. Our new electronics system consists mostly of off the shelf components fitting into a standard electronics board with traditional input and output connectors.

LRAD Corporation developed custom manufacturing equipment used to automate the production of our emitter sub-assemblies reducing the labor component and permitting higher volume production. We expect to build or have our emitters built by others under contract using this equipment. We plan to have our electronic boards produced by one or more contract manufacturers on a turnkey basis. We plan to purchase molded parts from suppliers and assemble and test final products at our facility or contract others for assembly. We may elect to utilize internal quality control personnel depending on circumstances as they arise.

Our technology is substantially different from proven, mass-produced sound transducer designs, and manufacturing and assembly involves new processes and materials. We believe arranging and maintaining quality manufacturing capacity will be essential to the performance of our products and the growth of our business.

### Selling, Marketing and Customers

We intend to sell our products using our executive officers and commissioned agents. We intend to employ one sales employee of LRAD Corporation but do not currently intend to develop an internal sales force but may do so in the future. We also intend to approach value added resellers (VARs) to use our sound solution in their products as well as engage distributors and independent selling representatives. There can be no assurance we can develop adequate distribution through these or other methods.

We intend to initially target the growing direct advertising market including digital signage, point-of-purchase and in-store networks. We believe our sound solution mitigates the effects of sound clutter often attributable to these products. In-store display advertising tends to irritate customers if too intrusive or loud and annoy workers due to repetition. Our products are being designed to achieve focused, controlled sound that targets only those customers situated in specified locations such that nearby customers and store clerks do not hear the message. We believe the ability to focus sound is a driving feature of our sound solution. We believe our technology offers a number of advantages:

delivery of more effective advertisements to store patrons;
ability to create a beam of sound and place it only where it is intended;
ability to manipulate or selectively position or diffuse the source of sound;
ability to deliver a beam of sound over longer distances than conventional speakers, such as down a grocery store aisle;
ability to penetrate other competing sounds; and
elimination of feedback from live microphones.

We expect additional sound applications to become evident as our products become more widely known and accepted. We intend to explore the uses of our speakers in the 3D sound market.

#### Warranties

We expect to warrant our products to be free from defects in materials and workmanship for a period up to one year from the date of purchase. The warranty will be generally a limited warranty, and in some instances impose certain shipping costs on the customer. We expect in most cases it will be more economical and effective to replace the defective part, whether emitter or electronics, rather than repair, but in the future we may establish repair warranty service directly or through others.

Early versions of LRAD Corporation's HSS products experienced excessive failures and warranty issues. The H450 has been more reliable, and we expect even greater reliability of our new product series because we generate higher sound levels at lower wattage producing less heat and stress on emitters. The new electronics system was designed with the knowledge gained from the H450 results.

### Competition

Our technologies and products compete with those of other companies. The consumer, commercial and government audio industry markets are fragmented and competitive and include numerous manufacturers with audio products that vary widely in price, quality and distribution channels. Many of our present and potential future competitors have, or may have, substantially greater resources to devote to further technological and new product developments.

We believe our new product will be the leading parametric sound system with limited direct competition. Companies such as Brown Innovations and others have employed domes and other techniques to try to focus or contain sound for directed sound applications such as point-of-sale. We do not believe these methods are directly competitive to our parametric sound solution in ease of use, cost and performance. Although others have attempted to use parametric speaker concepts or other concepts to produce directed sound, we do not believe they have progressed to directly competitive commercial products as compared to our planned product line. Holosonic Research Labs, Inc. produces a parametric speaker called the Audio Spotlight; Panphonics produces the Sound Shower directional speaker; Sennheiser Electronics has announced a parametric speaker product called the AudioBeam Master; and Mitsubishi has a parametric speaker product sold in Japan. These companies employ electrostatic and piezoelectric emitter devices or other concepts that we believe have lower output and are more expensive than our proprietary emitters, especially when combined with our new electronics system. However, these competitors or others may introduce products with features and performance competitive to our products.

### Seasonality

We do not expect to experience any significant seasonality trends. Seasonality trends may occur in the future.

### **Government Regulation**

Our electronic products are subject to various regulations and are required to meet the specifications of agencies such as the Federal Communications Commission (FCC). We believe we will be in substantial compliance with all current applicable regulations, and we expect to have all material governmental permits, licenses, qualifications and approvals required for our operations.

Our parametric sound technology is subject to control under the Radiation Control for Health and Safety Act of 1968, and the associated regulations promulgated by the Food and Drug Administration (FDA), as an electrical emitter of ultrasonic vibrations. Under the terms of such regulations, LRAD Corporation provided an abbreviated report to the FDA describing the technology. The FDA may respond to the report and request changes or safeguards to the technology, but it has not done so to date. We will also be required to notify the FDA in writing should a product be found to have a defect relating to safety of use due to the emission of electronic product radiation. We do not believe our technology poses any human health risks. However, it is possible that we, or one of our customers, could be required to modify the technology, or a product incorporating the technology, to comply with requirements that may be imposed by the FDA.

Our products will be produced to standard product safety requirements for sale in the United States and to similar requirements for sale in Europe and Canada. We expect to meet the electrical and other regulatory requirements for electronic systems or components we sell throughout the world.

### **Intellectual Property**

We have a substantial base of intellectual property assets including patents, pending patents, trademarks and trade craft and trade secrets such as know-how. We operate in an industry where innovations, investment in new ideas and protection of resulting intellectual property rights are important to success. We expect to rely on a variety of intellectual property protections for our products and technologies, including contractual obligations, and we intend to pursue a policy of vigorously enforcing such rights.

LRAD Corporation contributed to us 28 U.S. patents and 3 foreign patents. Our patents expire between 2016 and 2025. However, to maintain such patents we are required to pay periodic maintenance fees with nine patents requiring maintenance fees to be paid yet in 2010. We will evaluate all our patents with respect to our product offerings and determine which to maintain and which to let lapse as they become due for renewal. We have four pending patent applications that we evaluate periodically to determine which, if any, to continue to pursue for issuance. We also believe we own important tradecraft and trade secrets in the design and production of our piezo film ultrasonic emitter.

Mr. Norris, through Syzygy, has filed for patent coverage for the new electronics processing system and emitter improvements that will be exclusively licensed to us following the distribution date. Mr. Norris is an experienced inventor with multiple patents and believes the new system is novel and will result in one or more patents. If we enter into the proposed exclusive license, we will agree to pay for patent related expenses.

In addition to such factors as innovation, technological expertise and experienced personnel, we believe that a strong patent position is important to compete effectively in the sound reproduction industry. We have an ongoing policy of filing patent applications to seek protection for novel features of our products and technologies. Prior to the filing and granting of patents, our policy is to disclose key features to patent counsel and maintain these features as trade secrets prior to product introduction. Patent applications may not result in issued patents covering all important claims and could be denied in their entirety.

The electronics industry is characterized by frequent litigation regarding patent and other intellectual property rights. Others, including academic institutions and competitors, hold numerous patents in electronics and sound reproduction. Although we are not aware of any existing patents that would materially inhibit our ability to commercialize our sound technology; others may assert claims in the future. Such claims, with or without merit, may have a material adverse effect on our financial condition, results of operations or cash flows.

The validity of existing patents has not been adjudicated by any court. Competitors may bring legal action to challenge the validity of our existing or future patents or may attempt to circumvent the protection provided by such patents. The failure to obtain patent protection or the loss of patent protection on our existing and future technologies or the circumvention of our patents by competitors could have a material adverse effect on our ability to compete successfully.

We acquired four U.S. trade names from LRAD Corporation along with additional foreign rights to some of the same names. We intend to file for trade name and trademark protection when appropriate. Trade names or trademarks may not be successfully maintained, defended or protected.

Our policy is to enter into nondisclosure agreements with each employee and consultant or third party to whom any of our proprietary information is disclosed. These agreements prohibit the disclosure of confidential information to others, both during and subsequent to employment or the duration of the working relationship. These agreements may not prevent disclosure of confidential information or provide adequate remedies for any breach.

## **Employees and Executive Officers**

At the distribution date we will have two executive officers, Elwood G. Norris, our President and Chief Executive Officer, and James A. Barnes, Secretary and Treasurer. We have no other employees but intend to employ the current HSS sales employee of LRAD Corporation.

# Properties

We expect to lease approximately 4,500 square feet of office, testing, assembly and warehousing space at 1941 Ramrod Avenue, Suite #100, Henderson, Nevada 89014, pursuant to a one year lease at a monthly rate of \$4,000 that will only become an obligation of Parametric Sound at the distribution date. Syzygy has entered into the lease commencing July 1, 2010 for transfer to and use by Parametric Sound upon spin-off. We expect that this property will be sufficient to meet our needs for at least the next 12 months. We have sublet a portion of the space on a month-to-month basis for \$2,000 per month reducing the current monthly payment to \$2,000.

#### MANAGEMENT

### **Directors and Executive Officers**

Set forth below is information concerning those persons that will serve as executive officers and directors of Parametric Sound immediately following the distribution date.

Name	Age	Position(s)
Elwood G. Norris	71	Director, President and Chief Executive Officer
Daniel Hunter	59	Director
James A. Barnes	55	Secretary and Treasurer

After the spin-off, when and as business requires and funds are available, we may hire or appoint additional executive officers. We have no understanding or arrangements regarding any additional executive officers to be appointed on or after the spin-off date.

*Elwood G. Norris* will serve as a director and as President and Chief Executive Officer following the distribution date. He was a director of LRAD Corporation from August 1980 to June 2010. He served as Chairman of LRAD Corporation's Board of Directors, an executive position, in which he served in a technical advisory role and acted as a product spokesman from September 2000 to April 2009. From 1988 to November 1999, he was a director and Chairman of e.Digital Corporation, a public company engaged in electronic product development, licensing and sales. During that period, he also held various other executive officer positions at e.Digital. From August 1989 to October 1999, he served as director and held various executive officer positions with Patriot Scientific Corporation, a public company engaged in intellectual property licensing. Since 2000, he has been a director of AirScooter Corporation, a non-reporting public company. He is an inventor of there are 50 U.S. patents, primarily in the fields of electrical and acoustical engineering, and is a frequent speaker on innovation to corporations and government organizations. He was the inventor of the HyperSonic Sound (HSS) technology. Mr. Norris is one of the owners of Syzygy, but has no employment or management relationship with Syzygy. Mr. Norris expends, and expects following the spin-off to expend, minimal time on Syzygy matters.

Daniel Hunter, will be a director on the distribution date. He was a director of LRAD Corporation from May 2001 to March 2010. Mr. Hunter has been a licensed certified public accountant for the past 34 years. He obtained his accounting degree from the University of Utah in 1973. For the past 29 years, Mr. Hunter has operated his own law offices specializing in business and tax law. He obtained his J.D. from the University of Seattle in 1978.

James A. Barnes will be Secretary and Treasurer on the distribution date. He has been President of Sunrise Capital, Inc., a private venture capital and consulting firm since 1984. He participated in the recapitalization of LRAD Corporation and the founding of e.Digital Corporation, Patriot Scientific Corporation and other technology companies.. Since 1999, he has been Manager of Syzygy Licensing LLC, a private technology invention and licensing company owned with Mr. Norris. Since 2000, he has also been a director and Secretary of AirScooter Corporation a non-reporting public company. He previously practiced as a certified public accountant and management consultant with Ernst & Ernst (1976-1977), Touche Ross & Co. (1977-1980) and as a principal in J. McDonald & Co. Ltd., Phoenix, Arizona (1980-1984). He graduated from the University of Nebraska with a B.A. Degree in Business Administration in 1976 and is a certified public accountant. Mr. Barnes is expected to devote approximately 20 hours per week to Parametric Sound.

## Key Personnel

We expect to enter into an employment, sales representation or consulting relationship with David Pratt after the distribution. Since August 2003, Mr. Pratt has been Director of Business Development responsible for HSS business and sales development at LRAD Corporation. He has developed contacts and relationships with top audio/video distributors and resellers and nurtured relationships with media companies, advertisers and brand managers, retailers and network service providers throughout the digital signage industry.

#### **Board of Directors**

After the spin-off, we expect that Parametric Sound's Board of Directors will consist of two members, one of whom will be an independent director. Each director will hold office, in accordance with the Certificate of Incorporation and Bylaws of Parametric Sound, until the next annual meeting of stockholders and until his or her successor is duly elected and qualified.

Effective as of the spin-off date, Elwood G. Norris and Daniel Hunter will serve as directors of Parametric Sound. We expect that each of Mr. Norris and Mr. Hunter will continue as a director of Parametric Sound, and we have no additional current arrangements regarding any additional directors to be appointed to our Board of Directors on or after the spin-off date.

#### Director Independence

For a director to be considered "independent," the Board must affirmatively determine that the director has no material relationship with Parametric Sound (directly or as a partner, stockholder or officer of an organization that has a relationship with Parametric Sound). In each case, the Board considers all relevant facts and circumstances.

We expect during 2010 to designate at least one additional director such that at least two of our directors will be independent.

## **Committees of the Board of Directors**

During 2010, we plan to add at least one additional Board member and then we expect our Board of Directors will establish an Audit Committee and a Compensation Committee to assist it with its responsibilities. We expect all members of the Audit and Compensation Committees will meet the criteria for independence as established under the Sarbanes-Oxley Act of 2002. Each of the planned Committees is described in greater detail below. The Board will establish written charters for each of the Committees when formed, which will be available on our web site located at <u>www.parametricsound.com</u>.

## Audit Committee

We expect to designate members of our Audit Committee sometime in 2010. The principal duties of the Audit Committee under its written charter will include: (i) responsibilities associated with our external and internal audit staffing and planning; (ii) accounting and financial reporting issues associated with our financial statements and filings with the SEC; (iii) financial and accounting organization and internal controls; (iv) auditor independence and approval of non-audit services; and (v) "whistle-blower" procedures for reporting questionable accounting and audit practices.

The Audit Committee charter will require that the Committee be comprised of at least two directors, each of whom must be independent under the standards of the Sarbanes-Oxley Act of 2002. In addition, each member of the Audit Committee will be financially literate, and at least one member will have sufficient accounting or financial management expertise to qualify as an "audit committee financial expert," as determined by the Board in accordance with SEC rules.

## Compensation Committee

We expect to designate members of our Compensation Committee sometime in 2010. The principal duties of the Compensation Committee under its charter will include: (i) ensuring that a succession plan for the Chief Executive Officer is in place; (ii) reviewing management's recommendations for executive officers and making recommendations to the Board of Directors; (iii) approving compensation for the Chief Executive Officer; (iv) reviewing and approving compensation policies and practices for other executive officers including their annual salaries; (v) reviewing and approving major changes in employee benefit plans; (vi) reviewing short and long-term

incentive plans and equity grants; (vii) recommending to the full Board changes to the compensation of the independent members of the Board of Directors; and (viii) reviewing the Compensation Discussion and Analysis to be included in our annual report or proxy statement and, if appropriate, issuing its report recommending to the Board of Directors its inclusion of the Compensation Discussion and Analysis in our annual report or proxy statement. The Compensation Committee charter will require that the Committee be comprised of at least two independent directors. The scope of authority delegated to the Compensation Committee by the Board of Directors is to decide whether or not to accept, reject or modify our management's proposals for annual compensation awards to our executive officers. The Compensation Committee also has the authority to recommend the amount of compensation to be paid to our non-management directors.

## DIRECTOR COMPENSATION

We have not yet established arrangements to compensate our directors for their services to us following the separation. However, we expect that compensation for our non-employee directors will be comprised of an annual cash retainer and an equity award in the form of stock option grants.

### EXECUTIVE COMPENSATION

#### **Compensation of our Named Executive Officers**

We have identified Elwood G. Norris and James A. Barnes as our named executive officers. Our named executive officers for 2010 could change, as we may hire or appoint new executive officers. We will develop our own compensation plans and programs and anticipate that each of our executive officers will be covered by these programs following the spin-off.

An entity controlled by Mr. Norris and Mr. Barnes will receive a royalty as described above in "Business—License" in consideration of technology licensed. No other compensation has been determined as of the date hereof. We expect that Mr. Norris and Mr. Barnes may be compensated in their roles as executive officers commencing after the spin-off as determined by our Board of Directors.

### Description of the 2010 Stock Option Plan

Promptly following the distribution, Parametric Sound intends to adopt the Parametric Sound Corporation 2010 Stock Option Plan. The following is a description of the purpose and certain of the provisions of our 2010 Stock Option Plan as it will be adopted following the distribution. The summary is qualified in its entirety by reference to the complete text of the 2010 Stock Option Plan, which is filed as an exhibit hereto.

Purpose of the 2010 Stock Option Plan. The purpose of the 2010 Stock Option Plan will be to provide additional incentive to our directors, officers, employees and consultants who will be primarily responsible for our management and growth. Each option will be designated at the time of grant as either a non-qualified stock option (a "NQSO") or an incentive stock option (an "ISO").

The Internal Revenue Code requires that ISOs be granted pursuant to an option plan that receives stockholder approval within one year of its adoption. We may seek shareholder approval of the 2010 Stock Option Plan to permit us to grant ISOs under the 2010 Stock Option Plan. The benefits to be derived from the 2010 Stock Option Plan, if any, are not quantifiable or determinable.

Administration of the 2010 Stock Option Plan. Following its adoption, the 2010 Stock Option Plan will be administered by our Board of Directors, or by any committee that we may in the future form and to which our Board of Directors may delegate the authority to perform such functions (in either case, the "Administrator"). Our Board of Directors will appoint and remove members of the committee in its discretion in accordance with applicable laws. In the event that we establish such a committee and it is required to comply with Rule 16b-3 under the Exchange Act and Section 162(m) of the Code, the committee will, in our Board of Directors" discretions, be comprised solely of "non-employee directors" within the meaning of said Rule 16b-3 and "outside directors" within the meaning of Section 162(m) of the Code. Notwithstanding the foregoing, the Administrator may delegate non-discretionary administrative duties to such Parametric Sound employees as it deems proper and our Board of Directors, in its absolute discretion, may at any time and from time to time exercise any and all rights and duties of the Administrator under the 2010 Stock Option Plan.

Subject to the other provisions of the 2010 Stock Option Plan, the Administrator will have the authority, in its discretion, to: (i) grant options; (ii) determine the fair market value of the common stock subject to options; (iii) determine the exercise price of options granted; (iv) determine the persons to whom, and the time or times at which, options will be granted, and the number of shares subject to each option; (v) interpret the 2010 Stock Option Plan; (vi) prescribe, amend and rescind rules and regulations relating to the 2010 Stock Option Plan; (vii) determine the terms and provisions of each option granted (which need not be identical), including but not limited to, the time or times at which options will be exercisable; (viii) with the consent of the optionee, modify or amend any option; (x) authorize any person to execute on our behalf any instrument evidencing the grant of an option; and (xi) make all other determinations deemed necessary or advisable for the administration of the 2010 Stock Option Plan. The Administrator may delegate non-discretionary administrative duties to such Parametric Sound employees as it deems proper.

Shares of Stock Subject to the 2010 Stock Option Plan. Subject to the conditions outlined below, following adoption of the 2010 Stock Option Plan, the total number of shares of stock which will be available for issuance under options granted pursuant to the 2010 Stock Option Plan will not exceed 3,000,000 shares of Parametric Sound common stock, \$0.001 par value per share. The number of shares of common stock subject to options granted pursuant to the 2010 Stock Option Plan may be adjusted under certain conditions. If Parametric Sound common stock is changed by reason of a stock split, reverse stock split, stock dividend, recapitalization, combination or reclassification, appropriate adjustments will be made by our Board of Directors in (i) the number and class of stock subject to the 2010 Stock Option Plan, and (ii) the exercise price of each outstanding option; provided, however, that we will not be required to issue fractional shares as a result of any such adjustments. Each such adjustment will be subject to approval by our Board of Directors in its sole discretion.

In the event of the proposed dissolution or liquidation of Parametric Sound, the Administrator will notify each optionee at least 30 days prior to such proposed action. To the extent not previously exercised, all options will terminate immediately prior to the consummation of such proposed action; provided, however, that the Administrator, in the exercise of its sole discretion, may permit exercise of any options prior to their termination, even if such options were not otherwise exercisable. In the event of a merger or consolidation of Parametric Sound with or into another corporation or entity in which we do not survive, or in the event of a sale of all or substantially all of our assets in which our stockholders receive securities of the acquiring entity or an affiliate thereof, all options will be assumed or equivalent options will be substituted by the successor corporation (or other entity) or a parent or subsidiary of such successor corporation (or other entity); provided, however, that if such successor does not agree to assume the options or to substitute equivalent options therefor, the Administrator, in the exercise of its sole discretion, may permit the exercise of any of the options prior to consummation of such event, even if such options were not otherwise exercisable.

Participation. Every person who at the date of grant of an option is an employee of Parametric Sound or any of our Subsidiaries (as defined below) is eligible to receive NQSOs or ISOs (following stockholder approval of the 2010 Stock Option Plan) under the 2010 Stock Option Plan. Every person who at the date of grant is a consultant to, or non-employee director of, Parametric Sound or any of our Subsidiaries (as defined below) is eligible to receive NQSOs under the 2010 Stock Option Plan. The term "Subsidiary" as used in the 2010 Stock Option Plan means a subsidiary corporation as defined in the applicable provisions (currently Section 424(f)) of the Code. The term "employee" includes an officer or director who is an employee of Parametric Sound. The term "consultant.

Option Price. The exercise price of a NQSO will be not less than 85% of the fair market value of the stock subject to the option on the date of grant. To the extent required by applicable laws, rules and regulations, the exercise price of a NQSO granted to any person who owns, directly or by attribution under the Code (currently Section 424(d)), stock possessing more than 10% of the total combined voting power of all classes of stock of Parametric Sound or any of our Subsidiaries (a "10% Stockholder") will in no event be less than 110% of the fair market value of the stock covered by the option at the time the option is granted. The exercise price of an ISO will be determined in accordance with the applicable provisions of the Code and will in no event be less than the fair market value of the stock covered by the option at the time the option is granted. The exercise price of an ISO granted to any 10% Stockholder will in no event be less than 110% of the fair market value of the stock covered by the option at the time the option is granted. The exercise price of an ISO granted to any 10% Stockholder will in no event be less than 110% of the fair market value of the stock covered by the option at the time the option is granted.

Term of the Options. The Administrator, in its sole discretion, will fix the term of each option, provided that the maximum term of an option will be ten years. ISOs granted to a 10% Stockholder will expire not more than five years after the date of grant. The 2010 Stock Option Plan provides for the earlier expiration of options in the event of certain terminations of employment of the holder.

Restrictions on Grant and Exercise. Except with the express written approval of the Administrator which approval the Administrator is authorized to give only with respect to NQSOs, no option granted under the 2010 Stock Option Plan will be assignable or otherwise transferable by the optionee except by will or by operation of law. During the life of the optionee, an option will be exercisable only by the optionee.

Termination of the 2010 Stock Option Plan. The 2010 Stock Option Plan will become effective upon adoption by our Board or Directors following the distribution; provided, however, that if stockholder approval of the 2010 Stock Option Plan is not obtained within twelve months after adoption by our Board of Directors, all ISOs granted pursuant to the 2010 Stock Option Plan will be treated as NQSOs. Options may be granted and exercised under the 2010 Stock Option Plan only after there has been compliance with all applicable federal and state securities laws. The 2010 Stock Option Plan will terminate within ten years from the date of its adoption by our Board of Directors.

*Termination of Employment.* If for any reason other than death or permanent and total disability, an optionee ceases to be employed by Parametric Sounds or any of our Subsidiaries (such event, a "termination"), options held at the date of termination (to the extent then exercisable) may be exercised in whole or in part at any time within three months of the date of such termination, or such other period of not less than 30 days after the date of such termination as is specified in the option Agreement or by amendment thereof (but in no event after the expiration date of the option); provided, however, that if such exercise of the option vould result in liability for the optionee under Section 16(b) of the Exchange Act, then such three-month period auto matically will be extended until the tenth day following the last date upon which optionee has any liability under Section 16(b) (but in no event after the expiration date). If an optionee dies or becomes permanently and totally disabled (within the meaning of Section 22(e)(3) of the Code) while employed by Parametric Sound or any of our Subsidiaries or within the period that the option remains exercisable after termination, options then held (to the extent then exercisable) may be exercised, in whole or in part, by the optionee, by the optionee's personal representative or by the person to whom the option is transferred by devise or the laws of descent and distribution, at any time within 12 months after the death or 12 months after the permanent and total disability of the optionee or any longer period specified in the option Agreement or by amendment thereof (but in no event after the expiration date). "Employment" includes service as a director or as a consultant. For purposes of the 20 10 Stock Option Plan, an optionee's employment will not be deemed to terminate by reason of sick leave, military leave or other leave of absence approved by the Administrator, if the period of any such leave does not exceed 90 days or, if longer, if the optionee's right to re

Amendments to the 2010 Stock Option Plan. Our Board of Directors may at any time amend, alter, suspend or discontinue the 2010 Stock Option Plan. Without the consent of an optionee, no amendment, alteration, suspension or discontinuance may adversely affect outstanding options except to conform the 2010 Stock Option Plan and ISOs granted under the 2010 Stock Option Plan to the requirements of federal or other tax laws relating to ISOs. No amendment, alteration, suspension or discontinuance will require stockholder approval unless (i) stockholder approval is required to preserve incentive stock option treatment for federal income tax purposes or (ii) our Board of Directors otherwise concludes that stockholder approval is advisable.

Tax Treatment of the Options. Under the Code, neither the grant nor the exercise of an ISO is a taxable event to the optione (except to the extent an optionee may be subject to alternative minimum tax); rather, the optione is subject to tax only upon the sale of the common stock acquired upon exercise of the ISO. Upon such a sale, the entire difference between the amount realized upon the sale and the exercise price of the option will be taxable to the optionee. Subject to certain holding period requirements, such difference will be taxed as a capital gain rather than as ordinary income. Optionees who receive NQSOs will be subject to taxation upon exercise of such options on the spread between the fair market value of the common stock on the date of exercise and the exercise price of such options. This spread is treated as ordinary income to the optionee, and we are permitted to deduct as an employee expense a corresponding amount. NQSOs do not give rise to a tax preference item subject to the alternative minimum tax.

## Potential Payments Upon Termination, Death, Disability, or Retirement

We have no executive employee contracts at this time. Every officer and employee is an at will employee. The royalties to Syzygy, controlled by Mr. Norris and Mr. Barnes, are unrelated to employment or their roles as executive officers and will continue upon any termination, death, disability or retirement.

## SECURITY OWNERSHIP BY CERTAIN BENEFICIAL OWNERS AND MANAGEMENT

All of the outstanding shares of our common stock are, and will be, prior to the distribution, held beneficially and of record by LRAD Corporation. The following table sets forth information concerning shares of our common stock projected to be beneficially owned immediately after the distribution date by:

•	each person or entity known by us to be the beneficial owner of 5% or more of the outstanding shares of Parametric Sound common stock;
•	each person who we currently anticipate will be one of our directors at the time of the distribution;
	each person who we currently anticipate will be one of our named executive officers at the time of the distribution; and
•	all persons who we currently anticipate will be our directors and executive officers at the time of the distribution as a group.

The projected share amounts in the table below are based on the number of shares of LRAD Corporation common stock owned by each person or entity on August 17, 2010, as adjusted to reflect the distribution ratio of one share of our common stock for every two shares of LRAD Corporation common stock, or approximately 15,305,728 shares of common stock. To our knowledge, except as otherwise indicated in the footnotes below, each person or entity has sole voting and investment power with respect to the shares of common stock set forth opposite such person's or entity's name. Beneficial ownership is determined in accordance with the rules of the SEC and generally includes voting or investment power with respect to the securities. Warrants that provide for participation in the distribution that are vested or a re scheduled to vest within 60 days are deemed to be outstanding and to be beneficially owned by the persons holding the warrants for the purpose of computing the percentage ownership of the person.

Title of Class	Name and Address of Beneficial Owner	Amount and Nature of Be Ownership	neficial	Percent of Class
Common Stock	Austin W. Marxe and David M. Greenhouse	1,598,592	(1)	10.4%
	527 Madison Avenue, Suite 2600			
	New York, New York 10022			
Common Stock	Elwood G. Norris	1,985,312	(2)	13.0%
	1941 Ramrod Avenue, Suite #100			
	Henderson, Nevada 89014			
Common Stock	Daniel Hunter	95,750	(3)	*
	1941 Ramrod Avenue, Suite #100			
	Henderson, Nevada 89014			
Common Stock	James A. Barnes	235,587	(4)	1.5%
	1941 Ramrod Avenue, Suite #100			
	Henderson, Nevada 89014			
	All directors and executive	2,316,649		15.1%
	officers as a group (3 persons)			

less than 1%.

- (1) Beneficial joint ownership by Mr. Marxe and Mr. Greenhouse is based on information provided by the stockholder as of February 12, 2010. Consists of 1,598,592 shares. These shares are in the following entities: 729,896 shares owned by Special Situations Fund III QP, L.P., 255,213 shares -owned by Special Situations Private Equity Fund, L.P., 85,019 shares owned by Special Situations Technology Fund, L.P., and 528,464 shares owned by Special Situations Technology Fund II, L.P. MGP Advisors Limited Partnership, or MGP, is the general partner of the Special Situations Fund III QP, L.P. and AWM Investment Company, Inc., or AWM, is the general partner of MGP. SST Advisers, L.L.C., or SSTA, is the general partner of the Special Situations Technology Fund, L.P. and Special Situations Technology Fund II, L.P. MG Advisers, L.L.C., or MG, is the general partner of the Special Situations Private Equity Fund, L.P. AWM is the investment adviser to Special Situations Fund III QP, L.P., Special Situations Technology Fund II, L.P. and Special Situations Technology Fund II, L.P. MG Advisers, L.L.C., or MG, is the general partner of the Special Situations Private Equity Fund, L.P. AWM is the investment adviser to Special Situations Fund III QP, L.P., Special Situations Technology Fund II, L.P. and Special Situations Private Equity Fund, L.P. Austin W. Marxe and David M. Greenhouse are the principal owners of MGP, AWM, SSTA and MG, and are principally responsible for the selection, acquisition, voting and disposition of the portfolio securities by each investment adviser on behalf of its fund. Both Messrs. Marxe and Greenhouse share voting and dispositive power with respect to shares held by these stockholders.
- (2) Includes 1,962,815 shares held by a family trust for which Mr. Norris serves as trustee and 22,497 shares held by an investment company controlled by Mr. Norris.
- (3) Includes 3,000 shares held by a personal Individual Retirement Account.
- (4) Includes 12,500 shares held by Sunrise Capital, Inc., 114,837 shares held by Sunrise Management, Inc. Profit Sharing Plan, 102,500 shares held by Palermo Trust, 3,000 shares by a personal retirement plan and 2,250 shares by a personal retirement plan of his spouse. Mr. Barnes is President of Sunrise Capital, Inc. and Trustee of Sunrise Management, Inc. Profit Sharing Plan, the Palermo Trust and his personal retirement plan. He disclaims any beneficial interest in the 2,250 shares held in his spouse's personal retirement plan.

## OUR RELATIONSHIP WITH LRAD CORPORATION AFTER THE SPIN-OFF

#### General

In connection with the spin-off, we and LRAD Corporation will enter into a Separation and Distribution Agreement, which we refer to as the "Separation Agreement," and other ancillary agreements to complete the separation of our business from LRAD Corporation and to distribute our common stock to LRAD Corporation stockholders. These agreements will govern the relationship between us and LRAD Corporation after the distribution and will also provide for the allocation of taxes and other liabilities and obligations attributable to periods prior to the distribution. These agreements will have been prepared before the distribution, and will reflect agreement between affiliated parties established without arms-length negotiation. However, we believe that the terms of this agreement will equitably reflect the benef its and costs of our ongoing relationship with LRAD Corporation.

The expected terms of these agreements, which are subject to change prior to the spin-off, are summarized below. We may enter into other agreements with LRAD Corporation prior to or concurrently with the separation that would relate to other aspects of our relationship with LRAD Corporation following the spin-off. Following the separation, we may enter into other commercial agreements with LRAD Corporation from time to time, the terms of which will be determined at those relevant times.

Copies of these agreements described below are filed as exhibits to our Form 10, of which this information statement is a part. The summaries of the material agreements are qualified in their entireties by reference to the full text of the agreements. We encourage you to read the full text of these material agreements.

## Separation and Distribution Agreement

The Separation and Distribution Agreement will set forth the agreement between us and LRAD Corporation with respect to the principal corporate transactions required to effect our separation from LRAD Corporation; the distribution of our shares to LRAD Corporation stockholders; and other agreements governing the relationship between LRAD Corporation and us following the separation. LRAD Corporation will only consummate the spin-off if specified conditions are met. These conditions are intended to include, among others, final approval of the distribution given by the Board of Directors of LRAD Corporation, and the actions and filings necessary or appropriate under Federal and state securities laws and state blue sky laws of the United States (and any comparable laws under any foreign jurisdictions) in connection with the distribution shall have been taken and, where applicable, become effective or accepted.

Even if these conditions are satisfied, other events or circumstances could occur that could impact the timing or terms of the spin-off or LRAD Corporation's ability or plans to consummate the spin-off. As a result of these factors, the spin-off may not occur and, if it does occur, it may not occur on the terms or in the manner described, or in the timeframe currently contemplated.

## The Contribution

In connection with the distribution, LRAD Corporation has contributed or will contribute to us certain business assets, as described in this information statement. It will effect this contribution by transferring or assigning assets to us. LRAD Corporation will have no interest in our assets and business and will have no obligation with respect to our liabilities after the distribution. Similarly, we will have no interest in the assets of LRAD Corporation's other business and will have no obligation with respect to the liabilities of LRAD Corporation's retained business after the distribution.

#### The Distribution

Following the satisfaction or waiver of all conditions to the distribution as set forth in the Separation Agreement, LRAD Corporation will deliver to the distribution agent certificates representing all of the outstanding shares of our common stock. LRAD Corporation will instruct the distribution agent to distribute those shares on September 27, 2010 or as soon thereafter as practicable, so that each LRAD Corporation stockholder will receive one share of our common stock for every two shares of LRAD Corporation common stock as such stockholder owns as of the record date of the spin-off. *Liabilities and Indemnification* 

Following the spin-off, we will be liable for all liabilities and obligations relating to arising out of or resulting from ownership or use of our assets as conducted at any time on or after the distribution date or arising out of any legal action that is primarily related to the operation of our business following the distribution and as otherwise set forth in the Separation Agreement.

Following the spin-off, LRAD Corporation will be liable for all liabilities and obligations of LRAD Corporation and its subsidiaries arising, or related to the period, prior to the distribution (including but not limited to product warranty liabilities and returns related to sales prior to the distribution or continued sales for outstanding orders at the distribution date) and as otherwise set forth in the Separation Agreement.

Generally, LRAD Corporation will indemnify us, and we will indemnify LRAD Corporation, for losses related to the failure of the other to pay, perform or otherwise discharge, any of the its liabilities and obligations set forth in the Separation Agreement.



#### HSS Customers and Business

LRAD Corporation will fulfill any H450 orders that are received through the separation date. LRAD Corporation may thereafter accept and continue to fulfill continuing purchase orders for a current project with Cardinal Health through the completion of such project. LRAD Corporation shall not pursue new H450 or HSS business after the separation date but may accept unsolicited follow-on orders for the H450 product from current customers through December 31, 2010. LRAD Corporation will provide warranty services for its HSS customers through any warranty term. LRAD Corporation may retain completed inventory or parts to produce the necessary units for the Cardinal Health business and for follow-on business and warranty replacements. All other HSS inventory shall be assigned to Parametric Sound at separation and other parts shall be assigned as and when determined unnecessary to meet the above requirements.

### Sharing of Manufacturing Tooling and Equipment

Parametric will agree to permit LRAD Corporation to use the manufacturing equipment that is part of the assets contributed by LRAD Corporation to Parametric Sound in the distribution, until December 31, 2010 to manufacture, using components and parts owned by LRAD Corporation, the additional H450 products that LRAD Corporation may require to fulfill orders and warranty obligations as permitted hereunder. In addition to the foregoing, the Separation Agreement contemplates that each of LRAD Corporation and Parametric Sound will cooperate with the other in good faith to provide for the manufacturing needs of such other party.

## Expenses

When the Board of Directors of LRAD Corporation conditionally approved the spin-off in April 2010, Mr. Norris agreed to advance third-party fees, costs and expenses in connection with the transaction. Should the transaction not be completed then he and not LRAD Corporation will be responsible for such costs unless LRAD Corporation terminates the spin-off other than for good reason, as defined in the Separation Agreement, whereupon LRAD Corporation will reimburse Mr. Norris. Except as expressly set forth in the Separation Agreement, should the separation be completed then all third-party fees, costs and expenses paid or incurred in connection with the transactions by Mr. Norris, directly or through Syzygy, will be repaid by us at or after the distribution.

#### Termination

The Separation Agreement will provide that it may be terminated by LRAD Corporation at any time prior to the distribution date.

#### Services Agreement

In connection with the spin-off, we expect to operate as independent public companies and we do not expect to contract with LRAD Corporation for any specific services. However, we and LRAD Corporation will cooperate with respect to certain administrative, manufacturing know-how transfer and other services reasonably necessary following the distribution date

### **Employee Benefits**

We do not expect more than one possible employee transferring from LRAD Corporation to us. Accordingly, there is no agreement for sharing or continuing any employee benefits between the parties.

## **Tax Sharing Agreement**

In connection with the spin-off, we and LRAD Corporation will enter into a tax sharing agreement, which sets forth the responsibilities of LRAD Corporation and us with respect to, among other things, liabilities for federal, state, local and foreign taxes for periods before and including the spin-off, the preparation and filing of tax returns for such periods and disputes with taxing authorities regarding taxes for such periods. LRAD Corporation will be generally responsible for federal, state, local and foreign income taxes for periods before and including the spin-off. We will be generally responsible for all other taxes relating to our business. We and LRAD Corporation will each generally be responsible for managing those disputes that relate to the taxes for which each of us is responsible and, under certain circum stances, may jointly control any dispute relating to taxes for which both of us are responsible.

#### **Dispute Resolution**

The Separation Agreement will contain provisions that govern the resolution of disputes, controversies or claims that may arise between us and LRAD Corporation. In the event of any dispute or disagreement between us and LRAD Corporation as to the interpretation of any provision of the Separation Agreement (or the performance of obligations hereunder), we and LRAD Corporation will promptly meet in a good faith effort to resolve the dispute. If the officers do not agree upon a decision within 30 days after reference of the matter to them, each of the parties will submit any controversy, dispute or claim arising out of or relating in any way to the Separation Agreement or the transactions arising hereunder for arbitration in San Diego, California, and such arbitration shall be the sole remedy for such monetary claims. The fees and expenses of the arbitrator(s) will be shared equally by us and LRAD Corporation.

### CERTAIN RELATIONSHIPS AND RELATED PARTY TRANSACTIONS

Elwood G. Norris resigned as a director of LRAD Corporation on June 4, 2010 and will serve as our President and Chief Executive Officer at the distribution date. Daniel Hunter, a director of LRAD Corporation from May 2001 to March 2010, will serve as a director of our company.

On June 4, 2010, LRAD Corporation entered into a consulting agreement with Mr. Norris, whereby LRAD Corporation will pay Mr. Norris \$5,000 per quarter in exchange for business, management and advisory services which may be requested from time to time. The term of this agreement shall extend through May 31, 2012. LRAD Corporation was previously obligated to pay Mr. Norris a 2% royalty on net sales from certain of its technologies, including HSS. The royalty obligation discontinued in October 2008 under the terms of Mr. Norris' amended employment agreement dated November 5, 2008. No royalties were paid or recorded under this agreement in the fiscal years ended September 30, 2009, 2008 or 2007, as these royalties were immaterial and were waived by Mr. Norris. The amounts of the royalties waived were \$0, \$14,266 and \$26,112 for the years ended September 30, 2009, 2008 and 2007, respectively.

Following the distribution, we intend to enter into a license agreement with Syzygy. Mr. Norris is majority owner and Mr. Barnes is manager and a minority owner. Mr. Norris has assigned to Syzygy certain patent pending technology and trade secrets related to a new, improved and more cost-effective method of processing media input to create parametric sound output for parametric emitter devices such as those employed by us. He has also invented improvements to the emitters complementary with the new processing and control electronics. We have agreed to reimburse patent, testing and prototype costs incurred to date (an amount estimated at \$115,000) and to pay future patent related costs. These prior costs include up to a maximum of \$25,000 for Mr. Norris' time in producing and testing prototypes and preparing for p roduction.

The license will provide for future royalties of not more than 5% of revenues from products employing the technology. The term of the license will extend for a period of 20 years or the life of any resulting patent, whichever is greater. We will enter into the license only following the distribution. Thereafter, the license will revert to the licensor if we do not use commercially reasonable efforts to pursue the parametric sound business.

James A. Barnes owns 35% of Syzygy and as its managing member may also be considered a promoter as he was active with Mr. Norris in initiating the spin-off transaction and organizing our company. The consulting costs associated with the spin-off being reimbursed to Syzygy include \$25,000 payable to Sunrise Capital, Inc. for the services of Mr. Barnes to the distribution date.

For a list of LRAD Corporation's fiscal 2009 related person transactions, please see LRAD Corporation's 2010 Proxy Statement, filed with the SEC on January 25, 2010.

#### LEGAL PROCEEDINGS

We are not aware of any pending or threatened legal proceedings in which we are involved. In addition, we are not aware of any pending or threatened legal proceedings in which entities affiliated with our officers, directors or beneficial owners are involved with respect to our operations.

#### DESCRIPTION OF OUR CAPITAL STOCK

Upon the completion of the spin-off, we will be authorized to issue 50,000,000 shares of our common stock, \$0.001 par value per share, and 1,000,000 shares of preferred stock, \$0.001 par value per share. The following description of our capital stock is subject to and qualified in its entirety by our Certificate of Incorporation and Bylaws, which are included as exhibits to the registration statement on Form 10 of which this information statement is a part, and by the provisions of applicable Nevada law.

## Authorized and Outstanding Capital Stock

Immediately following the spin-off, our authorized capital stock will consist of 50,000,000 shares of common stock, par value \$0.001 per share, and 1,000,000 shares of preferred stock, par value \$0.001 per share. Based on the approximately 30,611,456 shares of LRAD Corporation common stock that we expect to be outstanding on the record date, and a distribution ratio of one share of our common stock for every two shares of LRAD Corporation common stock, we will have approximately 15,305,728 shares of common stock outstanding immediately following the spin-off. We will have no shares of preferred stock outstanding immediately following the spin-off. The actual number of shares to be distributed will be determined on the record date.

#### Common Stock

Before the spin-off, there were 100 shares of our common stock outstanding, all of which were held of record by LRAD Corporation and will be cancelled after the spin-off. The holders of our common stock are entitled to one vote per share on all matters to be voted upon by our stockholders. Subject to preferences that may be applicable to any future outstanding preferred stock, the holders of our common stock are entitled to receive ratably such dividends, if any, as may be declared from time to time by our Board of Directors out of funds legally available for that purpose. See "*Dividend Policy*" on page 27. In the event of our liquidation, dissolution or winding-up, the holders of our common stock are entitled to share ratably in all assets remaining a fter payment of liabilities, subject to prior distribution rights of preferred stock, if any, then outstanding. The holders of our common stock have no preemptive or conversion rights or other subscription rights. There are no redemption or sinking fund provisions applicable to our common stock.

### Preferred Stock

Our Board of Directors has the authority, without action by our stockholders, to designate and issue preferred stock in one or more series and to designate the rights, preferences and privileges of each series, which may be greater than the rights of our common stock. It is not possible to state the actual effect of the issuance of any shares of our preferred stock upon the rights of holders of our common stock until our Board of Directors determines the specific rights of the holders of our preferred stock. However, the effects might include, among other things:

- restricting dividends on our common stock;
  - diluting the voting power of our common stock;
  - impairing the liquidation rights of our common stock; or
    - delaying or preventing a change in control of our company without further action by our stockholders.

At the closing of the spin-off, no shares of our preferred stock will be outstanding, and we have no present plans to issue any shares of our preferred stock.

### Authorized but Unissued Capital Stock

Nevada law does not require stockholder approval for any issuance of authorized shares. These additional shares may be used for a variety of corporate purposes, including future public offerings, to raise additional capital or to facilitate acquisitions. One of the effects of the existence of unissued and unreserved common stock or preferred stock may be to enable our Board of Directors to issue shares to persons friendly to current management, which issuance could render more difficult or discourage an attempt to obtain control of our company by means of a merger, tender offer, proxy contest or otherwise, and thereby protect the continuity of our management and possibly deprive the stockholders of opportunities to sell their shares of common stock at prices higher than prevailing market prices.

#### Certificate of Incorporation; Bylaws

Our certificate of incorporation and bylaws contain provisions that could make more difficult the acquisition of Parametric Sound by means of a tender offer, a proxy contest or otherwise. These provisions are summarized below.

Undesignated Preferred Stock. The authorization of our undesignated preferred stock makes it possible for our Board of Directors to issue our preferred stock with voting or other rights or preferences that could impede the success of any attempt to change control of us. These and other provisions may have the effect of deferring hostile takeovers or delaying changes of control of our management.

Size of Board and Vacancies. Newly created directorships resulting from any increase in our authorized number of directors or any vacancies on the Board of Directors resulting from death, resignation, disqualification, removal or other causes and any newly created directorships resulting from any increase in the number of directors, shall unless the Board of Directors determines by resolution that any such vacancies or newly created directorships shall be filled by stockholder vote, be filled only by the affirmative vote of a majority of the directors then in office, even though less than a quorum of the Board of Directors.

No Cumulative Voting. Our certificate of incorporation and bylaws do not provide for cumulative voting in the election of directors.

Stockholder Meetings. Our bylaws provide that special meetings of the stockholders may be called only by our chairman, our chief executive officer or at the direction of our Board of Directors.

#### Nevada Laws

The Nevada Business Corporation Law contains a provision governing "*Acquisition of Controlling Interest.*" This law provides generally that any person or entity that acquires 20% or more of the outstanding voting shares of a publicly-held Nevada corporation in the secondary public or private market may be denied voting rights with respect to the acquired shares, unless a majority of the disinterested stockholders of the corporation elects to restore such voting rights in whole or in part. The control share acquisition act provides that a person or entity acquires "*control shares*" whenever it acquires shares that, but for the operation of the c ontrol share acquisition act, would bring its voting power within any of the following three ranges: (1) 20 to 33 1/3%, (2) 33 1/3 to 50%, or (3) more than 50%. A "*control share acquisition*" is generally defined as the direct or indirect acquisition of either ownership or voting power associated with issued and outstanding control shares. The stockholders or Board of Directors of a corporation may elect to exempt the stock of the corporation from the provisions of the control share acquisition act. The control share acquisition act through adoption of a provision to that effect in the articles of incorporation or bylaws of the corporation. Our articles of incorporation and bylaws do not exempt our common stock from the control share acquisition act. The control share acquisition act is applicable only to shares of "*Issuing Corporations*" as defined by the act. An Issuing Corporation is a Nevada corporation, which; (1) has 200 or more stockholders, with at least 100 of such stockholders being both stockholders of record and residents of Nevada; and (2) does business in Nevada directly or through an affiliated corporation.

At this time, we do not believe we have 100 stockholders of record resident of Nevada. Therefore, the provisions of the control share acquisition act do not apply to acquisitions of our shares and will not until such time as these requirements have been met. At such time as they may apply to us, the provisions of the control share acquisition act may discourage companies or persons interested in acquiring a significant interest in or control of our company, regardless of whether such acquisition may be in the interest of our stockholders.

The Nevada "Combination with Interested Stockholders Statute" may also have an effect of delaying or making it more difficult to effect a change in control of our company. This statute prevents an "interested stockholder" and a resident domestic Nevada corporation from entering into a "combination," unless certain conditions are met. The statute defines "combination" to include any merger or consolidation with an "interested stockholder," or any sale, lease, exchange, mortgage, pl edge, transfer or other disposition, in one transaction or a series of transactions with an "interested stockholder" having; (1) an aggregate market value equal to 5 percent or more of the aggregate market value of the assets of the corporation; (2) an aggregate market value equal to 5 percent or more of the aggregate market value of all outstanding shares of the corporation; or (3) representing 10 percent or more of the earning power or net income of the corporation. An "interested stockholder" means the beneficial owner of 10 percent or more of the voting shares of a resident domestic corporation, or a matifiliate or associate thereof. A corporation affected by the statute may not engage in a "combination" within three years after the interested stockholder acquires its shares unless the combination or p urchase is approved by the Board of Directors before the interested stockholder acquired such shares. If approval is not obtained, then after the expiration of the three-year period, the business combination may be consummated with the approval of the Board of Directors or a majority of the voting power held by disinterested stockholder within the three years immediately preceding the date of the announcement of the combination or in the transaction in which he became an interested stockholder, whichever is higher; (2) the market value per common share on the date of announcement of the combination or in the transaction in which he became an interested stockholder, whichever is higher; (2) the market value per common share on the da

#### LIMITATION OF LIABILITY AND INDEMNIFICATION OF DIRECTORS AND OFFICERS

Our officers and directors are indemnified as to personal liability as provided by the Nevada Revised Statutes ("NRS"), our articles of incorporation and our bylaws. Section 78.7502 of the NRS provides that a corporation may eliminate personal liability of an officer or director to the corporation or its stockholders for breach of fiduciary duty as an officer or director provided that such indemnification is limited if such party acted in good faith and in a manner which he reasonably believed to be in or not opposed to the best interest of the corporation.

Our articles of incorporation and our bylaws state that we shall indemnify our directors and officers to the fullest extent not prohibited by the NRS. Except under limited circumstances, our bylaws provide that we will advance to any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative, by reason of the fact that he is or was a director or officer, prior to the final disposition of the proceeding, promptly following request therefore, all expenses incurred by any director or officer in connection with such proceeding upon receipt of an undertaking by or on behalf of such person to repay said mounts if it should be determined ultimately that such person is not entitled to be indemnified. In addition, to the fullest extent permitted by the NRS, we may purchase insurance on behalf of any person required or permitted to be indemnified pursuant to our bylaws.

Insofar as indemnification for liabilities arising under the Securities Act may be permitted to directors, officers and controlling persons of a small business issuer pursuant to the foregoing provisions, or otherwise, we have been advised that in the opinion of the SEC such indemnification is against public policy as expressed in the Securities Act and is, therefore, unenforceable.

In the event that a claim for indemnification against these types of liabilities, other than the payment by us of expenses incurred or paid by a director, officer or controlling person in the successful defense of any action, suitor proceeding, is asserted by a director, officer or controlling person in connection with the securities being registered, we will submit the question of whether indemnification by us is against public policy to an appropriate court and will be governed by the final adjudication of the case.

### DESCRIPTION OF INDEBTEDNESS

We expect to fund a minimum of \$350,000 and a maximum of \$750,000 of subordinated note financing with accompanying stock purchase warrants at the distribution date. We have received nonbinding, verbal indications of interest from seven individual accredited investors to provide an aggregate of \$475,000 in cash financing and from Syzygy to convert \$200,000 of the amounts owed to Syzygy. We have received a written commitment from Mr. Norris that at least \$350,000 will be funded at the distribution date through conversion of amounts owed to Syzygy and additional cash on the same terms as other investors. The proceeds of this financing are intended to fund the spin-off and related costs described above and provide initial working capital to start the process of bringing our new product line to production. Conversion or repayment of amounts owed to Syzygy will not represent cash available for working capital. After payment of incurred costs, we expect to have approximately \$35,000 for working capital at the distribution date assuming only the minimum financing of \$350,000.

Our indebtedness is expected to consist of 8% subordinated notes due on or before September 27, 2011, one year after the distribution date. We are obligated to issue warrants to purchase between 700,000 and 1,500,000 shares of our common stock in connection with this financing. See "*Capitalization and Financing*" on page 27.

We expect that we will need to refinance this debt and/or arrange additional debt or equity or equity-based financing in the first 12 months after the distribution date. Management has not yet determined the amounts required in future financings, which will depend in part on the time required to bring the new product line to production and customer acceptance of the new product. There is no assurance we will be successful in arranging financing for our company or the terms of any such financing. Any future financing could be dilutive to existing Parametric Sound stockholders. See "*Risk Factors*" beginning on page 11.

## WHERE YOU CAN FIND MORE INFORMATION

We have filed with the SEC a registration statement on Form 10 with respect to the shares of our common stock to be received by the stockholders of LRAD Corporation in the spin-off. This information statement does not contain all of the information set forth in the Form 10 registration statement and the exhibits to the Form 10 registration statement. For further information with respect to Parametric Sound and the shares of our common stock, reference is hereby made to the Form 10 registration statement, including its exhibits. Statements made in this information statement relating to the contents of any contract, agreement or other documents are not necessarily complete and you should refer to the exhibits attached to the registration statement for copies of the actual contract, agreement or other document, with each such statement being qualified in all respects by reference to the document to which it refers. You may review a copy of the Form 10 registration statement, including its exhibits, at the SEC's public reference room, located at 100 F Street, N.E., Washington, D.C. 20549. You may obtain copies of all or any part of these materials from the SEC upon the payment of certain fees prescribed by the SEC. You may obtain further information on the operation of the public reference room by calling the SEC at 1-800-SEC-0330. In addition, copies of the Form 10 registration statement and related documents may be obtained through the SEC Internet address at http://www.sec.gov.

As a result of the spin-off, we will become subject to the information and reporting requirements of the Securities Exchange Act of 1934 and, in accordance with the Exchange Act, will file reports, proxy statements and other information may be inspected and copied at the public reference facilities of the SEC listed above. You also will be able to obtain copies of this material from the public reference facilities of the SEC as described above, or inspect them without charge at the SEC's web site.

In addition, we intend to furnish holders of our common stock with annual reports containing consolidated financial statements audited by an independent accounting firm.

## Parametric Sound Corporation (A Component of LRAD Corporation) Index to Historical Financial Statements

Audited Financial Statements:	
Report of Independent Registered Public Accounting Firm	F-2
Balance Sheets as of September 30, 2009 and 2008	F-3
Statements of Operations for the Years Ended September 30, 2009 and 2008	F-4
Statements of Changes in LRAD Corporation Net Investment for the Years Ended September 30, 2009 and 2008	F-5
Statements of Cash Flows for the Years Ended September 30, 2009 and 2008	F-6
Notes to Financial Statements	F-7 – F-19
Unaudited Interim Financial Statements:	
Preface	F-20
Balance Sheets as of June 30, 2010 and September 30, 2009	F-21
Statements of Operations for the Nine Months Ended June 30, 2010 and 2009	F-22
Statements of Cash Flows for the Nine Months Ended June 30, 2010 and 2009	F-23
Notes to Unaudited Interim Financial Statements	F-24 – F-29

### To the Board of Directors and Stockholders of LRAD Corporation:

We have audited the accompanying balance sheets of Parametric Sound Corporation (a component of LRAD Corporation) as of September 30, 2009 and 2008, and the related statements of operations, changes in LRAD Corporation net investment and cash flows for each of the years then ended. These financial statements are the responsibility of the management of LRAD Corporation. Our responsibility is to express an opinion on the financial statements based on our audits.

We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. We were not engaged to perform an audit of Parametric Sound Corporation's internal control over financial reporting. Our audit included consideration of internal control over financial reporting as a basis for designing audit procedures that were appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Parametric Sound Corporation's internal control over financial reporting. Accordingly, we express no such opinion. An audit also includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the financial statements. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of Parametric Sound Corporation (a component of LRAD Corporation) as of September 30, 2009 and 2008, and the results of its operations and its cash flows for the years then ended, in conformity with accounting principles generally accepted in the United States of America.

The accompanying financial statements have been prepared assuming that Parametric Sound Corporation (a component of LRAD Corporation) will continue as a going concern. As discussed in Note 1 to the financial statements, Parametric Sound Corporation has incurred significant recurring net losses through September 30, 2009 and 2008 and will require new financing. These factors raise substantial doubt about Parametric Sound Corporation's ability to continue as a going concern. Management's plans as to these matters are described in Note 1. The accompanying financial statements do not include any adjustments that might result from the outcome of this uncertainty.

## /S/ SQUAR, MILNER, PETERSON, MIRANDA & WILLIAMSON, LLP

San Diego, California June 23, 2010

# Parametric Sound Corporation (A Component of LRAD Corporation) Balance Sheets

	September 30,			
		2009		2008
ASSETS				
Current assets:				
Accounts receivable, less allowance of \$222,864 and				
\$225,403 for doubtful accounts	\$	29,910	\$	69,677
Inventories, net		422,457		954,792
Total current assets		452,367		1,024,469
Equipment and tooling, net		-		1,134
Patents, net		553,645		637,856
Total assets	\$	1,006,012	\$	1,663,459
LIABILITIES AND LRAD CORPORATION NET INVESTMENT				
Current liabilities:				
Accounts payable	\$	4,662	\$	9,888
Accrued liabilities		78,146		65,378
Total current liabilities		82,808		75,266
Commitments and contingencies				
LRAD Corporation, net investment		923,204		1,588,193
Total liabilities and LRAD Corporation net investment	\$	1,006,012	\$	1,663,459

See accompanying notes to financial statements

# Parametric Sound Corporation (A Component of LRAD Corporation) Statements of Operations

	Years Ended September 30,			er 30,
		2009		2008
Revenues:				
Product sales	\$	586,693	\$	713,294
Other revenue	*	20,413	+	30,050
Total revenues		607,106		743,344
Cost of revenues		638,763		811,845
Gross loss		(31,657)		(68,501)
Operating expenses:				
Selling, general and administrative		839,971		1,340,649
Research and development		114,798		740,582
Total operating expenses		954,769		2,081,231
Loss from operations		(986,426)		(2,149,732)
Net loss	¢	(986,426)	\$	(2,149,732)
	φ	(980,420)	¢	(2,145,752)

See accompanying notes to financial statements

# Parametric Sound Corporation (A Component of LRAD Corporation) Statements of Changes in LRAD Corporation Net Investment

Balance at September 30, 2007	\$ 2,362,545
Net loss for the year	(2,149,732)
Net transfers from parent	 1,375,380
Balance at September 30, 2008	\$ 1,588,193
Net loss for the year	(986,426)
Net transfers from parent	 321,437
Balance at September 30, 2009	\$ 923,204

See accompanying notes to financial statements

# Parametric Sound Corporation (A Component of LRAD Corporation) Statements of Cash Flows

	Years Ended Septe		er 30,
	2009		2008
Increase (Decrease) in Cash and Cash Equivalents:	 	_	
Operating Activities:			
Net loss	\$ (986,426)	\$	(2,149,732)
Adjustments to reconcile net loss to net cash provided by (used in) operating activities:			(, , , ,
Depreciation and amortization	69,620		113,215
Provision for doubtful accounts receivable	(2,538)		704
Warranty provision	(5,437)		(23,904)
Inventory obsolescence	69,079		106,548
Share-based compensation	447,893		797,805
Loss on impairment of patents	46,312		202,170
Changes in assets and liabilities:			
Accounts receivable	42,305		42,612
Inventories	463,256		526,679
Accounts payable	(5,226)		(85,815)
Warranty settlements	(929)		(5,598)
Accrued liability	19,134		14,124
Net cash provided by (used in) operating activities	157,043		(461,192)
Investing Activities:			
Patent costs paid	(30,587)		(116,384)
Net cash used in investing activities	 (30,587)		(116,384)
Financing Activities:			
Net change in investment from LRAD Corporation	(126,456)		577,576
Net cash provided by (used in) financing activities	(126,456)		577,576
Net increase (decrease) in cash and cash equivalents	-		-
Cash and cash equivalents, beginning of year	-		-
Cash and cash equivalents, end of year	\$ -	\$	

See accompanying notes to financial statements

## NOTE 1-DESCRIPTION OF BUSINESS AND SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

In April 2010, the board of directors of LRAD Corporation ("LRAD") approved a plan to separate its HyperSonic Sound ("HSS") product line into a new independent, stand-alone company. In a special meeting of stockholders held June 2, 2010, the proposal to separate the HSS business from LRAD was approved, and on June 2, 2010, LRAD created a new wholly owned subsidiary, hereinafter referred to as Parametric Sound Corporation ("Parametric Sound" or the "Company"), into which the HSS business and substantially all of the assets of the business and associated intellectual property rights were contributed. The spin-off will separate the Parametric Sound subsidiary from LRAD Corporation. Consummation of the spin-off is subject to certain conditions, including final approval by the LRAD board of directors, and the effectiveness of the registration statement filed with the Securities and Exchange Commission. Following the spin-off, each holder of LRAD Corporation common stock will receive a dividend of one share of Parametric Sound common stock for every two shares of LRAD Corporation common stock held on the record date. LRAD Corporation will continue to operate its existing LRAD and SoundSaber businesses, excluding the HSS business to be held by Parametric Sound.

The HSS product utilizes a proprietary parametric speaker technology that generates sound along an air column using ultrasonic frequencies above the normal range of hearing. HSS products are compatible with standard media players and the highly directional sound beam localizes or shines sound where intended by users. The Company's product offering consists primarily of the HSS 450 sound system with principal markets being North America, Europe and Asia.

## Basis of Accounting

These financial statements include the assets, liabilities and results of operations of the component of LRAD Corporation that constitutes the HSS business to be separated. These financial statements are prepared as if Parametric Sound existed and owned this business in all periods presented, and have been prepared using LRAD Corporation's historical carrying value of assets and liabilities. Historically, the HSS business in LRAD Corporation operated as a product line and not a separate segment and not as an independent stand-alone business. For each of the periods presented, Parametric Sound was fully integrated with LRAD Corporation, including production, product development, accounting, finance, treasury, payroll, legal services and investor relations.

The accompanying financial statements include all revenues and cost of revenues directly attributable to the HSS business. However, in circumstances where costs are shared, management used certain estimates to allocate expenses incurred by LRAD Corporation on behalf of the business included in the financial statements. The most significant expenses are as follows:

- Direct costs of HSS research and development, sales and administrative personnel are included along with associated stock-based compensation and allocated employee benefits on a specific identification basis.
- HSS patent amortization costs, including impairments, are included in research and development costs.
- Certain shared administrative salaries and costs including accounting, payroll, human resources, information technology and other services along with legal, auditing and other administrative costs were allocated based on estimates of the proportionate cost incurred by LRAD Corporation related to Parametric Sound.
- · Other costs including facility and occupancy costs were allocated proportionately based on revenues.

Management believes that the assumptions and methods of allocation used underlying the financial statements are reasonable in all material respects. However, the costs as allocated to the Company are not necessarily indicative of the costs that would have been incurred if the Company operated as a stand-alone entity. Therefore, the financial statements included herein may not necessarily be indicative of the financial position, results of operations, changes in stockholders' equity and cash flows of the Company to be expected in the future or what they would have been had the Company been a separate stand-alone entity during the periods presented.

All cash balances are retained and considered those of LRAD Corporation consistent with the planned spin-off.

#### Going Concern

The financial statements have been prepared on a going concern basis which contemplates the realization of assets and the satisfaction of liabilities in the normal course of business for the foreseeable future. Parametric Sound had substantial net losses of \$986,426 and \$2,149,732 for the years ended September 30, 2009 and 2008, respectively and the Company had no sources of financing except for that provided by LRAD Corporation. At separation, LRAD Corporation is retaining all cash and accounts receivable and the Company will be reliant on obtaining new debt or equity financing sufficient to sustain operations until profitability can be achieved. These factors raise substantial doubt about the Company's ability to continue as a going concern.

The Company's plans for raising initial required financing include the issuance of subordinated promissory notes in the amount of \$350,000 - \$750,000. Of this amount, the Company's Chairman and Chief Executive Officer has agreed to, at a minimum guarantee that at least \$350,000 is funded through the conversion of amounts owed by the Company to Syzygy (see Note 10) and additional cash on the same terms as other investors in the subordinated promissory note financing. There is no assurance that Parametric Sound will be successful in raising sufficient funds to sustain its operations for twelve months or beyond.

The continuation of the Company as a going concern is dependent on its ability to develop revenues and to obtain financing from outside sources. Should the Company be unable to obtain required financing, it may have to curtail operations, which may have a material adverse effect on its financial position and results of operations. The accompanying financial statements do not include any adjustments that would be necessary should the Company be unable to continue as a going concern and, therefore, be required to liquidate its assets and discharge its liabilities in other than the normal course of business and at amounts different from those reflected in the accompanying financial statements.

#### Use of Estimates

The preparation of financial statements in conformity with accounting principles generally accepted in the United States of America requires management to make estimates and assumptions (e.g., reserves for accounts receivable and inventory, valuation of patents and warranty reserves) that affect the reported amounts of assets and liabilities, and disclosure of contingent assets and liabilities at the date of the financial statements and affect the reported amounts of revenues and expenses during the reporting period. Actual results could materially differ from those estimates.

#### Fair Value of Financial Instruments

The carrying amounts of accounts receivables, accounts payable and accrued liabilities approximate fair values due to the short maturity of these instruments.

#### Accounts Receivable and Allowance for Doubtful Accounts

The Company carries its accounts receivable at their face amounts, less an allowance for doubtful accounts. On a periodic basis, the Company evaluates its accounts receivable and establishes an allowance for doubtful accounts for estimated losses considering the following factors when determining if collection of a receivable is reasonably assured: customer credit-worthiness, past transaction history with the customer, current economic industry trends and changes in customer payment terms. If the Company has no previous experience with the customer, the Company may obtain reports from various credit organizations to ensure that the customer has a history of paying its creditors. The Company may also request financial information to ensure that the customer has the means of making payment. If these factors do not indica te collection is reasonably assured, revenue is deferred until collection becomes reasonably assured, which is generally upon receipt of cash. If the financial condition of the Company's customers were to deteriorate, adversely affecting their ability to make payments, additional allowances would be required. The Company determines allowances on a case-by-case basis.

## Contract Manufacturers

The Company employs contract manufacturers for production of certain components and sub-assemblies. The Company may provide parts and components to such parties from time to time but recognizes no revenue or markup on such transactions. The Company performs assembly of products in-house using components and sub-assemblies from a variety of contract manufacturers and suppliers.

## Inventories

Inventories are valued at the lower of cost or net realizable value. Cost is determined on a first-in, first-out basis. Inventory is comprised of raw materials, assemblies and finished products intended for sale to customers. The Company periodically makes judgments and estimates regarding the future utility and carrying value of inventory. The carrying value of inventory is periodically reviewed and impairments, if any, are recognized when the expected future benefit is less than carrying value. The Company has inventory reserves for estimated obsolescence or unmarketable inventory which is equal to the difference between the cost of inventory and the estimated market value based upon assumptions about future demand and market conditions. The Company increa sed its inventory reserve by \$69,079 and \$106,548 during the years ended September 30, 2009 and 2008, respectively, for excess HSS components.

## Equipment and Depreciation

Equipment is stated at cost. Depreciation on machinery and equipment is computed over the estimated useful lives of three to five years using the straight-line method. Upon retirement or disposition of equipment, the related cost and accumulated depreciation or amortization is removed and a gain or loss is recorded.

### **Intangibles**

Patents and trademarks are carried at cost and are amortized over their estimated useful lives, which have been estimated to be 15 years. The carrying value of intangibles is periodically reviewed and impairments, if any, are recognized when the expected future benefit to be derived from an individual intangible asset is less than its carrying value. The Company wrote off \$46,312 and \$202,170, of previously capitalized patent costs during the years ended September 30, 2009 and 2008, respectively.

#### Revenue Recognition

The Company derives its revenue primarily from product sales. Product sales are recognized in the periods that products are shipped to customers (FOB shipping point) or when product is received by the customer (FOB destination), when the fee is fixed and determinable, when collection of resulting receivables is probable and there are no remaining obligations on the part of the Company.

## Shipping and Handling Costs

Shipping and handling costs are included in cost of revenues. The amount of shipping and handling costs invoiced to customers is included in revenue. Estimated shipping and handling costs were \$25,540 and \$35,706 for the fiscal years ended September 30, 2009 and 2008, respectively.

## Research and Development Costs

Research and development costs are expensed as incurred.

## Warranty Reserves

The Company warrants its products to be free from defects in materials and workmanship for a period of one year from the date of purchase. The warranty is generally a limited warranty. The Company currently provides direct warranty service.

The Company establishes a warranty reserve based on anticipated warranty claims at the time revenue from product sales is recognized. Factors affecting warranty reserve levels include the number of units sold and anticipated cost of warranty repairs and anticipated rates of warranty claims. The Company evaluates the adequacy of the provision for warranty costs each reporting period. See Note 6 for additional information regarding warranties.



## Income Taxes

The Company accounts for its income taxes under the asset and liability method. Under this method, deferred tax assets and liabilities are determined based on temporary differences between financial statement and tax basis of assets and liabilities and net operating loss and credit carry-forwards using enacted tax rates in effect for the year in which the differences are expected to reverse. Valuation allowances are established when it is more likely than not that some portion of the deferred tax assets will not be realized.

The Company was not a separate legal entity and, therefore, was not subject to taxation by federal and state authorities for the periods presented; however, the income tax benefit would have been nil in the accompanying financial statements if the Company was filing a separate return. The historical net operating loss carryforwards and tax credits generated by Parametric Sound will remain with LRAD Corporation subsequent to the separation.

## Earnings Per Share

Common stock and stock equivalents represent ownership in the parent LRAD Corporation. As Parametric Sound has no common stock or stock equivalents issued or outstanding, there is no earnings per share calculation included in the Parametric Sound financial statements.

### Comprehensive Loss

Comprehensive loss consists of net loss and other gains and losses affecting stockholders' equity that under US generally accepted accounting principles are excluded from reported net loss. There were no differences between net loss and comprehensive loss for any of the periods presented.

#### Impairment of Long-Lived Assets

Long-lived assets and identifiable intangibles held for use are reviewed for impairment whenever events or changes in circumstances indicate that the carrying amount may not be recoverable. If the sum of undiscounted expected future cash flows is less than the carrying amount of the asset or if changes in facts and circumstances indicate, an impairment loss is recognized and measured using the asset's fair value.

## Share-Based Compensation

The Company recognized share-based compensation expense related to stock options issued to employees over the vesting term of the stock-based instrument based on the grant date fair value.

#### Recent Accounting Pronouncements

In June 2009, through the issuance of Financial Accounting Standards Board (FASB) Statement No. 168, "The FASB Accounting Standards Codification and the Hierarchy of Generally Accepted Accounting Principles" (the ASC or Codification), there became a single official source of authoritative generally accepted accounting principles in the United States (US GAAP), superseding existing FASB, American Institute of Certified Public Accountants, Emerging Issues Task Force (EITF), and related literature. There is now only one level of authoritative US GAAP. All other literature will be considered non-authoritative. The Codification does not change US GAAP; instead, it introduces a new structure that is organized in an easily accessible, user-friendly online research system. The Codification is effective for r interim and annual periods ending on or after September 15, 2009. The Company applied the Codification beginning in the fourth quarter of fiscal 2009. The adoption of the Codification did not have an effect on the Company's consolidated financial statements. This new standard is now codified under ASC 105, "Generally Accepted Accounting Principles." In circumstances where revisions or updates to existing standards are required, the FASB will issue an Accounting Standards Update (ASU) on the ASC topic. All of the Company's financial statement disclosures have been updated in accordance with the Codification. SEC pronouncements are still authoritative for public companies.

Effective April 1, 2009, the Company adopted three accounting standard updates that were intended to provide additional application guidance and enhanced disclosures regarding fair value measurements and impairments of securities established in ASC 820, "Fair Value Measurements and Disclosures" (formerly SFAS 157). They also provide additional guidelines for estimating fair value in accordance with fair value accounting. The first update, as codified in ASC 820-10-65, "Fair Value Measurements and Disclosures" (formerly SFAS 157-2), provides additional guidelines for estimating fair value in accordance with fair value accounting. The second accounting update, as codified in ASC 320-10-65, "Debt and Equity Securities" (formerly FSP FAS 115-2 and FAS 124-2), changes accounting requirements for other-than-temporary-impairment for debt securities by replacing the current requirement that a holder have the positive intent and ability to hold an impaired security to recovery in order to conclude an impairment was temporary with a requirement that an entity conclude it does not intend to sell an impaired security and it will not be required to sell the security offore the recovery of its amortized cost basis. The third accounting update, as codified in ASC 825-10-65, "Financial Instruments" (formerly FSP FAS No. 107-1 and APB 28-1), increases the frequency of fair value disclosures. These updates were effective for fiscal years and interim periods ended after June 15, 2009. The adoption of these accounting updates did not have any impact on the Company's consolidated financial statements.

In February 2008, the FASB issued guidance, as codified in ASC 820-10, "Fair Value Measurements and Disclosures", that delayed the effective date of fair value measurements accounting for all non-financial assets and non-financial liabilities, except for items that are recognized or disclosed at fair value in the financial statements on a recurring basis (at least annually), until the beginning of the first quarter of fiscal 2010. These include goodwill and other non-amortizable intangible assets. The Company adopted this guidance effective October 1, 2008. The adoption of this update to non-financial assets and liabilities did not have any impact on the Company's consolidated financial statements.

In April 2008, the Financial Accounting Standards Board ("FASB") issued authoritative guidance, as codified in ASC 350-30, "Goodwill and other; General Intangibles Other Than Goodwill" (formerly FSP SFAS 142-3). This guidance provides factors that should be considered in developing the renewal or extension assumptions used to determine the useful life of a recognized intangible asset. This guidance also requires expanded disclosure related to the determination of intangible asset useful lives. It is effective for fiscal years beginning after December 15, 2008. Earlier adoption was not permitted. The adoption of this guidance on October 1, 2009 is not expected to have any impact on the Company's financial statements.

In June 2008, the FASB issued authoritative guidance, as codified in ASC 815-40, "Derivatives and Hedging; Contracts in Entity's Own Equity" (formerly EITF Issue No. 07-5). This guidance provides a two-step model to be applied in determining whether a financial instrument or an embedded feature is indexed to an issuer's own stock and thus able to qualify for the scope exception codified in ASC 815-10 (formerly SFAS No. 133 paragraph 11(a)). The Company adopted this guidance effective October 1, 2009 and it did not have any impact on Parametric Sound's financial statements.

In May 2009, the FASB issued ASC 855-10, "Subsequent Events" (formerly SFAS No. 165). ASC 855-10 establishes general standards of accounting for and disclosure of events that occur after the balance sheet date but before financial statements are issued or are available to be issued. ASC 855-10 was effective for fiscal years and interim periods ending after June 15, 2009. The implementation of this standard did not have a material impact on the Company's consolidated financial statements. (See Note 11).

Effective October 1, 2009, the Company is required to adopt new standards regarding business combinations issued by the FASB in December 2007, as codified under ASC 805, "Business Combinations" (formerly FAS 141R) that require the acquisition method to be applied to all transactions and other events in which an entity obtains control over one or more other businesses, requires the acquirer to recognize the fair value of all assets and liabilities acquired, even if less than one hundred percent ownership is acquired, and establishes the acquisition date fair value as measurement date for all assets and liabilities assumed. For the Company, this accounting update is effective on a prospective basis for all business combinations for which the acquisition date is on or after October 1, 2009. Since the Company is not contemplating any business combinations it does not presently expect any impact of adoption on its financial statements.

In December 2007, the FASB issued authoritative guidance, as codified in ASC 810-10, "Consolidation" (formerly SFAS 160) that establishes accounting and reporting standards for the noncontrolling interest in a subsidiary and for the deconsolidation of a subsidiary. This guidance clarifies that a non-controlling interest in a subsidiary is an ownership interest in the consolidated financial statements. This guidance is effective for the Company's fiscal years beginning after December 15, 2008. The provisions are applied prospectively upon adoption except for the presentation and disclosure requirements that are applied retrospectively. The Company has no non-controlling interests and accordingly the adoption of this guidance effective October 1, 2009 is not expected to have a material impact on the Company's financial statements.

In September 2009, the FASB issued Accounting Standards Update (ASU) No. 2009-13, "Revenue Recognition (Topic 605)—Multiple Deliverable Arrangements" (formerly EITF Issue 08-1). This guidance updates the existing multiple-element revenue arrangements guidance currently included under ASC 605-25, which originated primarily from the guidance in EITF Issue No. 00-21, "Revenue Arrangements with Multiple Deliverables." The revised guidance primarily provides two significant changes: 1) eliminates the need for objective and reliable evidence of the fair value for the undelivered element in order for a delivered item to be treated as a separate unit of accounting, and 2) eliminates the residual method to allocate the arrangement consideration. In addition, the guidance also expands the discl osure requirements for revenue recognition. ASU 2009-13 will be effective for the first annual reporting period beginning on or after June 15, 2010, with early adoption permitted provided that the revised guidance is retroactively applied to the beginning of the year of adoption. The Company is currently assessing the future impact of this new accounting update to its financial statements.

In October 2009, the FASB issued ASU 2009-14, "Software (Topic 985)-Certain Revenue Arrangements that Include Software Elements" (formerly EITF Issue 09-3) and changes the accounting model for revenue arrangements that include both tangible products and software elements. Under this guidance, tangible products containing software components and nonsoftware components that function together to deliver the tangible product's essential functionality are excluded from the software revenue guidance in Subtopic 985-605, "Software-Revenue Recognition". In addition, hardware components of a tangible product containing software components are always excluded from the software revenue guidance. The guidance in this ASU is effective prospectively for revenue arrangements entered into or material ly modified in fiscal years beginning on or after June 15, 2010. Early adoption is permitted. The adoption of this standard is not expected to have an impact on the Company's financial position and results of operations since this accounting standard update provides only implementation and disclosure amendments.

In January 2010, the FASB issued ASU No. 2010-06, "Fair Value Measurements and Disclosures (Topic 820): Improving Disclosures about Fair Value Measurements." This ASU requires certain new disclosures and clarifies certain existing disclosure requirements about fair value measurement described in Subtopic 820-10, for the purpose of improving these disclosures and increasing the transparency in financial reporting. This standard is effective for reporting periods beginning after December 15, 2009, except for disclosure requirements relating to purchases, sales, issuances and settlements in the roll forward of activity in level 3 fair value measurements which are not effective until reporting periods beginning after December 15, 2010. The adoption of this standard is not expected to have a material impact on the Company's financial statements.

In February 2010, the FASB issued ASU No. 2010-09, "Subsequent Events (Topic 855): Amendments to Certain Recognition and Disclosure Requirements." The amendments in the ASU remove the requirement for an SEC filer to disclose a date through which subsequent events have been evaluated in both issued and revised financial statements. The amendments in the ASU were effective upon issuance on February 24, 2010. The Company's adoption of this accounting standards update did not impact the Company's financial statements.

In April 2010, the FASB issued ASU No. 2010-17, "Revenue Recognition-Milestone Method (Topic 605): Milestone Method of Revenue Recognition." This ASU provides guidance on defining a milestone and determining when it may be appropriate to apply the milestone method of revenue recognition for research or development transactions. This ASU is effective on a prospective basis for milestones achieved in fiscal years, and interim periods within those years, beginning on or after June 15, 2010. The Company is currently assessing the future impact of this new accounting update to its financial statements.

## 2. INVENTORIES, NET

Inventories are stated at the lower of cost, which approximates actual costs on a first in, first out cost basis, or market.

Inventories consisted of the following:

Years Ended September 30,	2009		2008
Finished goods	\$ 102,10	4	\$ 490,420
Work in process		-	9,385
Raw materials	1,760,65	6	1,826,211
	1,862,76	0	 2,326,016
Reserve for obsolescence	(1,440,30	3)	(1,371,224)
Total, net	\$ 422,45	7	\$ 954,792

The Company relies on one supplier for film for its HSS product. The Company's ability to manufacture its HSS product could be adversely affected if it were to lose a sole source supplier and was unable to find an alternative supplier.

The reserve for obsolescence increased by \$69,079 and \$106,548 in the years ended September 30, 2009 and 2008, respectively, due to excess HSS components.

# 3. EQUIPMENT AND TOOLING

Equipment consists of the following:

Years Ended September 30,	2009	2008
Equipment and tooling	\$ 179,835	\$ 179,835
Office equipment	 1,401	 1,401
	 181,236	181,236
Accumulated depreciation	 (181,236)	 180,102
Equipment and tooling, net	\$ _	\$ 1,134

Depreciation expense was \$1,134 and \$35,551 for the years ended September 30, 2009 and 2008, respectively.

## 4. PATENTS

Patents consist of the following:

Years Ended September 30,	2009			2008		
Cost	\$	1,015,572		\$	1,044,541	
Accumulated amortization		(461,927	)		(406,685	)
Patents, net	\$	553,645		\$	637,856	

Aggregate amortization expense for the Company's patents was \$68,486 and \$77,664 during the years ended September 30, 2009 and 2008, respectively. In addition to amortization, the Company wrote off \$46,312 and \$202,170 of impaired patent costs during the years ended September 30, 2009 and 2008, respectively. Additional patents may be written off in the future if it is deemed that certain patents are no longer consistent with the business strategy after the spin-off.

## Estimated Amortization Expense Years Ended September 30,

Estimated 7 mortal addin Expense Tears Ended September 50,	
2010	\$ 66,114
2011	\$ 65,940
2012	\$ 65,940
2013	\$ 65,940
2014	\$ 65,940
Thereafter	\$ 223,771

## 5. INCOME TAXES

For all historic periods reported in the financial statements, Parametric Sound maintained valuation allowances against its net deferred tax assets, including net operating loss carryforwards, because it was more likely than not that the deferred taxes would not be realized. The net operating losses (NOLs) generated by Parametric Sound are consolidated within the LRAD Corporation tax return and it is anticipated that all NOL carryforwards and research and development credits generated by Parametric Sound will be retained by LRAD Corporation upon the separation of the companies. As the amounts of cumulative expenses in the financial statements not allowed for federal and state income tax purposes were not sufficient enough to result in p ositive taxable income Parametric Sound recorded no income tax expense in 2009 and 2008 from the book losses incurred.

The effective tax rate of Parametric Sound after the separation and in the future could vary from that of LRAD Corporation. Upon separation, Parametric Sound as a new separate legal entity will not benefit from any of the carryforward tax attributes from prior periods including net operating loss carryforwards. Given that the tax rate is zero and that attributes will not carryforward, details of deferred tax assets, valuation allowances and the related rate reconciliation are not presented.

## 6. ACCRUED LIABILITIES

Accrued liabilities consisted of the following:

	Sej	September 30, 2009			nber 30, 008
Payroll and related	\$	48,757		\$	29,623
Warranty reserve		29,389			35,755
Total	\$	78,146		\$	65,378

The Company establishes a warranty reserve based on anticipated warranty claims at the time product revenue is recognized. Factors affecting warranty reserve levels include the number of units sold, anticipated cost of warranty repairs and anticipated rates of warranty claims. The Company evaluates the adequacy of the provision for warranty costs each reporting period.

Details of the estimated warranty liability are as follows:

Years Ended September 30,		2009		2008			
Beginning balance		\$	35,755		\$	65,257	
Warranty provision			(5,437	)		(23,904	)
Warranty settlements			(929	)		(5,598	)
Ending balance	-	\$	29,389		\$	35,755	

At distribution, LRAD Corporation will be responsible for warranty claims related to prior product sales and resulting from sales from the completion of any outstanding orders as of the distribution date.

The Company is obligated to pay a minimum of \$1.00 per unit royalty on one electronic component used in the HSS products. Royalty expenses incurred were \$1,254 and \$1,408 in the fiscal years ended September 30, 2009 and 2008, respectively. If Parametric Sound uses this component in its products, it will be required to pay this royalty.

## 7. LRAD CORPORATION NET INVESTMENT

The financial statements of Parametric Sound represent the HSS product business of LRAD Corporation. Because neither a separate subsidiary nor a direct ownership relationship existed, LRAD Corporation's investment in Parametric Sound is shown as parent company equity in lieu of a stockholders' equity classification in the financial statements.

Net transactions include intercompany transactions with LRAD Corporation. These intercompany transactions represent the cash contributions or distributions related to the operating transactions between LRAD Corporation and Parametric Sound. No cash or accounts receivable will be transferred at the distribution date and the Company will ultimately be responsible for all spin-off related costs (see Note 10). LRAD Corporation will retain certain inventory sufficient to complete outstanding customer orders and to support outstanding warranty obligations (see Note 10).

## 8. SHARE-BASED COMPENSATION

The Company expects to adopt a stock compensation plan following the effectiveness of the spin-off that will provide grants for incentive stock options or non-qualified stock options to directors, employees and/or consultants.

The following information regarding outstanding stock options and share-based compensation has been derived from equity awards granted under LRAD Corporation equity compensation plans to individuals whose salary and benefits were allocated to the Company in the accompanying financial statements.



The weighted average estimated fair value of employee stock options granted to these individuals during the year ended September 30, 2009 and 2008 was \$0.23 per share and \$0.92 per share, respectively, using the Black-Scholes option-pricing model with the following weighted-average assumptions (annualized percentages):

	Year Endeo	l September 30,
	2009	2008
Volatility	71.0% - 83.0%	71.0%
Risk-free interest rate	1.30% - 2.46%	2.79% - 3.49%
Forfeiture rate	20.0%	20.0%
Dividend yield	0.0%	0.0%
Expected life in years	3.4 - 4.9	3.4 - 4.9

The dividend yield of zero is based on the fact that LRAD Corporation has never paid cash dividends and has no present intention to pay cash dividends. Expected volatility is based on the historical volatility of LRAD Corporation's common stock over the period commensurate with the expected life of the options. The risk-free interest rate is based on rates published by the Federal Reserve Board. The expected life is based on observed and expected time to post-vesting exercise. The expected forfeiture rate is based on past experience and employee retention data. Forfeitures are estimated at the time of the grant and revised in subsequent periods if actual forfeitures differ from those estimates or if the estimated forfeiture rate is updated. Such amounts are recorded as a cumulative adjustment in the period in which the estimate is changed.

As of September 30, 2009, there was \$4,144 of total unrecognized compensation costs related to outstanding LRAD Corporation stock options granted to personnel whose salary and benefit costs were allocated to the company. This amount is expected to be recognized over a weighted-average period of 0.9 years, unless the employee leaves LRAD Corporation employment prior to that time. To the extent the forfeiture rate is different from what the Company anticipated, stock-based compensation related to these awards will be different from the Company's expectations. There will be no expense incurred by Parametric Sound Corporation after the distribution date related to these stock options.

## Stock Option Summary Information

A summary of activity for LRAD Corporation stock options issued to personnel whose salary and benefit costs were allocated to the Company as of September 30, 2009 and 2008 is presented below:

	Number of Shares	hted Average ercise Price
Fiscal 2008:		
Outstanding October 1, 2007	1,151,090	\$ 4.27
Granted	115,000	\$ 2.05
Canceled/expired	(313,090)	\$ 3.18
Outstanding September 30, 2008	953,000	\$ 4.08
Exercisable at September 30, 2008	690,500	\$ 4.24
Fiscal 2009:		
Outstanding October 1, 2008	953,000	\$ 4.08
Granted	40,000	\$ 0.52
Canceled/expired	(953,000)	\$ 4.00
Exercised	(5,833)	\$ 0.48
Outstanding September 30, 2009	34,167	\$ 2.80
Exercisable September 30, 2009	27,500	\$ 3.24
Weighted average fair value of options granted during the year		\$ 0.23

The aggregate intrinsic value for both options outstanding and options exercisable relating to options issued to personnel whose salary and benefits were allocated to the Company at September 30, 2009 was \$5,375 and \$-0-, respectively. The aggregate intrinsic value represents the difference between the Company's closing stock price on the last day of trading during the year, which was \$1.77 per share, and the exercise price multiplied by the number of applicable options. The total intrinsic value of options exercised during 2009 and 2008 was \$7,758 and \$-0-, respectively.

The following table summarizes information about stock options outstanding at September 30, 2009 relating to options issued to personnel whose salary and benefits were allocated to the Company:

		Weighted Average	Weighted		Weighted
Range of		Remaining	Average		Average
Exercise	Number	Contractual	Exercise	Number	Exercise
Prices	Outstanding	Life	Price	Exercisable	Price
\$0.48-\$1.88	19,167	3.56	\$.58	12,500	\$ 1.88
\$4.37	15,000	2.38	\$ .37	15,000	\$ 4.37
\$0.48-\$4.37	34,167	3.04	\$ 2.80	27,500	\$ 3.24

The Company recorded non-cash share-based compensation expense relating to options issued to personnel whose salary and benefits were allocated to the Company of \$447,893 and \$797,805, respectively, for the fiscal years ended September 30, 2009 and 2008. The amounts of share-based compensation expense are classified in the consolidated statements of operations as follows:

Years Ended September 30,	2009	2008
Cost of revenue	\$ 3,229	\$ 1,513
Selling, general and administrative	444,664	664,321
Research and development	 -	 131,971
Total	\$ 447,893	\$ 797,805

## 9. MAJOR CUSTOMERS AND SUPPLIERS

## Major Customers

For the fiscal year ended September 30, 2009, revenues from three customers accounted for 27%, 15% and 14% of total revenues. No other single customer represented more than 10% of total revenues. For the fiscal year ended September 30, 2008, revenues from three customers accounted for 16%, 13% and 12% of total revenues with no other single customer accounting for more than 10% of total revenues. One large customer, Cardinal Health, Inc., which accounted for 27% of sales in the year ended September 30, 2009, will continue to be supplied by LRAD Corporation through completion of their current project.

## Suppliers

The Company has a number of components and sub-assemblies produced by outside suppliers, some of which are sourced from a single supplier, which can magnify the risk of shortages and decrease the Company's ability to negotiate with suppliers on the basis of price. In particular, the Company depends on its HSS piezo-film supplier to provide expertise and materials used in the Company's proprietary HSS emitters. If supplier shortages occur, or quality problems arise, then production schedules could be significantly delayed or costs significantly increased, which could in turn have a material adverse effect on the Company's financial condition, results of operation and cash flows.

### Segment and Related Information

The Company business consists of only one product. Accordingly, the Company operates in one reportable segment.

The following table summarizes revenues by geographic region. Revenues are attributed to countries based on customer location.

Years Ended September 30,	2009	2008
Revenues		
United States	\$ 266,654	\$ 149,913
Other	340,452	593,431
Total	\$ 607,106	\$ 743,344

# **10. RELATED PARTY TRANSACTIONS**

The financial statements have been prepared on a stand-alone basis and are derived from the consolidated financial statements and accounting records of LRAD Corporation.

## Allocation of Expenses

The financial statements include expense allocations for certain functions provided by LRAD Corporation, including, but not limited to, general corporate expenses related to finance, legal, information technology, human resources, employee benefits and incentives. These expenses have been allocated to the Company on the basis of direct usage when identifiable, with the remainder allocated on the basis of revenue or other measure. During the fiscal years ended September 30, 2009 and 2008, the Company was allocated \$839,971 and \$1,340,649, respectively, of general corporate expenses incurred by LRAD Corporation which is included within selling, general and administrative expenses in the statements of operations. During the fiscal years ended September 30, 2009 and 2008, the Company was allocated \$114,798 and \$740,582, re spectively, of research and development expenses incurred by LRAD Corporation which is included within research and development expenses incurred by LRAD Corporation which is included within research and development expenses incurred by LRAD Corporation which is included within research and development expenses incurred by LRAD Corporation which is included within research and development expenses incurred by LRAD Corporation which is included within research and development expenses in the statements of operations. During the fiscal years ended September 30, 2009 and 2008, the Company was allocated \$73,063 and \$140,022, respectively, of manufacturing overhead expenses for warehousing, materials management and production management, which is included within cost of sales.

The expense allocations have been determined on a basis that both the Company and LRAD Corporation consider to be a reasonable reflection of the utilization of services provided or the benefit received by the Company during the periods presented. The allocations may not, however, reflect the expense the Company would have incurred as an independent, publicly-traded company for the periods presented. Actual costs that may have been incurred if the Company had been a stand-alone company would depend on a number of factors, including the chosen organization structure, what functions are outsourced or performed by employees and strategic decisions made in areas such as information technology and infrastructure.

## Separation Agreement

In connection with the spin-off, the Company and LRAD Corporation expect to enter into a Separation and Distribution Agreement, (the "Separation Agreement") and some other ancillary agreements to complete the separation of the business from LRAD Corporation and to distribute Parametric Sound common stock to LRAD Corporation stockholders. LRAD Corporation shall retain sufficient inventory and supplies to fulfill outstanding orders with current customers existing at the date of distribution (or received as a continuing order thereafter expected to extend not beyond December 31, 2010) and will be responsible for warranty, returns and related liabilities for such customers and sales. LRAD Corporation will not accept any new orders (other than a continuing order) from current customers after the distribution dat e. Unless otherwise agreed by the parties, the Company will not be obligated to fulfill any orders or continuing orders of LRAD Corporation.

#### Previously Licensed Technology

LRAD Corporation was previously obligated to pay Mr. Norris a 2% royalty on net sales from certain of its technologies, including HSS. The royalty obligation discontinued in October 2008 under the terms of Mr. Norris' amended employment agreement dated November 5, 2008. No royalties were paid or recorded under this agreement in the fiscal years ended September 30, 2009 or 2008, as these royalties were immaterial and were waived by Mr. Norris. The amounts of the royalties waived were \$0 and \$14,266 for the years ended September 30, 2009 and 2008, respectively.



## Syzygy Licensing LLC

Syzygy is an entity in which, Elwood G. Norris, a director and greater than 10% stockholder of LRAD Corporation, is majority owner. Mr. Norris is also expected to be the Company's President and Chief Executive Officer at the distribution date.

When the Board of Directors of LRAD Corporation conditionally approved the spin-off in April 2010, Mr. Norris through Syzygy agreed to advance third party fees, costs and expenses in connection with the transaction. Should the transaction not be completed then Syzygy and not LRAD Corporation will be responsible for such costs unless LRAD Corporation unilaterally terminates the spin-off other than for lack of stockholder approval, whereupon LRAD Corporation will reimburse Syzygy. Except as expressly set forth in the Separation Agreement, should the separation be completed then all third party fees, costs and expenses paid or incurred in connection with the transactions by Syzygy or by LRAD Corporation will be repaid by the Company at or after the distribution.

The Company also plans to enter into a license agreement effective upon distribution with Syzygy related to certain patent pending technology and trade secrets for a new method of controlling and processing media input to create parametric sound output for parametric emitter devices such as those employed in the HSS technology. The Company will pay no initial license fee nor issue any equity for this license but is expected to reimburse technology, patent, prototype and testing costs incurred to the distribution and to pay future patent related costs. The license is expected to provide for royalties of not more than 5% of net sales from products employing the technology.

## 11. SUBSEQUENT EVENTS

Subsequent events have been evaluated through June 23, 2010, which was the date the Company's financial statements were issued.

<b>F-</b> 1	19
-------------	----

## PARAMETRIC SOUND CORPORATION UNAUDITED INTERIM FINANCIAL STATEMENTS JUNE 30, 2010 AND 2009

## Preface

The preparation of the interim financial statements requires management to make use of estimates and assumptions that affect the reported amount of assets and liabilities, revenue and expenses and certain financial statement disclosures. Significant estimates in these unaudited interim financial statements include allowances for doubtful accounts receivable, net realizable value of inventories, useful lives of identifiable intangible assets, the evaluation of impairments of identifiable intangible assets, income tax and valuation reserves and the evaluation of warranty reserves. Actual results may differ from these estimates.

## Parametric Sound Corporation (A Component of LRAD Corporation) Balance Sheets

		June 30, 2010 (Unaudited)		2010 Septem		eptember 30, 2009	
ASSETS							
Current assets:							
Accounts receivable, less allowance of \$222,864 each period for doubtful accounts	\$	79,713	\$	29,910			
Inventories, net		110,032		422,457			
Total current assets		189,745		452,367			
Patents, net		473,489		553,645			
Total assets	\$	663,234	\$	1,006,012			
LIABILITIES AND LRAD CORPORATION NET INVESTMENT							
Current liabilities:							
Accounts payable	\$	5,907	\$	4,662			
Accrued liabilities		61,444		78,146			
Total current liabilities		67,351		82,808			
Commitments and contingencies							
LRAD Corporation, net investment		595,883		923,204			
Total liabilities and LRAD Corporation net investment	\$	663,234	\$	1,006,012			

See accompanying notes to financial statements

# Parametric Sound Corporation (A Component of LRAD Corporation) Statements of Operations (Unaudited)

	Nine months ended June 30,		
	 2010		2009
Revenues:			
Product sales	\$ 512,369	\$	375,476
Other revenue	7,709		17,519
Total revenues	520,078		392,995
Cost of revenues	441,403		414,341
Gross profit	 78,675		(21,346)
Operating expenses:			
Selling, general and administrative	215,348		666,117
Research and development	 90,207		95,169
Total operating expenses	305,555		761,286
Loss from operations	 (226,880)		(782,632)
Net loss	\$ (226,880)	\$	(782,632)

See accompanying notes to financial statements

## Parametric Sound Corporation (A Component of LRAD Corporation) Statements of Cash Flows (Unaudited)

	Nine Months Ended June 30,	
	2010	2009
Increase (Decrease) in Cash and Cash Equivalents:		
Operating Activities:		
Net loss	\$ (226,880) \$	(782,632)
Adjustments to reconcile net loss to net cash used in operating activities:		
Depreciation and amortization	49,906	51,794
Provision for doubtful accounts	-	7,263
Warranty provision	16,462	(10,285)
Inventory obsolescence	(26,159)	5,179
Share-based compensation	4,971	419,861
Loss on impairment of patents	40,301	43,737
Changes in assets and liabilities:		
Accounts receivable	(49,803)	(41,607)
Inventories	338,584	343,068
Accounts payable	1,245	(6,191)
Warranty settlements	(9,709)	(363)
Accrued liability	 (23,455)	(21,921)
Net cash provided by operating activities	 115,463	7,903
Investing Activities:		
Patent costs paid	(10,051)	(6,481)
Net cash used in investing activities	(10,051)	(6,481)
Financing Activities:		
Net change in investment from LRAD Corporation	(105,412)	(1,422)
Net cash provided by financing activities	 (105,412)	(1,422)
Net increase (decrease) in cash and cash equivalents		
Cash and cash equivalents, beginning of year	-	-
Cash and cash equivalents, end of year	\$ - \$	-

See accompanying notes to financial statements

## Note 1—Description of Business and Basis of Accounting/Presentation

In April 2010, the board of directors of LRAD Corporation approved a plan to separate its HyperSonic Sound ("HSS") product line into a new independent, stand-alone company. In a special meeting of stockholders held June 2, 2010, the proposal to separate the HSS business from LRAD was approved, and on June 2, 2010, LRAD created a new wholly owned subsidiary, hereinafter referred to as Parametric Sound Corporation ("Parametric Sound" or the "Company"), into which the HSS business and substantially all of the assets of the business and associated intellectual property rights were contributed. The spin-off will separate the Parametric Sound subsidiary from LRAD Corporation. Consummation of the spin-off is subject to certain conditions, including final approval by the LRAD board of directors, and the effectiveness of the registration statement filed with the Securities and Exchange Commission. Following the spin-off, each holder of LRAD Corporation common stock will receive a dividend of one share of Parametric Sound common stock for every two shares of LRAD Corporation common stock held on the record date. LRAD Corporation will continue to operate its existing LRAD and SoundSaber businesses, excluding the HSS business to be held by Parametric Sound.

The HSS product utilizes a proprietary parametric speaker technology that generates sound along an air column using ultrasonic frequencies above the normal range of hearing. HSS products are compatible with standard media players and the highly directional sound beam localizes or shines sound where intended by users. The Company's product offering consists primarily of the HSS 450 sound system with principal markets being North America, Europe and Asia.

## Basis of Accounting

The financial statements include the assets, liabilities and results of operations of the component of LRAD Corporation that constitutes the HSS business to be separated. These financial statements are prepared as if Parametric Sound existed and owned this business in all periods presented and have been prepared using LRAD Corporation's historical cost basis of assets and liabilities. Historically, the HSS business in LRAD Corporation operated as a product line and not a separate segment and not as an independent stand-alone business. For each of the periods presented, Parametric Sound was fully integrated with LRAD Corporation, including production, product development, accounting, finance, treasury, payroll, legal services and investor relations.

The accompanying financial statements include all revenues and cost of revenues directly attributable to the HSS business. However, in circumstances where costs are shared, management used certain estimates to allocate expenses incurred by LRAD Corporation on behalf of the business included in the financial statements. The most significant expenses are as follows:

- Direct costs of HSS research and development, sales and administrative personnel are included along with associated stock-based compensation and allocated employee benefits on a specific identification basis.
- HSS patent amortization costs, including impairments, are included in research and development costs.
- Certain shared administrative salaries and costs including accounting, payroll, human resources, information technology and other services along with legal, auditing and other administrative costs were allocated based on estimates of the proportionate cost incurred by LRAD Corporation related to Parametric Sound.
- · Other costs including facility and occupancy costs were allocated proportionately based on revenues.

Management believes that the assumptions and methods of allocation used underlying the financial statements are reasonable in all material respects. However, the costs as allocated to the Company are not necessarily indicative of the costs that would have been incurred if the Company operated as a stand-alone entity. Therefore the financial statements included herein may not necessarily be indicative of the financial position, results of operations, changes in stockholders' equity and cash flows of the Company to be expected in the future or what they would have been had the Company been a separate stand-alone entity during the periods presented.

All cash balances are retained and considered those of LRAD Corporation consistent with the planned spin-off.

The interim financial statements included herein are unaudited and have been prepared pursuant to the rules and regulations of the U.S. Securities and Exchange Commission, or SEC. The interim financial statements reflect all normal recurring adjustments which, in the opinion of management, are necessary to present fairly the financial position at June 30, 2010 and the results of operations and the cash flows for the interim periods presented. Although management believes the adjustments and disclosures made are adequate to make the information presented not misleading, certain information and footnote disclosures normally included in financial statements prepared in accordance with generally accepted accounting principles have been condensed or omitted pursuant to such rules or regulations. The accompanying balance she et as of September 30, 2009 has been derived from the Company's audited financial statements for the fiscal year ended September 30, 2009. The results of operations for the nine months ended June 30, 2010 are not necessarily indicative of the operating results expected for the full fiscal year. These interim financial statements should be read in conjunction with the audited financial statements and notes thereto included elsewhere in this information statement.

### Going Concern

The financial statements have been prepared on a going concern basis which contemplates the realization of assets and the satisfaction of liabilities in the normal course of business for the foreseeable future. Parametric Sound had net losses of \$226,880 and \$782,632 for the nine months ended June 30, 2010 and 2009, respectively and the Company had no sources of financing except for that provided by LRAD Corporation. At separation, LRAD Corporation is retaining all cash and accounts receivable and the Company will be reliant on obtaining new debt or equity financing sufficient to sustain operations until profitability can be achieved. These factors raise substantial doubt about the Company's ability to continue as a going concern.

The Company's plans for raising initial required financing include the issuance of subordinated promissory notes in the amount of \$350,000 - \$750,000. Of this amount, the Company's Chairman and Chief Executive Officer has agreed to, at a minimum guarantee that at least \$350,000 is funded through the conversion of amounts owed by the Company to Syzygy (see Note 9) and additional cash on the same terms as other investors in the subordinated promissory note financing. There is no assurance that Parametric Sound will be successful in raising sufficient funds to sustain its operations for twelve months or beyond.

The continuation of the Company as a going concern is dependent on its ability to develop revenues and to obtain financing from outside sources. Should the Company be unable to obtain required financing, it may have to curtail operations, which may have a material adverse effect on its financial position and results of operations. The accompanying financial statements do not include any adjustments that would be necessary should the Company be unable to continue as a going concern and, therefore, be required to liquidate its assets and discharge its liabilities in other than the normal course of business and at amounts different from those reflected in the accompanying financial statements.

## 2. Recent Accounting Pronouncements

There were no recent accounting pronouncements adopted by the Company during the nine months ended June 30, 2010 that had a material impact on the Company's financial statements, and there are no not yet effective recent accounting pronouncements that are expected to have a material impact on the Company's financial statements.



## 3. Inventories, net

Inventories are stated at the lower of cost, which approximates actual costs on a first in, first out basis, or estimated market. Inventories consisted of the following:

	ne 30, 010	 September 30, 2009
Finished goods	\$ 56,559	\$ 102,104
Raw materials	1,467,617	 1,760,656
	1,524,176	1,862,760
Reserve for obsolescence	(1,414,144)	 (1,440,303)
Total, net	\$ 110,032	\$ 422,457

## 4. Accrued Liabilities

Accrued liabilities consisted of the following:

	June 30,	September 30,		
	2010	 2009		
Payroll and related	\$ 25,302	\$ 48,757		
Warranty reserve	36,142	 29,389		
Total	\$ 61,444	\$ 78,146		

Changes in the warranty reserve for the nine months ended June 30, 2010 and 2009 were as follows:

Nine Months Ended June 30,	2010		2009
Beginning balance	\$ 29	389 \$	35,755
Warranty provision	16	462	(10,285)
Warranty settlements	(9)	709)	(363)
Ending balance	\$ 36	142 \$	25,107

#### 5. Patents

Patents consist of the following:

	June 30, 2010				September 30, 2009		
Cost	:	\$	942,886		\$	1,015,572	
Accumulated amortization			(469,397	)		(461,927	)
Patents, net		\$	473,489		\$	553,645	

Aggregate amortization expense for the Company's patents was \$49,906 and \$50,660 during the nine months ended June 30, 2010 and 2009, respectively. In addition to amortization, the Company wrote off \$40,301 and \$43,737 of impaired patent costs during the nine months ended June 30, 2010 and 2009, respectively. Additional patents may be written off in the future if it is deemed that certain patents are no longer consistent with the business strategy after the spin-off.

### 6. LRAD Corporation Net Investment

The financial statements of Parametric Sound represent the HSS product business of LRAD Corporation. Because neither a separate subsidiary nor a direct ownership relationship existed, LRAD Corporation's investment in Parametric Sound is shown as parent company equity in lieu of a stockholders' equity classification in the financial statements.

Net transactions include intercompany transactions with LRAD Corporation. These intercompany transactions represent the cash contributions or distributions related to the operating transactions between LRAD Corporation and Parametric Sound.

The following table summarizes changes in parent's equity components during the nine months ended June 30, 2010:

Net loss for the period Net transfers to parent	(226, 880)
Net transfers to parent	(220,000)
	 (100,441)
Balance at June 30, 2010	\$ 595,883

### 7. Share-Based Compensation

Pursuant to LRAD Corporation's 2005 Equity Incentive Plan, the Company recorded \$4,971 and \$419,861 of stock compensation expense for the nine months ended June 30, 2010 and 2009, respectively related to specific personnel whose salary and benefit costs were allocated to the Company. No options were granted to these personnel during the nine months ended June 30, 2010.

## 8. Major Customers and Suppliers

#### Major Customers

For the nine months ended June 30, 2010, revenues from two customers accounted for 40% and 13% of total revenues. No other single customer represented more than 10% of total revenues. For the nine months ended June 30, 2009, revenues from four customers accounted for 23%, 22%, 12% and 11% of total revenues with no other single customer accounting for more than 10% of total revenues. One large customer, Cardinal Health, Inc., which accounted for 40% of sales in the nine months ended June 30, 2010, will continue to be supplied by LRAD Corporation through completion of their current project.



## Suppliers

The Company has a number of components and sub-assemblies produced by outside suppliers, some of which are sourced from a single supplier, which can magnify the risk of shortages and decrease the Company's ability to negotiate with suppliers on the basis of price. In particular, the Company depends on its HSS piezo-film supplier to provide expertise and materials used in the Company's proprietary HSS emitters. If supplier shortages occur, or quality problems arise, then production schedules could be significantly delayed or costs significantly increased, which could in turn have a material adverse effect on the Company's financial condition, results of operation and cash flows.

### 9. Related Party Transactions

The financial statements have been prepared on a stand-alone basis and are derived from the consolidated financial statements and accounting records of LRAD Corporation.

#### Allocation of Expenses

The financial statements include expense allocations for certain functions provided by LRAD Corporation, including, but not limited to, general corporate expenses related to finance, legal, information technology, human resources, employee benefits and incentives. These expenses have been allocated to the Company on the basis of direct usage when identifiable, with the remainder allocated on the basis of revenue or other measure. During the nine months ended June 30, 2010 and 2009, the Company was allocated \$215,348 and \$666,117, respectively, of general corporate expenses incurred by LRAD Corporation which is included within selling, general and administrative expenses in the statements of operations. During the nine months ended June 30, 2010 and 2009, the Company was allocated \$215,348 and \$66,117, respectively, of and 2009, the Company was allocated \$90,207 and \$95,169, respectively, of research and development expenses incurred by LRAD Corporation which is included within research and development expenses in the statements of operations. During the nine months ended June 30, 2010 and 2009, the Company was allocated \$41,069 and \$33,294, respectively, of manufacturing overhead expenses for warehousing, materials management and production management, which is included within cost of sales.

The expense allocations have been determined on a basis that both the Company and LRAD Corporation consider to be a reasonable reflection of the utilization of services provided or the benefit received by the Company during the periods presented. The allocations may not, however, reflect the expense the Company would have incurred as an independent, publicly-traded company for the periods presented. Actual costs that may have been incurred if the Company had been a stand-alone company would depend on a number of factors, including the chosen organization structure, what functions are outsourced or performed by employees and strategic decisions made in areas such as information technology and infrastructure.

### Separation Agreement

In connection with the spin-off, the Company and LRAD Corporation expect to enter into a Separation and Distribution Agreement, (the "Separation Agreement") and some other ancillary agreements to complete the separation of the HSS business from LRAD Corporation and to distribute Parametric Sound common stock to LRAD Corporation stockholders. LRAD Corporation shall retain sufficient inventory and supplies to fulfill outstanding orders with current customers existing at the date of distribution (or received as a continuing order thereafter expected to extend not beyond December 31, 2010) and will be responsible for warranty, returns and related liabilities for such customers and sales. LRAD Corporation will not accept any new orders (other than a continuing order) from current customers after the distribution date. Unless otherwise agreed by the parties, the Company will not be obligated to fulfill any orders or continuing orders of LRAD Corporation.

### Syzygy Licensing LLC

Syzygy is an entity in which, Elwood G. Norris, a director and greater than 10% stockholder of LRAD Corporation, is majority owner. Mr. Norris is also expected to be the Company's President and Chief Executive Officer at the distribution date.

When the Board of Directors of LRAD Corporation conditionally approved the spin-off in April 2010, Mr. Norris through Syzygy agreed to advance third party fees, costs and expenses in connection with the transaction. Should the transaction not be completed then Syzygy and not LRAD Corporation will be responsible for such costs unless LRAD Corporation unilaterally terminates the spin-off other than for lack of stockholder approval, whereupon LRAD Corporation will reimburse Syzygy. Except as expressly set forth in the Separation Agreement, should the separation be completed then all third party fees, costs and expenses paid or incurred in connection with the transactions by Syzygy or by LRAD Corporation will be repaid by the Company at or after the distribution.

The Company also plans to enter into a license agreement effective upon distribution with Syzygy related to certain patent pending technology and trade secrets for a new method of controlling and processing media input to create parametric sound output for parametric emitter devices such as those employed in the HSS technology. The Company will pay no initial license fee nor issue any equity for this license but is expected to reimburse technology, patent, prototype and testing costs incurred to the distribution and to pay future patent related costs. The license is expected to provide for royalties of not more than 5% of net sales from products employing the technology.